Diabetic Bone Disease

Basic and Translational Research and Clinical Applications

Beata Lecka-Czernik John L. Fowlkes *Editors*



Diabetic Bone Disease

Beata Lecka-Czernik • John L. Fowlkes Editors

Diabetic Bone Disease

Basic and Translational Research and Clinical Applications



Editors Beata Lecka-Czernik, PhD Departments of Orthopaedic Surgery, Physiology and Pharmacology Center for Diabetes and Endocrine Research University of Toledo Toledo, OH, USA

John L. Fowlkes, MD Departments of Pediatrics, Pharmacology and Nutritional Sciences University of Kentucky Lexington, KY, USA

ISBN 978-3-319-16401-4 ISBN 978-3-319-16402-1 (eBook) DOI 10.1007/978-3-319-16402-1

Library of Congress Control Number: 2015945658

Springer Cham Heidelberg New York Dordrecht London

© Springer International Publishing Switzerland 2016

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

The publisher, the authors and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, express or implied, with respect to the material contained herein or for any errors or omissions that may have been made.

Printed on acid-free paper

Springer International Publishing AG Switzerland is part of Springer Science+Business Media (www.springer.com)

Preface

The World Health Organization (WHO) estimates that worldwide, diabetes occurs in more than 180 million people. Because the incidence of Type 1 diabetes mellitus (T1D) and Type 2 diabetes mellitus (T2D) is increasing globally, it is estimated that the number of people with diabetes will more than double by 2030. In parallel, it is anticipated that comorbid states associated with diabetes will also rise; thus, understanding and treating complications of diabetes will be a very high priority going forward in order to decrease morbidity and mortality, as well as to better control health care expenditures. Historically, most attention has been focused on four major complications known to afflict many individuals with T1DM and T2DM: retinopathy, neuropathy, nephropathy, and cardiovascular disease. However, epidemiological data now show that other tissues and organs may be significantly impacted by the diabetic state—and the skeletal system is now emerging as a primary target of diabetes-mediated damage (i.e., diabetic bone disease).

Studies have demonstrated that osteopenia and osteoporosis may be frequent complications of T1D, both in children and adults, and that T1D is associated with decreased bone density and increased fracture risk. In contrast to T1D, T2D has typically not been associated with osteopenia or osteoporosis and, in fact, has been more often associated with increased BMD. However, newer data show that bone quality and bone microarchitecture may be compromised in both conditions, suggesting that underlying mechanisms related to increased risk to fracture may be contributory to both forms of diabetes.

In this volume, we provide the reader with up-to-date information about what is currently known about diabetic bone disease and what are the challenges still facing the research and clinical care communities. In the first two chapters, the clinical and epidemiological data about diabetic bone disease is evaluated and reviewed for T1D and T2D, respectively. Chapter 3 discusses how the propensity to fracture in diabetic bone disease can impact fracture risk assessments and how it can be adjusted for using current clinically relevant fracture risk models. Chapter 4 provides a comprehensive overview of orthopedic complications observed in diabetes, and Chapter 5 focuses on the consequences of diabetes on periodontal disease. The utility

of skeletal biomarkers in assessing diabetic bone disease is reviewed in Chapter 6. Chapter 7 shows how drugs used to treat diabetes may also have skeletal consequences. Diabetes may fundamentally impact early progenitor cells of various bone lineages, and through this mechanism globally impact bone; Chapter 8 reviews the literature related to this possibility. How diabetes ultimately may impact the architecture, integrity, and quality of bone is discussed in Chapters 9–11.

As editors, we are truly indebted to the authors who have allowed us to catalogue their unique insights and expertise in diabetic bone disease into one comprehensive text. We hope the reader will find this volume, the first ever to be devoted specifically to diabetic bone disease, to be a useful and thought-provoking resource.

Toledo, OH, USA Lexington, KY, USA Beata Lecka-Czernik John L. Fowlkes

Contents

Par	t I Clinical and Pre-clinical Applications and Research				
1	Skeletal Deficits in Type 1 Diabetes Mellitus Kathryn M. Thrailkill	3			
2	Type 2 Diabetes Mellitus and Skeletal Health Ann V. Schwartz	25			
3	Fracture risk assessment in diabetes William D. Leslie and Stephen Hough	45			
4	Diabetes-Related Conditions and Complications in Orthopaedic Surgery Daniel J. Gehling and Nabil A. Ebraheim	71			
5	Impact of Diabetes on Periodontal Disease E. Xiao, Yingying Wu, and Dana T. Graves	95			
6	Biomarkers of Diabetic Bone Disease Mishaela R. Rubin	113			
7	Safety of Antidiabetic Therapies on Bone Beata Lecka-Czernik and Ann V. Schwartz	125			
8	Bone Marrow Stem Cells and Bone Turnover in Diabetic Disease Laura R. McCabe, Sandi Raehtz, Caitlyn VanGelderen, and Naiomy D. Rios-Arce	147			
Part II Biomechanics and Bone Quality					
9	Material Properties of Diabetic Bone Jeffry S. Nyman and Amy Creecy	183			

10	Bone Quality in Type 2 Diabetes Mellitus	211
	Joshua N. Farr and Sundeep Khosla	
11	Imaging of Diabetic Bone Structure Thomas M. Link and Ursula Heilmeier	225
Ind	ex	237

Contributors

Amy Creecy, MS Department of Biomedical Engineering, Vanderbilt University, Nashville, TN, USA

Nabil A. Ebraheim, MD Department of Orthopaedic Surgery, University of Toledo Medical Center, Toledo, OH, USA

Joshua N. Farr, PhD Division of Endocrinology, Metabolism, Nutrition and Diabetes, Department of Medicine, Mayo Clinic, Rochester, MN, USA

Robert and Arlene Kogod Center on Aging, Rochester, MN, USA

Daniel J. Gehling, MD Department of Orthopaedic Surgery, University of Toledo Medical Center, Toledo, OH, USA

Dana T. Graves, DDS, DMSc Department of Periodontics, School of Dental Medicine, University of Pennsylvania, Philadelphia, PA, USA

Ursula Heilmeier, MD Department of Radiology and Biomedical Imaging, University of California, San Francisco, San Francisco, CA, USA

Stephen Hough, MD Division of Endocrinology, Department of Medicine, Stellenbosch University, Tygerberg (deceased) Academic Hospital, Cape Town, Western Province, South Africa

Sundeep Khosla, MD Division of Endocrinology, Metabolism, Nutrition and Diabetes, Department of Medicine, Mayo Clinic, Rochester, MN, USA

Robert and Arlene Kogod Center on Aging, Rochester, MN, USA

Beata Lecka-Czernik, PhD Departments of Orthopaedic Surgery, Physiology and Pharmacology, Center for Diabetes and Endocrine Research, University of Toledo, Toledo, OH, USA

William D. Leslie, MD Department of Medicine, University of Manitoba, Winnipeg, MB, Canada

Thomas M. Link, MD, PhD Department of Radiology and Biomedical Imaging, University of California, San Francisco, San Francisco, CA, USA

Laura R. McCabe, PhD Departments of Physiology and Radiology, Michigan State University, East Lansing, MI, USA

Jeffry S. Nyman, PhD Department of Orthopaedic Surgery and Rehabilitation, Vanderbilt University Medical Center, Nashville, TN, USA

Tennessee Valley Healthcare System, Department of Veterans Affairs, Nashville, TN, USA

Sandi Raehtz, MS Departments of Physiology and Radiology, Michigan State University, East Lansing, MI, USA

Naiomy D. Rios-Arce, BS Departments of Physiology and Radiology, Michigan State University, East Lansing, MI, USA

Mishaela R. Rubin, MD Department of Medicine, Metabolic Bone Diseases Unit, College of Physicians and Surgeons Columbia University, New York, NY, USA

Ann V. Schwartz, PhD Department of Epidemiology and Biostatistics, University of California, San Francisco, San Francisco, CA, USA

Kathryn M. Thrailkill, MD Department of Pediatrics, University of Kentucky, Lexington, KY, USA

Caitlyn VanGelderen, BA Departments of Physiology and Radiology, Michigan State University, East Lansing, MI, USA

Yingying Wu Department of Periodontics, School of Dental Medicine, University of Pennsylvania, Philadelphia, PA, USA

State Key Laboratory of Oral Disease, West China School of Stomatology, Sichuan University, Chengdu, Sichuan, China

E. Xiao Department of Oral and Maxillofacial Surgery, Peking University School and Hospital of Stomatology, Beijing, China

Department of Periodontics, School of Dental Medicine, University of Pennsylvania, Philadelphia, PA, USA

Part I Clinical and Pre-clinical Applications and Research

Chapter 1 Skeletal Deficits in Type 1 Diabetes Mellitus

Kathryn M. Thrailkill

While type 1 diabetes mellitus (T1D) accounts for <10 % of all diabetes, studies suggest that the global incidence of T1D appears to be increasing by 2–3 % per year [1, 2], attributed to a variety of possible immune-modulatory factors which include societal changes in hygiene, infectious exposures, vitamin D deficiency, and/or infant diets [1]. Diagnosis of the disease most commonly occurs in the pediatric population [3], with peaks in presentation occurring at school entry (5–7 years) and puberty (10–14 years) [2]. In the USA, prevalence of T1D has recently been estimated at 1 in every 433 youth <20 years of age [4].

T1D is, foremost, a state of insulin deficiency due to the progressive, predominantly autoimmune-mediated, destruction of pancreatic beta-cells. As such, unlike type 2 diabetes (T2D), insulinopenia, rather than insulin secretory dysregulation, becomes the overriding phenotype of this disease and insulin replacement therapy is a necessity. In the last two decades, however, therapeutic options (insulin analogues, insulin pumps, continuous glucose sensors) [5–7] have advanced tremendously. And, in recent years, the use of insulin-pump therapy, in particular, continues to increase in an effort to more-tightly regulate glycemic control [8, 9]. Hence, understanding the protracted pathophysiology of *adult* diabetic bone disease in T1D requires an appreciation of the individual context in which this comorbidity has developed over an individual lifetime. The following pages will review skeletal deficits in T1D as they are currently understood, recognizing that the disease phenotype may change in the future, as treatment options continue to improve.

K.M. Thrailkill, M.D. (🖂)

Department of Pediatrics, University of Kentucky, Lexington, KY, USA e-mail: kathryn.thrailkill@uky.edu

[©] Springer International Publishing Switzerland 2016 B. Lecka-Czernik, J.L. Fowlkes (eds.), *Diabetic Bone Disease*, DOI 10.1007/978-3-319-16402-1_1

Human Bone Phenotype in T1D

Bone Mineral Density

In patients with T1D, an increased incidence of osteopenia and osteoporosis has been recognized for over three decades [10–14], occurring not only in adults, but in children as well [15–17]. Many more recent studies have since validated these early findings, demonstrating a reduced bone mineral density (BMD) in T1D [18–22]. Clinical factors associated with lower bone density include: male gender [18, 19, 23, 24]; longer duration of disease [14, 25]; younger age at diagnosis [26]; lower endogenous insulin or C-peptide levels [27]; low body mass index (BMI) [18, 28]; and possibly the presence of chronic diabetes comorbidities or associated autoimmunity [29]. Some studies also suggest that greater longitudinal decrements in BMD occur over time in males [24].

In most studies, poor glycemic control does not seem to be strongly associated with a reduced BMD [18–20, 22, 23, 30, 31], other than in the context of fracture risk (see below) or as a prerequisite for diabetes complications [28]. However, a single point-in-time HbA1c determination used by many of these studies does not reflect long-term hyperglycemia exposure. It is interesting, therefore, that in a few studies with repeated measures of HbA1c over time, chronically poor glycemic control does correlate with low BMD [32], even in children [33]. In addition, in studies of bone quality using phalangeal quantitative ultrasound (QUS) techniques, poor metabolic control is associated with lower QUS scores in premenopausal women [34]. Chronic hypercalciuria, as a consequence of chronic osmotic diuresis in patients with poorly controlled disease, also contributes to negative calcium balance and loss of bone mineral content [30]. Moreover, microvascular complications [35], including the presence of diabetic retinopathy [36], neuropathy [37, 38], and/or nephropathy [39], are associated with a greater occurrence of osteopenia or osteoporosis, in most, but not all studies [30, 40]. Together, these findings would seem to support the intuitive concept that chronically poor metabolic control, particularly of childhood onset [41], either directly or indirectly has some detrimental impact on the skeleton.

While both the axial and appendicular skeleton can be impacted, skeletal sites at greater risk for osteopenia or osteoporosis in males include the spine and femoral neck [18, 19, 23], whereas in females low BMD of the hip or femoral neck is prevalent [19, 30, 40]. Reduced BMD at distal limb sites has also been reported [32, 38]. Even so, a few studies with exclusively premenopausal female enrollment have demonstrated normal BMD in T1D, relative to healthy controls [42, 43].

Fracture Risk

T1D is also associated with an increased risk for fracture, higher than the risk in type 2 diabetes (T2D) [22], though the specific nature and circumstances of this risk are still undergoing scrutiny. Many initial studies examining the risk of fracture in

patients with diabetes either did not, or were not able to distinguish between T1D and T2D patients in their reported analyses [44-46]. Typically, these studies were conducted in older adults (>50 years of age), considered generally at greater risk for fracture, and diabetes was subcategorized only as to the presence or absence of insulin treatment. Hence, some studies demonstrated an increased risk for fracture [44, 46, 47], while others did not [45, 48]. However, among risk factors for hip fracture in >33,000 middle-aged adults in Sweden (~25-60 years), the strongest risk factor for both women (RR = 3.89; 95 % CI 1.69–8.93, p = 0.001) and men (RR = 6.13, 95 % CI 3.19–11.8, p=0.001) was diabetes [49], suggesting that the presence of diabetes was a major risk determinant for this age group. Similar findings had been reported years before in middle-aged Norwegian women and men [50]. To better delineate the relative risk for fracture in T1D, Table 1.1 summarizes a majority of studies conducted during the last 15 years specifically examining patients with T1D, compared with nondiabetics, from selected populations. Together, these studies demonstrate an unequivocally increased fracture risk at the hip, with most demonstrating a six to ninefold increase in relative risk. Moreover, the increase in hip fracture risk is greater than would be expected on the basis of BMD decrements alone [22, 51]. Two more recent studies examining mixed T1D and T2D populations using health-care databases in Canada [52] and Taiwan [53] have also confirmed an increased risk for hip fracture in persons with diabetes, though these studies were, again, predominantly composed of T2D patients.

Though only a very few studies have examined fracture risk at other skeletal sites [51, 54], an increased risk for vertebral fracture is also a consistent finding in studies that have quantified this. This includes the elevated prevalence of asymptomatic vertebral fractures in T1D, independent of BMD [51]. And, in one study, an approximate threefold increase in risk for all non-vertebral fractures was reported in men with T1D [55]. Even in case–control studies comparing lifetime fracture history at any site, an increased frequency of fracture is noted in persons with T1D (odds ratios of ~2) [30, 32]. And, as in other conditions, a more common site for osteoporotic fracture in adults with T1D remains limb fractures, related to accident or falls [44, 47, 56].

In diabetes, an increased risk for fracture has been associated with longer duration of disease [44]; then again, few studies of fracture risk in *pediatric* patients with T1D have been conducted. However, from one study with relatively small sample size, when comparing the history of any fracture in pediatric T1D patients \leq 13 years of age to age-matched control subjects, no significant increase in fracture occurrence was yet seen in these early years [57].

Whether fracture risk is influenced by long-term glycemic control remains unclear, as studies exist to both support and refute this idea (Table 1.1). Nevertheless, in T1D, the presence of diabetic microvascular complications [58], and specifically including retinopathy [44], and nephropathy [59], are associated with a higher risk for fracture, though this increased fracture risk might relate, in part, to the health consequences of these comorbidities (decreased visual acuity or kidney function). Certainly, in persons with long-standing T1D, fracture risk is amplified by an increased fall risk, as might result from hypoglycemic events, visual deficits, peripheral neuropathy, or physical disability [60].

						Age at enroll	Fracture		Associations, impact	
Authors	Years	Locations	Study methods	Ν	(F, M)	(years)	sites	RR [OR] ^a	of comorbidities	CGC
Zhukouskaya [51]	2013	Belarus	Case-control	82/82	F+M	20-55	Vertebral	4	NR ^b	No
Neumann [30]	2011	Germany	Case-control	128/77	F+M	20–70	Any	[2.6] (F)	Not associated	Yes
								[1.9] (M)	with microvascular complications	
Danielson [32]	2009	USA	Case-control	75/75	н	18-50	Any	[2.3]	Poor control	NR
		(IVI)							associated with low BMD	
Vestergaard [22]	2007	Mixed	Meta-analysis	5 studies	F+M	Ι	Hip	6.9	I	NR
Janghorbani [175]	2007	Mixed	Meta-analysis	6 studies	F+M	I	Hip	6.3	1	NR
Ahmed [55]	2006	Norway	Prospective	27,159	F+M	25–98	All non-vert.	3.1 (M)	I	NR
			cohort-6 years	(T1D=81)			hip	8.9 (F) 17.8 (M)		
Strotmeyer [40]	2006	USA (PA)	Case-control	67/237	F	35-55	Any	[1.9]	I	NR
Janghorbani [176]	2006	USA	Nurses' health study cohort	101,343 (T1D=292)	F	30–55	Hip	6.4	1	NR
Vestergaard [54]	2005	Denmark	Case-control	1745/2618	F+M	43±27	Hip	[1.7]	Hypoglycemia	NR
							Spine	[2.5]		
Miao [58]	2005	Sweden	Prospective	24,605 (all	F+M	20.7 ± 10.9	Hip	9.8 (F) 7.6	Microvascular,	NR
			cohort-10 years	T1D)				(W)	Cardiovascular complications	
Nicodemus [177]	2001	USA (Iowa)	Prospective cohort-11 years	32,089 (T1D=47)	Н	55–69	Hip	12.25	I	NR
Forsen [178]	1999	Norway	Prospective cohort—9 years	35,444 (T1D=54)	F+M	≥50	Hip	6.9 (F)	Impaired vision, stroke, motor deficits	No
^a RR is reported for r ^b NR not reported, CC	nost stu 3C fract	dies; [OR] in ure risk corre	dicates that odds rati elated with glycemic	o was reported control						

Table 1.1 Studies of fracture risk in type 1 diabetes

Recognizing that diabetic bone has a greater propensity for fracture than is predicted by BMD, attributes of diabetic bone quality have been examined. A role for the skeletal accumulation of advanced glycation end products (AGEs; see section "Advanced Glycosylation End Products," below) [61, 62], chronic hyperglycemia [30], oxidative stress [63], and microarchitectural bone defects [64] have all been proposed, and it is expected that the pathological mechanisms leading to bone fragility in T1D are multifactorial [65]. To date, however, most information concerning diabetic bone quality is provided by data from animal models. This topic is reviewed in greater detail in other chapters (see Chaps. 9 and 10).

Beyond fragility fractures, other skeletal complications also occur disproportionately in persons with T1D, including fracture-healing complications (nonunion, malunion) [66], Charcot osteoarthropathy [67], osteomyelitis, and diabetic foot syndrome. These orthopedic complications are discussed in Chap. 4.

Onset of Skeletal Pathology: Studies in Children and Adolescents

When focused on *pediatric* patients with T1D, measurements of bone mineral density by dual X-ray absorptiometry (DXA) alone have provided conflicting results [20], with specific reports demonstrating global reductions in BMD [68], sitespecific reductions in BMD [69, 70], or minimal to no effect on bone density [71, 72]. By providing only a two-dimensional estimate of BMD, however, DXA neglects parameters of bone size, bone geometry and bone compartment, which are critical components of the integrity of the growing skeleton. Other investigations using QUS or peripheral quantitative computed tomography (pQCT) provide additional information on bone mineral content, bone size, and bone structure. Collectively, these studies do suggest that cumulative changes in bone architecture are beginning early in childhood, particularly in those diagnosed with T1D at very young ages [73]. Compared with nondiabetic children, reductions in BMD [68, 74-78] and bone size, specifically total cross-sectional area (CSA) [73, 79] and cortical area [15, 80], are relatively consistent findings. Consequently, total bone mineral content of these generally smaller bones is also reduced in pediatric patients with T1D [80].

A mild delay in skeletal maturation, acquired after diagnosis, has been reported in some studies [81, 82]. In addition, the age of peak bone mass acquisition may be delayed in individuals with T1D [68, 72] and chronically poor glycemic control may negatively impact growth velocity [83]. However, unlike historical reports of short stature (i.e., Mauriac syndrome) in poorly controlled T1D [84], in the modern era of tighter glycemic management the impact of T1D on final height is minimal [73, 82, 85].

As with adults, the impact of T1D on the pediatric skeleton may again be genderdiscrepant. Several studies report a more significant impact in young males [80]. Additionally, in a study of females, 13–19 years of age, no differences in BMD,

Table 1.2 Pediatric risk factors for diabetic Image: Comparison of the second seco	Possible pediatric risk factors for DBD in adulthood	References
bone disease	Younger age at diagnosis	[73]
	Longer duration of disease	[41, 77, 179]
	Longer duration of poor glycemic control	[33, 41]
	Lower BMI	[18, 22, 28]
	Delayed skeletal maturation	[81]
	Male gender	[80]
	Accompanying celiac disease	[151, 154]
	Vitamin D deficiency	[110, 111]

measured by DXA, were seen when comparing T1D with control subjects [72]. Other comorbidities, including the diagnosis of associated autoimmune disorders during childhood [86], may also exacerbate the impact of DBD on the growing skeleton (see section "Effects of Ancillary Diagnoses: Celiac Disease, Autoimmune Thyroid Disease, Addison's Disease"). Possible pediatric risk factors for DBD are listed in Table 1.2. Evidence to date would suggest that in patients with one or more of these risk factors, earlier screening for DBD may be warranted.

Contributing Pathological Mechanisms

Decreased Bone Formation

Analysis of bone turnover markers in persons with T1D suggest that bone homeostasis is altered so as to create, predominantly, a state of lowered bone formation, osteoblast dysfunction, and low bone turnover. Numerous studies have demonstrated a decrease in serum osteocalcin (OC) concentration, a marker of bone formation, in children and adolescents [87-90], young adults [31], and middle-aged adults [32, 90–92] with T1D. Even at the time of diagnosis, OC levels in children with T1D appear to be lower than in healthy children [93]. Additionally, lower OC concentrations, typically, are associated with indices of poorer glycemic control [32, 88, 91, 94-96] implying a disease-specific effect. A decrease in serum concentrations of insulin-like growth factor-I (IGF-I), an anabolic regulator of osteoblast function, is also very common in patients of all ages with T1D [31, 97-99], and IGF-I levels correlate with residual beta-cell function, again implying a potential impact of disease severity on bone health in T1D [100]. Similarly, a decrease in serum levels of bone-specific alkaline phosphatase (bALP), a marker of active bone formation and osteoblast activity, has been reported in adults with T1D [84]. However, somewhat contrary to expectations for a state of low bone turnover, an increase in bALP has been reported by others [91, 98] although this has been postulated to reflect impaired osteoblast differentiation. Finally, serum levels of sclerostin, an osteocyte product that antagonizes the Wnt signaling pathway and hence inhibits

bone formation, are increased in T1D [101] perhaps more-so in women [102]. As a whole, these studies suggest that systemic markers of bone formation in T1D are generally indicative of a condition in which bone formation is reduced.

In T1D, an uncoupling of bone formation and bone resorption is also apparent. In young [31] and middle-aged [32] women and in men [103] markers of bone resorption, including urine N-terminal telopeptides (NTx), are comparable in concentration between T1D cases and controls. Similarly, urine deoxypyridinoline (DPD) [91, 104] and C-terminal telopeptide (CTX) [21, 104] values are comparable between patients and healthy controls. Osteoprotegerin (OPG), an inhibitor of osteoclast formation, and subsequently of bone resorption, has also been assessed in T1D. Plasma OPG concentrations and OPG mRNA expression [88] are typically, but not unequivocally [105], higher in persons with T1D. This has been demonstrated both in pediatric [57, 88] and in adult populations [21]. Whether this finding suggests a state of constrained bone resorption, however, is unclear since OPG levels are also increased in periodontal disease [106] and can be indicative of vascular pathology or endothelial dysfunction. Finally, average serum PTH concentrations in T1D are often elevated relative to a matched control population, but frequently not above normal ranges in patients without renal dysfunction or vitamin D deficiency [25, 107]. Taken together, it would appear that T1D is characterized best as a state of inappropriately lowered bone turnover which exists in conjunction with relative osteoblast dysfunction [90] and, hence, low bone formation [103].

Vitamin D Deficiency

Hypovitaminosis D is an important risk factor for decreased bone mineralization [108, 109]. Hence, studies have examined the prevalence of vitamin D deficiency or insufficiency among individuals with T1D, both in childhood and adulthood, and across a variety of geographic locations; many suggest that vitamin D insufficiency/deficiency is more common in T1D compared with the general population [110–112]. Additionally, circulating levels of 250HD are very often found to be significantly lower in case–control comparisons of T1D with the general population [107, 112–114], even when absolute levels do not meet diagnostic cut-points for deficiency. Even so, a 13-year Danish study of 250HD levels in 907 children with newly diagnosed T1D, compared with 896 nondiabetic siblings, a study intended to eliminate genetic and environmental confounding, showed no difference in vitamin D levels [95]. Hence, the nature of this relationship between vitamin D deficiency and T1D, as being a preexisting condition or an acquired condition, remains controversial.

Several mechanisms could contribute to vitamin D deficiency in T1D. By crosssectional analysis, data from National Health and Nutrition Examination Surveys (NHANES III and NHANES 2001–2006) identified an association between vitamin D deficiency and/or insufficiency and albuminuria [115, 116]. In addition, excess urinary loss of vitamin D binding protein (DBP) in T1D, particularly in persons with albuminuria, might contribute mechanistically to vitamin D deficiency. Consistent with this hypothesis, we have demonstrated a significant increase in the urinary excretion of DBP in persons with T1D compared with controls; additionally, urine DBP concentrations correlated with urinary albumin excretion, and vitamin D deficiency or insufficiency was again more prevalent in diabetic subjects with albuminuria [107]. Decreased serum DBP levels in T1D have also been reported [117]. Finally, population-specific genetic variation in the vitamin D axis has been proposed. Vitamin D receptor (VDR) gene polymorphisms (FokI), for instance, have been associated with vitamin D deficiency [118]. However, while polymorphisms in the VDR gene (e.g., FokI, BsmI, ApaI, TaqI) have inconsistently been linked to possible susceptibility for T1D in certain populations, VDR genotypes have *not* been associated with differences in bone turnover markers seen in T1D [119, 120] or with measurements or bone mineral density [120, 121].

While numerous studies have examined the prevalence of vitamin D insufficiency/deficiency in various T1D populations, clinical data examining a direct role for vitamin D in the pathogenesis of diabetic bone disease are scarce, and largely historical. In 1981, a study of 45 Caucasian children with T1D (age 7-18 years), reported no difference in circulating 25OHD levels between non-osteopenic vs. osteopenic diabetic patients (cortical thickness >2SD below the mean normal value) [122], though much less stringent glucometabolic control would have been standardof-care at the time. However, a more recent study of 58 children with T1D, ages 9-19 years, found that children with vitamin D deficiency and T1D did have lower lumbar spine BMD Z-score [89]. Additionally, serum 250HD levels have been shown to negatively correlate with serum collagen type 1 C-terminal propertide [123], yet positively correlate with GLA-carboxylated osteocalcin [124], perhaps inferring a positive effect of 25OHD on bone quality in T1D. Finally, some studies suggest that vitamin D insufficiency may also increase the risk for insulin resistance in T1D [111, 125], hence perhaps impeding the anabolic benefits of insulin signaling in bone [20] in these patients.

Advanced Glycosylation End Products

In the presence of hyperglycemia, the non-enzymatic addition of reduced sugar moieties to amine groups on both tissue-specific and circulating proteins occurs. The generation of one subtype, the advanced glycosylation end products (AGEs), in the milieu of chronic hyperglycemia and the interaction of AGEs with their receptors (RAGEs) are thought to play a significant role in the pathogenesis of diabetic complications [126] including diabetic cardiovascular disease [127], diabetic nephropathy [128], and diabetic retinopathy [129], via activation of pro-inflammatory pathways. While enzymatic cross-link formation between bone collagen molecules is an important element of the mechanical strength and material properties of healthy bone, the dysregulated accumulation of AGEs in bone is thought to negatively impact the integrity of skeletal tissues [130] possibly by causing

micro-damage, hindering osteoblast function and differentiation [131], reducing bone turnover, and/or by competitively inhibiting normal enzymatic cross-link formation [132]. For example, evidence exists to suggest that AGE accumulation is operative in age-related bone fragility [93].

Because diabetic bone has a greater propensity for fracture than is predicted by BMD deficits alone, reduced bone quality secondary to AGE accumulation [133] has also been hypothesized. However, in T1D this mechanism has not yet been established. Measurements of urine and serum pentosidine have been used as biomarkers for AGE accumulation in bone, particularly trabecular bone [134]. Adults with diabetes have increased levels of tissue AGEs compared with chronologically age-matched nondiabetics [96]; and, in older adults with type 2 diabetes, higher urine [135], and serum pentosidine [136] levels correlate with increased vertebral fracture prevalence. Similar data linking serum pentosidine levels and fracture risk in T1D have only recently been reported [137]. Nevertheless, serum AGE concentrations are clearly elevated in T1D during childhood [138], even during preschool and prepubertal years [139]. In addition, skin AGEs, estimated from measurements of skin intrinsic florescence (SIF), are increased in children with both T1D and T2D, to the extent that "approximately 4–6 years of diabetes exposure in some children may be sufficient to increase skin AGEs to levels that would naturally accumulate only after ~25 years of chronological aging" [140]. A similar elevation in SIF is observed in adults with T1D, and is clearly related to long-term glycemic control [141–143]. Gingival AGEs are also increased in T1D-associated periodontitis [144]. Even so, while studies to date suggest an association of AGEs with micro- and macrovascular complications of diabetes, future research will be required to establish a link between bone AGE biomarkers and skeletal phenotype in T1D-associated diabetic bone disease.

Effects of Ancillary Diagnoses: Celiac Disease, Autoimmune Thyroid Disease, Addison's Disease

The relationships between autoimmune disease and bone loss are multifactorial, with outcomes dictated by: (1) direct immune cell regulation of bone homeostasis; (2) disease-specific systemic or local inflammation (e.g., rheumatoid arthritis); (3) disease-specific systemic hormone alterations (e.g., hypothyroidism/hyperthyroidism); (4) ancillary organ damage; or (5) negative skeletal consequences of therapy (i.e., corticosteroids) (For Review, [145]). Certain autoimmune disorders in particular, including autoimmune thyroid disease (ATD), celiac disease (CD), and Addison's disease (AD) are more prevalent in persons with T1D, and all can independently confer detrimental effects on skeletal homeostasis, compounding the impact of diabetic bone disease. The prevalence of celiac autoimmunity or CD in persons with T1D has been estimated at between 4 and 12 % [86, 146–149] and osteoporosis, along with an increase in the risk of bone fractures, is present in the

majority of celiac patients [150]. It is interesting, therefore, that in children with T1D, high-titer seropositivity to celiac antigens in otherwise asymptomatic children is associated with lower bone mineralization [151]. Similarly, in studies of BMD in adults with T1D, a greater reduction in BMD, as well as a higher incidence of fractures [149], is seen among the subset of T1D patients with concurrent celiac autoimmunity [149, 152, 153]. In patients with T1D and active CD, a significant increase in the prevalence of osteopenia (compared to T1D alone) is seen in those with seropositivity and non-adherence to dietary gluten restrictions [154]. Finally, autoantibodies against osteoprotegerin have been reported in a few patients with celiac disease, possibly further contributing to osteoprosis [155]. Hence, in persons with a long-standing coexistence of T1D and CD, the risk for poor bone health is almost undoubtedly higher.

Autoimmune thyroid disease (ATD), either Hashimoto's disease or Grave's disease can occur in 12–30 % of persons with T1D [86, 147], likely related to a common genetic susceptibility for the two disorders [156]. Whether related to the primary hormonal disruption, or to the antithyroid autoimmunity [157], these thyroid disorders can independently contribute to a decrement in BMD. Prevalence rates for adrenal autoantibodies and/or Addison's disease (AD) among persons with T1D vary from 0.5 to 1.4 % [86, 147, 158]. While Addison's disease is a rare condition, AD is also associated with a higher prevalence of osteopenia or osteoporosis [159] and an increased risk of fracture [160], which is unrelated to glucocorticoid replacement therapy. At present, it is not clear whether the diagnosis of ATD or of AD in a person with T1D further increases the risk of osteoporosis or fracture; however, the combination of multiple autoimmune diseases in some persons with T1D may synergistically increase the risk of low bone mineral density [29].

Treatment of Diabetic Bone Disease in T1D

Because diabetic bone disease in type 1 diabetes represents a deficit in osteoblast function and bone formation, antiresorptive therapies for osteoporosis (e.g., bisphosphonates, denosumab) may be ineffective in this form of secondary osteoporosis, other than as combination therapy in conditions of increased bone resorption (i.e., postmenopausal females with T1D) [161]. Calcium and vitamin D supplementation, where indicated, is considered standard-of-care for osteoporosis treatment [162]. Nonetheless, 1 year of calcitriol supplementation in young adults with recent-onset T1D did not significantly change circulating markers of bone turnover, specifically levels of OC or beta-Crosslaps [160]. Moreover, very little information from comparative effectiveness studies is available on the treatment of osteoporosis in T1D. In animal models, intermittent PTH treatment has been shown to improve osteoblast survival and reverse diabetes-associated bone loss [163]. In clinical trials examining the effectiveness of injectable PTH (teriparatide) for treatment of osteoporosis, however, diabetes has typically been an exclusion criteria preventing

enrollment, though positive results with off-label use of teriparatide for nonunion fracture repair [164] or Charcot osteoarthropathy provide some early support for this concept. Clearly, additional clinical investigation in this area is needed.

Prevention of Diabetic Bone Disease in T1D

Greater than 90 % of peak adult bone mass is typically achieved by the end of the second decade of life, and ~40 % of this bone mass is acquired during the adolescent growth spurt; hence, adolescence is a period of particular importance for prevention strategies. Optimal T1D disease management in children and adolescents, according to best practice guidelines [3], would seem a prudent approach to minimizing the impact of T1D on skeletal tissues. This would include effectively managing blood glucose to achieve age-appropriate treatment goals for plasma glucose and HbA1c, as published [3]; additionally, adequate insulin replacement may, in and of itself, have beneficial anabolic effects on bone [20]. Routine screening for associated autoimmune disorders, such as celiac disease and hypothyroidism, as is recommended, is also a necessary component to prevention [3].

Optimal bone mineralization during puberty should also become an important objective for the pediatric clinician, utilizing the following general guidelines:

- Weight-bearing exercise during childhood and adolescence is an important determinant of bone mineral density [165, 166]. Regular physical exercise and sustained physical activity should be encouraged for all adolescents, in accordance with physical activity recommendations of the Centers for Disease Control [167], and as further delineated for persons with T1D by the American Diabetes Association [3]; specifically, it is recommended that children and adolescents perform 60 min or more of physical activity daily. Demonstrating the importance of exercise, a 9-month weight bearing physical activity program comparing 27 children with T1D and 32 healthy children demonstrated improved bone mineral accretion in T1D children, comparable in magnitude to the improvements seen in healthy children [160].
- Adequate daily intake of calcium is important. During puberty, normal skeletal
 mineralization is dependent upon calcium accrual; calcium acquisition/retention
 rates vary by gender and race, but are estimated at ~160–500 mg/day [168–171].
 To achieve this, the most recent Institute of Medicine dietary guidelines for the
 Recommended Dietary Allowance for calcium for adolescents is 1300 mg per
 day [172]. Recognizing that urinary calcium excretion is increased as a consequence of osmotic diuresis in T1D, recommendations for daily calcium intake in
 the adolescent with T1D may, in fact, be greater.
- Recommended Dietary Allowance (RDA) for vitamin D in children and adolescents 1–18 years of age is 600 IU daily [172]. Because vitamin D insufficiency is more common in persons with T1D, the periodic assessment of vitamin D status along with the diagnosis and treatment of vitamin D deficiency in this population are also important goals.

Consensus Recommendations

The FRAX[®] tool, or World Health Organization Fracture Risk Assessment tool, was developed to integrate BMD measurements and patient-specific clinical risk factors in determining an individual's 10-year probability of fracture. Current American Diabetes Association (ADA) Standards of Medical Care for fracture assessment are largely derived from studies of fracture risk in type 2 diabetes, which suggest that for a given FRAX[®] score, the risk of fracture is higher in persons with diabetes [3, 173]. ADA guidelines state that "it is appropriate to assess fracture history and risk factors in older patients with diabetes and recommend BMD testing if appropriate for the patient's age and sex" [3]. Additionally, the fracture risk prediction algorithm used in the United Kingdom (QFracture algorithm) has been recently updated to incorporate type 1 diabetes as a specific variable in their hazard assessment [174]. Guidelines for BMD testing in children and adolescents with T1D, however, do not currently exist.

Summary

Persons with type 1 diabetes, particularly those with long-standing disease, are at significantly increased risk for osteopenia or osteoporosis, fractures, and poor bone healing, collectively now termed diabetic bone disease. Many factors contribute to these deficits in skeletal integrity, including: (1) a disease-specific reduction in new bone formation; (2) an increased occurrence of vitamin D insufficiency; (3) a variety of bone-adverse comorbid conditions, including diabetic microvascular complications and associated autoimmune diseases; and, very possibly (4) glucose-related alterations in the material properties and mechanical strength of diabetic bone. Because T1D is most commonly diagnosed in childhood and adolescence, the early assessment of bone health and skeletal risk factors in youth with T1D will be important, in an effort to maximize the acquisition of peak bone mass in these patients, and to minimize the impact of T1D on the skeleton going forward.

References

- 1. Egro FM. Why is type 1 diabetes increasing? J Mol Endocrinol. 2013;51(1):R1–13. doi:10.1530/JME-13-0067.
- Atkinson MA, Eisenbarth GS, Michels AW. Type 1 diabetes. Lancet. 2014;383(9911):69–82. doi:10.1016/S0140-6736(13)60591-7.
- 3. American DA. Standards of medical care in diabetes 2014. Diabetes Care. 2014;37 Suppl 1:S14–80. doi:10.2337/dc14-S014.
- Pettitt DJ, Talton J, Dabelea D, Divers J, Imperatore G, Lawrence JM, et al. Prevalence of diabetes in U.S. Youth in 2009: The SEARCH for diabetes in youth study. Diabetes Care. 2014;37(2):402–8. doi:10.2337/dc13-1838.

- Kim HS, Shin JA, Chang JS, Cho JH, Son HY, Yoon KH. Continuous glucose monitoring: current clinical use. Diabetes Metab Res Rev. 2012;28 Suppl 2:73–8. doi:10.1002/dmrr.2346. Epub 2013/01/04.
- Shalitin S, Peter CH. Diabetes technology and treatments in the paediatric age group. Int J Clin Pract Suppl. 2011;170:76–82. doi:10.1111/j.1742-1241.2010.02582.x. Epub 2011/11/10.
- Ludvigsson J. Novel therapies in the management of type I diabetes mellitus. Panminerva Med. 2012;54(4):257–70. Epub 2012/11/06.
- Nathan DM, Bayless M, Cleary P, Genuth S, Gubitosi-Klug R, Lachin JM, et al. Diabetes control and complications trial/epidemiology of diabetes interventions and complications study at 30 years: advances and contributions. Diabetes. 2013;62(12):3976–86. doi:10.2337/ db13-1093.
- Golden SH, Sapir T. Methods for insulin delivery and glucose monitoring in diabetes: summary of a comparative effectiveness review. J Manag Care Pharm. 2012;18(6 Suppl):S1–17. Epub 2012/09/26.
- Mathiassen B, Nielsen S, Johansen JS, Hartwell D, Ditzel J, Rodbro P, et al. Long-term bone loss in insulin-dependent diabetic patients with microvascular complications. J Diabet Complications. 1990;4(4):145–9.
- Hui SL, Epstein S, Johnston Jr CC. A prospective study of bone mass in patients with type I diabetes. J Clin Endocrinol Metab. 1985;60(1):74–80. doi:10.1210/jcem-60-1-74.
- 12. Bouillon R. Diabetic bone disease. Calcif Tissue Int. 1991;49(3):155-60.
- McNair P, Christiansen C, Christensen MS, Madsbad S, Faber OK, Binder C, et al. Development of bone mineral loss in insulin-treated diabetes: a 1 1/2 years follow-up study in sixty patients. Eur J Clin Invest. 1981;11(1):55–9.
- Miazgowski T, Czekalski S. A 2-year follow-up study on bone mineral density and markers of bone turnover in patients with long-standing insulin-dependent diabetes mellitus. Osteoporosis Int. 1998;8(5):399–403.
- Santiago JV, McAlister WH, Ratzan SK, Bussman Y, Haymond MW, Shackelford G, et al. Decreased cortical thickness & osteopenia in children with diabetes mellitus. J Clin Endocrinol Metab. 1977;45(4):845–8. doi:10.1210/jcem-45-4-845.
- Rosenbloom AL, Lezotte DC, Weber FT, Gudat J, Heller DR, Weber ML, et al. Diminution of bone mass in childhood diabetes. Diabetes. 1977;26(11):1052–5.
- 17. Weber G, Beccaria L, de'Angelis M, Mora S, Galli L, Cazzuffi MA, et al. Bone mass in young patients with type I diabetes. Bone Miner. 1990;8(1):23–30.
- Hamilton EJ, Rakic V, Davis WA, Chubb SA, Kamber N, Prince RL, et al. Prevalence and predictors of osteopenia and osteoporosis in adults with Type 1 diabetes. Diabet Med. 2009;26(1):45–52. doi:10.1111/j.1464-5491.2008.02608.x.
- Hadjidakis DJ, Raptis AE, Sfakianakis M, Mylonakis A, Raptis SA. Bone mineral density of both genders in Type 1 diabetes according to bone composition. J Diabetes Complications. 2006;20(5):302–7. doi:10.1016/j.jdiacomp.2005.07.006.
- Thrailkill KM, Lumpkin Jr CK, Bunn RC, Kemp SF, Fowlkes JL. Is insulin an anabolic agent in bone? Dissecting the diabetic bone for clues. Am J Physiol Endocrinol Metab. 2005;289(5):E735–45. doi:10.1152/ajpendo.00159.2005.
- Alexopoulou O, Jamart J, Devogelaer JP, Brichard S, de Nayer P, Buysschaert M. Bone density and markers of bone remodeling in type 1 male diabetic patients. Diabetes Metab. 2006;32(5 Pt 1):453–8.
- 22. Vestergaard P. Discrepancies in bone mineral density and fracture risk in patients with type 1 and type 2 diabetes—a meta-analysis. Osteoporos Int. 2007;18(4):427–44. doi:10.1007/s00198-006-0253-4.
- Miazgowski T, Pynka S, Noworyta-Zietara M, Krzyzanowska-Swiniarska B, Pikul R. Bone mineral density and hip structural analysis in type 1 diabetic men. Eur J Endocrinol. 2007;156(1):123–7. doi:10.1530/eje.1.02309.
- Hamilton EJ, Rakic V, Davis WA, Paul Chubb SA, Kamber N, Prince RL, et al. A five-year prospective study of bone mineral density in men and women with diabetes: the Fremantle Diabetes Study. Acta Diabetol. 2012;49(2):153–8. doi:10.1007/s00592-011-0324-7.

- 25. Hamed EA, Faddan NH, Elhafeez HA, Sayed D. Parathormone 25(OH)-vitamin D axis and bone status in children and adolescents with type 1 diabetes mellitus. Pediatr Diabetes. 2011;12(6):536–46. doi:10.1111/j.1399-5448.2010.00739.x.
- Maser RE, Kolm P, Modlesky CM, Beck TJ, Lenhard MJ. Hip strength in adults with type 1 diabetes is associated with age at onset of diabetes. J Clin Densitom. 2012;15(1):78–85. doi:10.1016/j.jocd.2011.08.004.
- Lopez-Ibarra PJ, Pastor MM, Escobar-Jimenez F, Pardo MD, Gonzalez AG, Luna JD, et al. Bone mineral density at time of clinical diagnosis of adult-onset type 1 diabetes mellitus. Endocr Pract. 2001;7(5):346–51. doi:10.4158/EP.7.5.346.
- Eller-Vainicher C, Zhukouskaya VV, Tolkachev YV, Koritko SS, Cairoli E, Grossi E, et al. Low bone mineral density and its predictors in type 1 diabetic patients evaluated by the classic statistics and artificial neural network analysis. Diabetes Care. 2011;34(10):2186–91. doi:10.2337/dc11-0764.
- Lombardi F, Franzese A, Iafusco D, del Puente A, Esposito A, Prisco F, et al. Bone involvement in clusters of autoimmune diseases: just a complication? Bone. 2010;46(2):551–5. doi:10.1016/j.bone.2009.09.034.
- Neumann T, Samann A, Lodes S, Kastner B, Franke S, Kiehntopf M, et al. Glycaemic control is positively associated with prevalent fractures but not with bone mineral density in patients with type 1 diabetes. Diabet Med. 2011;28(7):872–5. doi:10.1111/j.1464-5491.2011.03286.x.
- Mastrandrea LD, Wactawski-Wende J, Donahue RP, Hovey KM, Clark A, Quattrin T. Young women with type 1 diabetes have lower bone mineral density that persists over time. Diabetes Care. 2008;31(9):1729–35. doi:10.2337/dc07-2426.
- Danielson KK, Elliott ME, LeCaire T, Binkley N, Palta M. Poor glycemic control is associated with low BMD detected in premenopausal women with type 1 diabetes. Osteoporos Int. 2009;20(6):923–33. doi:10.1007/s00198-008-0763-3.
- 33. Camurdan MO, Ciaz P, Bideci A, Demirel F. Role of hemoglobin A(1c), duration and puberty on bone mineral density in diabetic children. Pediatr Int. 2007;49(5):645–51. doi:10.1111/j.1442-200X.2007.02451.x.
- 34. Catalano A, Morabito N, Di Vieste G, Pintaudi B, Cucinotta D, Lasco A, et al. Phalangeal quantitative ultrasound and metabolic control in pre-menopausal women with type 1 diabetes mellitus. J Endocrinol Invest. 2013;36(5):347–251. doi:10.3275/8646.
- Munoz-Torres M, Jodar E, Escobar-Jimenez F, Lopez-Ibarra PJ, Luna JD. Bone mineral density measured by dual X-ray absorptiometry in Spanish patients with insulin-dependent diabetes mellitus. Calcif Tissue Int. 1996;58(5):316–9.
- 36. Campos Pastor MM, Lopez-Ibarra PJ, Escobar-Jimenez F, Serrano Pardo MD, Garcia-Cervigon AG. Intensive insulin therapy and bone mineral density in type 1 diabetes mellitus: a prospective study. Osteoporos Int. 2000;11(5):455–9.
- Rix M, Andreassen H, Eskildsen P. Impact of peripheral neuropathy on bone density in patients with type 1 diabetes. Diabetes care. 1999;22(5):827–31.
- Forst T, Pfutzner A, Kann P, Schehler B, Lobmann R, Schafer H, et al. Peripheral osteopenia in adult patients with insulin-dependent diabetes mellitus. Diabet Med. 1995;12(10):874–9.
- Smets YF, de Fijter JW, Ringers J, Lemkes HH, Hamdy NA. Long-term follow-up study on bone mineral density and fractures after simultaneous pancreas-kidney transplantation. Kidney Int. 2004;66(5):2070–6. doi:10.1111/j.1523-1755.2004.00986.x.
- 40. Strotmeyer ES, Cauley JA, Orchard TJ, Steenkiste AR, Dorman JS. Middle-aged premenopausal women with type 1 diabetes have lower bone mineral density and calcaneal quantitative ultrasound than nondiabetic women. Diabetes care. 2006;29(2):306–11.
- 41. Valerio G, del Puente A, Esposito-del Puente A, Buono P, Mozzillo E, Franzese A. The lumbar bone mineral density is affected by long-term poor metabolic control in adolescents with type 1 diabetes mellitus. Hormone Res. 2002;58(6):266–72.
- 42. Ingberg CM, Palmer M, Aman J, Arvidsson B, Schvarcz E, Berne C. Body composition and bone mineral density in long-standing type 1 diabetes. J Int Med. 2004;255(3):392–8.
- Gallacher SJ, Fenner JA, Fisher BM, Quin JD, Fraser WD, Logue FC, et al. An evaluation of bone density and turnover in premenopausal women with type 1 diabetes mellitus. Diabet Med. 1993;10(2):129–33.

1 Skeletal Deficits in Type 1 Diabetes Mellitus

- 44. Ivers RQ, Cumming RG, Mitchell P, Peduto AJ, Blue Mountains Eye Study. Diabetes and risk of fracture: the Blue Mountains Eye Study. Diabetes care. 2001;24(7):1198–203.
- 45. Melchior TM, Sorensen H, Torp-Pedersen C. Hip and distal arm fracture rates in peri- and postmenopausal insulin-treated diabetic females. J Intern Med. 1994;236(2):203–8.
- 46. Seeley DG, Kelsey J, Jergas M, Nevitt MC. Predictors of ankle and foot fractures in older women. The Study of Osteoporotic Fractures Research Group. J Bone Mine Res. 1996;11(9):1347–55. doi:10.1002/jbmr.5650110920.
- 47. Kelsey JL, Browner WS, Seeley DG, Nevitt MC, Cummings SR. Risk factors for fractures of the distal forearm and proximal humerus. The Study of Osteoporotic Fractures Research Group. Am J Epidemiol. 1992;135(5):477–89.
- Heath 3rd H, Melton 3rd LJ, Chu CP. Diabetes mellitus and risk of skeletal fracture. N Engl J Med. 1980;303(10):567–70. doi:10.1056/NEJM198009043031008.
- Holmberg AH, Johnell O, Nilsson PM, Nilsson JA, Berglund G, Akesson K. Risk factors for hip fractures in a middle-aged population: a study of 33,000 men and women. Osteoporos Int. 2005;16(12):2185–94. doi:10.1007/s00198-005-2006-1.
- 50. Meyer HE, Tverdal A, Falch JA. Risk factors for hip fracture in middle-aged Norwegian women and men. Am J Epidemiol. 1993;137(11):1203–11.
- Zhukouskaya VV, Eller-Vainicher C, Vadzianava VV, Shepelkevich AP, Zhurava IV, Korolenko GG, et al. Prevalence of morphometric vertebral fractures in patients with type 1 diabetes. Diabetes Care. 2013;36(6):1635–40. doi:10.2337/dc12-1355.
- Lipscombe LL, Jamal SA, Booth GL, Hawker GA. The risk of hip fractures in older individuals with diabetes: a population-based study. Diabetes Care. 2007;30(4):835–41. doi:10.2337/ dc06-1851.
- 53. Chen HF, Ho CA, Li CY. Increased risks of hip fracture in diabetic patients of Taiwan: a population-based study. Diabetes Care. 2008;31(1):75–80. doi:10.2337/dc07-1072.
- Vestergaard P, Rejnmark L, Mosekilde L. Relative fracture risk in patients with diabetes mellitus, and the impact of insulin and oral antidiabetic medication on relative fracture risk. Diabetologia. 2005;48(7):1292–9. doi:10.1007/s00125-005-1786-3.
- 55. Ahmed LA, Joakimsen RM, Berntsen GK, Fonnebo V, Schirmer H. Diabetes mellitus and the risk of non-vertebral fractures: the Tromso study. Osteoporos Int. 2006;17(4):495–500. doi:10.1007/s00198-005-0013-x.
- 56. Kennedy RL, Henry J, Chapman AJ, Nayar R, Grant P, Morris AD. Accidents in patients with insulin-treated diabetes: increased risk of low-impact falls but not motor vehicle crashes—a prospective register-based study. J Trauma. 2002;52(4):660–6.
- 57. Galluzzi F, Stagi S, Salti R, Toni S, Piscitelli E, Simonini G, et al. Osteoprotegerin serum levels in children with type 1 diabetes: a potential modulating role in bone status. Eur J Endocrinol. 2005;153(6):879–85. doi:10.1530/eje.1.02052.
- Miao J, Brismar K, Nyren O, Ugarph-Morawski A, Ye W. Elevated hip fracture risk in type 1 diabetic patients: a population-based cohort study in Sweden. Diabetes care. 2005;28(12):2850–5.
- Vestergaard P, Rejnmark L, Mosekilde L. Diabetes and its complications and their relationship with risk of fractures in type 1 and 2 diabetes. Calcif Tissue Int. 2009;84(1):45–55. doi:10.1007/s00223-008-9195-5.
- Schwartz AV, Sellmeyer DE. Diabetes, fracture, and bone fragility. Curr Osteoporos Rep. 2007;5(3):105–11.
- Yamagishi S. Role of advanced glycation end products (AGEs) in osteoporosis in diabetes. Curr Drug Targets. 2011;12(14):2096–102.
- Saito M, Fujii K, Soshi S, Tanaka T. Reductions in degree of mineralization and enzymatic collagen cross-links and increases in glycation-induced pentosidine in the femoral neck cortex in cases of femoral neck fracture. Osteoporos Int. 2006;17(7):986–95. doi:10.1007/ s00198-006-0087-0.
- Hamada Y, Fujii H, Fukagawa M. Role of oxidative stress in diabetic bone disorder. Bone. 2009;45 Suppl 1:S35–8. doi:10.1016/j.bone.2009.02.004.
- 64. Armas LA, Akhter MP, Drincic A, Recker RR. Trabecular bone histomorphometry in humans with type 1 diabetes mellitus. Bone. 2012;50(1):91–6. doi:10.1016/j.bone.2011.09.055.

- Merlotti D, Gennari L, Dotta F, Lauro D, Nuti R. Mechanisms of impaired bone strength in type 1 and 2 diabetes. Nutr Metab Cardiovasc Dis. 2010;20(9):683–90. doi:10.1016/j. numecd.2010.07.008.
- 66. Hernandez RK, Do TP, Critchlow CW, Dent RE, Jick SS. Patient-related risk factors for fracture-healing complications in the United Kingdom General Practice Research Database. Acta Orthop. 2012;83(6):653–60. doi:10.3109/17453674.2012.747054.
- Rogers LC, Frykberg RG. The Charcot foot. Med Clin North Am. 2013;97(5):847–56. doi:10.1016/j.mcna.2013.04.003.
- Heap J, Murray MA, Miller SC, Jalili T, Moyer-Mileur LJ. Alterations in bone characteristics associated with glycemic control in adolescents with type 1 diabetes mellitus. J Pediatr. 2004;144(1):56–62. doi:10.1016/j.jpeds.2003.10.066.
- 69. Gunczler P, Lanes R, Paoli M, Martinis R, Villaroel O, Weisinger JR. Decreased bone mineral density and bone formation markers shortly after diagnosis of clinical type 1 diabetes mellitus. J Pediatr Endocrinol Metab. 2001;14(5):525–8.
- Gunczler P, Lanes R, Paz-Martinez V, Martins R, Esaa S, Colmenares V, et al. Decreased lumbar spine bone mass and low bone turnover in children and adolescents with insulin dependent diabetes mellitus followed longitudinally. J Pediatr Endocrinol Metab. 1998;11(3):413–9.
- De Schepper J, Smitz J, Rosseneu S, Bollen P, Louis O. Lumbar spine bone mineral density in diabetic children with recent onset. Horm Res. 1998;50(4):193–6.
- Liu EY, Wactawski-Wende J, Donahue RP, Dmochowski J, Hovey KM, Quattrin T. Does low bone mineral density start in post-teenage years in women with type 1 diabetes? Diabetes Care. 2003;26(8):2365–9.
- Bechtold S, Dirlenbach I, Raile K, Noelle V, Bonfig W, Schwarz HP. Early manifestation of type 1 diabetes in children is a risk factor for changed bone geometry: data using peripheral quantitative computed tomography. Pediatrics. 2006;118(3):e627–34. doi:10.1542/ peds.2005-2193.
- 74. Lettgen B, Hauffa B, Mohlmann C, Jeken C, Reiners C. Bone mineral density in children and adolescents with juvenile diabetes: selective measurement of bone mineral density of trabecular and cortical bone using peripheral quantitative computed tomography. Horm Res. 1995;43(5):173–5.
- 75. Chobot AP, Haffke A, Polanska J, Halaba ZP, Deja G, Jarosz-Chobot P, et al. Bone status in adolescents with type 1 diabetes. Diabetologia. 2010;53(8):1754–60. doi:10.1007/s00125-010-1782-0.
- Heilman K, Zilmer M, Zilmer K, Tillmann V. Lower bone mineral density in children with type 1 diabetes is associated with poor glycemic control and higher serum ICAM-1 and urinary isoprostane levels. J Bone Miner Metab. 2009;27(5):598–604. doi:10.1007/ s00774-009-0076-4.
- 77. Damilakis J, Galanakis E, Mamoulakis D, Sbyrakis S, Gourtsoyiannis N. Quantitative ultrasound measurements in children and adolescents with: type 1 diabetes. Calcif Tissue Int. 2004;74(5):424–8. doi:10.1007/s00223-003-0164-8.
- Roe TF, Mora S, Costin G, Kaufman F, Carlson ME, Gilsanz V. Vertebral bone density in insulin-dependent diabetic children. Metabolism. 1991;40(9):967–71.
- Roggen I, Gies I, Vanbesien J, Louis O, De Schepper J. Trabecular bone mineral density and bone geometry of the distal radius at completion of pubertal growth in childhood type 1 diabetes. Horm Res Paediatr. 2013;79(2):68–74. doi:10.1159/000346686.
- Saha MT, Sievanen H, Salo MK, Tulokas S, Saha HH. Bone mass and structure in adolescents with type 1 diabetes compared to healthy peers. Osteoporos Int. 2009;20(8):1401–6. doi:10.1007/s00198-008-0810-0.
- Dost A, Rohrer T, Fussenegger J, Vogel C, Schenk B, Wabitsch M, et al. Bone maturation in 1788 children and adolescents with diabetes mellitus type 1. J Pediatr Endocrinol Metab. 2010;23(9):891–8.
- Messaaoui A, Dorchy H. Bone age corresponds with chronological age at type 1 diabetes onset in youth. Diabetes Care. 2009;32(5):802–3. doi:10.2337/dc08-2317.

- Myers SE, Albert SG, Haas MJ, Clifton D, Mooradian AD. Pubertal changes in serum leptin levels in adolescents with type 1 diabetes mellitus: a controlled longitudinal study. J Pediatr Endocrinol Metab. 2004;17(12):1653–62.
- Kim MS, Quintos JB. Mauriac syndrome: growth failure and type 1 diabetes mellitus. Pediatric Endocrinol Rev. 2008;5 Suppl 4:989–93. Epub 2008/10/23.
- Rosenbloom AL. Limited joint mobility in childhood diabetes: discovery, description, and decline. J Clin Endocrinol Metab. 2013;98(2):466–73. doi:10.1210/jc.2012-3776.
- Triolo TM, Armstrong TK, McFann K, Yu L, Rewers MJ, Klingensmith GJ, et al. Additional autoimmune disease found in 33% of patients at type 1 diabetes onset. Diabetes Care. 2011;34(5):1211–3. doi:10.2337/dc10-1756.
- 87. Abd El Dayem SM, El-Shehaby AM, Abd El Gafar A, Fawzy A, Salama H. Bone density, body composition, and markers of bone remodeling in type 1 diabetic patients. Scand J Clin Lab Invest. 2011;71(5):387–93. doi:10.3109/00365513.2011.573574.
- Loureiro MB, Ururahy MA, Freire-Neto FP, Oliveira GH, Duarte VM, Luchessi AD, et al. Low bone mineral density is associated to poor glycemic control and increased OPG expression in children and adolescents with type 1 diabetes. Diabetes Res Clin Pract. 2014. doi:10.1016/j.diabres.2013.12.018.
- Karaguzel G, Akcurin S, Ozdem S, Boz A, Bircan I. Bone mineral density and alterations of bone metabolism in children and adolescents with type 1 diabetes mellitus. J Pediatr Endocrinol Metab. 2006;19(6):805–14.
- Bouillon R, Bex M, Van Herck E, Laureys J, Dooms L, Lesaffre E, et al. Influence of age, sex, and insulin on osteoblast function: osteoblast dysfunction in diabetes mellitus. J Clin Endocrinol Metab. 1995;80(4):1194–202. doi:10.1210/jcem.80.4.7714089.
- Masse PG, Pacifique MB, Tranchant CC, Arjmandi BH, Ericson KL, Donovan SM, et al. Bone metabolic abnormalities associated with well-controlled type 1 diabetes (IDDM) in young adult women: a disease complication often ignored or neglected. J Am Coll Nutr. 2010;29(4):419–29.
- 92. Lumachi F, Camozzi V, Tombolan V, Luisetto G. Bone mineral density, osteocalcin, and bone-specific alkaline phosphatase in patients with insulin-dependent diabetes mellitus. Ann N Y Acad Sci. 2009;1173 Suppl 1:E64–7. doi:10.1111/j.1749-6632.2009.04955.x.
- Pater A, Sypniewska G, Pilecki O. Biochemical markers of bone cell activity in children with type 1 diabetes mellitus. J Pediatr Endocrinol Metab. 2010;23(1–2):81–6.
- 94. Lappin DF, Eapen B, Robertson D, Young J, Hodge PJ. Markers of bone destruction and formation and periodontitis in type 1 diabetes mellitus. J Clin Periodontol. 2009;36(8):634– 41. doi:10.1111/j.1600-051X.2009.01440.x.
- 95. Maddaloni E, D'Onofrio L, Lauria A, Maurizi AR, Strollo R, Palermo A, et al. Osteocalcin levels are inversely associated with Hba1c and BMI in adult subjects with long-standing type 1 diabetes. J Endocrinol Invest. 2014;37(7):661–6. doi:10.1007/s40618-014-0092-7.
- Wedrychowicz A, Stec M, Sztefko K, Starzyk JB. Associations between bone, fat tissue and metabolic control in children and adolescents with type 1 diabetes mellitus. Exp Clin Endocrinol Diabetes. 2014;122(8):791–5. doi:10.1055/s-0034-1375666.
- Jehle PM, Jehle DR, Mohan S, Bohm BO. Serum levels of insulin-like growth factor system components and relationship to bone metabolism in Type 1 and Type 2 diabetes mellitus patients. J Endocrinol. 1998;159(2):297–306.
- Moyer-Mileur LJ, Slater H, Jordan KC, Murray MA. IGF-1 and IGF-binding proteins and bone mass, geometry, and strength: relation to metabolic control in adolescent girls with type 1 diabetes. J Bone Miner Res. 2008;23(12):1884–91. doi:10.1359/jbmr.080713.
- AboElAsrar MA, Elbarbary NS, Elshennawy DE, Omar AM. Insulin-like growth factor-1 cytokines cross-talk in type 1 diabetes mellitus: relationship to microvascular complications and bone mineral density. Cytokine. 2012;59(1):86–93. doi:10.1016/j.cyto.2012.03.019.
- Hedman CA, Frystyk J, Lindstrom T, Chen JW, Flyvbjerg A, Orskov H, et al. Residual betacell function more than glycemic control determines abnormalities of the insulin-like growth factor system in type 1 diabetes. J Clin Endocrinol Metab. 2004;89(12):6305–9. doi:10.1210/ jc.2004-0572.

- 101. Neumann T, Hofbauer LC, Rauner M, Lodes S, Kastner B, Franke S, et al. Clinical and endocrine correlates of circulating sclerostin levels in patients with type 1 diabetes mellitus. Clin Endocrinol (Oxf). 2014;80(5):649–55. doi:10.1111/cen.12364.
- 102. Catalano A, Pintaudi B, Morabito N, Di Vieste G, Giunta L, Bruno ML, et al. Gender differences in sclerostin and clinical characteristics in type 1 diabetes mellitus. Eur J Endocrinol. 2014;171(3):293–300. doi:10.1530/EJE-14-0106.
- Kemink SA, Hermus AR, Swinkels LM, Lutterman JA, Smals AG. Osteopenia in insulindependent diabetes mellitus; prevalence and aspects of pathophysiology. J Endocrinol Invest. 2000;23(5):295–303.
- 104. Fassbender WJ, Godde M, Brandenburg VM, Usadel KH, Stumpf UC. Urinary bone resorption markers (deoxypyridinoline and C-terminal telopeptide of type I collagen) in healthy persons, postmenopausal osteoporosis and patients with type I diabetes. Adv Med Sci. 2009;54(1):1–6. doi:10.2478/v10039-009-0003-x.
- 105. Singh DK, Winocour P, Summerhayes B, Viljoen A, Sivakumar G, Farrington K. Low serum osteoprotegerin levels in normoalbuminuric type 1 diabetes mellitus. Acta Diabetol. 2010;47 Suppl 1:105–10. doi:10.1007/s00592-009-0146-z.
- 106. Antonoglou G, Knuuttila M, Nieminen P, Vainio O, Hiltunen L, Raunio T, et al. Serum osteoprotegerin and periodontal destruction in subjects with type 1 diabetes mellitus. J Clin Periodontol. 2013;40(8):765–70. doi:10.1111/jcpe.12129.
- 107. Thrailkill KM, Jo CH, Cockrell GE, Moreau CS, Fowlkes JL. Enhanced excretion of vitamin D binding protein in type 1 diabetes: a role in vitamin D deficiency? J Clin Endocrinol Metab. 2011;96(1):142–9. doi:10.1210/jc.2010-0980.
- Christodoulou S, Goula T, Ververidis A, Drosos G. Vitamin D and bone disease. BioMed Res Int. 2013;2013:396541. doi:10.1155/2013/396541.
- 109. Lips P, Hosking D, Lippuner K, Norquist JM, Wehren L, Maalouf G, et al. The prevalence of vitamin D inadequacy amongst women with osteoporosis: an international epidemiological investigation. J Intern Med. 2006;260(3):245–54. doi:10.1111/j.1365-2796.2006.01685.x.
- Thrailkill KM, Fowlkes JL. The role of vitamin D in the metabolic homeostasis of diabetic bone. Clin Rev Bone Miner Metab. 2013;11(1):28–37. doi:10.1007/s12018-012-9127-9.
- 111. The NS, Crandell JL, Lawrence JM, King IB, Dabelea D, Marcovina SM, et al. Vitamin D in youth with type 1 diabetes: prevalence of insufficiency and association with insulin resistance in the SEARCH Nutrition Ancillary Study. Diabet Med. 2013;30(11):1324–32. doi:10.1111/ dme.12297.
- 112. Bin-Abbas BS, Jabari MA, Issa SD, Al-Fares AH, Al-Muhsen S. Vitamin D levels in Saudi children with type 1 diabetes. Saudi Med J. 2011;32(6):589–92.
- 113. Cooper JD, Smyth DJ, Walker NM, Stevens H, Burren OS, Wallace C, et al. Inherited variation in vitamin D genes is associated with predisposition to autoimmune disease type 1 diabetes. Diabetes. 2011;60(5):1624–31. doi:10.2337/db10-1656.
- 114. Greer RM, Portelli SL, Hung BS, Cleghorn GJ, McMahon SK, Batch JA, et al. Serum vitamin D levels are lower in Australian children and adolescents with type 1 diabetes than in children without diabetes. Pediatr Diabetes. 2013;14(1):31–41. doi:10.1111/j.1399-5448.2012.00890.x.
- 115. de Boer IH, Ioannou GN, Kestenbaum B, Brunzell JD, Weiss NS. 25-Hydroxyvitamin D levels and albuminuria in the Third National Health and Nutrition Examination Survey (NHANES III). Am J Kidney Dis. 2007;50(1):69–77. doi:10.1053/j.ajkd.2007.04.015.
- 116. Diaz VA, Mainous 3rd AG, Carek PJ, Wessell AM, Everett CJ. The association of vitamin D deficiency and insufficiency with diabetic nephropathy: implications for health disparities. J Am Board Fam Med. 2009;22(5):521–7. doi:10.3122/jabfm.2009.05.080231.
- 117. Blanton D, Han Z, Bierschenk L, Linga-Reddy MV, Wang H, Clare-Salzler M, et al. Reduced serum vitamin D-binding protein levels are associated with type 1 diabetes. Diabetes. 2011;60(10):2566–70. doi:10.2337/db11-0576.
- Hamed EO, Abdel-Aal AM, Din AK, Atia MM. Vitamin D level and Fok-I vitamin D receptor gene polymorphism in Egyptian patients with type-1 diabetes. Egypt J Immunol. 2013;20(2):1–10.

- 119. Fassbender WJ, Goertz B, Weismuller K, Steinhauer B, Stracke H, Auch D, et al. VDR gene polymorphisms are overrepresented in German patients with type 1 diabetes compared to healthy controls without effect on biochemical parameters of bone metabolism. Horm Metab Res. 2002;34(6):330–7. doi:10.1055/s-2002-33262.
- 120. Gogas Yavuz D, Keskin L, Kiyici S, Sert M, Yazici D, Sahin I, et al. Vitamin D receptor gene BsmI, FokI, ApaI, TaqI polymorphisms and bone mineral density in a group of Turkish type 1 diabetic patients. Acta Diabetol. 2011;48(4):329–36. doi:10.1007/s00592-011-0284-y.
- 121. Kocabas A, Karaguzel G, Imir N, Yavuzer U, Akcurin S. Effects of vitamin D receptor gene polymorphisms on susceptibility to disease and bone mineral density in Turkish patients with type 1 diabetes mellitus. J Pediatr Endocrinol Metab. 2010;23(12):1289–97.
- 122. Frazer TE, White NH, Hough S, Santiago JV, McGee BR, Bryce G, et al. Alterations in circulating vitamin D metabolites in the young insulin-dependent diabetic. J Clin Endocrinol Metab. 1981;53(6):1154–9. doi:10.1210/jcem-53-6-1154.
- 123. Hampson G, Evans C, Petitt RJ, Evans WD, Woodhead SJ, Peters JR, et al. Bone mineral density, collagen type 1 alpha 1 genotypes and bone turnover in premenopausal women with diabetes mellitus. Diabetologia. 1998;41(11):1314–20.
- 124. Thrailkill KM, Jo CH, Cockrell GE, Moreau CS, Lumpkin Jr CK, Fowlkes JL. Determinants of under carboxylated and carboxylated osteocalcin concentrations in type 1 diabetes. Osteoporos Int. 2012;23(6):1799–806. doi:10.1007/s00198-011-1807-7.
- 125. Thnc O, Cetinkaya S, Kizilgun M, Aycan Z. Vitamin D status and insulin requirements in children and adolescent with type 1 diabetes. J Pediatr Endocrinol Metab. 2011;24(11–12): 1037–41.
- 126. Ramasamy R, Yan SF, Schmidt AM. The diverse ligand repertoire of the receptor for advanced glycation endproducts and pathways to the complications of diabetes. Vascul Pharmacol. 2012;57(5–6):160–7. doi:10.1016/j.vph.2012.06.004.
- 127. Ward MS, Fortheringham AK, Cooper ME, Forbes JM. Targeting advanced glycation endproducts and mitochondrial dysfunction in cardiovascular disease. Curr Opin Pharmacol. 2013;13(4):654–61. doi:10.1016/j.coph.2013.06.009.
- Sun YM, Su Y, Li J, Wang LF. Recent advances in understanding the biochemical and molecular mechanism of diabetic nephropathy. Biochem Biophys Res Commun. 2013;433(4):359–61. doi:10.1016/j.bbrc.2013.02.120.
- 129. Chen M, Curtis TM, Stitt AW. Advanced glycation end products and diabetic retinopathy. Curr Med Chem. 2013;20(26):3234–40.
- Vashishth D. The role of the collagen matrix in skeletal fragility. Curr Osteoporos Rep. 2007;5(2):62–6.
- 131. Okazaki K, Yamaguchi T, Tanaka K, Notsu M, Ogawa N, Yano S, et al. Advanced glycation end products (AGEs), but not high glucose, inhibit the osteoblastic differentiation of mouse stromal ST2 cells through the suppression of osterix expression, and inhibit cell growth and increasing cell apoptosis. Calcif Tissue Int. 2012;91(4):286–96. doi:10.1007/s00223-012-9641-2.
- 132. Saito M, Marumo K. Bone quality in diabetes. Front Endocrinol. 2013;4:72. doi:10.3389/ fendo.2013.00072.
- 133. Saito M, Marumo K. Collagen cross-links as a determinant of bone quality: a possible explanation for bone fragility in aging, osteoporosis, and diabetes mellitus. Osteoporos Int. 2010;21(2):195–214. doi:10.1007/s00198-009-1066-z.
- 134. Karim L, Tang SY, Sroga GE, Vashishth D. Differences in non-enzymatic glycation and collagen cross-links between human cortical and cancellous bone. Osteoporos Int. 2013;24(9):2441–7. doi:10.1007/s00198-013-2319-4.
- 135. Schwartz AV, Garnero P, Hillier TA, Sellmeyer DE, Strotmeyer ES, Feingold KR, et al. Pentosidine and increased fracture risk in older adults with type 2 diabetes. J Clin Endocrinol Metab. 2009;94(7):2380–6. doi:10.1210/jc.2008-2498.
- 136. Yamamoto M, Yamaguchi T, Yamauchi M, Yano S, Sugimoto T. Serum pentosidine levels are positively associated with the presence of vertebral fractures in postmenopausal women with type 2 diabetes. J Clin Endocrinol Metab. 2008;93(3):1013–9. doi:10.1210/jc.2007-1270.
- 137. Neumann T, Lodes S, Kastner B, Franke S, Kiehntopf M, Lehmann T, et al. High serum pentosidine but not esRAGE is associated with prevalent fractures in type 1 diabetes indepen-

dent of bone mineral density and glycaemic control. Osteoporos Int. 2014;25(5):1527–33. doi:10.1007/s00198-014-2631-7.

- 138. Kostolanska J, Jakus V, Barak L. HbA1c and serum levels of advanced glycation and oxidation protein products in poorly and well controlled children and adolescents with type 1 diabetes mellitus. J Pediatr Endocrinol Metab. 2009;22(5):433–42.
- 139. Chiarelli F, de Martino M, Mezzetti A, Catino M, Morgese G, Cuccurullo F, et al. Advanced glycation end products in children and adolescents with diabetes: relation to glycemic control and early microvascular complications. J Pediatr. 1999;134(4):486–91.
- 140. Shah S, Baez EA, Felipe DL, Maynard JD, Hempe JM, Chalew SA. Advanced glycation endproducts in children with diabetes. J Pediatr. 2013;163(5):1427–31. doi:10.1016/j. jpeds.2013.06.044.
- 141. Cleary PA, Braffett BH, Orchard T, Lyons TJ, Maynard J, Cowie C, et al. Clinical and technical factors associated with skin intrinsic fluorescence in subjects with type 1 diabetes from the Diabetes Control and Complications Trial/Epidemiology of Diabetes Interventions and Complications Study. Diabetes Technol Ther. 2013;15(6):466–74.
- 142. Genevieve M, Vivot A, Gonzalez C, Raffaitin C, Barberger-Gateau P, Gin H, et al. Skin autofluorescence is associated with past glycaemic control and complications in type 1 diabetes mellitus. Diabetes Metab. 2013;39(4):349–54. doi:10.1016/j.diabet.2013.03.003.
- 143. Aroda VR, Conway BN, Fernandez SJ, Matter NI, Maynard JD, Orchard TJ, et al. Crosssectional evaluation of noninvasively detected skin intrinsic fluorescence and mean hemoglobin a1c in type 1 diabetes. Diabetes Technol Ther. 2013;15(2):117–23. doi:10.1089/ dia.2012.0211.
- 144. Zizzi A, Tirabassi G, Aspriello SD, Piemontese M, Rubini C, Lucarini G. Gingival advanced glycation end-products in diabetes mellitus-associated chronic periodontitis: an immunohis-tochemical study. J Periodontal Res. 2013;48(3):293–301. doi:10.1111/jre.12007.
- 145. Schett G, David JP. The multiple faces of autoimmune-mediated bone loss. Nat Rev Endocrinol. 2010;6(12):698–706. doi:10.1038/nrendo.2010.190.
- 146. Holmes GK. Screening for coeliac disease in type 1 diabetes. Arch Dis Child. 2002;87(6):495-8.
- 147. Van den Driessche A, Eenkhoorn V, Van Gaal L, De Block C. Type 1 diabetes and autoimmune polyglandular syndrome: a clinical review. Neth J Med. 2009;67(11):376–87.
- 148. Ergur AT, Ocal G, Berberoglu M, Adiyaman P, Siklar Z, Aycan Z, et al. Celiac disease and autoimmune thyroid disease in children with type 1 diabetes mellitus: clinical and HLA-genotyping results. J Clin Res Pediatr Endocrinol. 2010;2(4):151–4. doi:10.4274/jcrpe.v2i4.151.
- 149. Joshi AS, Varthakavi PK, Bhagwat NM, Chadha MD, Mittal SS. Coeliac autoimmunity in type I diabetes mellitus. Arab J Gastroenterol. 2014;15(2):53–7. doi:10.1016/j.ajg.2014.04.004.
- 150. Lucendo AJ, Garcia-Manzanares A. Bone mineral density in adult coeliac disease: an updated review. Rev Esp Enferm Dig. 2013;105(3):154–62.
- 151. Artz E, Warren-Ulanch J, Becker D, Greenspan S, Freemark M. Seropositivity to celiac antigens in asymptomatic children with type 1 diabetes mellitus: association with weight, height, and bone mineralization. Pediatr Diabetes. 2008;9(4 Pt 1):277–84. doi:10.1111/j.1399-5448.2008.00386.x.
- 152. Lunt H, Florkowski CM, Cook HB, Whitehead MR. Bone mineral density, type 1 diabetes, and celiac disease. Diabetes Care. 2001;24(4):791–2.
- 153. Joshi A, Varthakavi P, Chadha M, Bhagwat N. A study of bone mineral density and its determinants in type 1 diabetes mellitus. J Osteoporos. 2013;2013:397814. doi:10.1155/2013/397814.
- 154. Valerio G, Spadaro R, Iafusco D, Lombardi F, Del Puente A, Esposito A, et al. The influence of gluten free diet on quantitative ultrasound of proximal phalanxes in children and adolescents with type 1 diabetes mellitus and celiac disease. Bone. 2008;43(2):322–6. doi:10.1016/j. bone.2008.04.004.
- 155. Riches PL, McRorie E, Fraser WD, Determann C, van't Hof R, Ralston SH. Osteoporosis associated with neutralizing autoantibodies against osteoprotegerin. N Engl J Med. 2009;361(15):1459–65. doi:10.1056/NEJMoa0810925.

- 156. Golden B, Levin L, Ban Y, Concepcion E, Greenberg DA, Tomer Y. Genetic analysis of families with autoimmune diabetes and thyroiditis: evidence for common and unique genes. J Clin Endocrinol Metab. 2005;90(8):4904–11. doi:10.1210/jc.2004-2236.
- 157. Amashukeli M, Korinteli M, Zerekidze T, Jikurauli N, Shanava S, Tsagareli M, et al. The negative correlation between thyrotropin receptor-stimulating antibodies and bone mineral density in postmenopausal patients with Graves' disease. J Investig Med. 2013;61(5): 842–7. doi:10.231/JIM.0b013e31828fcafb.
- 158. Barker JM, Ide A, Hostetler C, Yu L, Miao D, Fain PR, et al. Endocrine and immunogenetic testing in individuals with type 1 diabetes and 21-hydroxylase autoantibodies: Addison's disease in a high-risk population. J Clin Endocrinol Metab. 2005;90(1):128–34. doi:10.1210/jc.2004-0874.
- 159. Leelarathna L, Breen L, Powrie JK, Thomas SM, Guzder R, McGowan B, et al. Co-morbidities, management and clinical outcome of auto-immune Addison's disease. Endocrine. 2010;38(1):113–7. doi:10.1007/s12020-010-9359-8.
- 160. Bjornsdottir S, Saaf M, Bensing S, Kampe O, Michaelsson K, Ludvigsson JF. Risk of hip fracture in Addison's disease: a population-based cohort study. J Intern Med. 2011;270(2): 187–95. doi:10.1111/j.1365-2796.2011.02352.x.
- Mazziotti G, Bilezikian J, Canalis E, Cocchi D, Giustina A. New understanding and treatments for osteoporosis. Endocrine. 2012;41(1):58–69. doi:10.1007/s12020-011-9570-2.
- 162. Cosman F, de Beur SJ, LeBoff MS, Lewiecki EM, Tanner B, Randall S, et al. Clinician's guide to prevention and treatment of osteoporosis. Osteoporos Int. 2014;25(10):2359–81. doi:10.1007/s00198-014-2794-2.
- 163. Motyl KJ, McCauley LK, McCabe LR. Amelioration of type I diabetes-induced osteoporosis by parathyroid hormone is associated with improved osteoblast survival. J Cell Physiol. 2012;227(4):1326–34. doi:10.1002/jcp.22844.
- 164. Tamai K, Takamatsu K, Kazuki K. Successful treatment of nonunion with teriparatide after failed ankle arthrodesis for Charcot arthropathy. Osteoporos Int. 2013;24(10):2729–32. doi:10.1007/s00198-013-2367-9.
- 165. Pedersen BK, Saltin B. Evidence for prescribing exercise as therapy in chronic disease. Scand J Med Sci Sports. 2006;16 Suppl 1:3–63. doi:10.1111/j.1600-0838.2006.00520.x.
- 166. Lofgren B, Dencker M, Nilsson JA, Karlsson MK. A 4-year exercise program in children increases bone mass without increasing fracture risk. Pediatrics. 2012;129(6):e1468–76. doi:10.1542/peds.2011-2274.
- 167. Physical Activity Guidelines Advisory Committee report, 2008. To the Secretary of Health and Human Services. Part A: executive summary. Nutr Rev. 2009;67(2):114–20. doi:10.1111/j.1753-4887.2008.00136.x
- Vatanparast H, Bailey DA, Baxter-Jones AD, Whiting SJ. Calcium requirements for bone growth in Canadian boys and girls during adolescence. Br J Nutr. 2010;103(4):575–80. doi:10.1017/S0007114509992522.
- 169. Zhu K, Greenfield H, Zhang Q, Du X, Ma G, Foo LH, et al. Growth and bone mineral accretion during puberty in Chinese girls: a five-year longitudinal study. J Bone Miner Res. 2008;23(2):167–72. doi:10.1359/jbmr.071006.
- 170. Braun M, Palacios C, Wigertz K, Jackman LA, Bryant RJ, McCabe LD, et al. Racial differences in skeletal calcium retention in adolescent girls with varied controlled calcium intakes. Am J Clin Nutr. 2007;85(6):1657–63.
- 171. Braun M, Martin BR, Kern M, McCabe GP, Peacock M, Jiang Z, et al. Calcium retention in adolescent boys on a range of controlled calcium intakes. Am J Clin Nutr. 2006;84(2):414–8.
- 172. Abrams SA. Calcium and vitamin D requirements for optimal bone mass during adolescence. Curr Opin Clin Nutr Metab Care. 2011;14(6):605–9. doi:10.1097/MCO.0b013e32834b2b01.
- 173. Schwartz AV, Vittinghoff E, Bauer DC, Hillier TA, Strotmeyer ES, Ensrud KE, et al. Association of BMD and FRAX score with risk of fracture in older adults with type 2 diabetes. JAMA. 2011;305(21):2184–92. doi:10.1001/jama.2011.715.
- 174. Hippisley-Cox J, Coupland C. Derivation and validation of updated QFracture algorithm to predict risk of osteoporotic fracture in primary care in the United Kingdom: prospective open cohort study. BMJ. 2012;344:e3427. doi:10.1136/bmj.e3427.

- 175. Janghorbani M, Van Dam RM, Willett WC, Hu FB. Systematic review of type 1 and type 2 diabetes mellitus and risk of fracture. Am J Epidemiol. 2007;166(5):495–505. doi:10.1093/aje/kwm106.
- 176. Janghorbani M, Feskanich D, Willett WC, Hu F. Prospective study of diabetes and risk of hip fracture: the Nurses' Health Study. Diabetes Care. 2006;29(7):1573–8. doi:10.2337/ dc06-0440.
- 177. Nicodemus KK, Folsom AR, Iowa Women's Health Study. Type 1 and type 2 diabetes and incident hip fractures in postmenopausal women. Diabetes Care. 2001;24(7):1192–7.
- 178. Forsen L, Meyer HE, Midthjell K, Edna TH. Diabetes mellitus and the incidence of hip fracture: results from the Nord-Trondelag Health Survey. Diabetologia. 1999;42(8):920–5. doi:10.1007/s001250051248.
- 179. Chobot AP, Haffke A, Polanska J, Halaba ZP, Deja G, Jarosz-Chobot P, et al. Quantitative ultrasound bone measurements in pre-pubertal children with type 1 diabetes. Ultrasound Med Biol. 2012;38(7):1109–15. doi:10.1016/j.ultrasmedbio.2012.02.012.

Chapter 2 Type 2 Diabetes Mellitus and Skeletal Health

Ann V. Schwartz

Fractures, Type 2 Diabetes and an Aging Population

Fractures and type 2 diabetes (T2D) are both more prevalent with older age. At age 50 years, the lifetime risk of a hip fracture is estimated to be 17.5 % for women and 6 % for men in the United States [1]. The lifetime risk of a vertebral fracture is even higher [2]. Fractures exact a substantial public health toll among older adults [3]. Hip fractures in particular are associated with increased mortality and functional decline, but vertebral and other fractures also have substantial consequences [4-8]. The global trend towards an aging population is likely to produce profound changes in the number and geographic distribution of fractures. In the next decades, the number of older adults (60+) in the world population is expected to more than double, from 841 million in 2013 to 2 billion in 2050 [9]. The population of adults who are aged 80+ is projected to increase even more rapidly from 120 million in 2013 to 392 million in 2050. The proportion of older adults (60+) is expected to increase from 23 % in 2012 to 32 % in 2050 in the developed countries and from 9 to 19 % in developing countries. With these trends, the number of fractures is predicted to increase. The number of hip fractures is projected to reach 2.6 million by 2025 and 4.5 million in 2050 [10].

Type 2 diabetes affects over 25 % of older adults in the United States, including diagnosed and undiagnosed cases [11]. In the world population, a recent estimate indicates that over 15 % of adults age 55 years and older have T2D [12]. Given this high prevalence of T2D in older adults, efforts to prevent fractures must necessarily consider the epidemiology and underlying etiology of fracture in those with diabetes to guide effective prevention.

A.V. Schwartz, Ph.D. (🖂)

Department of Epidemiology and Biostatistics, University of California, San Francisco, 550 16th Street, 2nd Floor, Box 0560, UCSF Mission Hall, San Francisco, CA 94158, USA e-mail: aschwartz@psg.ucsf.edu

[©] Springer International Publishing Switzerland 2016

B. Lecka-Czernik, J.L. Fowlkes (eds.), *Diabetic Bone Disease*, DOI 10.1007/978-3-319-16402-1_2

Hip Fracture Incidence and Type 2 Diabetes

Hip Fracture Risk Higher with T2D

Type 2 diabetes is characterized by greater weight and by higher bone density [13]. Both of these factors are protective for most fractures, including hip fracture. Based on this information alone, one would expect reduced fracture risk in those with T2D. However, most studies have instead identified a higher risk of fracture associated with diabetes. The most extensive studies of the effects of T2D on fracture incidence have focused on hip fracture. Because nearly all cases of hip fracture are admitted to a hospital for care, this fracture outcome is easier to study in the large datasets necessary to compare fracture incidence in those with and without diabetes. In 2007 Vestergaard published a meta-analysis of hip fracture results that included eight studies and reported an age-adjusted summary relative risk for hip fracture of 1.38 (1.25–1.53), comparing those with and without T2D [14]. This increase in fracture risk with T2D occurred in spite of higher bone density in those with T2D. The estimated BMD *Z*-score, also assessed in this meta-analysis, was +0.41 for lumbar spine and +0.27 for total hip.

Most [15–21], but not all [22, 23], subsequent studies have reported increased rates of hip fracture with T2D in age-adjusted models. Among older women in WHI, the relative rate of hip fracture, comparing women with and without T2D, was 1.41 (1.17–1.70) [17]. Among Rochester MN residents, hip fracture was increased with T2D after 10 years of follow-up (standardized incidence ratio=1.5; 1.1–1.9) [16]. A study in Manitoba Canada found an increased hip fracture rate for diagnosed diabetes but not for newly identified diabetes [15]. In a cohort in Ontario Canada, women (HR=1.20; 1.16–1.24) and men (HR=1.22; 1.16–1.28) with diabetes had higher hip fracture rates than those without diabetes [19]. In contrast, a study in nursing home residents found no difference in hip fracture rates in age and weight-adjusted models (HR=0.90; 0.60–1.34) although, as discussed below, hip fracture rates were higher in those with diabetes after adjustment for BMD [22].

In a recent large cohort study from Scotland that reported hip fracture rates for the period 2005 through 2007, the age- and calendar-year adjusted results showed a small increased hip fracture rate with T2D in women (1.05; 95 % CI 1.01–1.10) and no evidence of increased hip fracture with T2D in men (0.97; 95 % CI 0.92–1.02) [18]. The authors discuss several possible reasons, other than chance, for the lower relative rates in their study compared with the higher associations reported in the Vestergaard meta-analysis and several subsequent studies. One possible influence is the increase in the proportion of diabetic adults who are overweight or obese, a phenomenon reported in Scotland. In the United States as well, the proportion of those who are obese among adults with T2D increased from 35 % in 1994 to 57 % in 2010, a change that would be expected to reduce hip fracture risk among those with T2D [24]. Better screening and earlier detection of diabetes in recent years would increase the proportion of diabetic patients with shorter duration of the disease, a factor that is associated with fracture risk. Indeed, in the Scotland cohort, those with diabetes
duration of 7 years or more had a higher relative rate of hip fracture (women: rate ratio 1.55; 1.38–1.75. men: rate ratio 1.25; 1.08–1.45) [18]. Another possible influence on the relative rate of hip fracture might be improvements in medical care with improved glycemic control and reductions in diabetic complications. These and other factors are likely to result in variation in age-adjusted associations between diabetes and fracture across countries and time periods.

Nearly all of the studies of diabetes and hip fracture included in the Vestergaard meta-analysis were conducted in Western populations where most of those with T2D are also overweight or obese. In contrast, in East Asian countries, a substantial proportion of those with T2D are normal weight. In this setting, T2D may be associated with a greater relative risk of fracture since fewer diabetic adults have the protective effects of higher BMI. Limited studies indicate that the age-adjusted relative risk is as strong or stronger compared with studies in Western countries. In Taiwan, diabetes was associated with hip fracture risk in women (HR 1.72; 1.66–1.78) and men (HR 1.28; 1.21–1.34) in models adjusted for age, geographic area and urbanization status, but not for BMI or BMD [20]. A study of diabetes and hip fracture in Singapore Chinese reported a rate ratio of 2.00 (1.73–2.31), adjusted for age, sex, dialect group and SES, but not for BMI or BMD [21].

The association between T2D and hip fracture may vary by fracture location although only limited data are currently available. In a study of hip fracture cases in the Netherlands, diabetes was more prevalent among those with a subtrochanteric or femoral shaft fracture, compared with a femoral neck or peritrochanteric fracture (OR = 3.62; 1.45-9.07). In older white women in the United States, the relative rate of intertrochanteric (multivariable adjusted HR = 1.76; 1.37-2.27) and subtrochanteric (3.25; 1.55-6.82), comparing T2D and nondiabetic women, was higher than the relative rate for femoral neck (1.20; 0.90-1.58) fractures [25].

Hip Fracture Risk Higher for a Given BMD in T2D

Some studies of T2D and hip fracture have been able to adjust for the higher BMI and/or BMD that characterizes T2D. These studies have generally found that T2D is associated with higher hip fracture risk *for a given BMD*. A meta-analysis (12 studies) by Janghorbani et al. assessed the relationship between T2D and hip fracture from this perspective, using reported results adjusted for BMI and, where data were available, BMD in those with T2D [26]. The summary relative risk for hip fracture reported in this meta-analysis was 1.7 (1.3–2.2) (Fig. 2.1). Results for subsequent studies with adjustment for BMD have been consistent with this meta-analysis. Among older women in WHI, hip BMD was available on a subset, and the BMD-adjusted model was consistent with increased hip fracture risk although not statistically significant (HR = 1.82; 0.90-3.64) [17]. In a study among nursing home residents, BMD-adjusted relative rate for hip fracture, comparing those with and without DM, was 1.46 (1.25-1.81) [22].



Fig. 2.1 Association between type 2 diabetes mellitus and risk of hip fracture in case–control and cohort studies. Each square shows the study-specific relative risk (RR) estimate (the size of the square reflects the study-specific statistical weight, that is, the inverse of the variance), and the horizontal line shows the related 95 confidence interval (CI). The *diamond* shows the summary RR estimate, and its width represents the corresponding 95 % CI. All statistical tests were two-sided. Statistical heterogeneity between studies was assessed with Cochran's Q test. Reprinted with permission from Janghorbani et al. [26]

Interaction with Age But Not Gender

There is limited evidence that the relationship between diabetes and hip fracture may be stronger at younger ages. A study in Manitoba, Canada, reported an increased rate of hip fracture with diabetes in those <65 years old (6.27; 95 % CI 3.62–10.87) that was greater (*p* for interaction=0.002) than the relative rate in those \geq 65 years old (2.22; 95 % CI 1.71–2.90) [27]. A study of diabetes and hip fracture in Taiwan reported a similar interaction between diabetes and age [20]. For gender, the association between diabetes and hip fracture appears to be similar for women and men. In the meta-analysis by Janghorbani et al., the association between T2D and hip fracture did not differ by gender (*p* for interaction=0.51). The summary relative risk was 2.1 (1.6–2.7) among women (eight studies), and 2.8 (1.2–6.6) among men (five studies) [26]. Similarly, in the Canadian cohort, there was no evidence of interaction by gender for diabetes and hip fracture [27].

Incidence of Any Fracture and Type 2 Diabetes

The two meta-analyses discussed earlier provided evidence of a modest increase in the risk of any fracture with T2D. As with hip fracture, studies of all fractures that have adjusted for the higher BMD associated with type 2 diabetes have generally found that those with T2D have an elevated risk *for a given BMD*. Vestergaard reported an age-adjusted relative risk of 0.96 (0.57–1.61), combining five studies, with strong evidence of heterogeneity (p < 0.01) [14]. When two studies reporting reduced fracture risk with T2D were excluded, heterogeneity was reduced (p=0.88), and the estimated relative risk indicated a modest increase in the risk of any fracture with T2D (1.19; 95 % CI 1.11–1.27). Janghorbani et al. reported an increase in non-vertebral fractures with T2D in results from eight studies, adjusted for BMI and/or BMD (adjusted RR 1.2; 95 % CI 1.01–1.5) [26]. Since these meta-analyses, the WHI study reported an increased risk of any fracture in women (age-adjusted RR 1.29; 1.20–1.38) [17], and a similar increased risk was reported for Rochester MN residents (Standardized incidence ratio=1.3; 1.2–1.4) [16]. In a cohort of older US men, risk of non-vertebral fracture was not increased in age-adjusted models (HR=1.12; 0.94–1.34) but was modestly elevated after adjustment for BMD (HR = 1.30; 1.09–1.54) [28].

A few studies have considered whether diabetes interacts with age, gender, or race for the outcome of any fracture. A recent study in the United States using NHANES data reported an interaction (p < 0.05) between diabetes and race for the outcome of non-skull fracture, comparing age- and sex-adjusted results for the relative rate of fracture associated with T2D in Mexican-American (HR=2.29; 1.41–3.73), non-Hispanic black (1.86; 1.05–3.30), and non-Hispanic white (HR=1.17; 0.89–1.52) participants [29]. The study found no evidence of interaction between diabetes and age or gender for the outcome of non-skull fracture. In WHI there was a suggestion of an increased relative rate of any fracture in black women (HR=1.33; 1.00–1.75) compared with non-Hispanic white women (HR 1.18; 1.08–1.29) [17]. In a smaller cohort of older black and white adults, there was no evidence of interaction between diabetes and gender or race for the outcome of any fracture [30].

The data available for specific fracture sites other than hip are more limited. Janghorbani et al. summarized results for studies published before 2006. With the exception of distal forearm fracture (summary RR=0.98; 95 % CI 0.8-1.2), the point estimates for the summary relative risks were modestly elevated (ankle 1.3, proximal humerus 1.3, vertebra 1.2, foot 1.3), but only the RR for foot fracture was statistically significant. Studies published since this meta-analysis with results for specific non-hip sites include the WHI cohort [17] and Rochester, Minnesota population [16]. Results from WHI were quite similar to the Janghorbani et al. metaanalysis with increased rates for all fracture sites considered with the exception of the lower arm/wrist [17]. Many vertebral fractures do not come to clinical attention, but can be identified on spine X-rays as morphometric vertebral fractures. Some studies of T2D and morphometric vertebral fractures have reported increased prevalence of vertebral fractures [31, 32], but others have not found evidence of an association [33, 34]. A study among postmenopausal women in Beijing found no difference in prevalence of vertebral fracture between women with and without diabetes (OR = 1.04; 0.58–1.88) [35]. However, when women were stratified by BMI, diabetes was associated with higher vertebral fracture prevalence in the non-obese $(BMI < 25 \text{ kg/m}^2)$ women (OR = 2.79; 1.16-6.68).

Risk Factors for Fracture in T2D

Traditional risk factors for fracture include lower bone density, lower BMI, and increased frequency of falls. These factors are also associated with fracture in those with T2D [28, 30, 36].

Bone Density

As noted above, T2D presents a paradox of increased fracture risk in spite of higher bone density. A meta-analysis by Vestergaard reported higher BMD, measured by dual X-ray absorptiometry (DXA) and expressed as Z-score, at the lumbar spine and total hip in those with T2D [14]. Higher BMD was also found by Ma et al. in a more recent meta-analysis of age-adjusted results at the lumbar spine, total hip, and femoral neck [13]. BMD at the radius did not differ by diabetes status. BMI is positively associated with BMD in T2D [13] and broader populations [37] and may account for at least some of the higher BMD with T2D. However, when Ma et al. calculated pooled estimates using results with further adjustment for BMI and other factors, BMD remained higher at the hip (Fig. 2.2) and spine (Fig. 2.3) in T2D. All



Fig. 2.2 Forest plot for mean femoral neck bone mineral density. Difference in means (g/cm²) and 95 % confidence interval for femoral neck bone mineral density between comparison groups with and without type 2 diabetes mellitus, stratified per study and gender. *Diamonds* represent joint estimate for subgroups of available studies for women (*upper*) and men (*middle*), respectively. Pooled estimate for all studies displayed with the *diamond* at the bottom [13]





Fig. 2.3 Forest plot for mean spine mineral density. Difference in means (g/cm²) and 95 % confidence interval for femoral neck bone mineral density between comparison groups with and without type 2 diabetes mellitus, stratified per study and gender. *Diamonds* represent joint estimate for subgroups of available studies for women (*upper*) and men (*middle*), respectively. Pooled estimate for all studies displayed with the *diamond* at the bottom [13]

but one of the studies were conducted in Western countries, characterized by a high prevalence of obesity with T2D. Notably, the one study conducted in East Asia that was included in this meta-analysis found no statistically significant difference in BMD by diabetes status at the femoral neck or lumbar spine while BMD at the radius was lower with T2D [38]. Other studies in East Asian countries have found lower [39–41], similar [42–44] and higher [42–46] BMD at the hip and/or spine in those with diabetes. Two studies in China have assessed the relationship between DM and BMD, stratified by BMI. Among postmenopausal women in Shenyang, Zhou et al. reported lower BMD at the femoral neck and total hip with DM among non-obese (BMI <25 kg/m²) women [47]. Among obese women, BMD was higher but not statistically different between those with and without DM. However, the Peking Vertebral Fracture Study among women in Beijing found higher spine BMD with DM among the non-obese and no difference by diabetes status in the obese women [35]. A recent study in the United States, using high resolution pQCT rather than DXA, found evidence of reduced cortical bone density and thickness with greater cortical porosity, in non-obese T2D women compared with controls or with obese T2D women [48].

A few studies have considered change in BMD in those with diabetes. The results appear to differ by skeletal site. Somewhat surprisingly, several have reported greater bone loss with T2D at the hip in spite of higher baseline BMD. In contrast, bone loss at the radius does not appear to differ with diabetes status. Results for spine BMD have been inconsistent. In the Study of Osteoporotic Fractures (SOF) among older white women, bone loss was more rapid in those with T2D at the total hip, femoral neck, spine, and calcaneus but was not different at the distal radius [49]. Greater bone loss at the femoral neck with diabetes was also reported among white women in a cohort study of older adults; bone loss did not differ by diabetes status in white or black men or black women [50]. In the placebo group of the Fracture Intervention Trial, women with diabetes had faster bone loss at the total hip [51]. In a longitudinal study of perimenopausal women, those with diabetes lost bone more rapidly at the total hip, but preserved bone relative to nondiabetic women at the spine [52]. Krakauer et al. reported no differences in bone loss at the radius by T2D status over 12.5 years of follow-up [53].

The relationship between bone density and fracture in T2D has been an important focus of research. Longitudinal studies have demonstrated that lower BMD is a risk factor for fracture in T2D as in broader populations [27, 54]. However, at any given BMD those with T2D have a higher risk of fracture than those without diabetes. This discrepancy has implications for fracture risk assessment, discussed in Chap. 3. It also implies that there are other factors contributing to fracture risk in T2D, beyond BMD. In broad terms, this additional fracture risk at a given BMD might be the result of increased frequency of falls or reduced bone quality that is not captured by BMD measurements, or both. Evidence that more frequent falls do not fully account for increased fracture risk with T2D (discussed in more detail below), combined with evidence from rodent models [55], has led to the conclusion that diabetic bone is more fragile for a given BMD. Understanding the aspects of bone that are affected by diabetes and that result in fragile bone has been an important focus of research on diabetes and skeletal health.

Body Size

Higher BMI is associated with lower risk of fracture in those with T2D [16, 36, 56]. A study of adults (40+ years old) with a DXA scan record in the Manitoba health registry made an explicit comparison of BMI as a risk factor for fracture in those with (n=6455) and without (n=55,958) diabetes [36]. BMI was higher in those with diabetes, but the relationship with fracture did not differ. In models adjusted for femoral neck BMD and other risk factors in the FRAX calculator, a 5 kg/m² increase in BMI was associated with a lower rate of hip fracture in those with (HR=0.81; 0.69–0.95) and without (HR=0.82; 0.76–0.89) diabetes (p for interaction=0.891). For the outcome of "major osteoporotic fracture" the relationship was also not statistically different (p for interaction=0.080). If anything, the relationship was stronger in those with (HR=0.90; 0.83–0.98) compared to those without (HR=0.98; 0.95–1.02) diabetes.

Falls

About 90 % of fractures are due to a fall although less than 5 % of falls result in a fracture [57]. Falls are associated with fracture risk in those with T2D just as in broader populations [28, 58]. Those with T2D have a moderately increased risk of falling. A meta-analysis of eight studies estimated an increased risk of 1.19 (95 % CI 1.08–1.31), comparing those with and without diabetes [59]. However, insulintreated patients appear to have a 2–3 times higher risk of falls compared with non-diabetic patients [60, 61]. In addition, several [62–64] although not all [65] studies of serious fall injuries resulting in a hospital or emergency room visit have reported higher incidence for T2D patients. These studies did not include information on hypoglycemia, discussed below.

This increased frequency of falls has been proposed as an explanation for the increased fracture risk observed among T2D for a given BMD. Importantly, as noted earlier, several studies of diabetes and fracture have included data on falls and have been able to address this question. In these observational studies, more frequent falls did not fully account for the increased risk of fracture observed with T2D [17, 33, 58, 66]. Higher fracture risk associated with T2D persisted even with adjustment for increased frequency of falls in these cohorts.

Diabetes-Related Risk Factors for Fracture

Investigations into diabetes-related risk factors for fracture have assessed the possible contributions of diabetes duration, presence of diabetes-related complications, glycemic control, and diabetes medications. Evidence that diabetes medications influence skeletal health is discussed in Chap. 7.

Diabetes Duration

Those with longer duration of diabetes have a higher risk of fracture [15, 16, 18, 19, 21, 33, 66–69]. For example, in a longitudinal cohort of Chinese in Singapore, the relative rate of hip fracture was 1.40 (1.08–1.82) in diabetic participants with duration of less than 5 years, compared with nondiabetic participants, and 2.66 (2.04–3.47) in diabetic participants with duration of 15+ years [21]. Longer duration of diabetes also appears to be associated with greater frequency of falls [70]. Reasons for this higher risk of fractures and falls with greater duration of diabetes-related complications.

Glycemic Control

The effect of glycemic control on fracture risk, BMD, and falls remains poorly understood and controversial. On the one hand, reducing A1C levels is a standard goal of diabetes care that has been shown to reduce microvascular complications [71]. It is reasonable to assume that improved control might also have positive effects on bone health. On the other hand, lower A1C levels increase the frequency of hypoglycemic episodes which may increase the risk of falls and fractures. These hypotheses have been tested in a randomized trial, the Action to Control Cardiovascular Risk in Diabetes (ACCORD), a comparison of intensive and standard glycemic control in a cohort of patients with long-term T2D at increased risk of cardiovascular disease age 40-79 years [72]. The median achieved A1C in ACCORD was 6.4 % in the intensive and 7.5 % in the standard glycemic control groups [73]. In an ancillary study, ACCORD BONE, assessment of incident fractures and falls was added to the trial. This trial found no difference in fracture rates between the two treatment groups (HR = 1.04; 0.86-1.27) [74]. These results may have been influenced by the greater degree of TZD use in the intensive glycemic control group. As discussed in Chap. 7, fracture rates are doubled in women, but not men, using a TZD [75]. However, considering the fracture results only among men, whose fracture rates are less likely to be affected by TZD use, there was no evidence of a difference in fracture rates between the intensive and standard glycemic control groups (HR=0.93; 0.70-1.25).

The trial also considered the effect of intensive versus standard glycemic control on falls. There were more hypoglycemic episodes in the intensive treatment group [73], but there was no overall effect of intensive control on the rate of falls (rate ratio=1.10; 0.84–1.43). In analyses stratified by age, there was no evidence of increased falls with intensive control among those 65–79 years old (HR=0.75; 0.55–1.01). In sum, the ACCORD trial provides evidence that reducing average A1C to 6.4 % does not reduce or increase fracture or fall risk compared with an average A1C of 7.5 % in older adults (40–79 years old). While the intensive treatment did not reduce fractures, it was also safe with regard to fractures and falls.

The ACCORD trial does not address the effects of *poor* glycemic control compared with standard (or intensive) control on fractures or falls. This question has been considered recently in several observational studies [76–79]. Three longitudinal studies have reported increased fracture risk with poor control. The largest study, using health registry data in Taiwan, included 1514 hip fracture cases in T2D patients [76]. Those with baseline A1C levels of 9–10 % (HR = 1.24; 1.02–1.49) and 10 % + (HR = 1.32; 1.09–1.58) had increased rates of hip fracture compared with patients whose A1C level was 6–7 %. There was a suggestion of a higher rate in those with A1C <6 % (HR = 1.19; 0.97–1.45) compared with 6–7 %, but the difference was not statistically significant. Those with a baseline A1C of 7–8 % had a slight increase in hip fracture rate (HR = 1.07; 0.92–1.25) compared with 6–7 %. This result is generally consistent with the report from the ACCORD BONE trial comparing fracture rates in men in the standard (median A1C 7.5 %) and intensive (median A1C 6.4 %) control groups (HR = 1.08; 0.80–1.43) [74]. In the Rotterdam cohort of older adults, baseline measurements of fructosamine were converted to A1C equivalent units; the median A1C was 7.5 % among those with type 2 diabetes [77]. Diabetic participants with A1C \geq 7.5 % had a higher rate of fracture, in spite of higher BMD, compared with diabetic participants with A1C <7.5 % (HR=1.54; 95 % CI 1.04–2.29, adjusted for age, sex, height, and weight). The fracture incidence was 31.1 per 1000 person-year in those with A1C \geq 7.5 % and 23.0 per 1000 person-year for those with A1C <7.5 %, an absolute difference in fracture rates of 8 per 1000 person-years. A second longitudinal study, based on the Atherosclerosis Risk in Communities (ARIC) Study, compared rates of fractures identified through hospitalizations [78]. Among those with diagnosed diabetes, the average A1C was 8.3 (SD 2.3). Those with A1C \geq 8 % had a higher rate of fracture than those with lower A1C (HR=1.63; 95 % CI 1.09–2.44, adjusted for age, sex, race, BMI, and other factors). Ten-year cumulative incidence of hospitalized fracture was 4.9 % in those with A1C \geq 8 % and 4.4 % in those with A1C <8 %.

In contrast to these three studies that considered long-term effects of poor glycemic control on fracture risk, Puar et al. considered a somewhat different question, analyzing hip fracture risk within 3 months after assessment of A1C, among diabetic patients admitted to Changi General Hospital in Singapore over a 5-year period [79]. In this analysis of short-term effects of glycemic control, *lower* A1C in the previous 3 months was associated with reduced hip fracture risk. Compared with the reference group (A1C >8 %), diabetic patients with A1C <6 % (OR = 3.0; 2.0–4.5), A1C between 6.1 and 7 % (OR = 2.4; 95 % 1.7–3.2), or A1C between 7.1 and 8.0 % (OR = 1.2; 95 % CI 0.8–1.6) had higher odds of a hip fracture. In contrast, the ACCORD trial did not find increased fracture risk with sustained intensive glycemic control (median A1C 6.4 %) compared with standard control (median A1C 7.5 %) [74].

Studies of the effect of A1C on falls have yielded inconsistent results. In a study of incident falls among older white and African-American adults in the United States, there was no association between glycemic control and falls for those using an oral antidiabetes medication [80]. However, among those using insulin, low baseline A1C (≤ 6 %) was associated with increased risk of falling. A study of US adults 75 years of age and older also found increased risk of falls with lower A1C (≤ 7 %). In contrast, a study in London reported increased falls in older adults (65+ years) in those with poor glycemic control (A1C >8 %) [81]. A study in older African-American adults in the United States found no association between glycemic control, assessed with fructosamine, and falls [82]. In a study that was limited to falls resulting in an injury requiring hospitalization, poor glycemic control (A1C of 8 % or higher) was associated with increased risk among older adults with T2D [64].

The most appropriate A1C goal for treatment of diabetes in older adults remains controversial [71, 83, 84]. There are unanswered questions regarding the net benefit of a lower target in this age group, and the effects on falls and fractures are an important part of this equation. The ACCORD trial suggests that an A1C target of 6.4 % is safe with regards to falls and fractures but notably the trial did not include any participants over age 79 years. ACCORD also indicates that prevention of

fractures and falls is not a motivation for maintaining A1C levels below 7.5 %. Observational studies to date are limited but suggest that an A1C target of less than 8 % could reduce fractures.

Hypoglycemia

Severe hypoglycemia is relatively common in T2D [85, 86]. A recent survey in a California HMO found that about 10 % of T2D patients experience an episode of severe hypoglycemia during a year [86]. Severe episodes were more prevalent among those with near normal glycemia and those with poor control (A1C \geq 9 %). It is believed that hypoglycemic episodes lead to falls and increase the risk of fractures. However, there has been surprisingly little study of these associations. A study using Danish health registry data found higher fracture risk associated with a prior episode of hypoglycemia (HR=1.13; 1.00–1.26) [87]. In a study designed to assess the relationship between hypoglycemic events and fracture, using a healthcare claims database in the United States, T2D patients with a hypoglycemic event requiring medical care during a 1-year period of observation had an increased risk of a fall-related fracture during the same year (adjusted OR 1.70; 95 % CI 1.58–1.83) [88].

Diabetes-Related Complications

Diabetes duration may increase fracture risk in part through an increase in the prevalence of microvascular and macrovascular diabetic complications. Certainly in broader populations there is evidence that the manifestations of microvascular complications, particularly reduced vision [89] and kidney disease [90], contribute to fracture risk. Similarly, the manifestations of macrovascular complications increase fracture risk in broader populations, including stroke [91], myocardial infarction [92] and peripheral arterial disease [93].

Diabetic patients with multiple complications appear to be at higher risk of fracture, but results are mixed for the association between specific complications and fracture. A study using Danish registry data found increased risk of any fracture in T2 diabetic patients with eye disease (OR=2.1; 95 % CI 1.8–2.4), kidney disease (RO=2.0; 1.6–2.5), diabetic neuropathy (1.9; 1.6–2.2), or macrovascular complications (OR=1.9; 95 % CI 1.6–2.3) while T2 diabetes without complications was associated with a more modest increase in fracture risk (OR=1.4; 95 % CI 1.4–1.5), all compared with nondiabetic patients in unadjusted models [94]. However, after multiple adjustment, including for the presence of other complications, these estimates were attenuated. Those with multiple complications had an increased risk of fracture (adjusted OR=1.3; 95 % CI 1.2–1.5) as did those with uncomplicated T2 diabetes (adjusted OR=1.13; 95 % CI 1.06–1.22), but specific complications were not associated with fracture risk in adjusted models. A population-based study using

medical records in Rochester, Minnesota, assessed diabetic complications as risk factors for fracture in models limited to those with T2D [16]. Increased fracture risk was reported for neuropathy (age-adjusted HR 1.4; 95 % CI 1.2–1.7) but not for clinically diagnosed nephropathy (age-adjusted HR 1.1; 0.8–1.3) or retinopathy (age-adjusted HR 1.0; 0.8–1.2). However, renal failure was associated with higher fracture risk (age-adjusted HR = 1.6; 1.2–2.2).

A study using fundus photography to identify retinopathy, conducted as part of the Blue Mountains Eye Study in Australia, found increased fracture risk with diabetic retinopathy [68]. In a US study with direct measurements of monofilament detection and serum creatinine, Strotmeyer et al. reported higher fracture risk among diabetic participants with inability to detect 10 g monofilament, but no increased risk associated with high creatinine [58]. A study of risk factors for prevalent vertebral fractures among T2D women in Brazil found increased prevalence of fractures in those with diabetic retinopathy, identified by funduscopy, and those with lower creatinine clearance, but no difference in fracture prevalence based on clinical nephropathy or peripheral neuropathy [95].

For macrovascular complications, history of stroke is associated with higher fracture risk in older adults with T2D [58]. A mediation analysis, based on data from the Cardiovascular Health Study, found that ankle-arm index, a measure of peripheral arterial disease, accounted for a substantial portion of the higher hip fracture risk associated with diabetes [56].

Prediabetes and Fracture Risk

Those with glucose levels below the threshold defining diabetes but above normal levels have a higher risk of developing diabetes. Primarily on this basis, a category of "prediabetes" has been defined using fasting glucose, the oral glucose tolerance test, or A1C levels [71]. Prediabetes predicts the development of cardiovascular disease as well as the development of diabetes [96]. The effect of prediabetes on fracture risk has been assessed in a limited number of studies with inconsistent results. None of the studies have reported a statistically significant increased risk of fracture with prediabetes although power in each study was limited. Some have reported lower risk of fracture among those with elevated 2 h oral glucose tolerance test (OGTT) results, but elevated fasting glucose (FG) has not been associated with reduced risk of fracture. In the Rotterdam study, participants with impaired glucose tolerance based on OGTT (7.8-11.1 mmol/L) had a lower risk of non-vertebral fracture compared with nondiabetic participants (RR=0.80; 0.63-1.00) in multivariable models including BMD [66]. The Malmo Preventive Project also found evidence of lower osteoporotic fracture risk among those without diabetes in the highest quartile of OGTT but not among those with elevated fasting glucose (FG) [97]. The AusDiab study reported a similar pattern in women, but not men. Nondiabetic women in the highest quartile of OGTT had a reduced risk of a lowtrauma fracture (age and BMI-adjusted OR = 0.59; 0.40-0.88) but fracture risk did not differ across levels of FG [98]. For men, FG and OGTT (age and BMI-adjusted OR = 1.39; 0.60–3.26, comparing upper and lower quartiles of OGTT) were not associated with fracture risk. The Osteoporotic Fractures in Men (MrOS) study in the United States also found no association between prediabetes, defined by FG, and risk of non-vertebral fracture (age and BMD-adjusted HR = 1.04; 0.89–1.21) [28]. In contrast, in older US adults, fracture risk in those with impaired fasting glucose (110–125 mg/dL) was modestly elevated but not statistically different compared with participants who had normal fasting glucose in multivariable models including BMD (HR = 1.34; 0.67–2.67) [58]. A larger US study based on NHANES data similarly reported modestly elevated, but not statistically different, risk of any fracture in those with prediabetes, defined by A1C (5.7–6.4 %), compared with no diabetes (A1C <5.7 %) in multivariable models including BMI [29]. The adjusted HR for any fracture was 1.20 (95 % CI 0.96–1.51) in non-Hispanic whites and 1.42 (95 % CI 0.72–2.81) in Mexican-Americans.

Conclusion

Those with T2D are at increased risk of fracture, especially when their higher bone density is taken into account. The reasons for this higher fracture risk are not clearly understood. Observational studies exploring the role of falls as an intermediary have concluded that falls do not fully account for the increased fracture risk with T2D. Instead, the epidemiology of T2D and fracture points to aspects of bone strength—that cannot be measured with standard DXA—as the underlying factor responsible for increased fracture risk.

References

- Melton LJ, Chrischilles EA, Cooper C, Lane AW, Riggs BL. How many women have osteoporosis? J Bone Miner Res. 1992;7:1005–10.
- Cummings SR, Black DM, Rubin SM. Lifetime risks of hip, Colles' or vertebral fracture and coronary heart disease among white postmenopausal women. Arch Intern Med. 1989;149:2445–8.
- 3. Cauley JA. Public health impact of osteoporosis. J Gerontol A Biol Sci Med Sci. 2013;68: 1243–51.
- Fink HA, Ensrud KE, Nelson DB, Kerani RP, Schreiner PJ, Zhao Y, Cummings SR, Nevitt MC. Disability after clinical fracture in postmenopausal women with low bone density: the fracture intervention trial (FIT). Osteoporos Int. 2003;14:69–76.
- Bliuc D, Nguyen ND, Milch VE, Nguyen TV, Eisman JA, Center JR. Mortality risk associated with low-trauma osteoporotic fracture and subsequent fracture in men and women. JAMA. 2009;301:513–21.
- Hallberg I, Rosenqvist AM, Kartous L, Lofman O, Wahlstrom O, Toss G. Health-related quality of life after osteoporotic fractures. Osteoporos Int. 2004;15:834–41.
- 7. Edwards BJ, Song J, Dunlop DD, Fink HA, Cauley JA. Functional decline after incident wrist fractures—study of osteoporotic fractures: prospective cohort study. BMJ. 2010;341:c3324.

- 2 Type 2 Diabetes Mellitus and Skeletal Health
- Brenneman SK, Barrett-Connor E, Sajjan S, Markson LE, Siris ES. Impact of recent fracture on health-related quality of life in postmenopausal women. J Bone Miner Res. 2006;21:809–16.
- 9. World Population Prospects The 2012 Revision. Demographic Profiles. Vol II. New York: United Nations; 2013. p. 1–870.
- Gullberg B, Johnell O, Kanis JA. World-wide projections for hip fracture. Osteoporos Int. 1997;7:407–13.
- Centers for Disease Control and Prevention (CDC). Prevalence of diabetes and impaired fasting glucose in adults—United States, 1999–2000. MMWR Morb Mortal Wkly Rep. 2003;52:833–7.
- 12. Publisher is International Diabetes Federation. www.idf.org/diabetesatlas. Accessed Sept 2015.
- Ma L, Oei L, Jiang L, Estrada K, Chen H, Wang Z, Yu Q, Zillikens MC, Gao X, Rivadeneira F. Association between bone mineral density and type 2 diabetes mellitus: a meta-analysis of observational studies. Eur J Epidemiol. 2012;27:319–32.
- 14. Vestergaard P. Discrepancies in bone mineral density and fracture risk in patients with type 1 and type 2 diabetes—a meta-analysis. Osteoporos Int. 2007;18:427–44.
- Leslie WD, Lix LM, Prior HJ, Derksen S, Metge C, O'Neil J. Biphasic fracture risk in diabetes: a population-based study. Bone. 2007;40:1595–601.
- Melton 3rd LJ, Leibson CL, Achenbach SJ, Therneau TM, Khosla S. Fracture risk in type 2 diabetes: update of a population-based study. J Bone Miner Res. 2008;23:1334–42.
- Bonds DE, Larson JC, Schwartz AV, Strotmeyer ES, Robbins J, Rodriguez BL, Johnson KC, Margolis KL. Risk of fracture among women with type 2 diabetes: the women's health initiative observational study. J Clin Endocrinol Metab. 2006;91:3404–10.
- Hothersall EJ, Livingstone SJ, Looker HC, Ahmed SF, Cleland S, Leese GP, Lindsay RS, McKnight J, Pearson D, Philip S, Wild SH, Colhoun HM. Contemporary risk of hip fracture in type 1 and type 2 diabetes: a national registry study from Scotland. J Bone Miner Res. 2014;29:1054–60.
- Lipscombe LL, Jamal SA, Booth GL, Hawker GA. The risk of hip fractures in older individuals with diabetes: a population-based study. Diabetes Care. 2007;30:835–41.
- Chen HF, Ho CA, Li CY. Increased risks of hip fracture in diabetic patients of Taiwan: a population-based study. Diabetes Care. 2008;31:75–80.
- Koh WP, Wang R, Ang LW, Heng D, Yuan JM, Yu MC. Diabetes and risk of hip fracture in the Singapore Chinese Health Study. Diabetes Care. 2010;33:1766–70.
- 22. Dobnig H, Piswanger-Solkner JC, Roth M, Obermayer-Pietsch B, Tiran A, Strele A, Maier E, Maritschnegg P, Sieberer C, Fahrleitner-Pammer A. Type 2 diabetes mellitus in nursing home patients: effects on bone turnover, bone mass, and fracture risk. J Clin Endocrinol Metab. 2006;91:3355–63.
- Martinez-Laguna D, Tebe C, Javaid MK, Nogues X, Arden NK, Cooper C, Diez-Perez A, Prieto-Alhambra D. Incident type 2 diabetes and hip fracture risk: a population-based matched cohort study. Osteoporos Int. 2015;26:827–33.
- Centers for Disease Control and Prevention. "Risk Factors for Complications." Retrieved from http://www.cdc.gov/diabetes/statistics/risk_factors_national.htm. Accessed September 2015.
- Napoli N, Schwartz AV, Palermo L, Jin JJ, Wustrack R, Cauley JA, Ensrud KE, Kelly M, Black DM. Risk factors for subtrochanteric and diaphyseal fractures: the study of osteoporotic fractures. J Clin Endocrinol Metab. 2013;98:659–67.
- Janghorbani M, Van Dam RM, Willett WC, Hu FB. Systematic review of type 1 and type 2 diabetes mellitus and risk of fracture. Am J Epidemiol. 2007;166:495–505.
- 27. Giangregorio LM, Leslie WD, Lix LM, Johansson H, Oden A, McCloskey E, Kanis JA. FRAX underestimates fracture risk in patients with diabetes. J Bone Miner Res. 2012;27:301–8.
- Napoli N, Strotmeyer ES, Ensrud KE, Sellmeyer DE, Bauer DC, Hoffman AR, Dam TT, Barrett-Connor E, Palermo L, Orwoll ES, Cummings SR, Black DM, Schwartz AV. Fracture risk in diabetic elderly men: the MrOS study. Diabetologia. 2014;57:2057–65.
- Looker AC, Eberhardt MS, Saydah SH. Diabetes and fracture risk in older U.S. adults. Bone. 2015. doi:10.1016/j.bone.2014.12.008.

- 30. Strotmeyer ES, Cauley JA, Schwartz AV, Nevitt MC, Resnick HE, Zmuda JM, Bauer DC, Tylavsky FA, de Rekeneire N, Harris TB, Newman AB. Diabetes is associated independently of body composition with BMD and bone volume in older white and black men and women: the health, aging, and body composition study. J Bone Miner Res. 2004;19:1084–91.
- Yamamoto M, Yamaguchi T, Yamauchi M, Kaji H, Sugimoto T. Diabetic patients have an increased risk of vertebral fractures independent of BMD or diabetic complications. J Bone Miner Res. 2009;24:702–9.
- 32. Kilpadi KL, Eldabaje R, Schmitz JE, Ehler B, Thames TA, Joshi AP, Simmons 3rd JW, Michalek JE, Fajardo RJ. Type 2 diabetes is associated with vertebral fractures in a sample of clinic- and hospital-based Latinos. J Immigr Minor Health. 2014;16:440–9.
- Schwartz AV, Sellmeyer DE, Ensrud KE, Cauley JA, Tabor HK, Schreiner PJ, Jamal SA, Black DM, Cummings SR. Older women with diabetes have an increased risk of fracture: a prospective study. J Clin Endocrinol Metab. 2001;86:32–8.
- 34. Hanley DA, Brown JP, Tenenhouse A, Olszynski WP, Ioannidis G, Berger C, Prior JC, Pickard L, Murray TM, Anastassiades T, Kirkland S, Joyce C, Joseph L, Papaioannou A, Jackson SA, Poliquin S, Adachi JD. Associations among disease conditions, bone mineral density, and prevalent vertebral deformities in men and women 50 years of age and older: cross-sectional results from the Canadian Multicentre Osteoporosis Study. J Bone Miner Res. 2003;18:784–90.
- 35. Jiajue R, Jiang Y, Wang O, Li M, Xing X, Cui L, Yin J, Xu L, Xia W. Suppressed bone turnover was associated with increased osteoporotic fracture risks in non-obese postmenopausal Chinese women with type 2 diabetes mellitus. Osteoporos Int. 2014;25:1999–2005.
- Leslie WD, Morin SN, Lix LM, Majumdar SR. Does diabetes modify the effect of FRAX risk factors for predicting major osteoporotic and hip fracture? Osteoporos Int. 2014;25:2817–24.
- Lloyd JT, Alley DE, Hawkes WG, Hochberg MC, Waldstein SR, Orwig DL. Body mass index is positively associated with bone mineral density in US older adults. Arch Osteoporos. 2014;9:175.
- 38. Majima T, Komatsu Y, Yamada T, Koike Y, Shigemoto M, Takagi C, Hatanaka I, Nakao K. Decreased bone mineral density at the distal radius, but not at the lumbar spine or the femo-ral neck, in Japanese type 2 diabetic patients. Osteoporos Int. 2005;16:907–13.
- 39. Xia J, Zhong Y, Huang G, Chen Y, Shi H, Zhang Z. The relationship between insulin resistance and osteoporosis in elderly male type 2 diabetes mellitus and diabetic nephropathy. Ann Endocrinol (Paris). 2012;73:546–51.
- 40. Xu L, Cheng M, Liu X, Shan P, Gao H. Bone mineral density and its related factors in elderly male Chinese patients with type 2 diabetes. Arch Med Res. 2007;38:259–64.
- 41. Chen H, Li X, Yue R, Ren X, Zhang X, Ni A. The effects of diabetes mellitus and diabetic nephropathy on bone and mineral metabolism in T2DM patients. Diabetes Res Clin Pract. 2013;100(2):272–6.
- 42. Shan PF, Wu XP, Zhang H, Cao XZ, Gu W, Deng XG, Gu C, Liao EY. Bone mineral density and its relationship with body mass index in postmenopausal women with type 2 diabetes mellitus in mainland China. J Bone Miner Metab. 2009;27:190–7.
- 43. Shan PF, Wu XP, Zhang H, Cao XZ, Yuan LQ, Liao EY. Age-related bone mineral density, osteoporosis rate and risk of vertebral fracture in mainland Chinese women with type 2 diabetes mellitus. J Endocrinol Invest. 2011;34:190–6.
- 44. Jung JK, Kim HJ, Lee HK, Kim SS, Shin CS, Kim JT. Fracture incidence and risk of osteoporosis in female type 2 diabetic patients in Korea. Diabetes Metab J. 2012;36:144–50.
- 45. Lau EM, Leung PC, Kwok T, Woo J, Lynn H, Orwoll E, Cummings S, Cauley J. The determinants of bone mineral density in Chinese men—results from Mr. Os (Hong Kong), the first cohort study on osteoporosis in Asian men. Osteoporos Int. 2006;17:297–303.
- Khoo CC, Woo J, Leung PC, Kwok A, Kwok T. Determinants of bone mineral density in older postmenopausal Chinese women. Climacteric. 2011;14:378–83.
- 47. Zhou Y, Li Y, Zhang D, Wang J, Yang H. Prevalence and predictors of osteopenia and osteoporosis in postmenopausal Chinese women with type 2 diabetes. Diabetes Res Clin Pract. 2010;90:261–9.

- 2 Type 2 Diabetes Mellitus and Skeletal Health
- 48. Furst J, Andreson LB, Zhang C, Nishiyama K, Fink D, Silverberg S, et al. Type 2 diabetes and obesity each contribute separately to adverse skeletal health: adverse effects on cortical bone microarchitecture. In: ASBMR 2014 Annual Meeting, Houston, TX.
- 49. Schwartz AV, Ewing SK, Porzig AM, McCulloch CE, Resnick HE, Hillier TA, Ensrud KE, Black DM, Nevitt MC, Cummings SR, Sellmeyer DE. Diabetes and change in bone mineral density at the hip, calcaneus, spine, and radius in older women. Front Endocrinol (Lausanne). 2013;4:62.
- Schwartz AV, Sellmeyer DE, Strotmeyer ES, Tylavsky FA, Feingold KR, Resnick HE, Shorr RI, Nevitt MC, Black DM, Cauley JA, Cummings SR, Harris TB. Diabetes and bone loss at the hip in older black and white adults. J Bone Miner Res. 2005;20:596–603.
- Keegan TH, Schwartz AV, Bauer DC, Sellmeyer DE, Kelsey JL. Effect of alendronate on bone mineral density and biochemical markers of bone turnover in type 2 diabetic women: the fracture intervention trial. Diabetes Care. 2004;27:1547–53.
- Khalil N, Sutton-Tyrrell K, Strotmeyer ES, Greendale GA, Vuga M, Selzer F, Crandall CJ, Cauley JA. Menopausal bone changes and incident fractures in diabetic women: a cohort study. Osteoporos Int. 2011;22:1367–76.
- 53. Krakauer JC, McKenna MJ, Buderer NF, Rao DS, Whitehouse FW, Parfitt AM. Bone loss and bone turnover in diabetes. Diabetes. 1995;44:775–82.
- 54. Schwartz AV, Vittinghoff E, Bauer DC, Hillier TA, Strotmeyer ES, Ensrud KE, Donaldson MG, Cauley JA, Harris TB, Koster A, Womack CR, Palermo L, Black DM. Association of BMD and FRAX score with risk of fracture in older adults with type 2 diabetes. JAMA. 2011;305:2184–92.
- Nyman JS. Effect of diabetes on the fracture resistance of bone. Clinic Rev Bone Miner Metab. 2013;11:38–48.
- 56. Strotmeyer ES, Kamineni A, Cauley JA, Robbins JA, Fried LF, Siscovick DS, Harris TB, Newman AB. Potential explanatory factors for higher incident hip fracture risk in older diabetic adults. Curr Gerontol Geriatr Res. 2011;2011:979270.
- Morrison A, Fan T, Sen SS, Weisenfluh L. Epidemiology of falls and osteoporotic fractures: a systematic review. Clinicoecon Outcomes Res. 2013;5:9–18.
- 58. Strotmeyer ES, Cauley JA, Schwartz AV, Nevitt MC, Resnick HE, Bauer DC, Tylavsky FA, de Rekeneire N, Harris TB, Newman AB. Nontraumatic fracture risk with diabetes mellitus and impaired fasting glucose in older white and black adults: the health, aging, and body composition study. Arch Intern Med. 2005;165:1612–7.
- Deandrea S, Lucenteforte E, Bravi F, Foschi R, La Vecchia C, Negri E. Risk factors for falls in community-dwelling older people: a systematic review and meta-analysis. Epidemiology. 2010;21:658–68.
- 60. Schwartz AV, Hillier TA, Sellmeyer DE, Resnick HE, Gregg E, Ensrud KE, Schreiner PJ, Margolis KL, Cauley JA, Nevitt MC, Black DM, Cummings SR. Older women with diabetes have a higher risk of falls: a prospective study. Diabetes Care. 2002;25:1749–54.
- Volpato S, Leveille SG, Blaum C, Fried LP, Guralnik JM. Risk factors for falls in older disabled women with diabetes: the women's health and aging study. J Gerontol A Biol Sci Med Sci. 2005;60:1539–45.
- Malmivaara A, Heliövaara M, Knekt P, Reunanen A, Aromaa A. Risk factors for injurious falls leading to hospitalization or death in a cohort of 19,500 adults. Am J Epidemiol. 1993;138:384–94.
- 63. Kennedy RL, Henry J, Chapman AJ, Nayar R, Grant P, Morris AD. Accidents in patients with insulin-treated diabetes: increased risk of low-impact falls but not motor vehicle crashes—a prospective register-based study. J Trauma. 2002;52:660–6.
- 64. Yau RK, Strotmeyer ES, Resnick HE, Sellmeyer DE, Feingold KR, Cauley JA, Vittinghoff E, De Rekeneire N, Harris TB, Nevitt MC, Cummings SR, Shorr RI, Schwartz AV. Diabetes and risk of hospitalized fall injury among older adults. Diabetes Care. 2013;36:3985–91.
- Herndon JG, Helmick CG, Sattin RW, Stevens JA, DeVito C, Wingo PA. Chronic medical conditions and risk of fall injury events at home in older adults. J Am Geriatr Soc. 1997;45:739–43.

- 66. de Liefde I, van der Klift M, de Laet CE, van Daele PL, Hofman A, Pols HA. Bone mineral density and fracture risk in type-2 diabetes mellitus: the Rotterdam Study. Osteoporos Int. 2005;16:1713–20.
- 67. Nicodemus KK, Folsom AR. Type 1 and type 2 diabetes and incident hip fractures in postmenopausal women. Diabetes Care. 2001;24:1192–7.
- Ivers RQ, Cumming RG, Mitchell P, Peduto AJ. Diabetes and risk of fracture: The Blue Mountains Eye Study. Diabetes Care. 2001;24:1198–203.
- 69. Janghorbani M, Feskanich D, Willett WC, Hu F. Prospective study of diabetes and risk of hip fracture: the Nurses' Health Study. Diabetes Care. 2006;29:1573–8.
- Quandt SA, Stafford JM, Bell RA, Smith SL, Snively BM, Arcury TA. Predictors of falls in a multiethnic population of older rural adults with diabetes. J Gerontol A Biol Sci Med Sci. 2006;61:394–8.
- American Diabetes Association. Standards of medical care in diabetes 2015. Diabetes Care. 2015;38(Suppl):S1–3.
- 72. Buse JB, Bigger JT, Byington RP, Cooper LS, Cushman WC, Friedewald WT, Genuth S, Gerstein HC, Ginsberg HN, Goff Jr DC, Grimm Jr RH, Margolis KL, Probstfield JL, Simons-Morton DG, Sullivan MD. Action to Control Cardiovascular Risk in Diabetes (ACCORD) trial: design and methods. Am J Cardiol. 2007;99:21i–33i.
- Gerstein HC, Miller ME, Byington RP, Goff Jr DC, Bigger JT, Buse JB, Cushman WC, Genuth S, Ismail-Beigi F, Grimm Jr RH, Probstfield JL, Simons-Morton DG, Friedewald WT. Effects of intensive glucose lowering in type 2 diabetes. N Engl J Med. 2008;358:2545–59.
- 74. Schwartz AV, Margolis KL, Sellmeyer DE, Vittinghoff E, Ambrosius WT, Bonds DE, Josse RG, Schnall AM, Simmons DL, Hue TF, Palermo L, Hamilton BP, Green JB, Atkinson HH, O'Connor PJ, Force RW, Bauer DC. Intensive glycemic control is not associated with fractures or falls in the ACCORD randomized trial. Diabetes Care. 2012;35:1525–31.
- Zhu ZN, Jiang YF, Ding T. Risk of fracture with thiazolidinediones: an updated meta-analysis of randomized clinical trials. Bone. 2014;68:115–23.
- 76. Li CI, Liu CS, Lin WY, Chen CC, Yang SY, Chen HJ, Lin CC, Li TC. Glycated hemoglobin level and risk of hip fracture in older people with type 2 diabetes: a competing risk analysis of Taiwan diabetes cohort study. J Bone Miner Res. 2015. doi:10.1002/jbmr.2462.
- 77. Oei L, Zillikens MC, Dehghan A, Buitendijk GH, Castano-Betancourt MC, Estrada K, Stolk L, Oei EH, van Meurs JB, Janssen JA, Hofman A, van Leeuwen JP, Witteman JC, Pols HA, Uitterlinden AG, Klaver CC, Franco OH, Rivadeneira F. High bone mineral density and fracture risk in type 2 diabetes as skeletal complications of inadequate glucose control: the Rotterdam Study. Diabetes Care. 2013;36(6):1619–28.
- Schneider AL, Williams EK, Brancati FL, Blecker S, Coresh J, Selvin E. Diabetes and risk of fracture-related hospitalization: the Atherosclerosis Risk in Communities Study. Diabetes Care. 2012;36(5):1153–8.
- 79. Puar TH, Khoo JJ, Cho LW, Xu Y, Chen YT, Chuo AM, Poh CB, Ng JM. Association between glycemic control and hip fracture. J Am Geriatr Soc. 2012;60:1493–7.
- Schwartz AV, Vittinghoff E, Sellmeyer DE, Feingold KR, de Rekeneire N, Strotmeyer ES, Shorr RI, Vinik AI, Odden MC, Park SW, Faulkner KA, Harris TB. Diabetes-related complications, glycemic control, and falls in older adults. Diabetes Care. 2008;31:391–6.
- Tilling LM, Darawil K, Britton M. Falls as a complication of diabetes mellitus in older people. J Diabetes Complications. 2006;20:158–62.
- Miller DK, Lui LY, Perry HM, Kaiser FE, Morley JE. Reported and measured physical functioning in older inner-city diabetic African Americans. J Gerontol A Biol Sci Med Sci. 1999;54:M230–6.
- 83. Du YF, Ou HY, Beverly EA, Chiu CJ. Achieving glycemic control in elderly patients with type 2 diabetes: a critical comparison of current options. Clin Interv Aging. 2014;9:1963–80.
- Lipska KJ, Ross JS, Miao Y, Shah ND, Lee SJ, Steinman MA. Potential overtreatment of diabetes mellitus in older adults with tight glycemic control. JAMA Intern Med. 2015;175(3):356–62.
- 85. Leese GP, Wang J, Broomhall J, Kelly P, Marsden A, Morrison W, Frier BM, Morris AD. Frequency of severe hypoglycemia requiring emergency treatment in type 1 and type 2 diabetes: a population-based study of health service resource use. Diabetes Care. 2003;26:1176–80.

- Lipska KJ, Warton EM, Huang ES, Moffet HH, Inzucchi SE, Krumholz HM, Karter AJ. HbA1c and risk of severe hypoglycemia in type 2 diabetes: the Diabetes and Aging Study. Diabetes Care. 2013;36:3535–42.
- Vestergaard P, Rejnmark L, Mosekilde L. Relative fracture risk in patients with diabetes mellitus, and the impact of insulin and oral antidiabetic medication on relative fracture risk. Diabetologia. 2005;48:1292–9.
- Johnston SS, Conner C, Aagren M, Ruiz K, Bouchard J. Association between hypoglycaemic events and fall-related fractures in Medicare-covered patients with type 2 diabetes. Diabetes Obes Metab. 2012;14:634–43.
- Dargent-Molina P, Favier F, Grandjean H, Baudoin C, Schott AM, Hausherr E, Meunier PJ, Breart G. Fall-related factors and risk of hip fracture: the EPIDOS prospective study. Lancet. 1996;348:145–9.
- Nickolas TL, Leonard MB, Shane E. Chronic kidney disease and bone fracture: a growing concern. Kidney Int. 2008;74:721–31.
- Cummings SR, Nevitt MC, Browner WS, Stone K, Fox KM, Ensrud KE, Cauley J, Black D, Vogt TM. Risk factors for hip fracture in white women. Study of osteoporotic fractures research group. N Engl J Med. 1995;332:767–73.
- Gerber Y, Melton 3rd LJ, Weston SA, Roger VL. Association between myocardial infarction and fractures: an emerging phenomenon. Circulation. 2011;124:297–303.
- 93. Collins TC, Ewing SK, Diem SJ, Taylor BC, Orwoll ES, Cummings SR, Strotmeyer ES, Ensrud KE. Peripheral arterial disease is associated with higher rates of hip bone loss and increased fracture risk in older men. Circulation. 2009;119:2305–12.
- Vestergaard P, Rejnmark L, Mosekilde L. Diabetes and its complications and their relationship with risk of fractures in type 1 and 2 diabetes. Calcif Tissue Int. 2009;84:45–55.
- 95. Viegas M, Costa C, Lopes A, Griz L, Medeiro MA, Bandeira F. Prevalence of osteoporosis and vertebral fractures in postmenopausal women with type 2 diabetes mellitus and their relationship with duration of the disease and chronic complications. J Diabetes Complications. 2011;25:216–21.
- Selvin E, Steffes MW, Zhu H, Matsushita K, Wagenknecht L, Pankow J, Coresh J, Brancati FL. Glycated hemoglobin, diabetes, and cardiovascular risk in nondiabetic adults. N Engl J Med. 2010;362:800–11.
- Holmberg AH, Nilsson PM, Nilsson JA, Akesson K. The association between hyperglycemia and fracture risk in middle age. A prospective, population-based study of 22,444 men and 10,902 women. J Clin Endocrinol Metab. 2008;93:815–22.
- Gagnon C, Magliano DJ, Ebeling PR, Dunstan DW, Zimmet PZ, Shaw JE, Daly RM. Association between hyperglycaemia and fracture risk in non-diabetic middle-aged and older Australians: a national, population-based prospective study (AusDiab). Osteoporos Int. 2010;21:2067–74.

Chapter 3 Fracture Risk Assessment in Diabetes

William D. Leslie and Stephen Hough

Introduction

As highlighted in earlier chapters, diabetes mellitus is associated with an increased risk for low-trauma fractures [1, 2]. In the case of Type 1 diabetes (T1DM) this is at least partially mediated through lower bone mineral density (BMD) as reflected by routine clinical measurements such as dual energy X-ray absorptiometry (DXA) [2]. The situation with Type 2 diabetes (T2DM) is clearly more complicated since BMD measurements are typically increased. In view of the differences in underlying pathophysiology and how this may mediate its effects on subsequent fractures, it is unlikely that a single approach for fracture risk assessment will be equally applicable to T1DM and T2DM. Given the preponderance of T2DM among older individuals, the segment of the population at highest risk for osteoporotic fractures, and the BMD-fracture paradox alluded to earlier, this chapter will emphasize considerations in T2DM.

In the absence of a fragility fracture, osteoporosis is diagnosed from bone mineral density (BMD) measured with dual X-ray absorptiometry (DXA). The World Health Organization operational definition of osteoporosis is a BMD that lies 2.5 standard deviations (SD) or more below the average mean value for young healthy women (*T*-score ≤ -2.5) based upon a standardized reference site (the femoral neck) and reference population (National Health and Nutrition Examination Survey [NHANES] III data for White women aged 20–29 years) [3–5]. This definition

W.D. Leslie, M.D. (🖂)

S. Hough, M.D.

B. Lecka-Czernik, J.L. Fowlkes (eds.), *Diabetic Bone Disease*, DOI 10.1007/978-3-319-16402-1_3

Department of Medicine (C5121), University of Manitoba, 409 Tache Avenue, Winnipeg, MB, Canada R2H 2A6 e-mail: bleslie@sbgh.mb.ca

Division of Endocrinology, Department of Medicine, Tygerberg (deceased) Academic Hospital, Stellenbosch University, Cape Town, Western Province, South Africa

[©] Springer International Publishing Switzerland 2016

serves as a reference standard for older adults independent of sex, ethnicity, and underlying conditions including obesity and diabetes. Despite the deceptive simplicity of a BMD-based approach to osteoporosis management, many studies show that most fractures occur in individuals who have a BMD *T*-score above the defining cutoff for osteoporosis [6–9]. This suboptimal performance of BMD alone for fracture prediction has led to the development of new risk prediction algorithms that estimate fracture probability by integrating the effects of multiple risk factors for fracture.

Fracture Risk Assessment in the General Population

At the present time, no fracture prediction tool has been developed for the diabetes population. This is not unique to diabetes, however, and is a limitation for risk assessment in other complex conditions such as organ transplantation, chronic kidney disease, and glucocorticoid-induced osteoporosis. Therefore, we start with a review of screening and fracture risk assessment systems developed for the general population before considering their applicability to patients with diabetes. Broadly speaking, there are tools to identify individuals with osteoporotic BMD, and tools that integrate BMD with other clinical risk factors to identify individuals at high risk for fracture in order to address the suboptimal performance of BMD alone [6-9]. These tools were systematically reviewed by Rubin et al. [10] and therefore we will focus on those that had the highest levels of evidence for their use and at least one independent assessment. Of a total of 48 tools, only 6 had been tested more than once in a population-based setting with acceptable methodology (defined a Quality Assessment Tool for Diagnostic Accuracy Studies [QUADAS] score above 60 % [11]). Modelbased fracture prediction algorithms include: the World Health Organization FRAX tool, the Garvan Fracture Risk Calculator, and the OResearch Database's OFracture. The basic components of these tools are summarized in Table 3.1. Discrimination (the model's ability to distinguish between individuals who do or do not experience the event of interest) and calibration (agreement between observed and predicted event rates for groups of individuals) are key performance aspects in risk prediction. Procedures for development and validation of fracture prediction models are reviewed elsewhere [12–16]. There was no consistent evidence that more complex tools had better performance characteristics than simpler tools, however, the paucity of head-to-head comparisons limits any definitive conclusions. Larger, high-quality studies with different case mixes should address this important question.

Simple Screening Tools

Tools developed to identify individuals with low BMD (e.g., SCORE, OST, ORAI [17–19]) do not provide a direct estimate of fracture probability, though some of these have also been shown to stratify fracture risk [20–23]. SCORE, ORAI, and

Screening tool, URL	Risk factors	Outputs	
SCORE (Simple Calculated Risk Estimation Score) [19]	• Age, weight, previous fracture, estrogen use, rheumatoid arthritis, race	• Risk score for femoral neck <i>T</i> -scores ≤ -2.0	
OST (Osteoporosis Self- Assessment Tool) [18]	• Age, weight • Risk score for femoral neck T -scores ≤ -2.5		
ORAI (Osteoporosis Risk Assessment Instrument) [17]	• Age, weight, current estrogen use	 Risk score for femoral neck <i>T</i>-scores ≤ -2.0 	
Prediction tool, URL	Risk factors	Outputs	
FRAX (Fracture Risk Assessment Tool) [27], www.shef.ac.uk/FRAX	• Age, sex, BMI	10 year major osteoporotic fracture (clinical vertebrae, hip, forearm, proximal humerus)	
	 Prior fragility fracture, glucocorticoid use ≥3 months, secondary osteoporosis, rheumatoid arthritis, parental hip fracture, current cigarette smoking, alcohol intake of ≥3 units/day (yes/no) Femoral neck BMD or <i>T</i>-score (optional) 	• 10 year hip fracture	
Garvan Fracture Risk Calculator (Dubbo nomogram) [40, 41], www.garvan.org.au/ bone-fracture-risk	• Age, sex	 5 or 10 year any osteoporotic fracture (hip, clinical vertebrae, wrist, metacarpal, humerus, scapula, clavicle, distal femur, proximal tibia, patella, pelvis, and sternum) 	
	 Number of fractures after age 50 (none, 0, 1, 2, ≥3) Number of falls in the previous 12 months (none, 0, 1, 2, ≥3) Femoral neck BMD (or <i>T</i>-score) or weight (if BMD not entered) 	• 5- or 10-year hip fracture	

 Table 3.1
 Screening and fracture risk prediction tools for the general population

(continued)

Screening tool, URL	Risk factors	Outputs
QFracture-2013 [43, 44], www.qfracture.org	• Age, sex, 10 ethnic origins	• 1–10 year osteoporotic fracture (clinical spine, hip, distal forearm, humerus fracture)
	• Height, weight	• 1–10-year hip
	 Smoking (4 levels), alcohol intake (5 levels), diabetes (type 1, type 2), previous fracture, parental osteoporosis or hip fracture, living in a nursing or care home, history of falls, dementia, cancer, asthma/ COPD, cardiovascular disease, chronic liver disease, chronic kidney disease, Parkinson's disease, rheumatoid arthritis/SLE, malabsorption, endocrine problems, epilepsy or anticonvulsant use, antidepressant use, steroid use, HRT use 	fracture

Table 3.1 (continued)

BMI body mass index, *BMD* bone mineral density, *HRT* hormone replacement therapy, *COPD* chronic obstructive pulmonary disease

OST have been validated as screening tools for BMD testing, and outperform a simple body weight criterion [24, 25]. In the review of Rubin et al. [10], none of the tools performed consistently better than others when tested in external validation studies, and simple tools with fewer risk factors (i.e., OST, ORAI) did as well as more complex tools with more risk factors (i.e., SCORE).

FRAX (www.shef.ac.uk/FRAX)

FRAX was developed by the WHO Collaborating Centre for Metabolic Bone Diseases to estimate an individual's 10-year probability of major osteoporotic fracture (MOF, composite of clinical spine, hip, forearm, proximal humerus) and hip fracture [26]. The input variables were selected following a series of meta-analyses using data from nine prospective international population-based cohorts [27]. In addition to age, sex, and body mass index (BMI), additional clinical risk factors (CRFs) for fractures include prior fragility fracture, a parental history of hip fracture, prolonged use of glucocorticoids, rheumatoid arthritis, current cigarette



Fig. 3.1 Sample screenshot for FRAX[®] (US Caucasian tool). Ten-year probability for major osteoporotic fracture is 18 % and for hip fracture is 2.9 % in a woman age 65 years, weight 82 kg, height 165 cm, previous fracture, and femoral neck *T*-score -2.1. (Note that more than one fracture, type 2 diabetes or fall in the prior year does not affect the calculation)

smoking, alcohol intake of three or more units/day, and secondary osteoporosis (Fig. 3.1). Femoral neck BMD is an optional input that can refine the risk estimate, though even in its absence FRAX performs very well [28, 29]. Interactions among CRFs are also incorporated into the FRAX algorithm. More recently, specific adjustments were developed that can be applied to FRAX-derived risk scores to accommodate discordantly lower or higher lumbar spine BMD (more than 1 SD difference from femoral neck BMD) or glucocorticoid doses that are above or below average (average use defined as daily 2.5–7.5 mg prednisone-equivalent) [30, 31].

In survival analysis, the time at which a subject experiences an event of interest may be altered by another event, known as competing risk events [32]. For fracture, competing death is particularly important to consider in order to produce unbiased estimates of fracture risk since, following death, fracture is no longer possible. FRAX adjusts for competing mortality, and this is unique among the risk prediction models. Individuals may have equivalent hazards for fracture but if they differ in terms of hazard for death then this will affect the 10-year fracture probability. For example, smoking is a risk factor for fracture but also increases the risk for death. Thus, the increased mortality associated with smoking reduces the importance of smoking as a risk factor for fracture. Ten-year major fracture probability tends to increase with age to peak around 80–85 years and then declines as the death hazard rises faster than the fracture hazard (Table 3.2). Failure to account for competing mortality has been shown to overestimate major fracture probability by 15–56 % and hip fracture probability by 17–36 % in those with high mortality [33].

Table 3.2 Relative change in fracture probability versus no other risk factors from type 1 diabetes,type 2 diabetes, and inflammatory arthropathy (rheumatoid arthritis or SLE) with QFracture®-2013or from rheumatoid arthritis with FRAX

Age	40	50	60	70	80	90	Mean
	Women-major osteoporotic fractures						
QFracture-type 1 diabetes	1.88	1.88	1.91	1.86	2.25	2.74	2.08
QFracture-type 2 diabetes	1.25	1.25	1.28	1.25	1.25	1.24	1.25
QFracture-rheumatoid arthritis or SLE	1.25	1.31	1.34	1.32	1.31	1.30	1.30
FRAX-rheumatoid arthritis (no BMD)	1.35	1.35	1.37	1.36	1.41	1.45	1.38
FRAX—rheumatoid arthritis $(T$ -score $-2.5)$	1.33	1.32	1.35	1.29	1.29	1.29	1.31
	Men-	major o	osteopor	otic fra	ctures		
QFracture-type 1 diabetes	2.20	2.33	2.30	2.61	3.38	3.34	2.69
QFracture-type 2 diabetes	1.20	1.33	1.20	1.28	1.23	1.24	1.25
QFracture-rheumatoid arthritis or SLE	1.40	1.50	1.50	1.56	1.54	1.52	1.50
FRAX-rheumatoid arthritis (no BMD)	1.32	1.33	1.39	1.41	1.53	1.58	1.43
FRAX-rheumatoid arthritis	1.32	1.34	1.29	1.30	1.29	1.33	1.31
(<i>T</i> -score –2.5)							
	Women—hip fracture						
QFracture-type 1 diabetes			4.83	4.36	4.13	3.73	4.27
QFracture-type 2 diabetes			1.67	1.55	1.53	1.51	1.56
QFracture-rheumatoid arthritis or SLE			1.83	1.64	1.65	1.61	1.68
FRAX-rheumatoid arthritis (no BMD)			1.71	1.74	1.74	1.67	1.71
FRAX — rheumatoid arthritis $(T$ -score $-2.5)$			1.42	1.39	1.41	1.39	1.40
	Men-	hip frac	ture				
QFracture-type 1 diabetes			5.33	4.70	4.55	4.24	4.71
QFracture-type 2 diabetes			1.33	1.30	1.31	1.33	1.32
QFracture-rheumatoid arthritis or SLE			2.00	1.90	1.86	1.86	1.90
FRAX-rheumatoid arthritis (no BMD)			2.00	1.69	1.76	1.77	1.81
FRAX—rheumatoid arthritis (<i>T</i> -score –2.5)			1.41	1.42	1.40	1.41	1.41

In recognition of the large international variability in fracture and mortality rates [34], population-specific FRAX tools are customized to the fracture and mortality epidemiology in a specific region, with the most recent version containing over 50 countries [26]. Minimum data requirements for constructing a new FRAX tool are sex and age-specific mortality and hip fracture rates (5 year subgroups). In many countries, such data are relatively easy to obtain. In contrast, non-hip fracture data considered by FRAX (clinical spine, distal forearm, proximal humerus) are difficult to accurately collect at the population level. Where high-quality data are not available, country-specific FRAX tools can be calibrated under the assumption that the ratio of these non-hip to hip fracture rates is similar to that observed in the Swedish population [35, 36]. Whether these ratios are applicable in all populations has been questioned, but currently they provide a valuable reference standard [37].

Fracture discrimination with FRAX was initially assessed in 9 primary derivation cohorts (46,340 subjects with 189,852 person years follow-up) and then in 11 additional validation cohorts (230,486 persons with 1,208,528 person years of follow-up) [38]. Risk stratification with FRAX including BMD was superior to FRAX without BMD or to BMD alone. In the primary derivation cohorts, the gradient of risk for hip fracture increased from 1.84 to 2.91 (area under the curve [AUC] from 0.67 to 0.78) with the inclusion of BMD, and for MOF increased from 1.55 to 1.61 (AUC from 0.62 to 0.63) with the inclusion of BMD. In the validation cohorts, the averaged hip fracture gradient of risk (1.83 without BMD and 2.52 with BMD) and AUC (0.66 without BMD and 0.74 with BMD) was similar to that of the derivation cohorts but the gradient of risk for other osteoporotic fractures (1.53 without BMD, 1.57 with BMD) and AUC (0.60 without BMD and 0.62 with BMD) was slightly lower.

A large number of studies have performed independent assessments of FRAX to predict subsequent fracture [16], but studies differ widely in their sample size, methodology (particularly incorporation of competing mortality risk) and techniques used to assess the performance of the fracture prediction tool (discrimination versus calibration), which can affect the validity of these validation studies [39]. As a general rule, these studies confirm the validity of FRAX for fracture risk assessment but highlight the importance of using high-quality fracture data to ensure accurate calibration of the FRAX tool.

Garvan Fracture Risk Calculator (www.garvan.org.au/bone-fracture-risk)

The Dubbo Osteoporosis Epidemiology Study (DOES) was initiated in 1989 and involves follow-up of over 3500 participants. Using information on 426 clinical fractures in women (96 hip) and 149 clinical fractures in men (31 hip) excluding digits, 5- and 10-year fracture probability nomograms were constructed [40, 41]. Inputs include age, sex, femoral neck BMD (optional), history of prior fractures after age 50 years (none, 0, 1, 2, 3, or more) and history of falls in the previous 12 months (none, 0, 1, 2, 3, or more) (Fig. 3.2). If femoral neck BMD is not available, then weight is used as a proxy. Risk factors that are relatively uncommon in the general population (e.g., glucocorticoid use and specific medical conditions) are not included. The model has only been calibrated for the Australian population and does not include an explicit competing mortality risk adjustment. However, the algorithm has been independently validated in one North American population of women and men [42].

QFracture (www.qfracture.org)

The largest prospective database for osteoporotic fracture prediction is from England and Wales using patients from 357 general practices for derivation and patients from 178 practices for validation in the initial analysis (QResearch Database) [43]. This

INSTITUTE	
FRACTURE RISK CALCULATOR	
Fill out the following to estimate vo	ur fracture rick
Full Name	
(optional) Sex?	0.111
Contraction of the second s	Female
Age	65
Fractures since the age of 50	2
Falls over last 12 months	1
Do you have a	Yes
Bone Mineral Density (BMD) measurement?	© No
T-scores	-2.1
	OR
Densitometer	 by DXA GE Lunar by DXA Hologic
Actual BMD	g/cm²

Fig. 3.2 Sample screenshot for Garvan fracture risk calculator. Ten-year probability for major osteoporotic fracture is 53.7 % and for hip fracture is 28.4 % in a woman age 65 years, two previous fractures, one previous fall, and femoral neck *T*-score -2.1. (Note that weight 82 kg, height 165 cm, and type 2 diabetes does not affect the calculation)

provided more than one million women and more than one million men age 30–85 years in the derivation cohort with 24,350 incident osteoporotic fractures in women (9302 hip fractures) and 7934 osteoporotic fractures in men (5424 hip fractures). The risk calculator includes numerous CRFs, but not BMD (Fig. 3.3). It provides outputs of any osteoporotic fracture (hip, wrist, or spine) and hip fracture over a user selected follow-up period from 1 year to 10 years. The QFracture algorithm was updated in 2012, with inclusion of a number of new CRFs, removal of several others, and the addition of humerus fractures as one of the osteoporotic fractures [44]. In addition to age, sex, and ethnicity (10 different ethnic origins) the algorithm includes smoking status (4 levels), alcohol consumption (5 levels), diabetes (2 levels, T1DM or T2DM), previous fracture, parental osteoporosis or hip fracture, living in a nursing or care home, history of falls, dementia, cancer, asthma/COPD,

3 Fracture Risk Assessment in Diabetes



Fig. 3.3 Sample screenshot for QFracture[®]-2013 risk calculator. Ten-year probability for major osteoporotic fracture is 7.9 % and for hip fracture is 3.7 % in a White woman age 65 years, weight 82 kg, height 165 cm, type 2 diabetes, previous fracture, history of falls. (Note that more than one fracture, and femoral neck *T*-score does not affect the calculation)

cardiovascular disease, chronic liver disease, chronic kidney disease, Parkinson's disease, rheumatoid arthritis/SLE, malabsorption, endocrine problems, epilepsy or anticonvulsant use, antidepressant use, steroid use, HRT use, height, and weight.

The 2012 version reported very good performance for osteoporotic fracture prediction (AUC 0.79 in women and 0.71 in men) and excellent performance for hip fracture prediction (AUC 0.89 in women and 0.88 in men). An independent validation study was performed using patients from 364 general practices in the UK Health Improvement Network (THIN) database (2.2 million adults aged 30–85 years with 25,208 osteoporotic and 12,188 hip fractures) [45]. The validation cohort gave AUC discrimination for osteoporotic fracture of 0.82 in women and 0.74 in men, and for hip fracture of 0.89 in women and 0.86 in men. Calibration plots adhered closely to the line of identity. QFracture explained 63 % of the variation in hip fracture risk in women and 60 % of the variation in men (49 and 38 % for osteoporotic fracture risk).

A small retrospective comparison of FRAX and QFracture was conducted in 246 postmenopausal women aged 50–85 years with recent low trauma fracture and 338 nonfracture control women from six centers in Ireland and the UK [46]. AUC for fracture discrimination were similar in QFracture and FRAX (0.668 versus 0.665) and also for hip fractures (0.637 versus 0.710). The striking difference with AUC

measures from the THIN database validation study is unexplained. The broad age range used in the initial validation work may have inflated the performance measures since osteoporotic fractures are unlikely before age 50, and additional assessments of QFracture in older women and men are needed.

Risk Factors for Fracture in Diabetes

General Risk Factors

Individual risk factors for fracture in individuals with diabetes can be divided into those that are applicable to the general population and those that are specific to diabetes. Most of the risk factors for osteoporotic fractures that apply to the general population, particularly those that are included in the most common risk assessment tools (Table 3.1), have been shown to predict fracture in individuals with diabetes [47]. For example, older age, lower BMI and previous osteoporotic fracture help to identify patients with diabetes who are at greater fracture risk [47]. BMD measurement from DXA provides a robust estimate of fracture risk, increasing 1.4-2.6 fold for every SD decrease in BMD in the general population [48, 49] and in those with diabetes [47]. Two reports from the population-based clinical BMD repository for Manitoba, Canada, have examined whether individual risk factors behave the same in those with and without diabetes, or whether there is effect moderation (interaction) [47, 50]. For MOF, no significant effect moderation was observed for FRAX clinical risk factors. Prior fracture was examined in relation to fracture site, and also behaved similarly in those with and without diabetes. Prior vertebral fracture was associated with the highest risk for subsequent MOF, while prior ankle fracture (which is common in individuals with diabetes) was not associated with an increased risk for subsequent fracture. Age was a significant effect modifier for hip fracture risk, however. Specifically, younger individuals with diabetes were at much higher risk for hip fracture than were those of similar age without diabetes and this difference narrowed with age (adjusted hazard ratio age <60 years: 4.67 [95 % CI 2.76-7.89], age 60-69 years: 2.68 [1.77-4.04], age 70-79 years: 1.57 [1.20-2.04], age >80 years: 1.42 [1.10-1.99]; P-interaction <0.001). Unfortunately, this study was unable to distinguish T1DM and T2DM which differ substantially in their pathophysiology as reviewed in previous chapters. Other studies have noted much higher risk for T1DM than T2DM which would be expected to be relatively more prevalent in younger versus older diabetic [1, 2].

Obesity

It is impossible to overlook the parallel between the BMD-fracture paradox in T2DM and in obesity, and implications for fracture risk assessment. Historically, obesity has been considered a protective factor for osteoporosis but recent

observations have called this into question based upon reports that obesity can be protective, neutral, or a risk factor for osteoporosis and related fractures [51-54]. Mechanisms whereby obesity might be associated with fractures are poorly defined but potentially include: biomechanical effects (increased skeletal loading), metabolic changes (fat-derived adipocytokines, inflammatory mediators), and statistical (failure to adequately consider variable collinearity, differential effects of lean versus fat mass) [54-57]. For some sites (particularly the upper arm), obesity appears to be a risk factor for some fractures even after adjustment for higher BMD [52, 58]. A recent meta-analysis examined the association between BMI and fracture risk in prospective cohorts from 25 countries (398,610 women average age 63 years, 2.2 million person-years follow-up) with a reported 22 % prevalence of obesity [58]. Risk for MOF and hip fractures decreased monotonically with increasing BMI when the modifying effect of BMD was excluded. When BMD was considered, the effect of BMI was greatly reduced and, at least for MOF, showed a biphasic response such that fracture risk decreased as BMI increased to the normal range and then slightly increased for overweight and obese individuals. Obese women were at lower risk for MOF when BMD was not considered (hazard ratio 0.87 for BMI 35 kg/m² versus 25 kg/m²) but at higher risk when BMD was considered (HR 1.16). Obesity was still protective against hip fractures, even when adjusted for higher BMD, possibly reflecting energy absorption from soft tissue padding [59].

However, weight associations become more complicated when separated into their lean mass and fat mass components. BMI does not distinguish the very different skeletal effects from lean versus fat tissue where significant nonlinearities have been observed. Most studies find that lean mass is positively associated with higher BMD, whereas fat mass shows neutral or negative effects though there are exceptions to this generalization and a dual (biphasic) effect has even been reported [57, 60]. A meta-analysis found that among premenopausal women lean mass exerted a greater effect on femoral neck BMD than fat mass (r=0.45 versus r=0.30), whereas in postmenopausal women the effects of lean mass and fat mass were similar (r=0.33 versus r=0.31) [57]. A recent population-based study found that the effect of BMI on fracture risk was largely mediated by its effect on BMD, with little if any direct effect of BMI on fracture risk [61]. This awaits confirmation in additional larger studies where the mediating effects of other body composition variables can be studied.

Biomechanical models have not yet been incorporated into fracture prediction tools. Based upon engineering principles, such as beam theory, these have the potential to incorporate the opposing effects of greater BMD (protective) versus increased skeletal loading from falls (risk factor) that occur during specific fall configurations (boundary conditions). Melton et al. [62] compared estimated bone load to bone strength ratios in 49 T2DM patients to age and sex-matched controls. Hip BMD from DXA was greater in diabetic subjects due to greater trabecular volumetric BMD (vBMD) while cortical vBMD was similar. Derived bone strength measures were generally better in diabetic subjects, but bone loads were higher from

their greater weight, such that load to strength ratios (i.e., factor-of-risk) were similar. A subsequent analysis of older men with (n=190) and without (n=981)T2DM found that although patients with T2DM had lower bone bending strength at the mostly cortical bone midshaft sites of the radius and tibia after adjusting for body weight (-2 to -5%, p < 0.05) despite the lack of difference in cortical vBMD at these sites. Indices of femoral neck strength in diabetic versus nondiabetic women (adjusted for age, race/ethnicity, menopausal stage, body mass index, smoking, physical activity, calcium and vitamin D supplementation, and study site) showed lower composite strength indices for compression, bending, or impact despite higher femoral neck BMD [63]. Furthermore, there was an inverse relationship between insulin resistance (homeostasis model) and all three strength indices. These studies suggest that patients with T2DM do not benefit from elevated BMD in terms of improved bone load to strength ratios. However, these predictions may not always align with clinical observations. In a large registry-based cohort (40,050 women and 3600 men), the femoral strength index (FSI) showed a progressive decline with increasing fat mass index but did not improve fracture prediction over conventional FRAX probability measurements [56]. The FSI does not consider soft tissue padding effects, which may significantly attenuate the fracture risk relationship [59]. Additional work is needed to evaluate these indices derived from biomechanical principles in individuals with obesity and diabetes.

Importantly, some measure of body size (weight or body mass index [BMI]) is included in each of the systems summarized in Table 3.1. Given that obesity strongly predisposes to T2DM, one would anticipate that these tools would be insensitive to the effect of T2DM and would underestimate risk. A demonstration of the differing approaches to including BMI (or weight equivalent) on fracture risk assessment is illustrated in Fig. 3.4. When BMD is not included, FRAX predicts a linear decrease in fracture probability with increasing BMI. A similar relationship is seen with the Garvan FRC (which uses weight when BMD is not available) and OFracture risk (which does not have a BMD input). When BMD is included in the FRAX calculation, BMI has a very different relationship such that fracture probability increases with greater BMI up to 25 kg/m², and subsequently declines. The lower fracture probability for individuals with BMI much less than 25 kg/m² likely relates to the effect of competing mortality since underweight is associated with a higher likelihood of death from comorbid conditions. The protective effect of BMI in the overweight and obese range that is independent of BMD may reflect soft tissue padding [59].

Together, the foregoing observations challenge conventional concepts of weight as a risk factor for osteoporosis and fractures, and beg the question of why something that is associated with preserved BMD would be a risk factor for fracture. Potential explanations include greater falls risk, adverse changes on skeletal biomechanics, alterations in energy homeostasis, inflammation, insulin resistance, overt T2DM or their associated treatments and comorbidities are potential contributors [52, 64]. Despite these questions, lean and fat tissue mass indices did not have an independent effect on MOF or hip fracture risk when adjusted for FRAX scores [56].



Fig. 3.4 Effect of increasing body mass index (BMI) or weight-equivalent (height 165 cm) on fracture predictions from US White FRAX (with and without BMD), Garvan FRC (without BMD), and QFracture[®]-2013 (without BMD) for a 65-year-old woman with *T*-score –3.0 and no other risk factors. (a) Non-hip fractures (MOF for FRAX and QFracture[®]-2013, any fracture for Garvan FRC). (b) Hip fractures

Of potential relevance to T2DM, Premaor et al. [65] examined the question of whether FRAX was applicable to obese older women using 6049 white women from the US Study of Osteoporotic Fractures (SOF) cohort. Fracture discrimination from AUC was similar in obese and nonobese women. Calibration was good in both groups for prediction of MOF using FRAX with BMD, but hip fracture risk was found to be underestimated, most markedly among obese women in the lowest category for FRAX probability with BMD (though based on only four predicted versus nine observed hip fractures).

Diabetes-Specific Risk Factors

It is logical to consider whether factors specific to diabetes should be included in the risk assessment since these may mediate some of the excess fracture risk. This could include increased risk for falls and fractures related to diabetic complications (including retinopathy, neuropathy), hypoglycemic episodes, or the adverse effect of some medications (most specifically the thiazolidinediones) [66, 67]. Indeed, there are population-based data that duration of diabetes, which correlates with long-term complications, is associated with increased fracture risk [68]. Use of insulin (HR 1.3) and neuropathy (HR 1.3) are independent risk factors for fracture in T2DM [69]. The effect of insulin may not be causally related to the higher fracture risk as patients on insulin tend to have more severe dysglycemia and are at higher risk for complications and falls, which have not been incorporated into risk assessment tools for individuals with diabetes. In part, this relates to the requirements for large cohorts for the derivation of the appropriate weight to be accorded the risk factor and verification that it improves fracture discrimination and calibration. The challenge of developing a diabetes-specific risk assessment tool becomes moot if it does not enter widespread clinical practice. Primary care physicians are already overwhelmed with the number of clinical practice guidelines and risk assessment tools, and the availability of more than one fracture risk assessment tool may generate confusion and inactivity rather than an improvement in care. Therefore, methods that can be incorporated into existing risk assessment tools with the minimum of effort on the part of practicing physicians are likely to be more successful in terms of enhanced clinical care.

Risk Prediction in Diabetes

Simple Screening Tools

To date, none of the BMD screening tools referred to above (SCORE, OST, ORAI [17–19]) have been specifically evaluated in diabetes. Moreover, they each include weight in the calculation given its strong correlation with BMD. Therefore, it is very unlikely that these tools would be satisfactory for use in individuals with T2DM where increased fracture risk is not mediated through lower BMD or lower weight. It is possible that these tools could be modified to incorporate T1DM and/or T2DM with a numeric "penalty" included in the score, but these would need to be worked out.

Multifactorial Models for Absolute Fracture Prediction

As noted earlier, FRAX does not include a diabetes input variable. T1DM is considered as one of the secondary osteoporosis causes and only affects fracture probability when BMD is not included in the calculation [27]. When BMD is available for the FRAX calculation, the secondary osteoporosis variable does not affect the fracture probability calculation since it is assumed that these conditions mediate their effect through lower BMD. All secondary osteoporosis conditions are assigned the same weight in the FRAX calculation (equal to rheumatoid arthritis) which may not be applicable to T1DM, particularly among younger individuals.

The predictive performance of FRAX in patients with diabetes has been evaluated in two large studies. Schwartz et al. [70] combined data in older communitydwelling adults (9449 women and 7436 men) from three prospective observational studies with adjudicated fracture outcomes (Study of Osteoporotic Fractures; Osteoporotic Fractures in Men Study; and Health, Aging, and Body Composition study). Of 770 women with T2DM, 84 experienced a hip fracture and 262 a nonspine fracture; among 1199 men with T2DM, 32 experienced a hip fracture and 133 a nonspine fracture. Although both femoral neck BMD and FRAX score were associated with fracture risk in participants with DM, for a given T-score and age or for a given FRAX score, participants with T2DM had a higher fracture risk than those without diabetes. Similar results were seen in a large BMD registry analysis from Manitoba, Canada [71]. Among 3518 men and women with diabetes and 36,085 nondiabetics aged >50 years at the time of BMD, diabetes was a significant risk factor for subsequent MOF (hazard ratio [HR]=1.61, 95 % confidence interval [CI] 1.42–1.83) after controlling for age, sex, medication use, and FRAX risk factors including BMD. Similar results were seen after adjusting for FRAX probability directly (HR = 1.59, 95 % CI 1.40-1.79). Diabetes was also associated with significantly higher risk for hip fractures (p < 0.001). FRAX underestimated observed major osteoporotic and hip fracture risk in diabetics (adjusted for competing mortality) but demonstrated good concordance with observed fractures for nondiabetics. An even larger discrepancy between observed and predict fracture risk was seen if competing mortality was not considered. Smaller cross-sectional studies have shown similar findings [72, 73]. A multicenter cross-sectional study found that 974 patients with T2DM had mean FRAX scores lower than 777 control subjects despite a greater number of previous fractures [72]. Of interest, HbA1c and hypoglycemia were significantly associated with higher FRAX scores independent of other covariates. Together, these studies indicate that FRAX systematically underestimates fracture risk in patients with T2DM despite the competing effect of increased mortality associated with diabetes.

The Garvan FRC has not been evaluated in relation to diabetes. Based upon its construction, it is likely to underestimate fracture risk in T2DM for the same reasons that affect FRAX. This discrepancy would be amplified by not considering competing mortality, but potentially attenuated by considering falls which is one of the mediating factors between diabetes and fracture. Therefore, it remains important to directly assess how the Garvan FRC performs in patients with T2DM.

Only the QFracture tool has direct inputs for diabetes, stratified as T1DM and T2DM. The magnitude of the adjustment is illustrated in Fig. 3.4. For T1DM, the average effect (as a ratio versus an individual with no other risk factors) across all ages and sexes is 2.37 for MOF and 4.46 for hip fracture in T1DM, 1.25 for MOF, and 1.45 for hip fracture in T2DM. The effect of T2DM (compared with an individual

with no other risk factors) is relatively constant across the age spectrum of women and men. The relative effect of T1DM on MOF increases with age (from 1.88 to 2.74 in women, from 2.20 to 3.34 in men) while the effect on hip fracture risk shows little variation with age or sex. Although QFracture appears to be well calibrated for the general population, it has not been specifically evaluated with regard to calibration in individuals with diabetes nor does it explicitly consider competing mortality.

Comparative Performance

Figures 3.1, 3.2, and 3.3 illustrate some of the differences between the fracture prediction systems discussed above for a 65-year-old obese woman (BMI 30.1 kg/m²) with femoral neck *T*-score -2.1 and having the following CRFs: two previous fragility fractures (distal radius and humerus), T2DM and one fall in the prior year. None of the systems uses all of the available information and this contributes to the range in fracture predictions. For example, 10-year hip fracture probability varies from 2.9 % for FRAX, to 3.7 % for QFracture-2013, to 21.2 % for the Garvan calculator (18 % MOF, 7.9 % MOF, and 48.1 % any fracture, respectively). It is clear that where an important CRF is not captured by a tool, clinical judgment must be used to make some estimate of the importance of the missing information.

Identification of High Risk in Diabetes

Clinical Practice Guidelines

Validated prognostic models for fracture risk assessment can guide clinicians and individuals in understanding the risk of suffering an osteoporosis-related fracture and inform their decision making to mitigate this risk. Integration of risk prediction tools into clinical practice guidelines can in turn support better clinical decision making and improved patient outcomes.

The lack of significant interaction (effect modification) means that conventional risk factors, including BMD and BMI, remain valid in the population with diabetes [47, 50]. For example, all other factors being equal, a 70-year-old woman with T2DM and BMD *T*-score -2.5 or BMI 22 kg/m² is at higher risk than a woman with the same risk profile who has BMD *T*-score -1.0 or BMI 28 kg/m². Based upon the expectation and empirical observations that fracture prediction tools underestimate fracture risk in diabetes (particularly T2DM), it is reasonable to assume that an individual with diabetes who meets the high risk (intervention) threshold for the general population should also be considered at high risk. Similarly, since prior fracture and low BMD predict fractures in individuals with diabetes as well as in individuals without diabetes (no effect modification), these can also be used as

indicators of high risk (intervention). For example, under the National Osteoporosis Foundation (NOF) guidelines, prior low-trauma fracture (vertebral or hip), osteoporotic BMD (*T*-score -2.5 or lower), or high FRAX probability for low bone mass (osteopenic) individuals (MOF 20 % or greater, hip 3 % or greater) would be considered intervention thresholds for the general population [74]. An individual with diabetes (T1DM or T2DM) could likewise be considered for treatment using the same thresholds. Since QFracture was calibrated for use in the UK, its applicability to other populations remains uncertain. However, within the UK (or countries with similar fracture and mortality rates), QFracture may be a good tool for use in individuals with T1DM or T2DM although this tool does not incorporate BMD in the calculation.

Potential Risk Adjustments for Diabetes

Although appealing, whether it is possible to develop simple adjustments to FRAX or the Garvan FRC that would accommodate the risk attributable to diabetes is uncertain. Several options on tempering clinical judgment with the existing output of the FRAX models have been proposed [75]. In the context of a specific FRAXbased intervention thresholds (e.g., NOF-based 20 % probability of MOF or 3 % probability of hip fracture [74]), a patient with diabetes below but close to the intervention threshold such as the one in Fig. 3.2 (MOF probability 18 %, hip fracture probability 2.9 %) could be recommended for treatment. (The upward revision of a patient with a fracture probability already exceeding the treatment threshold has little clinical utility in making a decision to treat.) A second measure might be to model the potential impact of diabetes on FRAX using a percentage adjustment to the probability score in much the same way that has been done when considering the effects of glucocorticoid dose or lumbar spine BMD on fracture probability [30, 31]. A third option is to use the existing input variable rheumatoid arthritis as a substitute for diabetes. As shown in Table 3.2 and Fig. 3.5, the weight attributed to rheumatoid arthritis in FRAX is similar to the weight of rheumatoid arthritis or SLE in QFracture and T2DM in QFracture, but considerably less than the effect of T1DM in QFracture. Therefore, this simple adjustment would not adequately address underestimation in fracture risk for T1DM. The prevalence of individuals with both rheumatoid arthritis and diabetes in the population is quite low, therefore the rheumatoid arthritis input to FRAX is usually available for individuals with diabetes.

Whether a similar approach would work for the Garvan FRC has not yet been evaluated. It is conceivable that either the fracture input or the falls input (which can accommodate 0, 1, 2, or 3+ events) could be used by incrementing the input that would be used in the absence of diabetes. For example, in an individual with no prior fractures and one fall in the last year, incrementing the falls input to 2 for T2DM and to 3+ for T1DM could conceptually be considered but requires careful study to determine the need and accuracy of such an adjustment.



Fig. 3.5 Comparative effects of type 1 diabetes, type 2 diabetes, and inflammatory arthropathy (rheumatoid arthritis or SLE) versus no other risk factors on 10-year probability for major osteo-porotic fracture (*upper panel*) and hip fracture (*lower panel*) with QFracture[®]-2013

Lumbar Spine Trabecular Bone Score

The trabecular bone score (TBS) is a novel texture parameter that evaluates pixel gray-level variations in the spine DXA image and is related to bone microarchitecture and fracture risk, providing information independent of BMD [76–79]. TBS is not a direct physical measurement of bone microarchitecture, but rather an overall score computed by the projection of the 3D structure onto a 2D plane [77]. TBS is calculated as the slope of the log-log transform of the 2D variogram from the projected DXA image (calculated as the sum of the squared gray-level differences between pixels at a specific distance). While the TBS result is given for each vertebra, the usually reported TBS value represents the average of L1–L4. TBS has the potential to discern differences between DXA scans that show similar BMD measurements. A recent review by Silva et al. [80] concluded that: (1) TBS gives lower values in postmenopausal women and in men with previous fragility fractures than their nonfractured counterparts; (2) TBS is complementary to data available by

lumbar spine DXA measurements; (3) TBS results are lower in women who have sustained a fragility fracture but in whom DXA does not indicate osteoporosis or even osteopenia; (4) TBS predicts fracture risk as well as lumbar spine BMD measurements in postmenopausal women; (5) efficacious therapies for osteoporosis differ in the extent to which they influence the TBS; (6) TBS is associated with fracture risk in individuals with conditions related to reduced bone mass or bone quality.

Although there is accumulating evidence that TBS is able to discriminate between fracture and nonfracture subjects and predict future fracture risk in the general population [80], it is less certain whether TBS is useful in specific conditions that modify fracture risk. Several studies have evaluated TBS in individuals with conditions or diseases related to increased fracture risk, and TBS was associated with fragility fracture in subjects with diabetes, rheumatoid arthritis, primary hyperparathyroidism, adrenal incidentaloma, individuals on long-term glucocorticoid therapy and chronic kidney disease [80].

To date, four published studies have evaluated the ability of lumbar spine TBS to account for the increased fracture risk in diabetes. A retrospective cohort study using BMD results (Prodigy, GE HealthCare) from a large clinical registry for the province of Manitoba, Canada [81]. This included 29,407 women >50 years with baseline DXA examinations, among whom 2356 had diagnosed diabetes. Diabetes was associated with higher BMD at all sites but lower lumbar spine TBS in unadjusted and adjusted models (all p < 0.001). The adjusted odds ratio (aOR) for a measurement in the lowest versus highest tertile was less than 1 for BMD (all p < 0.001) but was increased for lumbar spine TBS (aOR 2.61, 95 % CI 2.30-2.97). During mean 4.7 years of observation, MOF were identified in 175 women (7.4 %) with and 1493 (5.5 %) without diabetes (p < 0.001). Lumbar spine TBS was a BMDindependent predictor of fracture, and predicted fractures in those with diabetes (adjusted HR 1.27, 95 % CI 1.10-1.46) and without diabetes (HR 1.31, 95 % CI 1.24–1.38). The effect of diabetes on fracture was reduced when lumbar spine TBS was added to a regression model, but was paradoxically increased from adding BMD measurements.

Similar effects of T2DM were seen in a subsequent retrospective analysis of 56 postmenopausal women with T2DM and 61 women without DM or impaired glucose tolerance [67] using equipment from the second major DXA manufacturer (Hologic). T2DM was associated with higher BMD at all sites but lower lumbar spine TBS (all p < 0.05). The adjusted odds ratio (aOR) for a measurement in the lowest versus the highest tertile was less than 1 for BMD (p < 0.05) but was increased for lumbar spine TBS (aOR 2.39, 95 % CI 2.22–2.81). In a small cross-sectional study, Dhaliwal et al. [82] examined lumbar spine BMD (iDXA, GE HealthCare) and TBS in 57 women with T2DM and 43 women without diabetes, ages 30–90 years. Mean TBS was lower in T2DM (p=0.013), irrespective of age, while mean BMD was higher (p=0.001). Within the T2DM group, TBS was higher in subjects with good glycemic control (A1c ≤ 7.5 %) compared to those with poor glycemic control (A1c ≥ 7.5 %). Rubin et al. [83] reported the association of TBS with trabecular heterogeneity by high resolution pQCT (HRpQCT) and biochemical markers of bone turnover in 14 postmenopausal T2DM women and 21 age-matched
controls. Despite similar lumbar spine BMD, TBS was significantly lower in T2DM versus controls (p=0.01). Consistent with this, trabecular heterogeneity (SD of 1/Tb Number) at the tibia was greater in T2DM (p=0.05). The authors concluded that while BMD does not differ from controls in T2DM, TBS is worse and is consistent with the greater trabecular heterogeneity seen on HRpQCT.

Notwithstanding these promising reports, some notes of caution are warranted. Spine TBS is (negatively) affected by obesity and tissue thickness [84], though recent changes to the algorithm have greatly attenuated this effect [85]. Increased soft tissue thickness may have the same effect on TBS as noise, i.e., lower TBS. The TBS algorithm includes an adjustment for BMI that is optimized for BMI ranges from 15 to 35 kg/m², so that the assessment of TBS may not be as reliable in subjects with a BMI beyond these limits. The use of BMI as a proxy for soft tissue thickness is limited since it can overestimate adiposity in subjects with a high lean body mass, and underestimate adiposity in subjects with low lean body mass. Furthermore, higher BMI does not distinguish abdominal weight accumulation (which would directly affect spine TBS) from weight accumulation at other sites. Truncal obesity in T2DM could contribute to the lower apparent TBS in diabetes, though this does not negative the fact that TBS is still able to predict fractures in those with diabetes. While results from clinical studies have confirmed the fracturediscriminating ability of TBS in a substantial number of postmenopausal women, data in men are still preliminary with only one cross-sectional [86] and one longitudinal study of fractures in men [87]. A final limitation for the use of TBS in the clinical practice is the lack of a well-established TBS cutoff point that defines normal and abnormal TBS values, and how this would be used clinically in conjunction with other data. Methods to incorporate this information into risk assessment are still evolving, but the feasibility of a TBS-based adjustment to FRAX probability has been reported [88]. Additional work is needed to extend this preliminary observation to hip fracture prediction with validation in independent cohorts.

In summary, lumbar spine TBS predicts osteoporotic fractures in those with diabetes, and captures a larger portion of the diabetes-associated fracture risk than BMD [81]. In T2DM, TBS is lower and associated with poor glycemic control [82]. Abnormal trabecular microarchitecture may help explain the paradox of increased fractures at a higher BMD in T2DM [83]. Further studies are needed to better understand the relationship of T2DM with glycemic control and trabecular bone quality.

Emerging and Future Methods

There are many additional fracture risk predictors that could be considered in individuals with diabetes that are not currently incorporated in any of the fracture prediction systems discussed earlier. Some of these are based upon advanced imaging such as HRpQCT, such as cortical porosity, or minimally invasive measurements of tissue properties, such as microindentation [67, 89]. Such methods are likely to remain as research methods for the foreseeable future. The imaging method most likely to be helpful in routine clinical practice is lumbar spine TBS where several studies now indicate that this captures some of the skeletal effect of T2DM that is missed by conventional BMD [67, 81–83]. The use of biochemical parameters as discussed elsewhere (such as osteoblast- and osteoclast-derived bone turnover markers, advanced glycation end products such as pentosidine, measures of diabetes control, serum sclerostin, insulin-like growth factor-1), or clinical markers of diabetes-related complications (neuropathy, retinopathy, nephropathy, microvascular) represent an alternative direction based upon clinical parameters [66, 67, 90, 91]. Incorporation of falls risk into FRAX (already considered by Garvan FRC and QFracture) is another option. Meanwhile, the need for improved methods for identification of fracture risk in older adults with diabetes is an important priority for osteoporosis research.

Disclosures WDL (all fees paid to facility): Speaker bureau: Amgen, Eli Lilly, Novartis. Research grants: Amgen, Genzyme. SH: None.

References

- 1. Janghorbani M, Van Dam RM, Willett WC, Hu FB. Systematic review of type 1 and type 2 diabetes mellitus and risk of fracture. Am J Epidemiol. 2007;166:495–505.
- 2. Vestergaard P. Discrepancies in bone mineral density and fracture risk in patients with type 1 and type 2 diabetes—a meta-analysis. Osteoporos Int. 2007;18:427–44.
- Kanis JA, Melton III LJ, Christiansen C, Johnston CC, Khaltaev N. The diagnosis of osteoporosis. J Bone Miner Res. 1994;9:1137–41.
- 4. Looker AC, Wahner HW, Dunn WL, et al. Updated data on proximal femur bone mineral levels of US adults. Osteoporos Int. 1998;8:468–89.
- Kanis JA, McCloskey EV, Johansson H, Oden A, Melton III LJ, Khaltaev N. A reference standard for the description of osteoporosis. Bone. 2008;42:467–75.
- 6. Siris ES, Chen YT, Abbott TA, et al. Bone mineral density thresholds for pharmacological intervention to prevent fractures. Arch Intern Med. 2004;164:1108–12.
- 7. Schuit SC, van der Klift M, Weel AE, et al. Fracture incidence and association with bone mineral density in elderly men and women: the Rotterdam Study. Bone. 2004;34:195–202.
- Stone KL, Seeley DG, Lui LY, et al. BMD at multiple sites and risk of fracture of multiple types: long-term results from the Study of Osteoporotic Fractures. J Bone Miner Res. 2003; 18:1947–54.
- Cranney A, Jamal SA, Tsang JF, Josse RG, Leslie WD. Low bone mineral density and fracture burden in postmenopausal women. CMAJ. 2007;177:575–80.
- Rubin KH, Friis-Holmberg T, Hermann AP, Abrahamsen B, Brixen K. Risk assessment tools to identify women with increased risk of osteoporotic fracture: complexity or simplicity? A systematic review. J Bone Miner Res. 2013;28:1701–17.
- 11. Whiting P, Rutjes AW, Reitsma JB, Bossuyt PM, Kleijnen J. The development of QUADAS: a tool for the quality assessment of studies of diagnostic accuracy included in systematic reviews. BMC Med Res Methodol. 2003;3:25.
- 12. Steyerberg EW. Clinical prediction models: a practical approach to development, validation, and updating. New York: Springer; 2008.
- Altman DG, Vergouwe Y, Royston P, Moons KG. Prognosis and prognostic research: validating a prognostic model. BMJ. 2009;338:1432–5.

- 14. Royston P, Moons KG, Altman DG, Vergouwe Y. Prognosis and prognostic research: developing a prognostic model. BMJ. 2009;338:1373–7.
- 15. Steyerberg EW, Vickers AJ, Cook NR, et al. Assessing the performance of prediction models: a framework for traditional and novel measures. Epidemiology. 2010;21:128–38.
- Leslie WD, Lix LM. Comparison between various fracture risk assessment tools. Osteoporos Int. 2014;25:1–21.
- Cadarette SM, Jaglal SB, Kreiger N, McIsaac WJ, Darlington GA, Tu JV. Development and validation of the Osteoporosis Risk Assessment Instrument to facilitate selection of women for bone densitometry. CMAJ. 2000;162:1289–94.
- 18. Koh LK, Sedrine WB, Torralba TP, et al. A simple tool to identify asian women at increased risk of osteoporosis. Osteoporos Int. 2001;12:699–705.
- Lydick E, Cook K, Turpin J, Melton M, Stine R, Byrnes C. Development and validation of a simple questionnaire to facilitate identification of women likely to have low bone density. Am J Manag Care. 1998;4:37–48.
- 20. Morin S, Tsang JF, Leslie WD. Weight and body mass index predict bone mineral density and fractures in women aged 40 to 59 years. Osteoporos Int. 2009;20:363–70.
- Gourlay ML, Powers JM, Lui LY, Ensrud KE. Clinical performance of osteoporosis risk assessment tools in women aged 67 years and older. Osteoporos Int. 2008;19:1175–83.
- 22. Rud B, Hilden J, Hyldstrup L, Hrobjartsson A. The Osteoporosis Self-Assessment Tool versus alternative tests for selecting postmenopausal women for bone mineral density assessment: a comparative systematic review of accuracy. Osteoporos Int. 2009;20:599–607.
- Schwartz EN, Steinberg DM. Prescreening tools to determine who needs DXA. Curr Osteoporos Rep. 2006;4:148–52.
- Cadarette SM, McIsaac WJ, Hawker GA, et al. The validity of decision rules for selecting women with primary osteoporosis for bone mineral density testing. Osteoporos Int. 2004;15: 361–6.
- Cadarette SM, Jaglal SB, Murray TM, McIsaac WJ, Joseph L, Brown JP. Evaluation of decision rules for referring women for bone densitometry by dual-energy X-ray absorptiometry. JAMA. 2001;286:57–63.
- Kanis JA, Oden A, Johansson H, Borgstrom F, Strom O, McCloskey E. FRAX and its applications to clinical practice. Bone. 2009;44:734–43.
- 27. Kanis JA, on behalf of the World Health Organization Scientific Group. (2007). Assessment of osteoporosis at the primary health-care level. Technical Report. Published by the University of Sheffield. http://www.shef.ac.uk/FRAX/pdfs/WHO_Technical_Report.pdf
- Kanis JA, McCloskey E, Johansson H, Oden A, Leslie WD. FRAX((R)) with and without bone mineral density. Calcif Tissue Int. 2012;90:1–13.
- 29. Leslie WD, Morin S, Lix LM, et al. Fracture risk assessment without bone density measurement in routine clinical practice. Osteoporos Int. 2012;23:75–85.
- Leslie WD, Lix LM, Johansson H, Oden A, McCloskey E, Kanis JA. Spine-hip discordance and fracture risk assessment: a physician-friendly FRAX enhancement. Osteoporos Int. 2011;22:839–47.
- 31. Kanis JA, Johansson H, Oden A, McCloskey EV. Guidance for the adjustment of FRAX according to the dose of glucocorticoids. Osteoporos Int. 2011;22:809–16.
- Satagopan JM, Ben-Porat L, Berwick M, Robson M, Kutler D, Auerbach AD. A note on competing risks in survival data analysis. Br J Cancer. 2004;91:1229–35.
- Leslie WD, Lix LM, Wu X. Competing mortality and fracture risk assessment. Osteoporos Int. 2013;24:681–8.
- Kanis JA, Oden A, McCloskey EV, Johansson H, Wahl DA, Cooper C. A systematic review of hip fracture incidence and probability of fracture worldwide. Osteoporos Int. 2012;23: 2239–56.
- 35. Kanis JA, Johnell O, Oden A, et al. Long-term risk of osteoporotic fracture in Malmo. Osteoporos Int. 2000;11:669–74.
- 36. Kanis JA, Oden A, Johnell O, Jonsson B, De Laet C, Dawson A. The burden of osteoporotic fractures: a method for setting intervention thresholds. Osteoporos Int. 2001;12:417–27.

- Lam A, Leslie WD, Lix LM, Yogendran M, Morin SN, Majumdar SR. Major osteoporotic to hip fracture ratios in canadian men and women with Swedish comparisons: a population-based analysis. J Bone Miner Res. 2014;29:1067–73.
- Kanis JA, Oden A, Johnell O, et al. The use of clinical risk factors enhances the performance of BMD in the prediction of hip and osteoporotic fractures in men and women. Osteoporos Int. 2007;18:1033–46.
- Kanis JA, Oden A, Johansson H, McCloskey E. Pitfalls in the external validation of FRAX. Osteoporos Int. 2012;23:423–31.
- 40. Nguyen ND, Frost SA, Center JR, et al. Development of a nomogram for individualizing hip fracture risk in men and women. Osteoporos Int. 2007;18:1109–17.
- Nguyen ND, Frost SA, Center JR, et al. Development of prognostic nomograms for individualizing 5-year and 10-year fracture risks. Osteoporos Int. 2008;19:1431–44.
- 42. Langsetmo L, Nguyen TV, Nguyen ND, et al. Independent external validation of nomograms for predicting risk of low-trauma fracture and hip fracture. CMAJ. 2011;183:E107–14.
- Hippisley-Cox J, Coupland C. Predicting risk of osteoporotic fracture in men and women in England and Wales: prospective derivation and validation of QFractureScores. BMJ. 2009;339:b4229.
- 44. Hippisley-Cox J, Coupland C. Derivation and validation of updated QFracture algorithm to predict risk of osteoporotic fracture in primary care in the United Kingdom: prospective open cohort study. BMJ. 2012;344, e3427.
- 45. Collins GS, Mallett S, Altman DG. Predicting risk of osteoporotic and hip fracture in the United Kingdom: prospective independent and external validation of QFractureScores. BMJ. 2011;342:d3651.
- 46. Cummins NM, Poku EK, Towler MR, O'Driscoll OM, Ralston SH. Clinical risk factors for osteoporosis in Ireland and the UK: a comparison of FRAX and QFractureScores. Calcif Tissue Int. 2011;89:172–7.
- 47. Leslie WD, Morin SN, Lix LM, Majumdar SR. Does diabetes modify the effect of FRAX risk factors for predicting major osteoporotic and hip fracture? Osteoporos Int. 2014;25(12): 2817–24.
- Johnell O, Kanis JA, Oden A, et al. Predictive value of BMD for hip and other fractures. J Bone Miner Res. 2005;20:1185–94.
- Marshall D, Johnell O, Wedel H. Meta-analysis of how well measures of bone mineral density predict occurrence of osteoporotic fractures. BMJ. 1996;312:1254–9.
- 50. Fraser LA, Pritchard J, Ioannidis G, et al. Clinical risk factors for fracture in diabetes: a matched cohort analysis. J Clin Densitom. 2011;14:416–21.
- Nielson CM, Srikanth P, Orwoll ES. Obesity and fracture in men and women: an epidemiologic perspective. J Bone Miner Res. 2012;27:1–10.
- 52. Compston JE, Flahive J, Hosmer DW, et al. Relationship of weight, height, and body mass index with fracture risk at different sites in postmenopausal women: the Global Longitudinal study of Osteoporosis in Women (GLOW). J Bone Miner Res. 2014;29:487–93.
- Gower BA, Casazza K. Divergent effects of obesity on bone health. J Clin Densitom. 2013;16: 450–4.
- 54. Reid IR. Fat and bone. Arch Biochem Biophys. 2010;503:20-7.
- 55. Gonnelli S, Caffarelli C, Nuti R. Obesity and fracture risk. Clin Cases Miner Bone Metab. 2014;11:9–14.
- 56. Leslie WD, Orwoll ES, Nielson CM, et al. Estimated lean mass and fat mass differentially affect femoral bone density and strength index but are not FRAX independent risk factors for fracture. J Bone Miner Res. 2014;29:2511–9.
- 57. Ho-Pham LT, Nguyen UD, Nguyen TV. Association between lean mass, fat mass, and bone mineral density: a meta-analysis. J Clin Endocrinol Metab. 2014;99:30–8.
- Johansson H, Kanis JA, Oden A, et al. A meta-analysis of the association of fracture risk and body mass index in women. J Bone Miner Res. 2014;29:223–33.
- Nielson CM, Bouxsein ML, Freitas SS, Ensrud KE, Orwoll ES. Trochanteric soft tissue thickness and hip fracture in older men. J Clin Endocrinol Metab. 2009;94:491–6.

- 60. Viljakainen HT, Pekkinen M, Saarnio E, Karp H, Lamberg-Allardt C, Makitie O. Dual effect of adipose tissue on bone health during growth. Bone. 2011;48:212–7.
- Chan MY, Frost SA, Center JR, Eisman JA, Nguyen TV. Relationship between body mass index and fracture risk is mediated by bone mineral density. J Bone Miner Res. 2014; 29(11):2327–35.
- Melton III LJ, Riggs BL, Leibson CL, et al. A bone structural basis for fracture risk in diabetes. J Clin Endocrinol Metab. 2008;93:4804–9.
- 63. Ishii S, Cauley JA, Crandall CJ, et al. Diabetes and femoral neck strength: findings from the hip strength across the Menopausal Transition Study. J Clin Endocrinol Metab. 2012;97: 190–7.
- 64. Sornay-Rendu E, Boutroy S, Vilayphiou N, Claustrat B, Chapurlat RD. In obese postmenopausal women, bone microarchitecture and strength are not commensurate to greater body weight: the Os des Femmes de Lyon (OFELY) Study. J Bone Miner Res. 2013;28:1679–87.
- Premaor M, Parker RA, Cummings S, et al. Predictive value of FRAX for fracture in obese older women. J Bone Miner Res. 2013;28:188–95.
- 66. Adami S. Bone health in diabetes: considerations for clinical management. Curr Med Res Opin. 2009;25:1057–72.
- 67. Jackuliak P, Payer J. Osteoporosis, fractures, and diabetes. Int J Endocrinol. 2014;2014:820615.
- Leslie WD, Lix LM, Prior HJ, Derksen S, Metge C, O'Neil J. Biphasic fracture risk in diabetes: a population-based study. Bone. 2007;40:1595–601.
- 69. Melton III LJ, Leibson CL, Achenbach SJ, Therneau TM, Khosla S. Fracture risk in type 2 diabetes: update of a population-based study. J Bone Miner Res. 2008;23:1334–42.
- 70. Schwartz AV, Vittinghoff E, Bauer DC, et al. Association of BMD and FRAX score with risk of fracture in older adults with type 2 diabetes. JAMA. 2011;305:2184–92.
- Giangregorio LM, Leslie WD, Lix LM, et al. FRAX underestimates fracture risk in patients with diabetes. J Bone Miner Res. 2012;27:301–8.
- 72. Carnevale V, Morano S, Fontana A, et al. Assessment of fracture risk by the FRAX algorithm in men and women with and without type 2 diabetes mellitus: a cross-sectional study. Diabetes Metab Res Rev. 2014;30:313–22.
- 73. Bhattoa HP, Onyeka U, Kalina E, et al. Bone metabolism and the 10-year probability of hip fracture and a major osteoporotic fracture using the country-specific FRAX algorithm in men over 50 years of age with type 2 diabetes mellitus: a case-control study. Clin Rheumatol. 2013;32:1161–7.
- National Osteoporosis Foundation. Clinician's guide to prevention and treatment of osteoporosis. Osteoporos Int. 2014;25(10):2359–81.
- 75. Leslie WD, Rubin MR, Schwartz AV, Kanis JA. Type 2 diabetes and bone. J Bone Miner Res. 2012;27:2231–7.
- Pothuaud L, Barthe N, Krieg MA, Mehsen N, Carceller P, Hans D. Evaluation of the potential use of trabecular bone score to complement bone mineral density in the diagnosis of osteoporosis: a preliminary spine BMD-matched, case-control study. J Clin Densitom. 2009;12: 170–6.
- Bousson V, Bergot C, Sutter B, Levitz P, Cortet B. Trabecular bone score (TBS): available knowledge, clinical relevance, and future prospects. Osteoporos Int. 2011;23:1489–501.
- 78. Hans D, Barthe N, Boutroy S, Pothuaud L, Winzenrieth R, Krieg MA. Correlations between trabecular bone score, measured using anteroposterior dual-energy X-ray absorptiometry acquisition, and 3-dimensional parameters of bone microarchitecture: an experimental study on human cadaver vertebrae. J Clin Densitom. 2011;14:302–12.
- Roux JP, Wegrzyn J, Boutroy S, Hans D, Chapurlat R. Relationship between trabecular bone score (TBS), bone mass and bicroarchitecture in human vertebrae: an ex vivo study. Osteoporos Int. 2012;23:S327.
- Silva BC, Leslie WD, Resch H, et al. Trabecular bone score: a noninvasive analytical method based upon the DXA image. J Bone Miner Res. 2014;29:518–30.
- Leslie WD, Aubry-Rozier B, Lamy O, Hans D. TBS (trabecular bone score) and diabetesrelated fracture risk. J Clin Endocrinol Metab. 2013;98:602–9.

- 3 Fracture Risk Assessment in Diabetes
- Dhaliwal R, Cibula D, Ghosh C, Weinstock RS, Moses AM. Bone quality assessment in type 2 diabetes mellitus. Osteoporos Int. 2014;25:1969–73.
- Rubin M, Shah A, Zhang C, et al. Trabecular bone assessment in type 2 diabetes mellitus. J Bone Miner Res. 2013;28 Suppl 1:S398.
- Leslie WD, Krieg MA, Hans D. Clinical factors associated with trabecular bone score. J Clin Densitom. 2013;16:374–9.
- 85. Leslie WD, Winzenrieth R, Majumdar SR, Lix LM, Hans D. Clinical performance of an updated version of trabecular bone score in men and women: The Manitoba BMD Cohort. J Bone Miner Res. 2014;28 Suppl 1:S297.
- Leib E, Winzenrieth R, Aubry-Rozier B, Hans D. Vertebral microarchitecture and fragility fracture in men: a TBS study. Bone. 2014;62:51–5.
- Leslie WD, Aubry-Rozier B, Lix LM, Morin SN, Majumdar SR, Hans D. Spine bone texture assessed by trabecular bone score (TBS) predicts osteoporotic fractures in men: the Manitoba Bone Density Program. Bone. 2014;67:10–4.
- Leslie WD, Johansson H, Kanis JA, et al. Lumbar spine texture enhances 10-year fracture probability assessment. Osteoporos Int. 2014;25:2271–7.
- Farr JN, Drake MT, Amin S, Melton III LJ, McCready LK, Khosla S. In vivo assessment of bone quality in postmenopausal women with type 2 diabetes. J Bone Miner Res. 2014;29: 787–95.
- Starup-Linde J, Eriksen SA, Lykkeboe S, Handberg A, Vestergaard P. Biochemical markers of bone turnover in diabetes patients-a meta-analysis, and a methodological study on the effects of glucose on bone markers. Osteoporos Int. 2014;25:1697–708.
- 91. Saito M, Kida Y, Kato S, Marumo K. Diabetes, collagen, and bone quality. Curr Osteoporos Rep. 2014;12:181–8.

Chapter 4 Diabetes-Related Conditions and Complications in Orthopaedic Surgery

Daniel J. Gehling and Nabil A. Ebraheim

Introduction

Diabetes mellitus (DM) is a significant clinical problem, with the treatment costs in 2012 estimated at \$245 billion in the USA alone [1]. This amount is a 41 % increase over the estimated \$174 billion in 2007 [2]. Medical expenditures account for the largest proportion of general costs. However, there are also many indirect costs of diabetes as a consequence of increased absenteeism, reduced productivity in the workplace, unemployment from disease-related disability, and productive capacity loss due to early mortality [1, 2]. In orthopaedics, patients with diabetes have a number of associated disorders, and these present a challenge as many have an increased hospital stay, higher risk of infection, and higher risk of complications after orthopaedic treatment.

The orthopaedic-related problems in diabetes are varied, and the true causal links between diabetes and the disorders are largely unknown. Hip fractures have a higher incidence in patients with diabetes, while ankle fractures are important in diabetic patients because there is a unique concern of treatment of patients with diabetes and the clearly associated worse outcomes. Infection with or without surgery is also a concern in diabetes. Plantar ulcers are discussed because there is an association of ulcers in diabetes and they are predictive of poor outcomes. Finally, there are a number of upper extremity disorders including trigger finger, carpal tunnel syndrome, Dupuytren's disease, and adhesive capsulitis that have clear associations with diabetes.

D.J. Gehling, M.D. (🖂) • N.A. Ebraheim, M.D.

Department of Orthopaedic Surgery, University of Toledo Medical Center, 3000 Arlington Avenue, Toledo, OH 43614, USA

e-mail: daniel.gehling@utoledo.edu; nabil.ebraheim@utoledo.edu

[©] Springer International Publishing Switzerland 2016

B. Lecka-Czernik, J.L. Fowlkes (eds.), *Diabetic Bone Disease*, DOI 10.1007/978-3-319-16402-1_4

Hip Fracture

Fractures around the hip in the general population are associated with decreased function and an increased mortality [3–12]. These fractures can be either intracapsular, usually femoral neck fractures, or extra-capsular, usually referred to as intertrochanteric or subtrochanteric fractures (Fig. 4.1) [13]. The cause of these fractures has largely been attributed to decreased bone mass, as most of these fractures are low energy falls in patients older than 65 years old [3]. In diabetic patients, the concern with hip fracture is the increased incidence, with a clear increase in type I DM, and a probable increase in those patients with type II DM [14].

The overall incidence of femoral neck fractures in the USA is between 0.4 and 1.0 % [15, 16], and the estimated cost of treatment of femoral neck fracture alone has been estimated at 19,000-23,000 for 1 year of treatment [17–19]. With the demographical aging of the population, the number of fractures is expected to increase, and it is reasonable to expect that cost of treatment will increase as well. Over 12 years in a British hospital, there has been a 10 % increase in admissions from hip fracture from 2000 to 2012, and it has been considered to be a nonlinear increase over time [20].

Type I DM has a definite effect on the incidence of hip fractures with reported relative risks of between 1.7 and 12 times higher than controls, with a suggested average of sixfold increase [21–24]. In addition, type I DM patients have hip fractures at a younger age on average, with a mean of 43 for women and 41 for men in one study. Almost 7 % of people with type I DM can be expected to have sustained a hip fracture by age 65 [7]. The mechanisms that cause these fractures in type I DM are unclear, but three main reasons have been speculated.

First, bone mineral density (BMD) in patients with DM type I is decreased as compared with controls [25]. One large meta-analysis estimates that the hip Z-scores are -0.37 from expected, and this decrease in BMD suggests a 1.42 increase in risk of hip fracture. However the relative risk is somewhere about sixfold over nondiabetic



Fig. 4.1 General classification of hip fractures

people. This discrepancy has suggested that other pathological mechanisms should be explored to help explain the increased risk of hip fracture in type I DM [26].

Second, type I DM is commonly associated with decreased proprioception, neuropathic changes, and retinopathy with increasing duration and severity of disease process. Each of these comorbidities can potentially contribute to falls, increasing the chances of hip fracture over time [7, 27, 28]. In addition, these disease processes could serve as a marker of severity, and may also have some contribution to pathological bone homeostasis. Supporting these hypotheses, the presence of ophthalmic, neurological, and cardiovascular complications increase the risks of hip fracture by 20, 41, and 29 times expected general population values. Furthermore, as compared to others with type I DM without complications, the risks are 1.3, 2.0, and 1.7, respectively [7]. Finally, hypoglycemia has been suggested as a possible cause, as type I DM is treated with various types of insulin, but all can lead to hypoglycemia [27, 29].

Compared to patients with type I DM, the increased incidence of hip fracture in patients with type II DM is less certain, but is likely. Conflicting evidence has suggested an increased risk, no change in risk, and even a decreased risk of fracture [10, 21, 28, 30–35]. However, the reports on the largest enrollments generally put the relative risk between 1.2 and 2.2, and most recent studies suggest an increased relative risk [14]. It is somewhat confusing that patients with type II DM also have increased BMD on average; however, some propose that the BMD difference is related to the confounding variable of obesity rather than solely a direct effect [25, 26, 30, 33, 34]. The duration of diabetes has also been suggested as a risk factor [31], but age is a significant confounding variable to be considered.

The medical treatment regimens for patients with diabetes have also been suggested as a possible risk factor, given a two times higher risk of hip fracture in patients with a HbA1c of <6.0 vs. those with HbA1c levels greater than 8 %. The tighter control of diabetes could result in more episodes of hypoglycemia, which in turn can potentially lead to more falls [27]. To support this hypothesis, the use of metformin and acarbose has been associated with a decreased risk of hip fractures in type II DM as compared to patients treated with insulin or sulfonylureas. Metformin and acarbose have a much smaller risk of hypoglycemia, and hypothetically may lead to less falls. Thiazolidinediones have also been implicated in higher rates of hip fracture, but not universally [21, 27, 36]. However, hypoglycemia as a potential reason for falls is controversial, as some research suggests that intensive glucose control does not increase the propensity for falls [37].

Treatment for hip fractures is almost exclusively operative, as nonsurgical treatments result in high rates of nonunion, pain, decubitus ulcers, pneumonia, deep venous thrombosis, and pulmonary embolus [38, 39]. For femoral neck fractures, treatment options are open reduction and internal fixation (ORIF), percutaneous screw fixation, hemiarthroplasty, and total hip arthroplasty (THA). Although somewhat controversial, there has been a trend towards total hip replacement in patients with femoral neck fracture despite many having intact cartilage. This trend is because the long-term outcomes after THA are superior to other methods of treatment, in spite of having some increase in early complications [40–43]. With ORIF, there are late complications of fracture nonunion and avascular necrosis of the femoral head. These complications are thought to be related to damage of the femoral head blood supply and the resulting hemarthrosis after displaced femoral neck fractures [44, 45]. In hip arthroplasty, the proximal segment of the femur is replaced and the problems of nonunion and avascular necrosis of the femoral head are no longer issues. For intertrochanteric and subtrochanteric hip fractures, the most accepted treatment is with ORIF, as the blood supply generally allows the fracture to heal, without significant risk of avascular necrosis [46].

Mortality rates after hip fracture at 1 year range from 18 to 33 % [3, 5, 8, 9, 47], and the rates remain higher than the general public for at least 10 years postoperatively [11, 12]. An analysis in the Danish registry noted that the 1 year mortality is nearly double that of those without femoral neck fracture, and that the excess mortality is approximately 1.8 % increased each year after. At 20 years follow-up, the survival is 57 % of what would be expected from the control group [11]. Patients with DM and hip fracture are at a higher risk of mortality than patients without DM, with 1-year rates as high as 32 % vs. 13 % of nondiabetic patients [3, 4]. In addition, DM is associated with an increased risk of cardiovascular, pressure ulcer, and infectious complications after hip fracture [4, 9, 48]. These complications of hip fracture treatment are thought to be the cause for increase in the risk of mortality, not just the presence of pre-existing diabetes [11].

Ankle Fracture

Fractures around the ankle are some of the most commonly seen fractures in the hospital setting and the incidence appears to be increasing, as a Finnish study has suggested a threefold increase in incidence by 2030 [49]. In diabetic patients, the concern is not necessarily an increased risk of fracture, but of treatment complications. Diabetic patients have significant increased risks of infection, malunion, neuropathic ankle, and amputation after ankle fractures, even those that are low energy injuries [50–58].

Ankle fractures consist of one or more fractures of the medial malleolus of the tibia, the lateral malleolus of the fibula (Fig. 4.2), and sometimes the posterior malleolus of the tibia [59]. A more significant, but related injury is a tibial plafond fracture. The plafond fracture is usually a higher energy injury with axial load, rather than a routine ankle fracture, which is a rotational injury [60]. The specific direction of forces that have created an ankle fracture classically determines the different fracture patterns [59]. Patient with complicated diabetes and no trauma may have Charcot or neuropathic arthropathy, and this type of presentation must be distinguished from those patients with a traumatic ankle fracture [55, 61, 62]. The ankle is an area of relative paucity of muscle tissue, and therefore the bony structures are covered with skin and subcutaneous tissue alone. Therefore, the associated soft tissue injury after ankle fracture is a more significant problem than with fractures in other locations in the body [63].

Fig. 4.2 Unstable ankle fracture



Treatment of ankle fractures can be either nonoperative or operative, depending on the stability of the ankle joint. The stability of ankle fractures is based on the integrity of the medial malleolus and the deltoid ligament. In an unstable ankle fracture, treatment is generally operative with anatomic reduction and fixation. In stable ankle fractures, the treatment is general nonoperative [64, 65].

Given the microvascular complications in diabetic patients, and with the relative paucity of soft tissue around the ankle, there is a great deal of concern with diabetic patients with unstable ankle fractures [66]. With nonoperative treatment, the complication rate in diabetic patients is high, with some studies stating an up to 83 % risk of complication in diabetics [57, 67]. Most of these complications relate to malunion and pressure sores from cast treatment, and the risks are understandably higher in those patients with neuropathy [50]. Although counter-intuitive, these studies are evidence that operative management of ankle fractures is indicated in those patients with more severe and complicated diabetes.

In diabetic patients treated with reduction and internal fixation, the complication rate is improved from nonoperative treatment, but still higher than controls at between 13 and 50 % [52, 56, 63, 68, 69]. This rate is compared to the reported risk of infections in controls of 1.4-7 % [55, 63, 68, 69]. The amputation rates are even more striking, with between 3.8 and 5 % of diabetics eventually ending with amputation after ankle fracture treatment, versus well under 1 % of nondiabetics [52, 56, 57, 67, 70, 71]. In open ankle fractures, the amputation rate in diabetic patients has been reported to be as high as 42 % [51], and is a strong predictor of increased complication rates overall [51, 70].

Peripheral neuropathy and microvascular complications are commonly cited as the underlying cause of these complications. First, the microvascular supply and the healing response are compromised in diabetic patients [66]. Second, peripheral neuropathy has loss of protective sensation in the foot and ankle, leading to pressure sores, noncompliance postoperatively, failure of fixation, and malunion of fractures. Indeed, patients with neuropathy have a reported 75 % complication rate after ORIF, and a 100 % complication rate treated with closed methods [50, 52, 55, 56, 67, 71]. Overall, those patients with peripheral neuropathy have a 15.5 odds ratio of a postoperative wound complication versus a control group [63].

Higher complication rates are seen in patients with HbA1c levels greater than 6.5 %, and optimization of blood glucose control is likely to be of benefit for overall outcomes [53, 72]. Emphasizing the importance of blood glucose control long-term, recent results suggest that the complication rate for those patients with uncomplicated diabetes is similar to nondiabetics [53]. More than 90 % of nondiabetic patients with ankle fractures will regain nearly all function after treatment, but only about 70 % of diabetic patients functionally recover, have longer hospital stays, and have higher overall hospital costs [54, 73]. There are two consensus recommendations for treatment of ankle fractures in diabetic patients. First, the ORIF fixation strength should be increased over what would be normally used for nondiabetic patients. Second, the postoperative nonweightbearing period should be lengthened over what would be used routinely, especially in those patients with complicated DM [58, 66].

Infection

Although there is a risk of infection associated with any orthopaedic surgery, patients with diabetes are at a higher risk of developing surgical site infections (SSIs) [51, 52, 57, 68, 74–77]. SSIs are associated with increased hospital stay, increased hospital cost, and reduced quality of life. A review of one hospital in the 1990s showed double mortality, double length of hospital stay, and double the cost expenditure in patients with SSIs compared to controls [78]. In orthopaedic surgery, SSIs are also associated with increased amputation rates, and worse functional outcomes [57, 74, 76, 77, 79–81].

SSIs are defined as superficial incisional, deep incisional, or organ/space infections. A superficial incisional SSI occurs within 30 days, and involves only the skin and subcutaneous tissues. These type of SSI are usually treated with antibiotics alone, or possibly a minor debridement if there is dehiscence of the incision. A deep incisional or organ/space SSI is one that occurs within 30 days or alternatively within 1 year if implants are in place [82]. A deep SSI usually requires aggressive management and almost always some form of surgical debridement.

Orthopaedic spine surgery has an infection rate of about 2 % in elective cases, but has been reported up to 5 % at 1 year following surgery for traumatic spine injury [74, 77, 83]. Diabetes is strongly associated with SSI in these patients, with

the risk being three times that of nondiabetic patients [74, 77, 83]. Furthermore, a preoperative HbA1c greater than 7.0 % is highly associated with SSI in spine surgery, with up to 35 % incidence in patients with uncontrolled DM [84]. Interestingly, increased serum glucose in patients without DM may also be an independent risk of SSI in spine surgery [74]. Correspondingly, in orthopaedic trauma, the stress-response increase in serum glucose also is associated with an increased risk of infection, even without history of diabetes [85–87]. Considering these studies, it appears that perioperative glycemic control is important to prevent SSI.

Periprosthetic infection (PPI) in total joint arthroplasty is also a serious epidemiologic issue. In 2003, there were about 200,000 total hip and 400,000 total knee arthroplasties performed in the USA. The projection to 2030 is a concerning 572,000 total hip and 3.5 million total knee arthroplasties annually [88]. A recent update seems to largely confirm those previous estimates [89]. In the general population, the arthroplasty infection rate is about 0.7 % [90]. Even this low rate of infection will generate a significant health burden of PPI, if the numbers of arthroplasty patients follows the predicted future trends.

Historically, rates of PPI in patients with diabetes have been as high as 7 % [76, 91]. An additional report has suggested a nearly fourfold increased risk of infection in those with diabetes [92]. However, a more recent report suggests a 1.2 times increase in the relative risk of PPI in diabetic patients [93]. Patients with uncontrolled DM preoperatively have a particularly high risk of PPI, urinary tract infection, transfusion, stroke, infection, and death [94]. In addition, even nondiabetic patients with a fasting preoperative glucose greater than 124 mg/dL have a significantly higher rate of PPI [95]. These studies strongly suggest preoperative DM optimization and glycemic control prior to orthopaedic procedures in order to prevent PPI and other complications.

Plantar Ulcerations

Plantar ulcers are a serious concern in patient with diabetes. Ulcers have predictive value for infection, sepsis, and amputation; and prophylactic care decreases patient morbidity and lowers expense of care [96]. The lifetime risk of plantar ulcer in diabetic patients has been estimated as high as 25 %, and the recurrence is 50 % at 3 years. In the past, up to 10 % of patients initially diagnosed with diabetes will have presented with an ulceration [97].

Peripheral neuropathy is the primary determinant of ulcer development, and 20–50 % of patients with diabetes have some degree of neuropathy [98–100]. With neuropathy, the loss of protective sensation in the foot allows an ulcer to start unchecked, as there is no painful feedback to the patient [101]. In addition, perhaps only half of patients with clinically significant diabetic neuropathy self-report neuropathic symptoms [98]. Therefore screening for sensation loss is the primary clinical method to determine those patients at risk for ulceration and to assign appropriate care measures accordingly [102, 103]. In addition to neuropathy, those

patients with minor foot trauma and deformity are at very high risk of ulceration. The triad of neuropathy, trauma, and deformity is present in up to two thirds of patients with ulcers [103].

Screening has been suggested in the primary care setting, and is most commonly recommended with a Semmes-Weinstein 5.07 monofilament, which is a simple and cost-effective diagnostic strategy [96, 97, 100, 104, 105]. Additional screening methods are evaluation of vibratory sensation, temperature sensation, the presence of an Achilles tendon reflex, and transcutaneous oxygenation measurements [96, 102, 104–107]. Prevention is targeted at patient education, appropriate foot wear, optimization of blood glucose control, smoking cessation, identification of deformity and orthopaedic consultation, identification of large vessel disease and vascular consultation, and consistent follow-up care [96, 105].

Treatment of ulcers, once formed, is targeted to offload the ulcer with appropriate foot wear, orthotics, limited weightbearing, or total contact casting [108–111]. Additionally, optimization of blood glucose control, cessation of smoking, orthopaedic debridement or deformity correction, and vascular intervention complete the multidisciplinary approach to treatment of ulceration [112–116]. This treatment comes at a great cost, as the treatment of single foot ulcer over 2 years is estimated at around \$28,000 [117, 118].

The time to successful healing of a plantar ulcer is usually on the scale of months, not days. The average healing time is variable, but many will heal by 2–3 months, with the more severe cases taking 4 months or longer [119, 120]. Most of the cross-sectional area of healing takes place at the beginning of treatment, but complete healing takes a much longer time [121]. The average time to healing is lower in those patients with smaller initial wounds, and those with more acute wounds [119, 120]. Transcutaneous oxygen tension may also predict ultimate healing potential [122]. Those ulcerations that ultimately are not able to be healed have very high incidences of osteomyelitis and eventual amputation. Of those diabetic patients who eventually end with amputation, up to 85 % had previous ulcerations [123–125]. In those ulcers with infection and ischemia, the outcomes are very poor. Patients with a deep wound, infection, and ischemia have an amputation rate of at least 50 %, and even a superficial wound can be 17-50 % [126].

Trigger Finger

Also known as stenosing tenosynovitis, trigger finger is an overuse disorder of the flexor tendon at the region of the A1 pulley. This pulley is the first in a set of pulleys in the finger that maintain the tendon–bone relationship while allowing the tendon to slide freely during movement [127]. The trigger finger is caused when there is a mismatch of tendon thickness to the internal diameter of the A1 pulley, and the tendon becomes stuck in the pulley after movement of the finger into flexion (Fig. 4.3) [127–129]. Patients commonly report using force to "pop" the finger back into extension.

Fig. 4.3 Trigger finger



The incidence of trigger finger is 7-20 % of patients with diabetes comparing to only about 1-2 % in nondiabetic patients [130–132]. Trigger finger is also more common in women than men, except for a possible equal association in type I DM [133, 134]. When associated with diabetes, it commonly presents with multiple digit involvement, with a mean of 2.2 digits vs. 1.3 digits in nondiabetic patients [135, 136]. The duration of diabetes and insulin dependence is also associated with an increased incidence of trigger finger [130, 131, 134, 137]. However, the overall effectiveness of glycemic control does not seem to be correlated with the prevalence of trigger digit [131, 137].

Treatment of trigger finger is similar among patients with and without diabetes. Non-operative treatment includes stretching exercises to the flexor tendon, especially before and after activity, or injections of corticosteroid into the region around or inside the A1 pulley [138, 139]. Diabetic patients may have less successful non-operative treatment overall, especially if insulin dependent [136]. Finally, surgical treatment is generally reserved for patients who fail non-operative management, and is almost always a release of the A1 pulley. The pulley is not absolutely necessary for proper function of the fingers, and has a reported 97 % success rate [140].

Dupuytren's Disease

Dupuytren's disease (DD) is a hand disorder that is primarily symptomatic in older individuals, but can have relatively asymptomatic signs in younger patients as well. It was described by and named after Baron Dupuytren in the 1830s, but he was not the first to describe it [141]. The disease clearly has a genetic component, but with incomplete penetrance, and is most common in men and in patients of Northern European ancestry. It is also associated with a number of other factors including a history of trauma, epilepsy, alcoholism, smoking, adhesive capsulitis, and diabetes [142, 143].

Symptoms of DD begin with palmar nodules, and later with pretendinous cords. Later, the cords begin to contract and cause flexion of the digits, limiting extension and causing flexion contractures (Fig. 4.4). These contractures are frequently the presenting symptoms, as they have a direct limitation of patient activity and overall hand function. The cords contain myofibroblasts and organize along predictable, but abnormal routes in the hand and fingers [144, 145]. Most often, the ring and small fingers are involved in nondiabetic patients, but any digit can be affected. Curiously, diabetic patients with DD are more likely to have long and ring finger involvement instead of ring and small finger involvement [133]. Also, the symptoms of the disease tend to be milder, and progression is slower in those diabetic patients than the general population [141]. Although men are more prone to develop the disease, diabetic women have a high chance of also being affected [133, 146].

The prevalence of DD in the general population is estimated to be 1-8 %, but depends on the diagnostic criteria used. Those patients who present with cords may be closer to 1 %, but those with palmar nodules are probably closer to 7–11 % of the population [130, 133]. The risk of developing Dupuytren's disease in diabetic



Fig. 4.4 Dupuytren's disease

81

patients is significantly higher than that of general population at between 19 and 42 % [130, 133, 141]. Dupuytren's disease has been reported to have incidence rates in type I diabetes patients comparable to type II diabetes patients, and correlates to the duration of diabetes [133, 134, 147, 148]. Impaired glycemic control as measured by HbA1c, mean glucose, and presence of diabetic comorbidities seem to correlate with the prevalence of DD [130, 141].

Noninvasive treatment is largely unsuccessful in improving the features of the disease. One minimally invasive treatment option is injection of collagenase followed by manipulation of the digits, and this method has been shown to be an effective treatment [149]. Percutaneous release of the cords is another treatment option that does not have the risk of wound complications postoperatively compared to open fasciectomy, but may not have as significant a correction overall [150, 151]. Open surgical treatment has historically been a complete or partial fasciectomy, which is aggressive removal of the cords of abnormal tissue. Complications of partial fasciectomy are incomplete correction, recurrence, infection, and wound problems. Even with surgical treatment, however, recurrence is up to 80 %, but only about 12 % have additional treatment from loss of function or recurrence [146]. Surgical treatment in diabetic patients does not seem to have an increased risk of complication [152].

Carpal Tunnel Syndrome

Carpal tunnel syndrome (CTS) is the compression of the median nerve within the carpal tunnel in the wrist, and it classically manifests as pain and paresthesias of the thumb, index, long, and the radial half of the ring finger. Symptoms are usually worse at nighttime, and patients frequently report being woken up from numbness in the hand. CTS is typically an overuse injury related to occupational and daily activities, and is more common in women [131, 153–155]. However, it can also present in cases with extrinsic compression of the carpal tunnel from masses, or acutely from trauma.

The prevalence of CTS in patients with diabetes has been estimated at 11-30 % [130, 133, 153, 156], and is dependent on the duration of diabetes. In contrast, the incidence is estimated at 2–3.8 % of the general population [155, 157]. Bilateral involvement of CTS is common in diabetic patients, but both sides are not necessarily symptomatic simultaneously. Type I DM patients have a high prevalence of CTS with increasing duration of disease, up to 85 % after 54 years of DM. However the prevalence does not seem to be associated with glycemic control [130, 158].

Overall, the basic science mechanism in which DM is associated with CTS is not well understood, but microangiopathy has been suggested as a likely contributing factor [159, 160]. However, nephropathy, retinopathy, and polyneuropathy have not been consistently identified as risk factors for CTS, possibly from the confounding effect of duration of diabetes [130, 157, 158].

Initial treatment of CTS is generally non-operative and initiated with splinting, especially at night. Injections into the carpal tunnel are sometimes advised in milder disease, and can sometimes be effective long-term [161]. Surgical treatment consists of a release of the transverse carpal ligament, which decompresses the median nerve and is a highly effective treatment of patients' symptoms [160, 162]. However, there may be a slightly less efficacious outcome in patients with DM [159, 160].

Adhesive Capsulitis

Adhesive capsulitis, also known as frozen shoulder, is a disorder of the shoulder characterized by pain and stiffness [163], and is commonly arbitrarily divided into phases [164–166]. The first phase is painful, without significant loss of shoulder range of motion. The second phase remains painful, but the shoulder becomes significantly stiffened. In the third phase, the shoulder remains stiff, but is less painful. Finally, the fourth phase is spontaneous resolution, which is sometimes mistakenly believed to happen in all patients [164, 165, 167].

Adhesive capsulitis is a term coined by Nevassier for a disorder of the shoulder in which the glenohumeral joint capsule thickens and has an adhesive appearance when opened to be released [163]. Although used interchangeably, frozen shoulder describes the clinical presentation of a stiff shoulder, either from a primary condition of capsular thickening or a secondary cause from another shoulder diagnosis such as rotator cuff pathology. Adhesive capsulitis is used as a description of the pathological condition of capsular thickening and hypertrophy resulting in shoulder stiffness [163, 164].

Adhesive capsulitis is associated with a number of other medical conditions including diabetes, both type 1 and 2 [166-171]. Other associated conditions reported are trauma, cervical spine disease, Dupuytren's disease, coronary heart disease, tuberculosis, carcinoma, hyperthyroidism, and epilepsy. However, the relationship to these medical comorbidities is unclear, including diabetes [171]. The incidence of adhesive capsulitis is up to 11-20 % in diabetic patients, in comparison to 2–5 % of control patients [171, 172]. Serum glycosylated hemoglobin levels have been studied, but the levels do not seem to correlate with a risk for adhesive capsulitis [166, 168]. Supporting this finding, histologically measured glycosylation of tissues does not seem to directly correlate with the risk of other secondary complications of diabetes including joint stiffness [173]. In type I diabetic patients, the prevalence of adhesive capsulitis is increased after 20 years duration of the disease process [168, 174]. Type II patients do not have a measurable association with duration of disease [168]. However, patients in the past were not commonly diagnosed with DM until years after the process actually commenced. DM patients with a history of myocardial infarction [168, 171] and autonomic neuropathy have been independently associated with an increased risk of adhesive capsulitis [168].

Diagnosis of adhesive capsulitis is based predominately on a loss of passive range of motion in a shoulder, and the exclusion of other diagnoses that may also cause a secondary loss of range of motion from pain [167, 171, 175]. A diagnostic injection into the subacromial space may aid in diagnosis to eliminate pain from other coexisting diagnoses such as rotator cuff pathology, as even in those cases the

shoulder range of motion is limited. Range of motion in adhesive capsulitis is limited in multiple directions, but especially in external rotation and abducted external rotation [167]. This ROM can be compared against the contralateral side to determine a patient's normal ROM, as it can differ among patients [176].

The natural history of adhesive capsulitis has commonly been cited as a selflimiting process. This citation would suggest that the condition resolves on its own, but careful review of past literature would suggest otherwise. Up to 60 % of patients followed for an average 7 years have some persistent symptoms and disability [177]. In addition, about 50 % of patients have incomplete resolution of symptoms at 1 year [164, 165, 167]. These persistent symptoms support early treatment in most patients.

Treatment of adhesive capsulitis is overall similar in diabetic and nondiabetic patients; however, the results are worse in diabetic patients. Diabetic patients have less range of motion, more pain, and more disability after treatment [169, 170]. Treatment options include physical therapy, intra-articular corticosteroid injection, manipulation under anesthesia, and surgical release [178].

One study of aggressive vs. gentle ROM would suggest that gentle ROM may be better than aggressive manipulation [179]. However, it has been unclear which intervention or combination of interventions are the most successful at treatment [180–182]. Corticosteroid injection may improve with pain, but it is unclear if the long-term outcomes are changed [165, 181, 182]. Oral corticosteroid does not appear to affect outcomes positively [165, 183, 184].

Manipulation under anesthesia is indicated in some cases that have failed a course of physical therapy [175]. This manipulation can be under regional or general anesthesia, and may be followed or preceded with an intra-articular injection of steroid solution. Follow-up with physical therapy immediately afterwards is considered standard. A complication to be avoided is fracture of the humerus, which if occurs, may require additional surgical fixation to continue physical therapy, as immobilization would likely result in treatment failure.

Finally, arthroscopic capsular release is indicated in those patients with 3–6 months of failure of other non-operative options, especially those with significantly stiff shoulders. The result of capsular release is marginally improved over manipulation, but is considered in refractory cases [185]. Risks of the procedure include fracture and axillary nerve injury, where the nerve is located in the axillary recess, immediately underneath the capsular tissue.

Conclusion

Diabetes mellitus is a significant medical issue that increases the risk of certain orthopaedic conditions and complications, and leads to less successful outcomes than in nondiabetic patients. Glycemic control is certainly recommended before any elective procedure, and has become more recognized to be a primary goal after traumatic injuries as well.

Hip fractures are a varied group of fractures around the hip, of which DM significantly increases the risks fracture, the risks of complications, and the risk of

mortality after treatment. Ankle fractures are a major source of complications in patients with diabetes after treatment, both non-operatively and operatively. Overall, operative treatment of unstable ankle fractures is superior to that of non-operative treatment. Some patients will ultimately end with amputation despite best efforts.

Infection risk is increased postoperatively after orthopaedic procedures, and there is an increasing emphasis on glycemic control perioperatively to reduce the risk of SSI. Plantar ulcerations are largely a complication of diabetic neuropathy, deformity, and minor trauma. Prevention is more cost-effective than treatment, as the presence of an ulcer is a predictor of ultimately poor outcomes.

Trigger finger has a high incidence in patients with DM, and more fingers are generally involved at presentation in patients with DM. Treatment options are no different than the general population, but successful treatment may be slightly less efficacious in patients with DM than in nondiabetic patients. Dupuytren's disease has a high prevalence in patients with DM, and there is a predilection for long finger and ring finger involvement rather than the small and ring finger involvement seen in the general population. DD is associated with the duration of diabetes, but the presentation in DM patients is less severe on the whole. The treatment of DD is similar in patients with or without DM. Carpal tunnel syndrome is a common orthopaedic hand issue in patients with diabetes, at least four times the general population incidence, and is associated with the duration of diabetes. The incidence of CTS may be even higher in type I DM than in type II DM. Treatment of CTS in DM and nondiabetic patients is not significantly different than the general population. Surgical treatment is usually successful, but possibly less improvement in patients with DM. Adhesive capsulitis has a high incidence in DM patients as well. The overall presentation is similar, but the treatments may be less successful overall. After treatment, patients with DM have more pain and stiffness on average vs. patients without DM.

Diabetic-associated orthopaedic conditions and complications are commonly seen in the general orthopaedic practice, and are challenging to treat. Although some of the conditions associated with DM do not depend on glycemic control, most orthopaedic complications of treatment do have increased risks in those patients with either acutely or chronically poor blood glucose control. In the future, the medical maintenance of DM and attention to perioperative glycemic control may be the most effective interventions to limit these set of orthopaedic complications in patients with DM.

References

- 1. American Diabetes Association. Economic costs of diabetes in the U.S. in 2012. Diabetes Care. 2013;36(4):1033–46.
- 2. American DA. Economic costs of diabetes in the U.S. In 2007. Diabetes Care. 2008;31(3):596–615.
- Belmont Jr PJ, Garcia EJ, Romano D, Bader JO, Nelson KJ, Schoenfeld AJ. Risk factors for complications and in-hospital mortality following hip fractures: a study using the National Trauma Data Bank. Arch Orthop Trauma Surg. 2014;134(5):597–604.

- 4. Gulcelik NE, Bayraktar M, Caglar O, Alpaslan M, Karakaya J. Mortality after hip fracture in diabetic patients. Exp Clin Endocrinol Diabetes. 2011;119(7):414–8.
- Hershkovitz A, Polatov I, Beloosesky Y, Brill S. Factors affecting mortality of frail hipfractured elderly patients. Arch Gerontol Geriatr. 2010;51(2):113–6.
- Hoang-Kim A, Busse JW, Groll D, Karanicolas PJ, Schemitsch E. Co-morbidities in elderly patients with hip fracture: recommendations of the ISFR-IOF hip fracture outcomes working group. Arch Orthop Trauma Surg. 2014;134(2):189–95.
- Miao J, Brismar K, Nyren O, Ugarph-Morawski A, Ye W. Elevated hip fracture risk in type 1 diabetic patients: a population-based cohort study in Sweden. Diabetes Care. 2005;28(12): 2850–5.
- Omsland TK, Emaus N, Tell GS, Magnus JH, Ahmed LA, Holvik K, et al. Mortality following the first hip fracture in Norwegian women and men (1999-2008). A NOREPOS study. Bone. 2014;63:81–6.
- Paksima N, Koval KJ, Aharanoff G, Walsh M, Kubiak EN, Zuckerman JD, et al. Predictors of mortality after hip fracture: a 10-year prospective study. Bull NYU Hosp Jt Dis. 2008; 66(2):111–7.
- Reyes C, Estrada P, Nogues X, Orozco P, Cooper C, Diez-Perez A, et al. The impact of common co-morbidities (as measured using the Charlson index) on hip fracture risk in elderly men: a population-based cohort study. Osteoporos Int. 2014;25(6):1751–8.
- Vestergaard P, Rejnmark L, Mosekilde L. Increased mortality in patients with a hip fractureeffect of pre-morbid conditions and post-fracture complications. Osteoporos Int. 2007;18(12): 1583–93.
- Vestergaard P, Rejnmark L, Mosekilde L. Loss of life years after a hip fracture. Acta Orthop. 2009;80(5):525–30.
- Marsh JL, Slongo TF, Agel J, Broderick JS, Creevey W, DeCoster TA, et al. Fracture and dislocation classification compendium - 2007: Orthopaedic Trauma Association classification, database and outcomes committee. J Orthop Trauma. 2007;21(10 Suppl):S1–133.
- Janghorbani M, Van Dam RM, Willett WC, Hu FB. Systematic review of type 1 and type 2 diabetes mellitus and risk of fracture. Am J Epidemiol. 2007;166(5):495–505.
- Abrahamsen B, Vestergaard P. Declining incidence of hip fractures and the extent of use of anti-osteoporotic therapy in Denmark 1997-2006. Osteoporos Int. 2010;21(3):373–80.
- Melton 3rd LJ, Crowson CS, O'Fallon WM. Fracture incidence in Olmsted County, Minnesota: comparison of urban with rural rates and changes in urban rates over time. Osteoporos Int. 1999;9(1):29–37.
- Haentjens P, Autier P, Barette M, Boonen S. The economic cost of hip fractures among elderly women. A one-year, prospective, observational cohort study with matched-pair analysis. Belgian Hip Fracture Study Group. J Bone Joint Surg Am. 2001;83-A(4):493–500.
- Haentjens P, Lamraski G, Boonen S. Costs and consequences of hip fracture occurrence in old age: an economic perspective. Disabil Rehabil. 2005;27(18–19):1129–41.
- 19. Lawrence TM, White CT, Wenn R, Moran CG. The current hospital costs of treating hip fractures. Injury. 2005;36(1):88–91. discussion 2.
- Baker PN, Salar O, Ollivere BJ, Forward DP, Weerasuriya N, Moppett IK, et al. Evolution of the hip fracture population: time to consider the future? A retrospective observational analysis. BMJ Open. 2014;4(4), e004405.
- Vestergaard P, Rejnmark L, Mosekilde L. Relative fracture risk in patients with diabetes mellitus, and the impact of insulin and oral antidiabetic medication on relative fracture risk. Diabetologia. 2005;48(7):1292–9.
- Ahmed LA, Joakimsen RM, Berntsen GK, Fonnebo V, Schirmer H. Diabetes mellitus and the risk of non-vertebral fractures: the Tromso study. Osteoporos Int. 2006;17(4):495–500.
- Nicodemus KK, Folsom AR. Iowa Women's Health S. Type 1 and type 2 diabetes and incident hip fractures in postmenopausal women. Diabetes Care. 2001;24(7):1192–7.
- Janghorbani M, Feskanich D, Willett WC, Hu F. Prospective study of diabetes and risk of hip fracture: the Nurses' Health Study. Diabetes Care. 2006;29(7):1573–8.

- Christensen JO, Svendsen OL. Bone mineral in pre- and postmenopausal women with insulindependent and non-insulin-dependent diabetes mellitus. Osteoporos Int. 1999;10(4):307–11.
- Vestergaard P. Discrepancies in bone mineral density and fracture risk in patients with type 1 and type 2 diabetes–a meta-analysis. Osteoporos Int. 2007;18(4):427–44.
- Puar TH, Khoo JJ, Cho LW, Xu Y, Chen YT, Chuo AM, et al. Association between glycemic control and hip fracture. J Am Geriatr Soc. 2012;60(8):1493–7.
- Lipscombe LL, Jamal SA, Booth GL, Hawker GA. The risk of hip fractures in older individuals with diabetes: a population-based study. Diabetes Care. 2007;30(4):835–41.
- 29. Neumann T, Samann A, Lodes S, Kastner B, Franke S, Kiehntopf M, et al. Glycaemic control is positively associated with prevalent fractures but not with bone mineral density in patients with type 1 diabetes. Diabet Med. 2011;28(7):872–5.
- De Liefde II, van der Klift M, de Laet CE, van Daele PL, Hofman A, Pols HA. Bone mineral density and fracture risk in type-2 diabetes mellitus: the Rotterdam Study. Osteoporos Int. 2005;16(12):1713–20.
- 31. Koh WP, Wang R, Ang LW, Heng D, Yuan JM, Yu MC. Diabetes and risk of hip fracture in the Singapore Chinese Health Study. Diabetes Care. 2010;33(8):1766–70.
- 32. Bhattoa HP, Onyeka U, Kalina E, Balogh A, Paragh G, Antal-Szalmas P, et al. Bone metabolism and the 10-year probability of hip fracture and a major osteoporotic fracture using the country-specific FRAX algorithm in men over 50 years of age with type 2 diabetes mellitus: a case-control study. Clin Rheumatol. 2013;32(8):1161–7.
- 33. Oei L, Zillikens MC, Dehghan A, Buitendijk GH, Castano-Betancourt MC, Estrada K, et al. High bone mineral density and fracture risk in type 2 diabetes as skeletal complications of inadequate glucose control: the Rotterdam Study. Diabetes Care. 2013;36(6):1619–28.
- 34. Garg R, Chen Z, Beck T, Cauley JA, Wu G, Nelson D, et al. Hip geometry in diabetic women: implications for fracture risk. Metabolism. 2012;61(12):1756–62.
- 35. Ivers RQ, Cumming RG, Mitchell P, Peduto AJ, Blue Mountains Eye Study. Diabetes and risk of fracture: The Blue Mountains Eye Study. Diabetes Care. 2001;24(7):1198–203.
- Loke YK, Singh S, Furberg CD. Long-term use of thiazolidinediones and fractures in type 2 diabetes: a meta-analysis. CMAJ. 2009;180(1):32–9.
- Schwartz AV, Margolis KL, Sellmeyer DE, Vittinghoff E, Ambrosius WT, Bonds DE, et al. Intensive glycemic control is not associated with fractures or falls in the ACCORD randomized trial. Diabetes Care. 2012;35(7):1525–31.
- Winter WG. Nonoperative treatment of proximal femoral fractures in the demented, nonambulatory patient. Clin Orthop Relat Res. 1987;218:97–103.
- Parker MJ, Myles JW, Anand JK, Drewett R. Cost-benefit analysis of hip fracture treatment. J Bone Joint Surg. 1992;74(2):261–4.
- Ravikumar KJ, Marsh G. Internal fixation versus hemiarthroplasty versus total hip arthroplasty for displaced subcapital fractures of femur–13 year results of a prospective randomised study. Injury. 2000;31(10):793–7.
- Yu L, Wang Y, Chen J. Total hip arthroplasty versus hemiarthroplasty for displaced femoral neck fractures: meta-analysis of randomized trials. Clin Orthop Relat Res. 2012;470(8): 2235–43.
- 42. Zi-Sheng A, You-Shui G, Zhi-Zhen J, Ting Y, Chang-Qing Z. Hemiarthroplasty vs primary total hip arthroplasty for displaced fractures of the femoral neck in the elderly: a metaanalysis. J Arthroplasty. 2012;27(4):583–90.
- 43. Burgers PT, Van Geene AR, Van den Bekerom MP, Van Lieshout EM, Blom B, Aleem IS, et al. Total hip arthroplasty versus hemiarthroplasty for displaced femoral neck fractures in the healthy elderly: a meta-analysis and systematic review of randomized trials. Int Orthop. 2012;36(8):1549–60.
- Crawfurd EJ, Emery RJ, Hansell DM, Phelan M, Andrews BG. Capsular distension and intracapsular pressure in subcapital fractures of the femur. J Bone Joint Surg. 1988;70(2):195–8.
- 45. Stromqvist B, Nilsson LT, Egund N, Thorngren KG, Wingstrand H. Intracapsular pressures in undisplaced fractures of the femoral neck. J Bone Joint Surg. 1988;70(2):192–4.

- 46. Kyle RF, Gustilo RB, Premer RF. Analysis of six hundred and twenty-two intertrochanteric hip fractures. J Bone Joint Surg Am. 1979;61(2):216–21.
- 47. Vestergaard P, Rejnmark L, Mosekilde L. Has mortality after a hip fracture increased? J Am Geriatr Soc. 2007;55(11):1720–6.
- 48. Norris R, Parker M. Diabetes mellitus and hip fracture: a study of 5966 cases. Injury. 2011;42(11):1313–6.
- Kannus P, Palvanen M, Niemi S, Parkkari J, Jarvinen M. Increasing number and incidence of low-trauma ankle fractures in elderly people: Finnish statistics during 1970-2000 and projections for the future. Bone. 2002;31(3):430–3.
- Connolly JF, Csencsitz TA. Limb threatening neuropathic complications from ankle fractures in patients with diabetes. Clin Orthop Relat Res. 1998;348:212–9.
- White CB, Turner NS, Lee GC, Haidukewych GJ. Open ankle fractures in patients with diabetes mellitus. Clin Orthop Relat Res. 2003;414:37–44.
- 52. Costigan W, Thordarson DB, Debnath UK. Operative management of ankle fractures in patients with diabetes mellitus. Foot Ankle Int. 2007;28(1):32–7.
- Wukich DK, Joseph A, Ryan M, Ramirez C, Irrgang JJ. Outcomes of ankle fractures in patients with uncomplicated versus complicated diabetes. Foot Ankle Int. 2011;32(2):120–30.
- 54. Ganesh SP, Pietrobon R, Cecilio WA, Pan D, Lightdale N, Nunley JA. The impact of diabetes on patient outcomes after ankle fracture. J Bone Joint Surg Am. 2005;87(8):1712–8.
- Jones KB, Maiers-Yelden KA, Marsh JL, Zimmerman MB, Estin M, Saltzman CL. Ankle fractures in patients with diabetes mellitus. J Bone Joint Surg. 2005;87(4):489–95.
- Blotter RH, Connolly E, Wasan A, Chapman MW. Acute complications in the operative treatment of isolated ankle fractures in patients with diabetes mellitus. Foot Ankle Int. 1999; 20(11):687–94.
- McCormack RG, Leith JM. Ankle fractures in diabetics. Complications of surgical management. J Bone Joint Surg Br. 1998;80(4):689–92.
- Jani MM, Ricci WM, Borrelli Jr J, Barrett SE, Johnson JE. A protocol for treatment of unstable ankle fractures using transarticular fixation in patients with diabetes mellitus and loss of protective sensibility. Foot Ankle Int. 2003;24(11):838–44.
- 59. Lauge-Hansen N. Fractures of the ankle. II. Combined experimental-surgical and experimental-roentgenologic investigations. Arch Surg. 1950;60(5):957–85.
- 60. Bonar SK, Marsh JL. Tibial plafond fractures: changing principles of treatment. J Am Acad Orthop Surg. 1994;2(6):297–305.
- Brodsky JW, Rouse AM. Exostectomy for symptomatic bony prominences in diabetic charcot feet. Clin Orthop Relat Res. 1993;296:21–6.
- Cofield RH, Morrison MJ, Beabout JW. Diabetic neuroarthropathy in the foot: patient characteristics and patterns of radiographic change. Foot Ankle. 1983;4(1):15–22.
- Miller AG, Margules A, Raikin SM. Risk factors for wound complications after ankle fracture surgery. J Bone Joint Surg Am. 2012;94(22):2047–52.
- 64. Donken CC, Verhofstad MH, Edwards MJ, van Laarhoven CJ. Twenty-one-year follow-up of supination-external rotation type II-IV (OTA type B) ankle fractures: a retrospective cohort study. J Orthop Trauma. 2012;26(8):e108–14.
- Gougoulias N, Khanna A, Sakellariou A, Maffulli N. Supination-external rotation ankle fractures: stability a key issue. Clin Orthop Relat Res. 2010;468(1):243–51.
- 66. Loder RT. The influence of diabetes mellitus on the healing of closed fractures. Clin Orthop Relat Res. 1988;232:210–6.
- Flynn JM, Rodriguez-del Rio F, Piza PA. Closed ankle fractures in the diabetic patient. Foot Ankle Int. 2000;21(4):311–9.
- Wukich DK, Lowery NJ, McMillen RL, Frykberg RG. Postoperative infection rates in foot and ankle surgery: a comparison of patients with and without diabetes mellitus. J Bone Joint Surg Am. 2010;92(2):287–95.
- Ovaska MT, Makinen TJ, Madanat R, Huotari K, Vahlberg T, Hirvensalo E, et al. Risk factors for deep surgical site infection following operative treatment of ankle fractures. J Bone Joint Surg Am. 2013;95(4):348–53.

- SooHoo NF, Krenek L, Eagan MJ, Gurbani B, Ko CY, Zingmond DS. Complication rates following open reduction and internal fixation of ankle fractures. J Bone Joint Surg Am. 2009;91(5):1042–9.
- Low CK, Tan SK. Infection in diabetic patients with ankle fractures. Ann Acad Med Singapore. 1995;24(3):353–5.
- Liu J, Ludwig T, Ebraheim NA. Effect of the blood HbA1c level on surgical treatment outcomes of diabetics with ankle fractures. Orthop Surg. 2013;5(3):203–8.
- 73. Egol KA, Tejwani NC, Walsh MG, Capla EL, Koval KJ. Predictors of short-term functional outcome following ankle fracture surgery. J Bone Joint Surg Am. 2006;88(5):974–9.
- Olsen MA, Nepple JJ, Riew KD, Lenke LG, Bridwell KH, Mayfield J, et al. Risk factors for surgical site infection following orthopaedic spinal operations. J Bone Joint Surg Am. 2008; 90(1):62–9.
- Meding JB, Reddleman K, Keating ME, Klay A, Ritter MA, Faris PM, et al. Total knee replacement in patients with diabetes mellitus. Clin Orthop Relat Res. 2003;416:208–16.
- Yang K, Yeo SJ, Lee BP, Lo NN. Total knee arthroplasty in diabetic patients: a study of 109 consecutive cases. J Arthroplasty. 2001;16(1):102–6.
- Chen S, Anderson MV, Cheng WK, Wongworawat MD. Diabetes associated with increased surgical site infections in spinal arthrodesis. Clin Orthop Relat Res. 2009;467(7):1670–3.
- Kirkland KB, Briggs JP, Trivette SL, Wilkinson WE, Sexton DJ. The impact of surgical-site infections in the 1990s: attributable mortality, excess length of hospitalization, and extra costs. Infect Control Hosp Epidemiol. 1999;20(11):725–30.
- 79. Bolognesi MP, Marchant Jr MH, Viens NA, Cook C, Pietrobon R, Vail TP. The impact of diabetes on perioperative patient outcomes after total hip and total knee arthroplasty in the United States. J Arthroplasty. 2008;23(6 Suppl 1):92–8.
- Mirzayan R, Itamura JM, Vangsness Jr CT, Holtom PD, Sherman R, Patzakis MJ. Management of chronic deep infection following rotator cuff repair. J Bone Joint Surg Am. 2000;82-A(8): 1115–21.
- Partanen J, Syrjala H, Vahanikkila H, Jalovaara P. Impact of deep infection after hip fracture surgery on function and mortality. J Hosp Infect. 2006;62(1):44–9.
- Mangram AJ, Horan TC, Pearson ML, Silver LC, Jarvis WR. Guideline for Prevention of Surgical Site Infection, 1999. Centers for Disease Control and Prevention (CDC) Hospital Infection Control Practices Advisory Committee. Am J Infect Control. 1999;27(2):97–132; quiz 3–4; discussion 96.
- Dubory A, Giorgi H, Walter A, Bouyer B, Vassal M, Zairi F, et al. Surgical-site infection in spinal injury: incidence and risk factors in a prospective cohort of 518 patients. Eur Spine J. 2015;24(3):543–54.
- 84. Hikata T, Iwanami A, Hosogane N, Watanabe K, Ishii K, Nakamura M, et al. High preoperative hemoglobin A1c is a risk factor for surgical site infection after posterior thoracic and lumbar spinal instrumentation surgery. J Orthop Sci. 2014;19(2):223–8.
- Karunakar MA, Staples KS. Does stress-induced hyperglycemia increase the risk of perioperative infectious complications in orthopaedic trauma patients? J Orthop Trauma. 2010;24(12):752–6.
- 86. Richards JE, Kauffmann RM, Obremskey WT, May AK. Stress-induced hyperglycemia as a risk factor for surgical-site infection in nondiabetic orthopedic trauma patients admitted to the intensive care unit. J Orthop Trauma. 2013;27(1):16–21.
- Richards JE, Kauffmann RM, Zuckerman SL, Obremskey WT, May AK. Relationship of hyperglycemia and surgical-site infection in orthopaedic surgery. J Bone Joint Surg Am. 2012;94(13):1181–6.
- Kurtz S, Ong K, Lau E, Mowat F, Halpern M. Projections of primary and revision hip and knee arthroplasty in the United States from 2005 to 2030. J Bone Joint Surg Am. 2007; 89(4):780–5.
- Kurtz SM, Ong KL, Lau E, Bozic KJ. Impact of the economic downturn on total joint replacement demand in the United States: updated projections to 2021. J Bone Joint Surg Am. 2014;96(8):624–30.

- Pulido L, Ghanem E, Joshi A, Purtill JJ, Parvizi J. Periprosthetic joint infection: the incidence, timing, and predisposing factors. Clin Orthop Relat Res. 2008;466(7):1710–5.
- England SP, Stern SH, Insall JN, Windsor RE. Total knee arthroplasty in diabetes mellitus. Clin Orthop Relat Res. 1990;260:130–4.
- Lai K, Bohm ER, Burnell C, Hedden DR. Presence of medical comorbidities in patients with infected primary hip or knee arthroplasties. J Arthroplasty. 2007;22(5):651–6.
- Bozic KJ, Lau E, Kurtz S, Ong K, Berry DJ. Patient-related risk factors for postoperative mortality and periprosthetic joint infection in medicare patients undergoing TKA. Clin Orthop Relat Res. 2012;470(1):130–7.
- Marchant Jr MH, Viens NA, Cook C, Vail TP, Bolognesi MP. The impact of glycemic control and diabetes mellitus on perioperative outcomes after total joint arthroplasty. J Bone Joint Surg Am. 2009;91(7):1621–9.
- 95. Jamsen E, Nevalainen P, Eskelinen A, Huotari K, Kalliovalkama J, Moilanen T. Obesity, diabetes, and preoperative hyperglycemia as predictors of periprosthetic joint infection: a single-center analysis of 7181 primary hip and knee replacements for osteoarthritis. J Bone Joint Surg Am. 2012;94(14), e101.
- 96. Pinzur MS, Slovenkai MP, Trepman E, Shields NN, Diabetes Committee of American Orthopaedic Foot and Ankle Society. Guidelines for diabetic foot care: recommendations endorsed by the Diabetes Committee of the American Orthopaedic Foot and Ankle Society. Foot Ankle Int. 2005;26(1):113–9.
- Boulton AJ, Vileikyte L, Ragnarson-Tennvall G, Apelqvist J. The global burden of diabetic foot disease. Lancet. 2005;366(9498):1719–24.
- Kumar S, Ashe HA, Parnell LN, Fernando DJ, Tsigos C, Young RJ, et al. The prevalence of foot ulceration and its correlates in type 2 diabetic patients: a population-based study. Diabet Med. 1994;11(5):480–4.
- Adler AI, Boyko EJ, Ahroni JH, Stensel V, Forsberg RC, Smith DG. Risk factors for diabetic peripheral sensory neuropathy. Results of the Seattle Prospective Diabetic Foot Study. Diabetes Care. 1997;20(7):1162–7.
- 100. Boulton AJ, Vinik AI, Arezzo JC, Bril V, Feldman EL, Freeman R, et al. Diabetic neuropathies: a statement by the American Diabetes Association. Diabetes Care. 2005;28(4):956–62.
- 101. Abbott CA, Carrington AL, Ashe H, Bath S, Every LC, Griffiths J, et al. The North-West Diabetes Foot Care Study: incidence of, and risk factors for, new diabetic foot ulceration in a community-based patient cohort. Diabet Med. 2002;19(5):377–84.
- Young MJ, Breddy JL, Veves A, Boulton AJ. The prediction of diabetic neuropathic foot ulceration using vibration perception thresholds. A prospective study. Diabetes Care. 1994; 17(6):557–60.
- 103. Reiber GE, Vileikyte L, Boyko EJ, del Aguila M, Smith DG, Lavery LA, et al. Causal pathways for incident lower-extremity ulcers in patients with diabetes from two settings. Diabetes Care. 1999;22(1):157–62.
- 104. Pham H, Armstrong DG, Harvey C, Harkless LB, Giurini JM, Veves A. Screening techniques to identify people at high risk for diabetic foot ulceration: a prospective multicenter trial. Diabetes Care. 2000;23(5):606–11.
- Singh N, Armstrong DG, Lipsky BA. Preventing foot ulcers in patients with diabetes. JAMA. 2005;293(2):217–28.
- 106. McNeely MJ, Boyko EJ, Ahroni JH, Stensel VL, Reiber GE, Smith DG, et al. The independent contributions of diabetic neuropathy and vasculopathy in foot ulceration. How great are the risks? Diabetes Care. 1995;18(2):216–9.
- Perkins BA, Olaleye D, Zinman B, Bril V. Simple screening tests for peripheral neuropathy in the diabetes clinic. Diabetes Care. 2001;24(2):250–6.
- Mueller MJ, Diamond JE, Sinacore DR, Delitto A, Blair 3rd VP, Drury DA, et al. Total contact casting in treatment of diabetic plantar ulcers. Controlled clinical trial. Diabetes Care. 1989;12(6):384–8.
- 109. Armstrong DG, Nguyen HC, Lavery LA, van Schie CH, Boulton AJ, Harkless LB. Off-loading the diabetic foot wound: a randomized clinical trial. Diabetes Care. 2001;24(6):1019–22.

- 110. Wu SC, Crews RT, Armstrong DG. The pivotal role of offloading in the management of neuropathic foot ulceration. Curr Diab Rep. 2005;5(6):423–9.
- 111. Saltzman CL, Zimmerman MB, Holdsworth RL, Beck S, Hartsell HD, Frantz RA. Effect of initial weight-bearing in a total contact cast on healing of diabetic foot ulcers. J Bone Joint Surg Am. 2004;86-A(12):2714–9.
- 112. Armstrong DG, Lavery LA, Vazquez JR, Short B, Kimbriel HR, Nixon BP, et al. Clinical efficacy of the first metatarsophalangeal joint arthroplasty as a curative procedure for hallux interphalangeal joint wounds in patients with diabetes. Diabetes Care. 2003;26(12):3284–7.
- 113. Mueller MJ, Sinacore DR, Hastings MK, Strube MJ, Johnson JE. Effect of Achilles tendon lengthening on neuropathic plantar ulcers. A randomized clinical trial. J Bone Joint Surg Am. 2003;85-A(8):1436–45.
- 114. Mueller MJ, Sinacore DR, Hastings MK, Lott DJ, Strube MJ, Johnson JE. Impact of achilles tendon lengthening on functional limitations and perceived disability in people with a neuropathic plantar ulcer. Diabetes Care. 2004;27(7):1559–64.
- 115. Lin SS, Lee TH, Wapner KL. Plantar forefoot ulceration with equinus deformity of the ankle in diabetic patients: the effect of tendo-Achilles lengthening and total contact casting. Orthopedics. 1996;19(5):465–75.
- 116. Piaggesi A, Schipani E, Campi F, Romanelli M, Baccetti F, Arvia C, et al. Conservative surgical approach versus non-surgical management for diabetic neuropathic foot ulcers: a randomized trial. Diabet Med. 1998;15(5):412–7.
- 117. Ramsey SD, Newton K, Blough D, McCulloch DK, Sandhu N, Reiber GE, et al. Incidence, outcomes, and cost of foot ulcers in patients with diabetes. Diabetes Care. 1999;22(3):382–7.
- Tennvall GR, Apelqvist J, Eneroth M. Costs of deep foot infections in patients with diabetes mellitus. Pharmacoeconomics. 2000;18(3):225–38.
- Margolis DJ, Kantor J, Santanna J, Strom BL, Berlin JA. Risk factors for delayed healing of neuropathic diabetic foot ulcers: a pooled analysis. Arch Dermatol. 2000;136(12):1531–5.
- 120. Zimny S, Schatz H, Pfohl M. The effects of ulcer size on the wound radius reductions and healing times in neuropathic diabetic foot ulcers. Exp Clin Endocrinol Diabetes. 2004;112(4):191–4.
- 121. Zimny S, Pfohl M. Healing times and prediction of wound healing in neuropathic diabetic foot ulcers: a prospective study. Exp Clin Endocrinol Diabetes. 2005;113(2):90–3.
- 122. Nouvong A, Hoogwerf B, Mohler E, Davis B, Tajaddini A, Medenilla E. Evaluation of diabetic foot ulcer healing with hyperspectral imaging of oxyhemoglobin and deoxyhemoglobin. Diabetes Care. 2009;32(11):2056–61.
- 123. Reiber GE. The epidemiology of diabetic foot problems. Diabet Med. 1996;13 Suppl 1:S6-11.
- 124. Adler AI, Boyko EJ, Ahroni JH, Smith DG. Lower-extremity amputation in diabetes. The independent effects of peripheral vascular disease, sensory neuropathy, and foot ulcers. Diabetes Care. 1999;22(7):1029–35.
- 125. Adler AI, Erqou S, Lima TA, Robinson AH. Association between glycated haemoglobin and the risk of lower extremity amputation in patients with diabetes mellitus-review and metaanalysis. Diabetologia. 2010;53(5):840–9.
- 126. Oyibo SO, Jude EB, Tarawneh I, Nguyen HC, Harkless LB, Boulton AJ. A comparison of two diabetic foot ulcer classification systems: the Wagner and the University of Texas wound classification systems. Diabetes Care. 2001;24(1):84–8.
- 127. Hueston JT, Wilson WF. The aetiology of trigger finger explained on the basis of intratendinous architecture. Hand. 1972;4(3):257–60.
- 128. Sampson SP, Badalamente MA, Hurst LC, Seidman J. Pathobiology of the human A1 pulley in trigger finger. J Hand Surg. 1991;16(4):714–21.
- 129. Sbernardori MC, Bandiera P. Histopathology of the A1 pulley in adult trigger fingers. J Hand Surg Eur Vol. 2007;32(5):556–9.
- 130. Pandey A, Usman K, Reddy H, Gutch M, Jain N, Qidwai S. Prevalence of hand disorders in type 2 diabetes mellitus and its correlation with microvascular complications. Annu Med Health Sci Res. 2013;3(3):349–54.

- 131. Kiani J, Goharifar H, Moghimbeigi A, Azizkhani H. Prevalence and risk factors of five most common upper extremity disorders in diabetics. J Res Health Sci. 2014;14(1):92–5.
- 132. Leden I, Schersten B, Svensson B, Svensson M. Locomotor system disorders in diabetes mellitus. Increased prevalence of palmar flexortenosynovitis. Scand J Rheumatol. 1983;12(3):260–2.
- 133. Chammas M, Bousquet P, Renard E, Poirier JL, Jaffiol C, Allieu Y. Dupuytren's disease, carpal tunnel syndrome, trigger finger, and diabetes mellitus. J Hand Surg. 1995;20(1): 109–14.
- Gamstedt A, Holm-Glad J, Ohlson CG, Sundstrom M. Hand abnormalities are strongly associated with the duration of diabetes mellitus. J Intern Med. 1993;234(2):189–93.
- 135. Blyth MJ, Ross DJ. Diabetes and trigger finger. J Hand Surg. 1996;21(2):244-5.
- Griggs SM, Weiss AP, Lane LB, Schwenker C, Akelman E, Sachar K. Treatment of trigger finger in patients with diabetes mellitus. J Hand Surg. 1995;20(5):787–9.
- 137. Yosipovitch G, Yosipovitch Z, Karp M, Mukamel M. Trigger finger in young patients with insulin dependent diabetes. J Rheumatol. 1990;17(7):951–2.
- 138. Freiberg A, Mulholland RS, Levine R. Nonoperative treatment of trigger fingers and thumbs. J Hand Surg. 1989;14(3):553–8.
- 139. Ryzewicz M, Wolf JM. Trigger digits: principles, management, and complications. J Hand Surg. 2006;31(1):135–46.
- 140. Turowski GA, Zdankiewicz PD, Thomson JG. The results of surgical treatment of trigger finger. J Hand Surg. 1997;22(1):145–9.
- 141. Noble J, Heathcote JG, Cohen H. Diabetes mellitus in the aetiology of Dupuytren's disease. J Bone Joint Surg. 1984;66(3):322–5.
- 142. Burge P. Genetics of Dupuytren's disease. Hand Clin. 1999;15(1):63-71.
- 143. Al-Qattan MM. Factors in the pathogenesis of Dupuytren's contracture. J Hand Surg. 2006;31(9):1527–34.
- 144. Seyhan H, Kopp J, Schultze-Mosgau S, Horch RE. Increased metabolic activity of fibroblasts derived from cords compared with nodule fibroblasts sampling from patients with Dupuytren's contracture. Plast Reconstr Surg. 2006;117(4):1248–52.
- 145. Kopp J, Seyhan H, Muller B, Lanczak J, Pausch E, Gressner AM, et al. N-acetyl-L-cysteine abrogates fibrogenic properties of fibroblasts isolated from Dupuytren's disease by blunting TGF-beta signalling. J Cell Mol Med. 2006;10(1):157–65.
- 146. Loos B, Puschkin V, Horch RE. 50 years experience with Dupuytren's contracture in the Erlangen University Hospital—a retrospective analysis of 2919 operated hands from 1956 to 2006. BMC Musculoskelet Disord. 2007;8:60–72.
- 147. Arkkila PE, Kantola IM, Viikari JS, Ronnemaa T, Vahatalo MA. Dupuytren's disease in type 1 diabetic patients: a five-year prospective study. Clin Exp Rheumatol. 1996;14(1):59–65.
- 148. Renard E, Jacques D, Chammas M, Poirier JL, Bonifacj C, Jaffiol C, et al. Increased prevalence of soft tissue hand lesions in type 1 and type 2 diabetes mellitus: various entities and associated significance. Diabete Metab. 1994;20(6):513–21.
- 149. Gilpin D, Coleman S, Hall S, Houston A, Karrasch J, Jones N. Injectable collagenase Clostridium histolyticum: a new nonsurgical treatment for Dupuytren's disease. J Hand Surg. 2010;35(12):2027–38. 12.
- 150. van Rijssen AL, Gerbrandy FS, Ter Linden H, Klip H, Werker PM. A comparison of the direct outcomes of percutaneous needle fasciotomy and limited fasciectomy for Dupuytren's disease: a 6-week follow-up study. J Hand Surg. 2006;31(5):717–25.
- 151. Nydick JA, Olliff BW, Garcia MJ, Hess AV, Stone JD. A comparison of percutaneous needle fasciotomy and collagenase injection for Dupuytren disease. J Hand Surg. 2013;38(12): 2377–80.
- 152. Eckerdal D, Nivestam A, Dahlin LB. Surgical treatment of Dupuytren's disease—outcome and health economy in relation to smoking and diabetes. BMC Musculoskelet Disord. 2014;15:117.
- 153. Comi G, Lozza L, Galardi G, Ghilardi MF, Medaglini S, Canal N. Presence of carpal tunnel syndrome in diabetics: effect of age, sex, diabetes duration and polyneuropathy. Acta Diabetol Lat. 1985;22(3):259–62.

- 154. Awada AA, Bashi SA, Aljumah MA, Heffernan LP. Carpal tunnel syndrome in type 2 diabetic patients. Neurosciences. 2000;5(4):219–22.
- 155. Atroshi I, Gummesson C, Johnsson R, Ornstein E, Ranstam J, Rosen I. Prevalence of carpal tunnel syndrome in a general population. JAMA. 1999;282(2):153–8.
- 156. Becker J, Nora DB, Gomes I, Stringari FF, Seitensus R, Panosso JS, et al. An evaluation of gender, obesity, age and diabetes mellitus as risk factors for carpal tunnel syndrome. Clin Neurophysiol. 2002;113(9):1429–34.
- 157. Perkins BA, Olaleye D, Bril V. Carpal tunnel syndrome in patients with diabetic polyneuropathy. Diabetes Care. 2002;25(3):565–9.
- 158. Singh R, Gamble G, Cundy T. Lifetime risk of symptomatic carpal tunnel syndrome in type 1 diabetes. Diabet Med. 2005;22(5):625–30.
- 159. Kennedy JM, Zochodne DW. Impaired peripheral nerve regeneration in diabetes mellitus. J Peripher Nerv Syst. 2005;10(2):144–57.
- 160. Ozkul Y, Sabuncu T, Kocabey Y, Nazligul Y. Outcomes of carpal tunnel release in diabetic and non-diabetic patients. Acta Neurol Scand. 2002;106(3):168–72.
- 161. Agarwal V, Singh R, Sachdev A, Wiclaff, Shekhar S, Goel D. A prospective study of the longterm efficacy of local methyl prednisolone acetate injection in the management of mild carpal tunnel syndrome. Rheumatology. 2005;44(5):647–50.
- 162. Mondelli M, Padua L, Reale F, Signorini AM, Romano C. Outcome of surgical release among diabetics with carpal tunnel syndrome. Arch Phys Med Rehabil. 2004;85(1):7–13.
- 163. Neviaser JS. Adhesive capsulitis of the shoulder: a study of the pathological findings in periarthritis of the shoulder. JBJS. 1945;27A(2):211–22.
- 164. Reeves B. The natural history of the frozen shoulder syndrome. Scand J Rheumatol. 1975;4(4):193-6.
- 165. Lloyd-Roberts GC, French PR. Periarthritis of the shoulder: a study of the disease and its treatment. Br Med J. 1959;1(5137):1569–71.
- 166. Thomas SJ, McDougall C, Brown ID, Jaberoo MC, Stearns A, Ashraf R, et al. Prevalence of symptoms and signs of shoulder problems in people with diabetes mellitus. J Shoulder Elbow Surg. 2007;16(6):748–51.
- 167. Moren-Hybbinette I, Moritz U, Schersten B. The clinical picture of the painful diabetic shoulder–natural history, social consequences and analysis of concomitant hand syndrome. Acta Med Scand. 1987;221(1):73–82.
- 168. Arkkila PE, Kantola IM, Viikari JS, Ronnemaa T. Shoulder capsulitis in type I and II diabetic patients: association with diabetic complications and related diseases. Ann Rheum Dis. 1996;55(12):907–14.
- 169. Fisher L, Kurtz A, Shipley M. Association between cheiroarthropathy and frozen shoulder in patients with insulin-dependent diabetes mellitus. Br J Rheumatol. 1986;25(2):141–6.
- 170. Milgrom C, Novack V, Weil Y, Jaber S, Radeva-Petrova DR, Finestone A. Risk factors for idiopathic frozen shoulder. Isr Med Assoc J. 2008;10(5):361–4.
- 171. Bridgman JF. Periarthritis of the shoulder and diabetes mellitus. Ann Rheum Dis. 1972; 31(1):69–71.
- 172. Pal B, Anderson J, Dick WC, Griffiths ID. Limitation of joint mobility and shoulder capsulitis in insulin- and non-insulin-dependent diabetes mellitus. Br J Rheumatol. 1986;25(2):147–51.
- 173. Vishwanath V, Frank KE, Elmets CA, Dauchot PJ, Monnier VM. Glycation of skin collagen in type I diabetes mellitus. Correlation with long-term complications. Diabetes. 1986;35(8): 916–21.
- 174. Cagliero E, Apruzzese W, Perlmutter GS, Nathan DM. Musculoskeletal disorders of the hand and shoulder in patients with diabetes mellitus. Am J Med. 2002;112(6):487–90.
- 175. Lundberg BJ. The frozen shoulder. Clinical and radiographical observations. The effect of manipulation under general anesthesia. Structure and glycosaminoglycan content of the joint capsule. Local bone metabolism. Acta Orthop Scand Suppl. 1969;119:1–59.
- 176. McCully SP, Kumar N, Lazarus MD, Karduna AR. Internal and external rotation of the shoulder: effects of plane, end-range determination, and scapular motion. J Shoulder Elbow Surg. 2005;14(6):602–10.

- 177. Shaffer B, Tibone JE, Kerlan RK. Frozen shoulder. J Bone Joint Surg Am. 1992;74(5): 738-46.
- 178. Levine WN, Kashyap CP, Bak SF, Ahmad CS, Blaine TA, Bigliani LU. Nonoperative management of idiopathic adhesive capsulitis. J Shoulder Elbow Surg. 2007;16(5):569–73.
- 179. Diercks RL, Stevens M. Gentle thawing of the frozen shoulder: a prospective study of supervised neglect versus intensive physical therapy in seventy-seven patients with frozen shoulder syndrome followed up for two years. J Shoulder Elbow Surg. 2004;13(5):499–502.
- Dacre JE, Beeney N, Scott DL. Injections and physiotherapy for the painful stiff shoulder. Ann Rheum Dis. 1989;48(4):322–5.
- Ryans I, Montgomery A, Galway R, Kernohan WG, McKane R. A randomized controlled trial of intra-articular triamcinolone and/or physiotherapy in shoulder capsulitis. Rheumatology. 2005;44(4):529–35.
- 182. Carette S, Moffet H, Tardif J, Bessette L, Morin F, Fremont P, et al. Intraarticular corticosteroids, supervised physiotherapy, or a combination of the two in the treatment of adhesive capsulitis of the shoulder: a placebo-controlled trial. Arthritis Rheum. 2003;48(3):829–38.
- 183. Buchbinder R, Hoving JL, Green S, Hall S, Forbes A, Nash P. Short course prednisolone for adhesive capsulitis (frozen shoulder or stiff painful shoulder): a randomised, double blind, placebo controlled trial. Ann Rheum Dis. 2004;63(11):1460–9.
- Binder A, Hazleman BL, Parr G, Roberts S. A controlled study of oral prednisolone in frozen shoulder. Br J Rheumatol. 1986;25(3):288–92.
- 185. Grant JA, Schroeder N, Miller BS, Carpenter JE. Comparison of manipulation and arthroscopic capsular release for adhesive capsulitis: a systematic review. J Shoulder Elbow Surg. 2013;22(8):1135–45.

Chapter 5 Impact of Diabetes on Periodontal Disease

E. Xiao, Yingying Wu, and Dana T. Graves

Diabetes mellitus is characterized by high blood glucose levels [1]. Type 1 diabetes mellitus (T1DM) caused by a deficiency of insulin is most commonly due to autoimmune destruction of pancreatic β -cells. Type 2 diabetes mellitus (T2DM) is caused by insulin resistance and a failure of β -cells to compensate for the reduced effect of insulin stimulation [2, 3].

Metabolic Changes in Diabetes That Contribute to Periodontal Bone Loss

The primary morbidity and mortality associated with DM is from diabetic complications. There are several underlying mechanisms that link diabetic complications including those that affect bone. In particular, advanced glycation end products,

E. Xiao

Y. Wu

Department of Periodontics, School of Dental Medicine, University of Pennsylvania, Philadelphia, PA 19104, USA

State Key Laboratory of Oral Disease, West China School of Stomatology Sichuan University, Chengdu, Sichuan 610041, China e-mail: yywdentist@163.com

D.T. Graves, D.D.S., D.M.Sc. (🖂) Department of Periodontics, School of Dental Medicine, University of Pennsylvania, Philadelphia, PA 19104, USA e-mail: dtgraves@dental.upenn.edu

Department of Oral and Maxillofacial Surgery, Peking University School and Hospital of Stomatology, Beijing 100081, China

Department of Periodontics, School of Dental Medicine, University of Pennsylvania, Philadelphia, PA 19104, USA e-mail: xiaoe1986@vip.163.com

[©] Springer International Publishing Switzerland 2016 B. Lecka-Czernik, J.L. Fowlkes (eds.), *Diabetic Bone Disease*, DOI 10.1007/978-3-319-16402-1_5

excessive production of reactive oxygen species (ROS), and enhanced cytokine expression such as tumor necrosis factor have been experimentally linked to diabetes-enhanced periodontal bone loss [4]. These underlined causes are also associated with several other diabetic complications.

Both systemic and local inflammation have been shown to be increased by diabetes. Macrophages in adipose tissue produce inflammatory mediators that contribute but are not solely responsible for insulin resistance and may promote diabetic complications [5]. Pro-inflammatory factors such as TNF- α , IL-1 β , IL-6, and IL-18 are reported to be increased in diabetes mellitus [6, 7] and contribute to destruction of beta cells in the pancreas and diabetic complications [8, 9].

Oxidative stress plays an important role in diabetic complications. Hyperglycemia leads to generation of ROS by overproduction of superoxides in mitochondria [10]. Increased intracellular ROS enhances inflammation [11]. Furthermore, increased ROS combined with depleted antioxidant defense mechanisms render tissues more susceptible to oxidative damage [12]. ROS also function as intracellular second messengers that regulate signaling cascades and modulate gene expression [13].

AGEs are formed by the nonenzymatic creation of glucose and other glycating compounds derived from both glucose and increased fatty acid oxidation. Proteins may be structurally modified by glycosylation thereby affecting their function. Alternatively, AGEs bind to AGE receptors inducing production of inflammatory mediators such as tumor necrosis alpha (TNF) and activation of NF- κ B [14]. There are a number of receptors for AGEs that are linked to increased inflammation including receptor for AGE (RAGE).

Impact of Diabetes on Bone

Diabetes negatively impacts bone metabolism and affects bone quantity and quality [15]. T1DM and T2DM have increased fracture risk. Bone mineral density (BMD) is reduced in T1DM [16]. T2DM has equal or increased BMD [17] but still has greater fracture risk [18]. Diabetes increases the severity and risk of periodontitis, the most common lytic disease of bone and a frequent complication of diabetes [9, 19]. Both increased bone resorption and reduced bone formation have been linked to the negative effect of diabetes on periodontal bone loss [20].

Periodontal Disease

Periodontitis involves the loss of supporting structure for the tooth consisting of connective tissue attachment and bone. Periodontitis is one of the most widespread oral diseases [21]. Severe periodontitis, which may result in tooth loss, is found in 5-20 % of most adult population worldwide. The 2010 National Health and Nutrition Examination Survey estimates that over 47 % of American adults have

periodontitis [22]. Almost 25 % of adults in Australia aged 35–54 years have moderate to severe periodontitis [23]. Children and adolescents also have forms of periodontitis such as aggressive periodontitis [24].

Under physiological conditions, the tooth is supported by alveolar bone and periodontal ligament in the tooth socket. The surface of the alveolar bone and periodontal ligament are covered by gingival connective tissue and epithelium. The periodontium consists of gingival tissue, periodontal ligament, and alveolar bone. Gingival tissues act as a physical barrier to reduce invasion by bacteria that form a biofilm on the tooth surface. Bacteria or their products penetrate the epithelial barrier of the gingiva to penetrate the connective tissue and thereby stimulate inflammation that induces an inflammatory response. The primary distinction between gingivitis and periodontitis is the irreversible loss of bone and connective tissue which attaches the gingiva to the tooth surface [20] (Fig. 5.1). Thus, gingivitis may cause reversible damage that is repaired whereas periodontitis involves irreversible damage to the periodontal tissues [20]. It is thought that an essential component of periodontitis is suppression of repair processes by inflammation. Periodontal disease includes both periodontitis and gingivitis. In some cases, periodontitis is referred to as moderate to severe periodontal disease. For the purpose of this review, we will use the term periodontitis to mean inflammation that causes irreversible bone loss.



Fig. 5.1 (*Right side*) Healthy periodontal tissue. (*Left half side*) Effects of periodontitis with subgingival bacterial plaque formation, gingival tissue inflammation, periodontal attachment loss, periodontal pocket formation, and alveolar bone loss

Pathogenic Mechanisms

The chronic inflammatory condition of periodontitis is induced by pathogenic biofilm or dental plaque, which accumulates on the tooth surface. Over 500 bacterial species have been detected in dental plaque; however, the composition of the causative bacterial species is still debated [25-28]. The classic periodontal pathogens are Gram-negative bacteria such as Porphyromonas gingivalis, Tannerella forsythia, and Treponema denticola [29]. Recent studies have identified other bacteria that may be pathogenic [30]. Although bacteria are necessary for periodontal disease to take place, a susceptible host is also needed [27]. The predisposing factors that make an individual susceptible have not yet been defined though periodontal disease increases considerably in aged humans and animals including dogs and mice and by conditions that increase inflammation such as diabetes [31]. The inflammatory process occurring in periodontitis is characterized by the infiltration of leukocytes, which limit the level of bacterial invasion and at the same time may be harmful to the periodontal tissue [31]. The destruction of periodontal ligament and bone is thought to be the result of a disruption of the homeostatic balance between the host response and bacteria that result in inflammation in close proximity to bone [31-33]. The host immune response to bacteria or their products stimulates production of osteoclastogenic factors by immune cells and cells of osteoblastic lineage, which then induce bone loss. Periodontal disease in humans and in experimental animal models is linked to both the innate and adaptive immune response. Several studies have reported that individuals with periodontitis exhibit increased levels of interleukin-1 (IL-1), tumor necrosis factor- α (TNF- α), and interleukin-6 (IL-6) in gingiva and crevicular fluid in the gingival sulcus [34]. Both genetic deletion and specific inhibition of these cytokines have been found to reduce periodontal disease progression [34–37]. Similarly, mediators of the adaptive immune response are elevated in individuals with periodontal disease and inhibition or genetic deletion of mediators such as receptor activator of nuclear factor kappa-B ligand (RANKL) and interferon- γ (IFN- γ) result in reduced periodontal disease progression [36, 38, 39]. Our laboratory has recently shown that the production of factors by osteoblasts and osteocytes also contributes to osteoclast formation and activity in periodontal disease (unpublished data). Evidence that the host response plays a critical role was also shown when treatment with a prostaglandin inhibitor reduced periodontitis-related bone loss [40] or inhibition of inflammatory cytokines such as IL-1 and TNF α [35, 37]. Thus, periodontitis is a complex disease, where multiple causal risk factors play simultaneous and interactive roles. Periodontal bone loss is affected by the bacterial biofilm which forms, ability of bacteria or their products to pass through the epithelial barrier into connective tissue, the status of the host response, and the presence of environmental stressors and/or systemic disease such as diabetes [41–43].

Impact of Diabetes on Periodontal Disease

Diabetes and periodontitis are two chronic diseases that are biologically linked [44, 45]. Periodontitis is one of the first clinical manifestations of diabetes [19]. Diabetes is an important risk factor for periodontitis [4, 46]. The risk of periodontitis is increased approximately 2–4 times in diabetic versus nondiabetic subjects [4, 47]. In one study, periodontitis was found in 60 % of T1DM patients compared to 15 % without diabetes [48]. Patients with diabetes are at higher risk of severe periodontitis compared with nondiabetic subjects [49]. A study in African Americans found 70 % T2DM patients had moderate periodontitis and 29 % had severe disease, which is significantly higher than the prevalence of 11 % without diabetes [50]. The severity of periodontitis is directly proportional to the level of glucose control [48, 51].

Impact of Diabetes on Gingiva/Gingivitis

Periodontitis is preceded by various stages of gingival inflammation referred to as gingivitis. Gingivitis in T1DM children and adolescents is twice that of matched control subjects [52]. Similarly, higher rates of gingival inflammation occur in T2DM adults compared to adults without diabetes. Nearly, 64 % of patients with T2DM have gingivitis compared with 50 % of subjects without diabetes [52]. The degree of metabolic control of diabetes is an important factor in the development and progression of gingivitis; good metabolic control significantly reduces gingivitis [53, 54].

Diabetes may affect periodontal disease through a number of different avenues including an impaired antibacterial defense. However, studies investigating whether diabetes causes a change in oral flora have had inconsistent results and the issue has not yet been settled. An alternative explanation is that diabetes alters the inflammatory response to periodontal pathogens. This is supported by studies examining the response to a well-defined inoculum of periodopathic bacteria [55]. Under normal circumstances, inflammation normally resolves through an active process regulated by cellular signals [56, 57]. However, resolution of the inflammation is impaired in the diabetic animals [58]. Thus, one explanation for enhanced risk of periodontal disease in diabetic individuals is the presence of a greater inflammatory response to bacteria on the tooth surface that stimulates the formation of osteoclasts and suppresses the repair process [55, 59].

In animal models, diabetes leads to increased production of TNF in the epithelium and connective tissue [60]. Periodontal infection causes an increase in apoptosis of gingival epithelial cells, fibroblasts, and bone lining cells that is significantly enhanced by diabetes in a caspase-3-dependent mechanism [60]. This may be important because diabetes-enhanced inflammation and apoptosis negatively impact the gingiva by causing a loss of epithelial barrier function and inhibiting repair processes [61, 62]. High levels of TNF- α can stimulate the expression of pro-apoptotic genes in diabetics [58, 63].

Impact of Diabetes on Periodontal Ligament/Loss of Attachment

The gingiva is attached to the root surface of teeth by collagen bundles that integrate with the tooth surface. Loss of this attachment is one of the hallmarks of periodontitis and occurs in conjunction with bone loss. Diabetes increases attachment loss, which is worsened by poor glycemic control [48, 64]. More than 25 % of T1DM patients with poor metabolic control have sites with moderate to severe signs of periodontitis compared to only 10 % of subjects with good metabolic control [52]. Moreover, the severity is proportional to the duration of diabetes [51, 65].

Impact of Diabetes on Alveolar Bone Loss

Periodontitis is the most common osteolytic disease in humans and is aggravated by diabetes [66]. Diabetes potentiates the severity of periodontitis and accelerates bone resorption. The number of sites with bone loss in poorly controlled T1DM individuals is twice that of nondiabetic subjects [67]. Animal studies demonstrate that periodontitis is increased threefold in T1DM rats compared to normal rats [68] and is significantly higher in T2DM rats [61]. The risk and degree of alveolar bone loss is positively correlated with the lack of metabolic control [69].

Effect of Diabetes on Osteoclasts in Periodontitis

Bone remodeling begins with the activity of osteoclasts, followed by new bone formation through the activity of osteoblasts. Under physiological conditions, the two activities are coupled, but the two processes are uncoupled in pathologic processes. Human studies generally indicate that diabetes mellitus increases osteoclastogenesis. Individuals with DM generally have increased systemic levels of bone resorption markers as indicated by higher circulating levels of tartrate-resistant acid phosphatase [70]. Animal studies show similar results [71]. T2DM rats have increased osteoclastic bone resorption in periodontal bone compared to normoglycemic controls [61]. Increased inflammation, ROS, and AGEs are thought to increase osteoclast activity.

Increased Inflammation Activate Osteoclasts in Diabetes

Diabetes has been shown to enhance osteoclast formation in inflammatory areas. Type 2 diabetic rats have a \sim 2–4-fold increase of the osteoclast number after bacterial infection by oral inoculation of a periodontal pathogen or ligature-induced periodontitis compared with control rats [59–61]. T1DM with periodontitis also have a 2–4-fold

increase in the number of osteoclasts compared to nondiabetic rats with periodontitis [72]. A higher degree of inflammation and a more persistent inflammatory response following periodontitis [55, 73] may lead to greater stimuli for osteoclastogenesis.

In diabetic mice, TNF- α , macrophage colony-stimulating factor, RANKL, and vascular endothelial growth factor-A (VEGF-A) are up-regulated which can directly promote osteoclast differentiation and activation [74, 75]. Diabetes increases TNF levels that has been shown to prevent downregulation of genes associated with host defense, apoptosis, cell signaling and activity, and coagulation/hemostasis/complement [58]. Similarly, patients with periodontitis and diabetes have significantly higher levels of IL-1 β , TNF- α , and prostaglandin E₂, which result in more prolonged osteoclast formation and activity [76, 77]. Enhancement of IL-17, IL-23 in periodontitis in type 1 diabetic subjects, and overexpression of IL-1β, IL-6 in type 2 diabetic patients have been reported, which result in enhanced osteoclastogenesis and prolonged duration of inflammatory responses [78, 79]. T2DM patients with periodontal disease have increased levels of TNF- α and IL-6, which was also associated with increased dyslipidemia and lipid peroxidation [80]. Increased fatty acid levels in diabetes mellitus may also enhance osteoclastogenesis [81]. In addition to increasing inflammation, diabetes also impairs the resolution of periodontal inflammation. The importance of resolving inflammation has been demonstrated by treatment of animals with periodontitis with resolvins [82] or by treatment of diabetic animals with TNF inhibitors [58, 59].

Increased ROS Activate Osteoclasts in Diabetes

High levels of ROS contribute to diabetes-related periodontitis. Invading bacteria trigger the release of cytokines and chemokines that induce neutrophil recruitment and activity and which subsequently release ROS in periodontal tissues [83, 84]. Neutrophils from diabetic patients produce more superoxide than neutrophils from normal subjects [14]. The imbalance between production of ROS and antioxidant defense results in increased oxidative stress [85]. The formation of AGEs also increases oxidative stress in the periodontal tissues. It has been shown that ROS such as superoxide and hydrogen peroxide activate osteoclasts and promote osteoclast formation [86]. A related process, lipid peroxidation, is also linked to increased periodontal disease T2DM and a greater inflammatory response in the periodontal tissues in humans [80, 87]. Patients with T2DM show elevated mitochondrial ROS which promotes RANKL-mediated osteoclast differentiation and function [88].

Increased AGEs Activate Osteoclasts in Diabetes

In vitro studies suggest that hyperglycemia predisposes to increased osteoclast formation [89]. AGEs also increase osteoclasts activity [90]. Osteoclast-like cells express RAGE, which serves as a positive factor to regulate the osteoclast formation [91].
Mice that lack the RAGE have increased bone mass and decreased osteoclast numbers compared to wild-type mice [91], supporting the concept that AGEs contribute to osteoclast formation in diabetes. Diabetes enhances the formation of AGEs in the periodontium and increases expression of RAGE [92]. The level of AGEs in the gingiva is increased in both type 1 and type 2 diabetes-associated periodontitis [93]. It has been shown that AGE–RAGE interaction on monocytes activates the transcription factor NF- κ B, which alters the phenotype of the monocyte/macrophage and results in the increased production of proinflammatory cytokines [94]. RAGE stimulation may contribute to osteoclastogenesis via increased expression of receptor activator of RANKL and downregulation of osteoprotegerin (OPG) [95].

RANKL interacts with its receptor on the surface of osteoclast precursors to induce osteoclast formation and activity. OPG inhibits osteoclast formation by binding to RANKL [96]. A number of studies focusing on osteoclastogenesis-related factors have reported elevated expression of RANKL and TNF in diabetes-associated periodontal tissues [59, 96]. Studies with animals suggest that RANKL/OPG ratios and the level of other inflammatory cytokines such as TNF are critical mediators for the enhanced osteoclastogenesis in diabetes in periodontal disease [59, 97]. TNF levels and high RANKL/OPG ratios in the periodontium in humans are negatively influenced by poor glycemic control in subjects with diabetes [80, 98]

Effect of Diabetes on Osteoblasts in Periodontitis

Bone resorption is followed by a period of bone formation, a coupling process that limits the amount of net bone loss, which occurs during the resolution of inflammation in the periodontium [20]. We found that T2DM rats do not generate a burst of bone formation that normally occurs following induction of periodontal disease [61]. Inflammation plays an important role in this effect by limiting repair of resorbed bone. This occurs by reducing osteoblast numbers through decreased proliferation of precursors and greater apoptosis of mature osteoblasts [59, 63]. These studies suggest a molecular basis for the negative impact of T2DM on bone by the effect of diabetes-enhanced inflammation on suppressing the expression of factors such as fibroblast growth factor or bone morphogenetic proteins that are needed for new bone formation.

Diabetes Inhibits Osteoblasts Differentiation and Function

Diabetes also interferes with the bone formation by reducing the expression of transcription factors that regulate osteoblast differentiation [99]. In T1DM and T2DM rats, osteoblasts exhibit lower alkaline phosphatase activity and mineralized matrix formation [100, 101]. Inflammation has a significant effect on bone [102, 103]. Inflammation impairs the function of bone-forming osteoblasts by suppressing mature osteoblast function such as the production of bone matrix [104]. One of the striking features of diabetes is elevated levels of inflammatory mediators, particularly TNF [76]. Diabetic animals have higher levels of TNF in bone, which is associated with reduced bone formation and repair. TNF blocks the differentiation of osteoblasts, where inflammation is thought to be present [105]. This is consistent with reports that TNF inhibits differentiation of osteoblasts in vitro [106, 107] and also interferes with bone morphogenetic protein signaling [108].

Periodontal infection-induced alveolar bone loss in diabetic subjects is accompanied by enhanced expression of RAGE and production of AGEs in the gingival tissue [109]. AGEs have been shown to interfere with osteoblast differentiation and induce apoptosis of osteoblasts in diabetes via the mitogen-activated protein kinase and cytosolic apoptotic pathway [110]. When AGEs are applied to osseous wounds in normal animals, the rate of healing is reduced in half, indicating that AGEs, which are elevated in diabetes, contribute to impaired bone formation [111]. In addition, RAGE is expressed at higher levels in osteoblasts in diabetic conditions rendering diabetic animals even more sensitive to the effects of AGEs [111].

Mesenchymal stem cells (MSCs) represent a precursor pool of osteoblasts that are bone-forming cells. Inflammation, which is elevated in diabetic bone healing [112], has a significant effect on reducing MSC differentiation [113]. A mechanism through which inflammation affects MSC is through induction of NF- κ B activation. Increased NF- κ B activity interferes with wnt-stimulated MSC differentiation by increasing beta-catenin degradation [102]. Moreover, TNF suppresses activation of the Osx promoter [114]. This interferes with MSC differentiation to osteoblasts since osterix is needed in early steps of differentiation. AGEs also inhibit MSC differentiation. One mechanism involves AGE-induced up-regulation of ROS in MSCs that leads to reduced MSC differentiation [115, 116]. In human MSCs and mouse stromal ST2 cells, AGEs suppress osteogenic differentiation [117]. T2DM mice have fewer MSCs and these MSCs appear to have poor homing capability to injury sites [118]. T1DM rats have more numerous apoptotic cells in the bone marrow, and the size of osteoprogenitor pool is significantly reduced [101]. Thus, diabetes reduces the number of progenitors and inhibits differentiation of MSC to osteoblasts.

Diabetes Promote Osteoblast Apoptosis

Apoptosis of osteoblasts is significantly increased by diabetes. Diabetes leads to the up-regulation of pro-apoptotic mediators including TNF- α , AGEs, and the formation of ROS [119]. TNF can induce apoptosis by binding to the TNF receptor-1, which triggers the initial events in apoptosis [120]. Some of the detrimental effects of diabetes-enhanced TNF- α levels may be due to the induction of cell death by triggering caspase activity. Caspases are a family of cysteine proteases that can act as either initiators (caspase-2, 8, and 9) or executioners (caspase-3, 6, and 7) of apoptosis [121]. Caspase-3 appears to play a central role in bacteria and lipopoly-saccharide (LPS)-mediated apoptosis [122, 123]. Additionally, TNF stimulates the

expression of several pro-apoptotic genes, many of which are regulated by the proapoptotic transcription factor, forkhead box-O1 (FOXO1) [124]. There is evidence that both diabetes and bacterial infection in periodontitis enhances apoptosis of osteoblastic cells to reduce osseous coupling [63, 125], which may involve stimulatory signals from both the innate and adaptive immune response [112, 126]. CMLcollagen, one of the AGEs found in bone, stimulates apoptosis of bone-lining cells in vivo and in various osteoblastic cell cultures mediated through RAGE [127] via the MAP kinase pathway [110].

The production of ROS is another mechanism of diabetes increasing apoptosis. Persistent inflammation and hyperglycemia leads to the cellular accumulation of ROS, which is linked to diabetic complications [66, 128]. Increased oxidative stress in periodontal tissue has been shown to lead to greater osteoblast apoptosis [129] and involve activation of caspase-3 [130].

Diabetes also increases loss of cells in the periodontal ligament from periodontal infection [61, 131]. This is significant since the periodontal ligament is a rich source of cells capable of differentiating into osteoblasts. Studies in diabetic animals indicate that diabetes causes a more than twofold induction of genes that regulate the apoptosis of osteoblasts and fibroblasts following bacterial infection and a fivefold increase in osteoblast apoptosis [122, 132].

Blocking apoptosis by treatment of diabetic rats with a caspase-3 inhibitor significantly increases the number of osteoblasts, which in turn leads to significantly greater amounts of new bone formation. Furthermore, the number of osteoclasts and their activity is increased by treatment with a caspase-3 inhibitor with the net effect increasing bone formation due to greater bone coupling. This is consistent with previous findings that a pancaspase inhibitor reduces apoptosis and increases new bone formation following bacterial infection [122]. Taken together the results indicate that bacterial infection in diabetic animals has a significant impact on periodontal disease through enhanced apoptosis of osteoblasts or their precursors.

Conclusion

In summary, individuals with diabetes mellitus have increased risk and severity of periodontal disease [9, 19, 133]. Periodontitis is one of the first clinical manifestations of diabetes [19]. Diabetes aggravates periodontitis by an increase in the inflammatory response to bacterial infection and reducing the capacity to down-regulate inflammation [9, 55]. There is a direct link between persistent hyperglycemia, an exaggerated inflammatory response to periodontal pathogens and periodontal bone loss [31, 66]. The impact of diabetes on the periodontium involves inflammation associated with both the innate and adaptive immune response [8, 31]. Diabetes-enhanced inflammation increases osteoclastogenesis and decreases reparative bone formation. A number of factors are increased by diabetes including RANKL, AGEs, ROS, and TNF that stimulate osteoclasts. Moreover, diabetes prolongs inflammation leading to longer periods on osteoclast activity as well as interfering with



Fig. 5.2 Impact of diabetic mellitus on periodontitis. Diabetes increases inflammation (increased cytokine production such as TNF α), and the production of AGEs and ROS that can affect both osteoclasts and osteoblasts. These factors may increase osteoclast formation and activity and inhibit osteoblast differentiation, activity, and survival

subsequent bone coupling. The negative effect of inflammation on bone coupling is likely to be an important factor in the disease process [59, 112] as inflammatory cytokines interferes with bone morphogenetic protein and Wnt signaling and also stimulates osteoblast apoptosis [104, 108, 134, 135]. The mechanisms by which diabetes affects periodontitis are summarized in Fig. 5.2.

References

- Bell GI, Polonsky KS. Diabetes mellitus and genetically programmed defects in beta-cell function. Nature. 2001;414(6865):788–91.
- Kahn SE. Clinical review 135: the importance of beta-cell failure in the development and progression of type 2 diabetes. J Clin Endocrinol Metab. 2001;86(9):4047–58.
- Cavaghan MK, Ehrmann DA, Polonsky KS. Interactions between insulin resistance and insulin secretion in the development of glucose intolerance. J Clin Invest. 2000;106(3):329–33.
- Preshaw PM, Bissett SM. Periodontitis: oral complication of diabetes. Endocrinol Metab Clin North Am. 2013;42(4):849–67.
- Cruz NG, Sousa LP, Sousa MO, Pietrani NT, Fernandes AP, Gomes KB. The linkage between inflammation and type 2 diabetes mellitus. Diabetes Res Clin Pract. 2013;99(2):85–92.
- 6. Johnson DR, O'Connor JC, Satpathy A, Freund GG. Cytokines in type 2 diabetes. Vitam Horm. 2006;74:405–41.
- Hotamisligil GS, Shargill NS, Spiegelman BM. Adipose expression of tumor necrosis factoralpha: direct role in obesity-linked insulin resistance. Science. 1993;259(5091):87–91.
- Nikolajczyk BS, Jagannathan-Bogdan M, Shin H, Gyurko R. State of the union between metabolism and the immune system in type 2 diabetes. Genes Immun. 2011;12(4):239–50.
- 9. Graves DT, Kayal RA. Diabetic complications and dysregulated innate immunity. Front Biosci. 2008;13:1227–39.
- Giacco F, Brownlee M. Oxidative stress and diabetic complications. Circ Res. 2010; 107(9):1058–70.

- 11. Pitocco D, Zaccardi F, Di Stasio E, Romitelli F, Santini SA, Zuppi C, et al. Oxidative stress, nitric oxide, and diabetes. Rev Diabet Stud. 2010;7(1):15–25.
- Niedowicz DM, Daleke DL. The role of oxidative stress in diabetic complications. Cell Biochem Biophys. 2005;43(2):289–330.
- 13. Evans JL, Goldfine ID, Maddux BA, Grodsky GM. Oxidative stress and stress-activated signaling pathways: a unifying hypothesis of type 2 diabetes. Endocr Rev. 2002;23(5): 599–622.
- Brownlee M. Biochemistry and molecular cell biology of diabetic complications. Nature. 2001;414(6865):813–20.
- 15. Valkusz Z. Diabetes and osteoporosis. Orv Hetil. 2011;152(29):1161-6.
- Holmberg AH, Johnell O, Nilsson PM, Nilsson J, Berglund G, Akesson K. Risk factors for fragility fracture in middle age. A prospective population-based study of 33,000 men and women. Osteoporos Int. 2006;17(7):1065–77.
- 17. Vestergaard P. Discrepancies in bone mineral density and fracture risk in patients with type 1 and type 2 diabetes—a meta-analysis. Osteoporos Int. 2007;18(4):427–44.
- Okazaki R. Management of osteoporosis in diabetes mellitus. Nihon Rinsho. 2009; 67(5):1003–10.
- 19. Lamster IB, Lalla E, Borgnakke WS, Taylor GW. The relationship between oral health and diabetes mellitus. J Am Dent Assoc. 2008;139(Suppl):19S-24.
- Graves DT, Li J, Cochran DL. Inflammation and uncoupling as mechanisms of periodontal bone loss. J Dent Res. 2011;90(2):143–53.
- Borrell LN, Papapanou PN. Analytical epidemiology of periodontitis. J Clin Periodontol. 2005;32 Suppl 6:132–58.
- Eke PI, Dye BA, Wei L, Thornton-Evans GO, Genco RJ. Prevalence of periodontitis in adults in the United States: 2009 and 2010. J Dent Res. 2012;91(10):914–20.
- 23. Dye BA. Global periodontal disease epidemiology. Periodontol 2000. 2012;58(1):10-25.
- Bascones-Martinez A, Gonzalez-Febles J, Sanz-Esporrin J. Diabetes and periodontal disease. Review of the literature. Am J Dent. 2014;27(2):63–7.
- Tanaka K, Miyake Y, Hanioka T, Arakawa M. Relationship between IL1 gene polymorphisms and periodontal disease in Japanese women. DNA Cell Biol. 2014;33(4):227–33.
- Tsaousoglou P, Nietzsche S, Cachovan G, Sculean A, Eick S. Antibacterial activity of moxifloxacin on bacteria associated with periodontitis within a biofilm. J Med Microbiol. 2014; 63(Pt 2):284–92.
- Pihlstrom BL, Michalowicz BS, Johnson NW. Periodontal diseases. Lancet. 2005; 366(9499):1809–20.
- Junemann S, Prior K, Szczepanowski R, Harks I, Ehmke B, Goesmann A, et al. Bacterial community shift in treated periodontitis patients revealed by ion torrent 16S rRNA gene amplicon sequencing. PLoS One. 2012;7(8):e41606.
- 29. Socransky S, Haffajee A, Cugini M, Smith C, Kent RJ. Microbial complexes in subgingival plaque. J Clin Periodontol. 1998;25:134–44.
- Griffen AL, Beall CJ, Campbell JH, Firestone ND, Kumar PS, Yang ZK, et al. Distinct and complex bacterial profiles in human periodontitis and health revealed by 16S pyrosequencing. ISME J. 2012;6(6):1176–85.
- Graves DT, Oates T, Garlet GP. Review of osteoimmunology and the host response in endodontic and periodontal lesions. J Oral Microbiol. 2011;3.
- Bostrom L, Bergstrom J, Dahlen G, Linder LE. Smoking and subgingival microflora in periodontal disease. J Clin Periodontol. 2001;28(3):212–9.
- 33. Hajishengallis G, Liang S, Payne MA, Hashim A, Jotwani R, Eskan MA, et al. Lowabundance biofilm species orchestrates inflammatory periodontal disease through the commensal microbiota and complement. Cell Host Microbe. 2011;10(5):497–506.
- 34. Garlet GP. Destructive and protective roles of cytokines in periodontitis: a re-appraisal from host defense and tissue destruction viewpoints. J Dent Res. 2010;89(12):1349–63.

- Assuma R, Oates T, Cochran D, Amar S, Graves D. IL-1 and TNF antagonists inhibit the inflammatory response and bone loss in experimental periodontitis. J Immunol. 1998; 160:403–9.
- 36. Baker P, Dixon M, Evans R, Dufour L, Johnson E, Roopenian D. CD4(+) T cells and the proinflammatory cytokines gamma interferon and interleukin-6 contribute to alveolar bone loss in mice. Infect Immun. 1999;67:2804–9.
- Delima A, Spyros K, Amar S, Graves DT. Inflammation and tissue loss caused by periodontal pathogens is reduced by IL-1 antagonists. J Infect Dis. 2002;186:511–6.
- Teng Y, Nguyen H, Gao X, Kong Y, Gorczynski R, Singh B, et al. Functional human T-cell immunity and osteoprotegerin ligand control alveolar bone destruction in periodontal infection. J Clin Invest. 2000;106:R59–67.
- Han X, Kawai T, Eastcott JW, Taubman MA. Bacterial-responsive B lymphocytes induce periodontal bone resorption. J Immunol. 2006;176(1):625–31.
- 40. Williams RC, Jeffcoat MK, Kaplan ML, Goldhaber P, Johnson HG, Wechter WJ. Flurbiprofen: a potent inhibitor of alveolar bone resorption in beagles. Science. 1985;227(4687):640–2.
- 41. Hajishengallis G. Aging and its impact on innate immunity and inflammation: implications for periodontitis. J Oral Biosci. 2014;56(1):30–7.
- Loos BG, Papantonopoulos G. Molecular biotypes for periodontal diseases? J Dent Res. 2013;92(12):1056–7.
- 43. Albert DA, Ward A, Allweiss P, Graves DT, Knowler WC, Kunzel C, et al. Diabetes and oral disease: implications for health professionals. Ann N Y Acad Sci. 2012;1255:1–15.
- 44. Bullon P, Newman HN, Battino M. Obesity, diabetes mellitus, atherosclerosis and chronic periodontitis: a shared pathology via oxidative stress and mitochondrial dysfunction? Periodontol 2000. 2014;64(1):139–53.
- 45. Tsai C, Hayes C, Taylor GW. Glycemic control of type 2 diabetes and severe periodontal disease in the US adult population. Community Dent Oral Epidemiol. 2002;30(3):182–92.
- 46. Loe H. Periodontal disease. The sixth complication of diabetes mellitus. Diabetes Care. 1993;16(1):329–34.
- 47. Apoorva SM, Sridhar N, Suchetha A. Prevalence and severity of periodontal disease in type 2 diabetes mellitus (non-insulin-dependent diabetes mellitus) patients in Bangalore city: an epidemiological study. J Indian Soc Periodontol. 2013;17(1):25–9.
- Poplawska-Kita A, Siewko K, Szpak P, Krol B, Telejko B, Klimiuk PA, et al. Association between type 1 diabetes and periodontal health. Adv Med Sci. 2014;59(1):126–31.
- 49. Leite RS, Marlow NM, Fernandes JK. Oral health and type 2 diabetes. Am J Med Sci. 2013;345(4):271–3.
- Fernandes JK, Wiegand RE, Salinas CF, Grossi SG, Sanders JJ, Lopes-Virella MF, et al. Periodontal disease status in Gullah African Americans with type 2 diabetes living in South Carolina. J Periodontol. 2009;80(7):1062–8.
- Al-Khabbaz AK. Type 2 diabetes mellitus and periodontal disease severity. Oral Health Prev Dent. 2014;12(1):77–82.
- 52. Ryan ME, Carnu O, Kamer A. The influence of diabetes on the periodontal tissues. J Am Dent Assoc. 2003;134 Spec No:34S–40.
- Albandar JM, Tinoco EM. Global epidemiology of periodontal diseases in children and young persons. Periodontol 2000. 2002;29:153–76.
- 54. Katagiri S, Nagasawa T, Kobayashi H, Takamatsu H, Bharti P, Izumiyama H, et al. Improvement of glycemic control after periodontal treatment by resolving gingival inflammation in type 2 diabetic patients with periodontal disease. J Diabetes Investig. 2012;3(4): 402–9.
- 55. Naguib G, Al-Mashat H, Desta T, Graves D. Diabetes prolongs the inflammatory response to a bacterial stimulus through cytokine dysregulation. J Invest Dermatol. 2004;123:87–92.
- 56. Spite M, Serhan CN. Novel lipid mediators promote resolution of acute inflammation: impact of aspirin and statins. Circ Res. 2010;107(10):1170–84.

- 57. Van Dyke TE. The management of inflammation in periodontal disease. J Periodontol. 2008;79(8 Suppl):1601–8.
- 58. Andriankaja OM, Galicia J, Dong G, Xiao W, Alawi F, Graves DT. Gene expression dynamics during diabetic periodontitis. J Dent Res. 2012;91(12):1160–5.
- Pacios S, Kang J, Galicia J, Gluck K, Patel H, Ovaydi-Mandel A, et al. Diabetes aggravates periodontitis by limiting repair through enhanced inflammation. FASEB J. 2012;26(4): 1423–30.
- 60. Kang J, de Brito Bezerra B, Pacios S, Andriankaja O, Li Y, Tsiagbe V, et al. Aggregatibacter actinomycetemcomitans infection enhances apoptosis in vivo through a caspase-3-dependent mechanism in experimental periodontitis. Infect Immun. 2012;80(6):2247–56.
- Liu R, Bal HS, Desta T, Krothapalli N, Alyassi M, Luan Q, et al. Diabetes enhances periodontal bone loss through enhanced resorption and diminished bone formation. J Dent Res. 2006;85(6):510–4.
- 62. Li S, Dong G, Moschidis A, Ortiz J, Benakanakere MR, Kinane DF, et al. P. gingivalis modulates keratinocytes through FOXO transcription factors. PLoS One. 2013;8(11):e78541.
- Pacios S, Andriankaja O, Kang J, Alnammary M, Bae J, de Brito Bezerra B, et al. Bacterial infection increases periodontal bone loss in diabetic rats through enhanced apoptosis. Am J Pathol. 2013;183(6):1928–35.
- Lalla E, Cheng B, Lal S, Kaplan S, Softness B, Greenberg E, et al. Diabetes mellitus promotes periodontal destruction in children. J Clin Periodontol. 2007;34(4):294–8.
- 65. Deshpande K, Jain A, Sharma R, Prashar S, Jain R. Diabetes and periodontitis. J Indian Soc Periodontol. 2010;14(4):207–12.
- 66. Chapple IL, Genco R, Working group 2 of the joint EFP/AAP workshop. Diabetes and periodontal diseases: consensus report of the Joint EFP/AAP Workshop on Periodontitis and Systemic Diseases. J Periodontol. 2013;84(4 Suppl):S106–12.
- Tervonen T, Karjalainen K, Knuuttila M, Huumonen S. Alveolar bone loss in type 1 diabetic subjects. J Clin Periodontol. 2000;27(8):567–71.
- Kim JH, Lee DE, Choi SH, Cha JH, Bak EJ, Yoo YJ. Diabetic characteristics and alveolar bone loss in streptozotocin- and streptozotocin-nicotinamide-treated rats with periodontitis. J Periodontal Res. 2014;49(6):792–800.
- Taylor GW, Burt BA, Becker MP, Genco RJ, Shlossman M, Knowler WC, et al. Non-insulin dependent diabetes mellitus and alveolar bone loss progression over 2 years. J Periodontol. 1998;69(1):76–83.
- 70. Suzuki K, Kurose T, Takizawa M, Maruyama M, Ushikawa K, Kikuyama M, et al. Osteoclastic function is accelerated in male patients with type 2 diabetes mellitus: the preventive role of osteoclastogenesis inhibitory factor/osteoprotegerin (OCIF/OPG) on the decrease of bone mineral density. Diabetes Res Clin Pract. 2005;68(2):117–25.
- Suzuki K, Ishida H, Takeshita N, Taguchi Y, Sugimoto C, Nosaka K, et al. Circulating levels of tartrate-resistant acid phosphatase in rat models of non-insulin-dependent diabetes mellitus. J Diabetes Complications. 1998;12(3):176–80.
- 72. Silva JA, Lopes Ferrucci D, Peroni LA, de Paula Ishi E, Rossa-Junior C, Carvalho HF, et al. Periodontal disease-associated compensatory expression of osteoprotegerin is lost in type 1 diabetes mellitus and correlates with alveolar bone destruction by regulating osteoclastogenesis. Cells Tissues Organs. 2012;196(2):137–50.
- 73. Graves DT, Naguib G, Lu H, Leone C, Hsue H, Krall E. Inflammation is more persistent in type 1 diabetic mice. J Dent Res. 2005;84:324–8.
- 74. Kayal RA, Tsatsas D, Bauer MA, Allen B, Al-Sebaei MO, Kakar S, et al. Diminished bone formation during diabetic fracture healing is related to the premature resorption of cartilage associated with increased osteoclast activity. J Bone Miner Res. 2007;22(4):560–8.
- Jeffcoate WJ, Game F, Cavanagh PR. The role of proinflammatory cytokines in the cause of neuropathic osteoarthropathy (acute Charcot foot) in diabetes. Lancet. 2005;366(9502): 2058–61.

- Alblowi J, Kayal RA, Siqueria M, McKenzie E, Krothapalli N, McLean J, et al. High levels of tumor necrosis factor-alpha contribute to accelerated loss of cartilage in diabetic fracture healing. Am J Pathol. 2009;175(4):1574–85.
- Graves DT, Liu R, Alikhani M, Al-Mashat H, Trackman PC. Diabetes-enhanced inflammation and apoptosis—impact on periodontal pathology. J Dent Res. 2006;85(1):15–21.
- Silva JA, Ferrucci DL, Peroni LA, Abrahao PG, Salamene AF, Rossa-Junior C, et al. Sequential IL-23 and IL-17 and increased Mmp8 and Mmp14 expression characterize the progression of an experimental model of periodontal disease in type 1 diabetes. J Cell Physiol. 2012;227(6):2441–50.
- 79. Duarte PM, de Oliveira MC, Tambeli CH, Parada CA, Casati MZ, Nociti Jr FH. Overexpression of interleukin-1beta and interleukin-6 may play an important role in periodontal breakdown in type 2 diabetic patients. J Periodontal Res. 2007;42(4):377–81.
- 80. Bastos AS, Graves DT, Loureiro AP, Rossa Junior C, Abdalla DS, Faulin TD, et al. Lipid peroxidation is associated with the severity of periodontal disease and local inflammatory markers in patients with type 2 diabetes. J Clin Endocrinol Metab. 2012;97(8):E1353–62.
- Drosatos-Tampakaki Z, Drosatos K, Siegelin Y, Gong S, Khan S, Van Dyke T, et al. Palmitic acid and DGAT1 deficiency enhance osteoclastogenesis, while oleic acid-induced triglyceride formation prevents it. J Bone Miner Res. 2014;29(5):1183–95.
- Hasturk H, Kantarci A, Goguet-Surmenian E, Blackwood A, Andry C, Serhan CN, et al. Resolvin E1 regulates inflammation at the cellular and tissue level and restores tissue homeostasis in vivo. J Immunol. 2007;179(10):7021–9.
- Gyurko R, Van Dyke TE. The role of polyunsaturated omega-3 fatty acid eicosapentaenoic acid-derived resolvin E1 (RvE1) in bone preservation. Crit Rev Immunol. 2014;34(4): 347–57.
- Omori K, Ohira T, Uchida Y, Ayilavarapu S, Batista Jr EL, Yagi M, et al. Priming of neutrophil oxidative burst in diabetes requires preassembly of the NADPH oxidase. J Leukoc Biol. 2008;84(1):292–301.
- 85. Thomas B, Ramesh A, Suresh S, Prasad BR. A comparative evaluation of antioxidant enzymes and selenium in the serum of periodontitis patients with diabetes mellitus type 2. Contemp Clin Dent. 2013;4(2):176–80.
- Garrett IR, Boyce BF, Oreffo RO, Bonewald L, Poser J, Mundy GR. Oxygen-derived free radicals stimulate osteoclastic bone resorption in rodent bone in vitro and in vivo. J Clin Invest. 1990;85(3):632–9.
- Collin HL, Sorsa T, Meurman JH, Niskanen L, Salo T, Ronka H, et al. Salivary matrix metalloproteinase (MMP-8) levels and gelatinase (MMP-9) activities in patients with type 2 diabetes mellitus. J Periodontal Res. 2000;35(5):259–65.
- Ha H, Kwak HB, Lee SW, Jin HM, Kim HM, Kim HH, et al. Reactive oxygen species mediate RANK signaling in osteoclasts. Exp Cell Res. 2004;301(2):119–27.
- Catalfamo DL, Britten TM, Storch DL, Calderon NL, Sorenson HL, Wallet SM. Hyperglycemia induced and intrinsic alterations in type 2 diabetes-derived osteoclast function. Oral Dis. 2013;19(3):303–12.
- 90. Miyata T, Kawai R, Taketomi S, Sprague SM. Possible involvement of advanced glycation end-products in bone resorption. Nephrol Dial Transplant. 1996;11 Suppl 5:54–7.
- Ding KH, Wang ZZ, Hamrick MW, Deng ZB, Zhou L, Kang B, et al. Disordered osteoclast formation in RAGE-deficient mouse establishes an essential role for RAGE in diabetes related bone loss. Biochem Biophys Res Commun. 2006;340(4):1091–7.
- 92. Brownlee M, Cerami A, Vlassara H. Advanced glycosylation end products in tissue and the biochemical basis of diabetic complications. N Engl J Med. 1988;318(20):1315–21.
- Zizzi A, Tirabassi G, Aspriello SD, Piemontese M, Rubini C, Lucarini G. Gingival advanced glycation end-products in diabetes mellitus-associated chronic periodontitis: an immunohistochemical study. J Periodontal Res. 2013;48(3):293–301.
- Mealey BL, Oates TW. Diabetes mellitus and periodontal diseases. J Periodontol. 2006; 77(8):1289–303.

- 95. Lamster IB. Diabetes and oral health. What's their relationship? Diabetes Self Manag. 2012;29(3):30, 32-4.
- 96. Lappin DF, Eapen B, Robertson D, Young J, Hodge PJ. Markers of bone destruction and formation and periodontitis in type 1 diabetes mellitus. J Clin Periodontol. 2009;36(8): 634–41.
- Mahamed DA, Marleau A, Alnaeeli M, Singh B, Zhang X, Penninger JM, et al. G(-) anaerobes-reactive CD4+ T-cells trigger RANKL-mediated enhanced alveolar bone loss in diabetic NOD mice. Diabetes. 2005;54(5):1477–86.
- Santos VR, Lima JA, Goncalves TE, Bastos MF, Figueiredo LC, Shibli JA, et al. Receptor activator of nuclear factor-kappa B ligand/osteoprotegerin ratio in sites of chronic periodontitis of subjects with poorly and well-controlled type 2 diabetes. J Periodontol. 2010; 81(10):1455–65.
- Lu H, Kraut D, Gerstenfeld L, Graves D. Diabetes interferes with the bone formation by affecting the expression of transcription factors that regulate osteoblast differentiation. Endocrinology. 2003;144:346–52.
- 100. Hamann C, Goettsch C, Mettelsiefen J, Henkenjohann V, Rauner M, Hempel U, et al. Delayed bone regeneration and low bone mass in a rat model of insulin-resistant type 2 diabetes mellitus is due to impaired osteoblast function. Am J Physiol Endocrinol Metab. 2011; 301(6):E1220–8.
- 101. Weinberg E, Maymon T, Moses O, Weinreb M. Streptozotocin-induced diabetes in rats diminishes the size of the osteoprogenitor pool in bone marrow. Diabetes Res Clin Pract. 2014;103(1):35–41.
- 102. Chang J, Liu F, Lee M, Wu B, Ting K, Zara JN, et al. NF-kappaB inhibits osteogenic differentiation of mesenchymal stem cells by promoting beta-catenin degradation. Proc Natl Acad Sci U S A. 2013;110(23):9469–74.
- Schett G. Effects of inflammatory and anti-inflammatory cytokines on the bone. Eur J Clin Invest. 2011;41(12):1361–6.
- 104. Chang J, Wang Z, Tang E, Fan Z, McCauley L, Franceschi R, et al. Inhibition of osteoblastic bone formation by nuclear factor-kappaB. Nat Med. 2009;15(6):682–9.
- 105. Pacifici R. The immune system and bone. Arch Biochem Biophys. 2010;503(1):41-53.
- 106. Gilbert L, He X, Farmer P, Boden S, Kozlowski M, Rubin J, et al. Inhibition of osteoblast differentiation by tumor necrosis factor-alpha. Endocrinology. 2000;141(11):3956–64.
- Lencel P, Delplace S, Hardouin P, Magne D. TNF-alpha stimulates alkaline phosphatase and mineralization through PPARgamma inhibition in human osteoblasts. Bone. 2011;48(2): 242–9.
- 108. Guo R, Yamashita M, Zhang Q, Zhou Q, Chen D, Reynolds DG, et al. Ubiquitin ligase Smurf1 mediates tumor necrosis factor-induced systemic bone loss by promoting proteasomal degradation of bone morphogenetic signaling proteins. J Biol Chem. 2008;283(34): 23084–92.
- 109. Lalla E, Lamster IB, Schmidt AM. Enhanced interaction of advanced glycation end products with their cellular receptor RAGE: implications for the pathogenesis of accelerated periodontal disease in diabetes. Ann Periodontol. 1998;3(1):13–9.
- Alikhani M, Alikhani Z, Boyd C, MacLellan CM, Raptis M, Liu R, et al. Advanced glycation end products stimulate osteoblast apoptosis via the MAP kinase and cytosolic apoptotic pathways. Bone. 2007;40(2):345–53.
- 111. Santana RB, Xu L, Chase HB, Amar S, Graves DT, Trackman PC. A role for advanced glycation end products in diminished bone healing in type 1 diabetes. Diabetes. 2003;52(6): 1502–10.
- 112. Liu R, Bal HS, Desta T, Behl Y, Graves DT. Tumor necrosis factor-alpha mediates diabetesenhanced apoptosis of matrix-producing cells and impairs diabetic healing. Am J Pathol. 2006;168(3):757–64.

- 113. Gilbert LC, Rubin J, Nanes MS. The p55 TNF receptor mediates TNF inhibition of osteoblast differentiation independently of apoptosis. Am J Physiol Endocrinol Metab. 2005;288(5): E1011–8.
- 114. Lu X, Beck Jr GR, Gilbert LC, Camalier CE, Bateman NW, Hood BL, et al. Identification of the homeobox protein Prx1 (MHox, Prrx-1) as a regulator of osterix expression and mediator of tumor necrosis factor alpha action in osteoblast differentiation. J Bone Miner Res. 2011;26(1):209–19.
- 115. Stolzing A, Sellers D, Llewelyn O, Scutt A. Diabetes induced changes in rat mesenchymal stem cells. Cells Tissues Organs. 2010;191(6):453–65.
- 116. Kume S, Kato S, Yamagishi S, Inagaki Y, Ueda S, Arima N, et al. Advanced glycation endproducts attenuate human mesenchymal stem cells and prevent cognate differentiation into adipose tissue, cartilage, and bone. J Bone Miner Res. 2005;20(9):1647–58.
- 117. Notsu M, Yamaguchi T, Okazaki K, Tanaka K, Ogawa N, Kanazawa I, et al. Advanced glycation end product 3 (AGE3) suppresses the mineralization of mouse stromal ST2 cells and human mesenchymal stem cells by increasing TGF-beta expression and secretion. Endocrinology. 2014;155(7):2402–10.
- 118. Shin L, Peterson DA. Impaired therapeutic capacity of autologous stem cells in a model of type 2 diabetes. Stem Cells Transl Med. 2012;1(2):125–35.
- 119. Ponugoti B, Dong G, Graves DT. Role of forkhead transcription factors in diabetes-induced oxidative stress. Exp Diabetes Res. 2012;2012:939751.
- 120. Graves DT, Liu R, Oates TW. Diabetes-enhanced inflammation and apoptosis: impact on periodontal pathosis. Periodontol 2000. 2007;45:128–37.
- 121. Chowdhury I, Tharakan B, Bhat GK. Caspases—an update. Comp Biochem Physiol B Biochem Mol Biol. 2008;151(1):10–27.
- 122. Al-Mashat HA, Kandru S, Liu R, Behl Y, Desta T, Graves DT. Diabetes enhances mRNA levels of proapoptotic genes and caspase activity, which contribute to impaired healing. Diabetes. 2006;55(2):487–95.
- 123. Alikhani M, Alikhani Z, He H, Liu R, Popek BI, Graves DT. Lipopolysaccharides indirectly stimulate apoptosis and global induction of apoptotic genes in fibroblasts. J Biol Chem. 2003;278(52):52901–8.
- Alikhani M, Alikhani Z, Graves DT. FOXO1 functions as a master switch that regulates gene expression necessary for tumor necrosis factor-induced fibroblast apoptosis. J Biol Chem. 2005;280(13):12096–102.
- 125. Fu YW, He HB, Ou JG. Osteoblast apoptosis in experimental diabetic periodontitis in rats. Hua Xi Kou Qiang Yi Xue Za Zhi. 2009;27(3):252–5, 259.
- 126. Behl Y, Siqueira M, Ortiz J, Li J, Desta T, Faibish D, et al. Activation of the acquired immune response reduces coupled bone formation in response to a periodontal pathogen. J Immunol. 2008;181(12):8711–8.
- 127. Hein G, Weiss C, Lehmann G, Niwa T, Stein G, Franke S. Advanced glycation end product modification of bone proteins and bone remodelling: hypothesis and preliminary immunohis-tochemical findings. Ann Rheum Dis. 2006;65(1):101–4.
- 128. Fu YW, He HB. Apoptosis of periodontium cells in streptozototocin- and ligature-induced experimental diabetic periodontitis in rats. Acta Odontol Scand. 2013;71(5):1206–15.
- 129. Ohnishi T, Bandow K, Kakimoto K, Machigashira M, Matsuyama T, Matsuguchi T. Oxidative stress causes alveolar bone loss in metabolic syndrome model mice with type 2 diabetes. J Periodontal Res. 2009;44(1):43–51.
- Wang GW, Klein JB, Kang YJ. Metallothionein inhibits doxorubicin-induced mitochondrial cytochrome c release and caspase-3 activation in cardiomyocytes. J Pharmacol Exp Ther. 2001;298(2):461–8.
- 131. Isaka J, Ohazama A, Kobayashi M, Nagashima C, Takiguchi T, Kawasaki H, et al. Participation of periodontal ligament cells with regeneration of alveolar bone. J Periodontol. 2001;72(3):314–23.

- 132. He H, Liu R, Desta T, Leone C, Gerstenfeld LC, Graves DT. Diabetes causes decreased osteoclastogenesis, reduced bone formation, and enhanced apoptosis of osteoblastic cells in bacteria stimulated bone loss. Endocrinology. 2004;145(1):447–52.
- 133. Lalla E, Papapanou PN. Diabetes mellitus and periodontitis: a tale of two common interrelated diseases. Nat Rev Endocrinol. 2011;7(12):738–48.
- 134. Walsh NC, Reinwald S, Manning CA, Condon KW, Iwata K, Burr DB, et al. Osteoblast function is compromised at sites of focal bone erosion in inflammatory arthritis. J Bone Miner Res. 2009;24(9):1572–85.
- 135. Walsh NC, Gravallese EM. Bone remodeling in rheumatic disease: a question of balance. Immunol Rev. 2010;233(1):301–12.

Chapter 6 Biomarkers of Diabetic Bone Disease

Mishaela R. Rubin

Introduction

Diabetes mellitus is the most common endocrine disorder with a prevalence of approximately 327 million worldwide [1]. Substantial evidence exists that in addition to the well-known complications of diabetes, such as neuropathy, nephropathy, and retinopathy, increased fracture risk is an important morbidity [2, 3]. Individuals with type 1 diabetes (T1D) have a very high risk of hip fracture, approximately six times greater than those without diabetes [2, 3]. Although the fracture risk is not as high in type 2 diabetes (T2D), it is nevertheless increased as well [4, 5]. A meta-analysis of 12 studies reported a relative risk of 1.7 (95 % CI: 1.3–2.2) for hip fracture in both men and women with T2D [2]. The risk of all clinical fractures was also increased, with a summary RR of 1.2 (95 % CI: 1.0–1.5) [2]. Subsequent studies have reported similar results [6, 7], with a direct association between the duration of diabetes and increased fracture risk [8]. Fractures in all diabetic individuals are particularly problematic because they are associated with poor fracture healing, greater morbidity [9, 10], and greater healthcare costs [11] as compared to nondiabetics.

Fracture risk can be explained by a decrease in measurement of bone mineral density (BMD) by dual energy X-ray absorptiometry in T1D but not in T2D [3]. In T1D, hip and spine BMD are reduced compared with normative reference populations [3] or with healthy controls [12]. As in broader populations, reduced BMD is associated with higher fracture prevalence among those with T1D [13]. In contrast, BMD is generally higher in those with T2D when compared to nondiabetics [3].

M.R. Rubin, M.D. (\boxtimes)

Department of Medicine, Metabolic Bone Diseases Unit, College of Physicians and Surgeons Columbia University, 630 West 168th Street, PH8W-864, New York, NY 10032, USA e-mail: mrr6@columbia.edu

[©] Springer International Publishing Switzerland 2016 B. Lecka-Czernik, J.L. Fowlkes (eds.), *Diabetic Bone Disease*,

DOI 10.1007/978-3-319-16402-1_6

In a meta-analysis, Vestergaard reported an increased Z-score of +0.41 at the spine and +0.27 at the hip associated with T2D [3]. Similarly, the WHO fracture risk assessment tool, FRAX, a key clinical instrument, has been shown to underestimate fracture risk in diabetes in several US study cohorts and in a large clinical cohort in Manitoba, Canada [4, 5]. In the Canadian study, diabetes was a significant predictor of subsequent major osteoporotic fracture (adjusted hazard ratio [aHR] = 1.61, 95 % confidence interval [CI] 1.42–1.83) and hip fractures (aHR 6.27, 95 % CI 3.62– 10.87 in those aged <65 years, aHR 2.22, 95 % CI 1.71–2.90 in those \geq 65 years) even when adjusted for competing mortality. These data have led to discussion of how to accommodate the risk associated with T2D in the FRAX algorithm [14].

The paradox of higher BMD in association with increased fractures in T2D could potentially be explained by more frequent trauma, as diabetes is associated with an increased frequency of falls. However, in studies of diabetes and fracture that controlled for fall frequency, diabetes still remained independently associated with increased fracture risk [6, 15]. Thiazolidinediones (TZD) use might also be considered as an explanation, since it has been proposed that these agents divert mesenchymal stem cells from the osteogenic to the adipocytic lineage and are associated with bone loss and increased fracture risk, particularly in women [16]. However, TZD use does not fully account for the increased risk of fracture observed with diabetes, since most studies included substantial observation time prior to the wide-spread use of these medications.

It is thus apparent that regardless of falls and TZD use, fracture risk is increased in both T1D and T2D. Yet BMD and FRAX, the clinical mainstays for predicting fractures, do not fully capture fracture risk in these populations. It is therefore imperative to identify biochemical markers in diabetes that can predict fracture risk independent of bone density assessment.

Glycemia

Chronic hyperglycemia, as reflected by HbA1c, is a marker of increased fracture risk in observational studies of patients with T2D. In the ARIC study, an increased risk of fractures was observed among diabetic individuals who had an average HbA1c ≥ 8 % as compared to diabetics with HbA1c <8 % (HR 1.63; 95 % CI 1.09–2.44) [17]. Similarly, in the Rotterdam study, participants with HbA1c levels ≥ 7.5 % had a 62 % higher fracture risk than diabetics with HbA1c levels <7.5 % (HR 1.62; 95 % CI 1.09–2.40), whereas those with HbA1c levels <7.5 % had a risk similar to those without diabetes [18]. However, chronic glycemia might not be consistently predictive of fracture risk. In one study, lower HbA1c levels were associated with increased hip fracture risk [19]. In addition, an ancillary analysis of fractures from the ACCORD study, which compared intensive vs. standard glycemic control [20], found no difference with lower HbA1c levels in the rate of first non-spine fracture (hazard ratio 1.04 [95 % CI 0.86–1.27]) over 3.8 years [21], suggesting that changes in HbA1c levels may not predict fracture risk. However, these data are not

conclusive because there was a high prevalence in ACCORD of TZD use, a known skeletal toxin [16], particularly in the intensive group, which may have obscured a relationship between fracture risk and HgbA1c .

Insulin

Whether insulin is a biomarker for diabetic bone disease is unclear. Insulin itself is typically considered to be anabolic for bone [22, 23], but higher endogenous insulin levels might also be associated with adverse skeletal effects such as decreases in cortical BMD and periosteal circumference [24]. Studies in T1D regarding the effect of exogenous insulin dosing have yielded mixed results [25, 26]. In T2D patients, insulin use does not appear to explain the higher rates of bone loss or the increased incidence of fractures [27]. In both T1D and T2D, higher insulin dosages may be markers of more severe disease, including increased inflammation and comorbidities that could lead to more bone damage [23].

Biochemical Markers of Bone Remodeling

Numerous lines of evidence suggest that diabetic bone disease is associated with alterations in biochemical markers of bone remodeling. High glucose levels, even in the absence of diabetes, can adversely affect bone formation. In healthy individuals, ingestion of 75 g of glucose leads to a decrease in markers of both bone formation and resorption [28] and in vitro data show that exposure to high glucose levels impairs osteoblast function [29-31]. With regard to T1D, in vitro data [30] and in vivo studies involving rodent models [32] indicate that bone formation is characteristically impaired, as shown by the expression of osteoblastic transcription factors such as RUNX2, biochemical markers, and histomorphometric indices [33]. The skeletal effect is generally rescued by insulin treatment and normalization of glycemia [34]. An association between T1D and low bone formation in clinical studies has also been shown [35–37]. In the largest histomorphometry study to date, iliac crest biopsies in 18 otherwise healthy subjects with T1D were compared with those from healthy age- and sex-matched nondiabetic control subjects [38]. Diabetic subjects, when compared to controls, had no significant differences in mineral apposition rate (MAR), mineralizing surface (MS/BS), osteoid maturation time (Omt), mineralizing osteoid (MS/OS), mineralization lag time (Mlt), bone formation rate (BFR/BS or BFR/BV), formation period (FP), remodeling period (Rm.P), or activation frequency (Ac.F) [38]. However, in a subset of diabetic patients who had fractured, dynamic variables such as BFR/BS, BFR/BV, and Ac.F tended to be lower in the fracturing subjects, perhaps indicating lower remodeling in those T1D subjects [38].

Decreased bone remodeling in T2D has also been demonstrated in a number of reports. Clinically, circulating biochemical markers of bone formation, including



Fig. 6.1 Histomorphometric changes in bone formation. (**a**, **b**) Tetracycline double-labeled bone biopsies in a 58-year-old T2D Caucasian woman (**a**) and a 57-year-old Caucasian female control (**b**). Bone formation is decreased in T2D with reduced mineralizing surface. The *arrows* highlight tetracycline uptake in the control subject and the absence of uptake in the diabetic subject. Adapted with permission from [46]

P1NP, osteocalcin [39, 40], and bone-specific alkaline phosphatase [41] have been found to be decreased in T2D. Osteocalcin levels were found to increase in diabetic men concomitant with improved glycemic control [42]. Impaired formation measures in T2D are associated with reductions in bone resorption markers including serum CTx [39–41, 43] and DPD [44]. The decrease in bone remodeling in T2D appears to be predictive of fracture risk regardless of BMD. In a study of 255 T2D women and 240 controls, T2D women with the combination of the lowest PTH and osteocalcin levels had nearly a fivefold increased risk of vertebral fractures independent of lumbar spine BMD [39].

Alterations in dynamic histomorphometry in T2D were reported in one study, but the numbers were very small (n=6 T2D patients; 2 female), and the results were confounded by selecting for low BMD and a problematical control group [45]. In a more recent pilot study, low bone formation was observed in 6 T2D postmenopausal women and 6 postmenopausal age-matched nondiabetic controls, where tetracy-cline double-labeled iliac crest bone biopsies showed virtually no uptake of label in diabetic subjects (Fig. 6.1), with reduced mineralizing surface, osteoid surface, and osteoblast surface (Fig. 6.2) [46]. These preliminary histological data corroborate the decrease in biochemical markers of bone turnover.

PTH

Levels of PTH, a key regulator of bone remodeling, are altered with glycemia and diabetes. In healthy subjects, a glucose load leads to a slight decrease in ionized calcium and an increase in PTH, after an initial temporary decrease [28]. In T1D, blunted PTH responses have been observed [47]. In T2D, levels of PTH tend to be 20–50 % lower than in controls, even in the setting of reduced eGFR, suggesting a



Fig. 6.2 Quantitative measures of bone formation were lower in T2D postmenopausal women than in controls. Adapted with permission from [46]

state of reduced PTH secretion [39, 40, 43]. The importance of PTH in diabetic bone disease has been corroborated by the beneficial skeletal effects of PTH treatment in T2D rats [48]. Specifically, PTH treatment partially reversed the adverse skeletal effects of T2D on bone mass, bone strength, and bone defect repair [48].

Nonclassical Bone Biomarkers

IGF-1, an anabolic factor which stimulates osteoblast proliferation, has been inversely associated with the risk and number of vertebral fractures in T2D postmenopausal women independent of BMD [40, 49]. Circulating osteogenic precursor (COP) cells [50] might also be a biomarker for diabetic bone disease. COP cells can be detected in the peripheral blood by flow cytometry using antibodies specific for the osteoblast matrix protein osteocalcin (OCN) [50, 51] or alkaline phosphatase [52] and are capable of mineralization in vitro [52]. Certain subpopulations of OCN+ cells (OCN+/CD133+/CD34-/KDR+, known as calcifying cells) were increased in subjects with an HbA1c in the prediabetic range [53], although in another report, peripheral blood mononuclear cells that were positive for osteocalcin were lower in postmenopausal women with T2D as compared to nondiabetic controls [46]. An additional potential biomarker of diabetic bone disease is RANKL. Kiechl et al. showed that increased levels of soluble RANKL were associated with the development of diabetes in 844 subjects (OR=3.37; 95 % CI: 1.63–6.97) [54]. Another novel bone marker in T2D may be sphingosine 1-phosphate (S1P), a lipid mediator which increases osteoclastogenesis by increasing RANKL [55]. S1P was found to be increased in T2D women (n=482) as compared to controls and was associated with increasing numbers of vertebral fractures and bone resorption markers [55]. Interestingly, this marker suggests an elevation in bone resorption in T2D, in contrast to the reports of reduced biochemical markers of bone resorption [39–41, 43].

Sclerostin

Sclerostin, an osteocyte product, is a negative regulator of bone formation which competes with the anabolic Wnt b-catenin pathway by binding to LRP5 or 6 [56]. In healthy adults, sclerostin levels are increased by factors including age, BMI, inactivity, bone mineral content, and possibly fractures [56]. It was first reported in 2012 that sclerostin levels were higher in 74 T2D women and men vs. 50 nondiabetic controls, and that higher levels correlated with age, male gender, and BMD [57]. In a different study which included T1D (n=43), T2D (n=40) and matched controls, sclerostin levels did not differ between T1D subjects and controls, but were twofold higher in T2D than in controls or T1D, even after adjusting for age and BMI [58]. These data suggest that the Wnt signaling pathway may be impaired in T2D, although no relationship between markers of bone formation and sclerostin was found [58]. Interestingly, T2D subjects also had lower PTH levels [58], raising the possibility that since PTH inhibits sclerostin [59], perhaps the higher sclerostin levels were due to a decrease in the usual inhibitory effect of PTH. A correlation between Wnt disruption and decreased osteoblast activity was further observed in 40 T2D postmenopausal women who, as compared to controls, had decreased β-catenin levels which correlated with lower BAP [41]. In the largest diabetes sclerostin study, higher sclerostin levels in 321 men and women with T2D were associated with an increased risk of vertebral fractures independent of lumbar spine BMD [60]. It could be posited from these data that because sclerostin is secreted from deeply embedded osteocytes that are in mature bone tissue, the higher sclerostin levels reflect an increase in the amount of aged bone mass that is more likely to accumulate microfractures independent of BMD. Further work will be needed to clarify this point. In a rat model of T2D, treatment with an anti-sclerostin antibody enhanced bone mass and reversed femoral defects [61], although this model did not fully replicate adult-onset diabetes because diabetes developed before skeletal maturity.

Advanced Glycation End Products

Increased levels of advanced glycation end products (AGEs) have been shown to be biomarkers for diabetic bone disease. In the setting of chronic hyperglycemia, AGEs accumulate in the organic bone matrix by a process known as nonenzymatic glycation (the Maillard reaction) [62–69]. Accumulation of AGEs in the organic bone matrix leads to more biomechanically brittle bone that has lost its toughness and is less able to deform before fracturing [63]. Circulating levels of pentosidine, one of the best studied AGEs, have been shown to correlate with bone AGE levels [70]. In 104 nondiabetic patients undergoing orthopedic surgery, plasma pentosidine levels correlated with cortical bone pentosidine [70]. In a nondiabetic population of 765 postmenopausal women, an increase in urinary pentosidine levels predicted a 20 % increase in vertebral and long bone fractures over 5 years [71]. However, pentosidine levels do not consistently distinguish diabetics and nondiabetics when measured in urine [72] or serum [73]. Nevertheless, pentosidine levels appear to provide information about diabetic bone fragility that is separate from BMD parameters. In 1000 patients followed for 7.5 years, urinary pentosidine levels in those with diabetes were associated with a 42 % increase in clinical fracture incidence [relative hazard, 1.42; 95 % confidence interval (CI), 1.10, 1.83] and a nearly sixfold increase in vertebral fracture prevalence, independently of BMD [72]. Similarly, in 153 Japanese men and women with T2D, serum pentosidine levels were significantly higher in the women who had vertebral fractures, independent of BMD (OR 2.50, CI: 1.09-5.73) [73]. With regard to direct measurement of pentosidine in diabetic bone, preliminary data in T1D patients showed that iliac crest bone biopsies in T1D who had fractured had higher pentosidine levels than controls, in association with a greater degree of mineralization [74]. In addition to pentosidine, circulating levels of another AGE, carboxy-methyl-lysine (CML), might indicate bone fragility. When 3373 nondiabetic patients were followed for 9 years, an increase in CML predicted a 27 % increase in hip fracture risk, independent of hip BMD [75].

Architectural Properties in T2D

In addition to alterations in remodeling and matrix properties, another factor that may contribute to the paradox of increased fractures despite normal areal BMD in T2D is microarchitectural abnormality. Increased cortical porosity, a key determinant of bone fragility [76], has been reported at the radius and tibia in female diabetics who have fractured, as measured by intra-cortical pore volume fraction via high-resolution peripheral quantitative computed tomography (HR-pQCT) [77]. In a recent community-based study of women and men, T2D and increased HbA1c levels were associated with deficits in cortical microstructure and density at the distal tibia [78]. Microarchitectural deficits in bone geometry might also explain reduced bone strength in T2D. Strength-to-load ratios (QCT) at the spine and femoral neck were not improved in older adults with T2D although areal BMD (DXA) was higher [79]. In a study of older men, volumetric BMD (pQCT) was higher but bone area was smaller at the distal radius and tibia [80]. Smaller cross-sectional area suggests that stimulation of periosteal apposition, normally observed with greater loading, may be reduced in diabetes. These data seem to suggest that the higher areal BMD in diabetics does not result in the expected biomechanical advantage and supports the likelihood of abnormalities in dynamic and material properties.

Conclusions

At a time when classical complications of diabetes mellitus are becoming less common due to improved glucose control, the skeleton has emerged as a target organ for disease complications. It is now well established that fracture risk is increased in both T1D and T2D. Yet BMD and FRAX, the clinical mainstays for predicting fractures, do not fully capture fracture risk in these populations. Identification of biomarkers in diabetic individuals that predict fracture risk, independent of bone density assessment, will help to diagnose and treat bone fragility in diabetes. Use of biomarkers in diabetes will hopefully offset serious skeletal challenges in this population as they age.

Acknowledgments Funding Source: Columbia University Irving Institute for Clinical and Translational Research CTSA/CTO pilot award

Disclosure: The author has nothing to declare

References

- 1. IDF diabetes atlas update. 2012. http://www.idf.org/diabetesatlas/5e/Update2012. Accessed 10 Jan 2013.
- Janghorbani M, Van Dam RM, Willett WC, Hu FB. Systematic review of type 1 and type 2 diabetes mellitus and risk of fracture. Am J Epidemiol. 2007;166(5):495–505.
- 3. Vestergaard P. Discrepancies in bone mineral density and fracture risk in patients with type 1 and type 2 diabetes–a meta-analysis. Osteoporos Int. 2007;18(4):427–44.
- 4. Schwartz AV, Vittinghoff E, Bauer DC, et al. Association of BMD and FRAX score with risk of fracture in older adults with type 2 diabetes. JAMA. 2011;305(21):2184–92.
- Giangregorio LM, Leslie WD, Lix LM, et al. FRAX underestimates fracture risk in patients with diabetes. J Bone Miner Res. 2012;27(2):301–8.
- Bonds DE, Larson JC, Schwartz AV, et al. Risk of fracture in women with type 2 diabetes: the Women's Health Initiative Observational Study. J Clin Endocrinol Metab. 2006;91(9):3404–10.
- Leslie WD, Lix LM, Prior HJ, Derksen S, Metge C, O'Neil J. Biphasic fracture risk in diabetes: a population-based study. Bone. 2007;40(6):1595–601.
- de Liefde II, van der Klift M, de Laet CE, van Daele PL, Hofman A, Pols HA. Bone mineral density and fracture risk in type-2 diabetes mellitus: the Rotterdam Study. Osteoporos Int. 2005;16(12):1713–20.
- Hernandez RK, Do TP, Critchlow CW, Dent RE, Jick SS. Patient-related risk factors for fracture-healing complications in the United Kingdom General Practice Research Database. Acta Orthop. 2012;83(6):653–60.
- Semel J, Gray JM, Ahn HJ, Nasr H, Chen JJ. Predictors of outcome following hip fracture rehabilitation. PM R. 2010;2(9):799–805.
- 11. Bass E, French DD, Bradham DD. A national perspective of Medicare expenditures for elderly veterans with hip fractures. J Am Med Dir Assoc. 2008;9(2):114–9.
- Pan H, Wu N, Yang T, He W. Association between bone mineral density and type 1 diabetes mellitus: a meta-analysis of cross-sectional studies. Diabetes Metab Res Rev. 2014;30(7):531–42.
- Neumann T, Samann A, Lodes S, et al. Glycaemic control is positively associated with prevalent fractures but not with bone mineral density in patients with Type 1 diabetes. Diabet Med. 2011;28(7):872–5.

- 14. Leslie WD, Rubin MR, Schwartz AV, Kanis JA. Type 2 diabetes and bone. J Bone Miner Res. 2012;27(11):2231–7.
- 15. Schwartz AV, Sellmeyer DE, Ensrud KE, et al. Older women with diabetes have an increased risk of fracture: a prospective study. J Clin Endocrinol Metab. 2001;86(1):32–8.
- Loke YK, Singh S, Furberg CD. Long-term use of thiazolidinediones and fractures in type 2 diabetes: a meta-analysis. CMAJ. 2009;180(1):32–9.
- 17. Schneider AL, Williams EK, Brancati FL, Blecker S, Coresh J, Selvin E. Diabetes and risk of fracture-related hospitalization: the Atherosclerosis Risk in Communities Study. Diabetes Care. 2013;36(5):1153–8.
- Oei L, Zillikens MC, Dehghan A, et al. High bone mineral density and fracture risk in type 2 diabetes as skeletal complications of inadequate glucose control: the Rotterdam Study. Diabetes Care. 2013;36(6):1619–28.
- Puar TH, Khoo JJ, Cho LW, et al. Association between glycemic control and hip fracture. J Am Geriatr Soc. 2012;60(8):1493–7.
- Action to Control Cardiovascular Risk in Diabetes Study Group, Gerstein HC, Miller ME, et al. Effects of intensive glucose lowering in type 2 diabetes. N Engl J Med. 2008;358(24):2545–59.
- Schwartz AV, Margolis KL, Sellmeyer DE, et al. Intensive glycemic control is not associated with fractures or falls in the ACCORD randomized trial. Diabetes Care. 2012;35(7):1525–31.
- Thrailkill KM, Lumpkin Jr CK, Bunn RC, Kemp SF, Fowlkes JL. Is insulin an anabolic agent in bone? Dissecting the diabetic bone for clues. Am J Physiol Endocrinol Metab. 2005;289(5):E735–45.
- Hofbauer LC, Brueck CC, Singh SK, Dobnig H. Osteoporosis in patients with diabetes mellitus. J Bone Miner Res. 2007;22(9):1317–28.
- 24. Sayers A, Lawlor D, Sattar N, Tobias J. The association between insulin levels and cortical bone: findings from a cross sectional analysis of pQCT parameters in adolescents. J Bone Miner Res. 2012;27(3):610–8.
- Campos PM, Lopez-Ibarra PJ, Escobar-Jimenez F, Serrano Pardo MD, Garcia-Cervigon AG. Intensive insulin therapy and bone mineral density in type 1 diabetes mellitus: a prospective study. Osteoporos Int. 2000;11:455–9.
- 26. Eller-Vainicher C, Zhukouskaya VV, Tolkachev YV, et al. Low bone mineral density and its predictors in type 1 diabetic patients evaluated by the classic statistics and artificial neural network analysis. Diabetes Care. 2011;34(10):2186–91.
- Khalil N, Sutton-Tyrrell K, Strotmeyer ES, et al. Menopausal bone changes and incident fractures in diabetic women: a cohort study. Osteoporos Int. 2011;22(5):1367–76.
- Clowes JA, Allen HC, Prentis DM, Eastell R, Blumsohn A. Octreotide abolishes the acute decrease in bone turnover in response to oral glucose. J Clin Endocrinol Metab. 2003; 88(10):4867–73.
- 29. Inaba M, Nishizawa Y, Mita K, et al. Poor glycemic control impairs the response of biochemical parameters of bone formation and resorption to exogenous 1,25-dihydroxyvitamin D3 in patients with type 2 diabetes. Osteoporos Int. 1999;9(6):525–31.
- Inaba M, Terada M, Koyama H, et al. Influence of high glucose on 1,25-dihydroxyvitamin D3-induced effect on human osteoblast-like MG-63 cells. J Bone Miner Res. 1995;10(7): 1050–6.
- Terada M, Inaba M, Yano Y, et al. Growth-inhibitory effect of a high glucose concentration on osteoblast-like cells. Bone. 1998;22(1):17–23.
- Botolin S, Faugere MC, Malluche H, Orth M, Meyer R, McCabe LR. Increased bone adiposity and peroxisomal proliferator-activated receptor-gamma2 expression in type I diabetic mice. Endocrinology. 2005;146(8):3622–31.
- McCabe L, Zhang J, Raehtz S. Understanding the skeletal pathology of type 1 and 2 diabetes mellitus. Crit Rev Eukaryot Gene Expr. 2011;21(2):187–206.
- Verhaeghe J, Van Herck E, van Bree R, Moermans K, Bouillon R. Decreased osteoblast activity in spontaneously diabetic rats. In vivo studies on the pathogenesis. Endocrine. 1997;7: 165–75.

- 35. Gunczler P, Lanes R, Paoli M, Martinis R, Villaroel O, Weisinger JR. Decreased bone mineral density and bone formation markers shortly after diagnosis of clinical type 1 diabetes mellitus. J Pediatr Endocrinol Metab. 2001;14(5):525–8.
- Gunczler P, Lanes R, Paz-Martinez V, et al. Decreased lumbar spine bone mass and low bone turnover in children and adolescents with insulin dependent diabetes mellitus followed longitudinally. J Pediatr Endocrinol Metab. 1998;11(3):413–9.
- 37. Masse PG, Pacifique MB, Tranchant CC, et al. Bone metabolic abnormalities associated with well-controlled type 1 diabetes (IDDM) in young adult women: a disease complication often ignored or neglected. J Am Coll Nutr. 2010;29(4):419–29.
- Armas LA, Akhter MP, Drincic A, Recker RR. Trabecular bone histomorphometry in humans with Type 1 Diabetes Mellitus. Bone. 2012;50(1):91–6.
- 39. Yamamoto M, Yamaguchi T, Nawata K, Yamauchi M, Sugimoto T. Decreased PTH levels accompanied by low bone formation are associated with vertebral fractures in postmenopausal women with type 2 diabetes. J Clin Endocrinol Metab. 2012;97(4):1277–84.
- Ardawi MS, Akhbar DH, Alshaikh A, et al. Increased serum sclerostin and decreased serum IGF-1 are associated with vertebral fractures among postmenopausal women with type-2 diabetes. Bone. 2013;56(2):355–62.
- 41. Gaudio A, Privitera F, Battaglia K, et al. Sclerostin levels associated with inhibition of the Wnt/β-catenin signaling and reduced bone turnover in type 2 diabetes mellitus. J Clin Endocrinol Metab. 2012;97(10):3744–50.
- Sayinalp S, Gedik O, Koray Z. Increasing serum osteocalcin after glycemic control in diabetic men. Calcif Tissue Int. 1995;57(6):422–5.
- 43. Dobnig H, Piswanger-Solkner JC, Roth M, et al. Type 2 diabetes mellitus in nursing home patients: effects on bone turnover, bone mass, and fracture risk. J Clin Endocrinol Metab. 2006;91(9):3355–63.
- 44. Gerdhem P, Isaksson A, Akesson K, Obrant KJ. Increased bone density and decreased bone turnover, but no evident alteration of fracture susceptibility in elderly women with diabetes mellitus. Osteoporos Int. 2005;16(12):1506–12.
- 45. Krakauer JC, McKenna MJ, Buderer NF, Rao DS, Whitehouse FW, Parfitt AM. Bone loss and bone turnover in diabetes. Diabetes. 1995;44(7):775–82.
- 46. Manavalan JS, Cremers S, Dempster DW, et al. Circulating osteogenic precursor cells in type 2 diabetes mellitus. J Clin Endocrinol Metab. 2012;97(9):3240–50.
- Schwarz P, Sorensen HA, Momsen G, Friis T, Transbol I, McNair P. Hypocalcemia and parathyroid hormone responsiveness in diabetes mellitus: a tri-sodium-citrate clamp study. Acta Endocrinol. 1992;126(3):260–3.
- Hamann C, Picke AK, Campbell GM, et al. Effects of parathyroid hormone on bone mass, bone strength, and bone regeneration in male rats with type 2 diabetes mellitus. Endocrinology. 2014;155(4):1197–206.
- 49. Kanazawa I, Yamaguchi T, Sugimoto T. Serum insulin-like growth factor-I is a marker for assessing the severity of vertebral fractures in postmenopausal women with type 2 diabetes mellitus. Osteoporos Int. 2011;22(4):1191–8.
- Eghbali-Fatourechi GZ, Modder UI, Charatcharoenwitthaya N, et al. Characterization of circulating osteoblast lineage cells in humans. Bone. 2007;40(5):1370–7.
- Rubin MR, Manavalan JS, Dempster DW, et al. Parathyroid hormone stimulates circulating osteogenic cells in hypoparathyroidism. J Clin Endocrinol Metab. 2011;96(1):176–86.
- 52. Undale A, Srinivasan B, Drake M, et al. Circulating osteogenic cells: characterization and relationship to rates of bone loss in postmenopausal women. Bone. 2010;47(1):83–92.
- 53. Flammer AJ, Gossl M, Li J, et al. Patients with an HbA1c in the prediabetic and diabetic range have higher numbers of circulating cells with osteogenic and endothelial progenitor cell markers. J Clin Endocrinol Metab. 2012;97(12):4761–8.
- 54. Kiechl S, Wittmann J, Giaccari A, et al. Blockade of receptor activator of nuclear factor-κB (RANKL) signaling improves hepatic insulin resistance and prevents development of diabetes mellitus. Nat Med. 2013;19(3):358–63.

- 55. Ardawi M, Akbar D, Rouzi A, Senani N, Ahmad A, Qari M. Elevated sphingosine 1-phosphate levels are associated with vertebral fractures in patients with type 2 diabetes mellitus. J Bone Miner Res. 2014. http://www.asbmr.org/education/AbstractDetail?aid=51ef5a65-8072-4369b2b7-c1381053da59. Accessed 11 Dec 2014.
- Canalis E. Wnt signalling in osteoporosis: mechanisms and novel therapeutic approaches. Nat Rev Endocrinol. 2013;9(10):575–83.
- 57. Garcia-Martin A, Rozas-Moreno P, Reyes-Garcia R, et al. Circulating levels of sclerostin are increased in patients with type 2 diabetes mellitus. J Clin Endocrinol Metab. 2012;97(1): 234–41.
- Gennari L, Merlotti D, Valenti R, et al. Circulating sclerostin levels and bone turnover in type 1 and type 2 diabetes. J Clin Endocrinol Metab. 2012;97(5):1737–44.
- 59. Costa AG, Cremers S, Rubin MR, et al. Circulating sclerostin in disorders of parathyroid gland function. J Clin Endocrinol Metab. 2011;96(12):3804–10.
- Yamamoto M, Yamauchi M, Sugimoto T. Elevated sclerostin levels are associated with vertebral fractures in patients with type 2 diabetes mellitus. J Clin Endocrinol Metab. 2013; 98(10):4030–7.
- Hamann C, Rauner M, Hohna Y, et al. Sclerostin antibody treatment improves bone mass, bone strength, and bone defect regeneration in rats with type 2 diabetes mellitus. J Bone Miner Res. 2013;28(3):627–38.
- 62. Saito M, Mori S, Mashiba T, Komatsubara S, Marumo K. Collagen maturity, glycation induced-pentosidine, and mineralization are increased following 3-year treatment with incadronate in dogs. Osteoporos Int. 2008;19(9):1343–54.
- 63. Tang SY, Allen MR, Phipps R, Burr DB, Vashishth D. Changes in non-enzymatic glycation and its association with altered mechanical properties following 1-year treatment with Risedronate or alendronate. Osteoporos Int. 2009;20(6):887–94.
- 64. Vashishth D. The role of the collagen matrix in skeletal fragility. Curr Osteoporos Rep. 2007;5(2):62–6.
- 65. Garnero P, Borel O, Gineyts E, et al. Extracellular post-translational modifications of collagen are major determinants of biomechanical properties of fetal bovine cortical bone. Bone. 2006;38(3):300–9.
- 66. Schalkwijk CG, Stehouwer CD, van Hinsbergh VW. Fructose-mediated non-enzymatic glycation: sweet coupling or bad modification. Diabetes Metab Res Rev. 2004;20(5):369–82.
- Monnier VM, Kohn RR, Cerami A. Accelerated age-related browning of human collagen in diabetes mellitus. Proc Natl Acad Sci U S A. 1984;81(2):583–7.
- 68. Schwartz AV. Diabetes mellitus: does it affect bone? Calcif Tissue Int. 2003;73(6):515-9.
- Reiser KM. Nonenzymatic glycation of collagen in aging and diabetes. Proc Soc Exp Biol Med. 1998;218(1):23–37.
- Odetti P, Rossi S, Monacelli F, et al. Advanced glycation end products and bone loss during aging. Ann N Y Acad Sci. 2005;1043:710–7.
- Tanaka S, Kuroda T, Saito M, Shiraki M. Urinary pentosidine improves risk classification using fracture risk assessment tools for postmenopausal women. J Bone Miner Res. 2011;26(11):2778–84.
- 72. Schwartz AV, Garnero P, Hillier TA, et al. Pentosidine and increased fracture risk in older adults with type 2 diabetes. J Clin Endocrinol Metab. 2009;94(7):2380–6.
- 73. Yamamoto M, Yamaguchi T, Yamauchi M, Yano S, Sugimoto T. Serum pentosidine levels are positively associated with the presence of vertebral fractures in postmenopausal women with type 2 diabetes. J Clin Endocrinol Metab. 2008;93(3):1013–9.
- 74. Farlay D, Armas L, Gineyts E, Recker R, Boivin G. Pentosidine and degree of mineralization are increased in bone from fractured-patients with type 1 diabetes mellitus. J Bone Miner Res. 2014;29 Suppl 1.
- 75. Barzilay JI, Buzkova P, Zieman SJ, et al. Circulating levels of carboxy-methyl-lysine (CML) are associated with hip fracture risk: the Cardiovascular Health Study. J Bone Miner Res. 2014;29(5):1061–6.

- Schaffler MB, Burr DB. Stiffness of compact bone: effects of porosity and density. J Biomech. 1988;21(1):13–6.
- 77. Patsch JM, Burghardt AJ, Yap SP, et al. Increased cortical porosity in type 2 diabetic postmenopausal women with fragility fractures. J Bone Miner Res. 2013;28(2):313–24.
- Samelson E, Bouxsein M, Brochin E, et al. Deficits in cortical bone density and microstructure in type 2 diabetes: Framingham HR-pQCT Study. J Bone Miner Res. 2014. http://www.asbmr. org/education/AbstractDetail?aid=717f3c4f-ac78-4141-be67-445acf5d4c9e. Accessed 12 Dec 2014.
- Melton III LJ, Riggs BL, Leibson CL, et al. A bone structural basis for fracture risk in diabetes. J Clin Endocrinol Metab. 2008;93(12):4804–9.
- Petit MA, Paudel ML, Taylor BC, et al. Bone mass and strength in older men with type 2 diabetes: the Osteoporotic Fractures in Men Study. J Bone Miner Res. 2010;25(2):285–91.

Chapter 7 Safety of Antidiabetic Therapies on Bone

Beata Lecka-Czernik and Ann V. Schwartz

Bone Remodeling

Maintenance of bone homeostasis throughout life relies on the bone remodeling process, which continually replaces old and damaged bone with new bone in order to maintain strength and elasticity [1]. In a healthy state, bone resorption is balanced with bone formation. Changes in the milieu of local and systemic factors may alter this balance leading to changes in the bone mass and/or bone biomechanical properties. Aging, estrogen deficiency, and metabolic diseases negatively affect bone mass and/or bone quality leading to the development of osteoporosis and increased fracture rate (Fig. 7.1).

Three types of cells are involved in bone remodeling: osteoclasts which resorb an old or damaged bone, osteoblasts which form new bone at the site of the resorbed cavity, and osteocytes which orchestrate the whole process. Osteoclasts and osteoblasts/osteocytes develop from two distinct populations of stem cells residing in the bone marrow, hematopoietic stem cells (HSC) and mesenchymal stem cells (MSC), respectively. Osteoclast differentiation is determined by both, factors produced by cells of osteoclast lineage and factors produced by other bone marrow cells including

B. Lecka-Czernik, Ph.D. (🖂)

Departments of Orthopaedic Surgery, Physiology and Pharmacology, Center for Diabetes and Endocrine Research, University of Toledo, 3000 Arlington Avenue, Toledo, OH 43614, USA e-mail: beata.leckaczernik@utoledo.edu

A.V. Schwartz, Ph.D.

Department of Epidemiology and Biostatistics, University of California San Francisco, 550 16th Street, San Francisco, CA 94158, USA e-mail: aschwartz@psg.ucsf.edu

[©] Springer International Publishing Switzerland 2016 B. Lecka-Czernik, J.L. Fowlkes (eds.), *Diabetic Bone Disease*, DOI 10.1007/978-3-319-16402-1_7



Fig. 7.1 Schematic representation of coupling between osteoblast, osteocyte, and osteoclast development and function. *MSC* mesenchymal stem cells, *HSC* hematopoietic stem cells

cells of osteoblast lineage [2]. Osteoclast recruitment from the HSC pool and their maturation is controlled by osteoblast-derived cytokines: M-CSF, IL-6, and RANKL. Osteoblasts originate in a marrow MSC compartment which also produces adipocytes [3, 4]. The commitment of MSC toward either the osteoblast or adipocyte lineage occurs by a stochastic mechanism [5]; lineage-specific transcription factors, such as Runx2, Dlx5, and Osterix for osteoblasts and PPAR γ 2 and C/EBPs for adipocytes are activated [6–11]. Activation of osteoblast-specific transcription factors is determined by a milieu of extracellular factors, which regulate the cellular activity of Wnt, TGF β /BMP, and IGF-1 signaling pathways [12]. Process of bone remodeling is controlled by osteocytes, which represent specialized cells of osteoblast lineage [13]. They are located inside of bone mineralized matrix and communicate with other osteocytes and bone marrow environment through the system of dendrite-like processes. Osteocytes control dynamics of bone remodeling process by secreting RANKL to control bone resorption and sclerostin to control bone formation [13].

Bone as an Integral Part of Energy Metabolism System

Bone is closely integrated with the system regulating energy balance. Organs involved in this regulation including brain, fat, gastrointestinal system, and pancreas are secreting hormones which in endocrine manner regulate both energy metabolism and bone mass. Their effect on bone is possible because bone marrow cells, both mesenchymal and hematopoietic lineage, are equipped with necessary receptors to respond to these signaling (Fig. 7.2).

Integration of bone metabolism with energy metabolism has been presented recently as a model which links anabolic effect of insulin signaling in osteoblasts



Fig. 7.2 Bone is a part of energy metabolism network. Factors and their receptors which mediate a cross-talk between bone and other organs involved in regulation of energy metabolism. *MCR* melanocortin receptor, *CBR* cannabinoid receptor, *LepR* leptin receptor, *AdipoR* adiponectin receptor, *GIPR* glucose inhibitory protein receptor, *GLP2R* glucagon inhibitory peptide receptor, β -ADR beta adrenergic receptor, *IR* insulin receptor, *IGFR* insulin growth factor receptor, *PPARy* peroxisome proliferator activator receptor γ

with bone turnover and regulation of insulin sensitivity in peripheral organs [14, 15]. Thus, in osteoblasts insulin signaling regulates an expression of Runx2 and osteocalcin production. In addition, insulin increases support for osteoclastogenesis by decreasing an expression of OPG, a decoy receptor for RANKL. As a result, insulin increases bone turnover and production of undercarboxylated osteocalcin, which in endocrine fashion regulates insulin release from β -cells in pancreas and production of adiponectin in fat tissue [14–17]. Although it is not clear whether this regulatory circuit is affected in diabetes, several studies suggest that patients with T2DM have decreased bone turnover [18–20]. If so, it would result in the decrease in osteocalcin production, especially its undercarboxylated form, which would lead to the attenuation of signaling responsible for increasing of insulin release from the pancreas and increasing fat sensitivity to insulin.

Anti-hyperglycemic Therapies and Their Effects on Bone

The most common form of diabetes is insulin-independent T2DM, which is characterized by insulin and glucose intolerance, and is associated with development of hyperglycemia and hyperinsulinemia. Therapies, either approved by FDA or in

Target	Mode of action	Class of drugs	Drugs	Skeletal effect
Insulin	Sensitizers	Biguanides	Metformin ^a	Neutral
		TZDs (PPARγ agonists)	Pioglitazone ^b , Rosiglitazone ^b	Bone loss; increased fractures
		Dual PPARα/ PPARγ agonists	Aleglitazar ^c	Unknown
	Secretagogues	K+ ATP	Sulfonylureas (e.g. Glyburide ^a)	Neutral
			Meglitinides (e.g. Nateglinide)	Unknown
		GLP-1 analogs	Exenatide, Liraglutide, Taspoglutide ^c , Albiglutide ^c , Lixisenatide ^c	Unknown
		DPP-4 inhibitors	Alogliptin ^c , Saxagliptin, Sitagliptin, Vildagliptin, Linagliptin	Decreased fractures
		Analogs/other insulins ^a	Insulin lispro, Insulin aspart, Insulin glargine	Increased fractures
Other	SGLT2 inhibitors		Canagliflozin, Dapagliflozin	Increased fractures
	Amylin analog		Pramlintide	Unknown
	Alpha-glucosidase inhibitors		Acarbose, Miglitol, Voglibose	Unknown

Table 7.1 Antidiabetic drugs and their effects on skeleton

^aWorld Health Organization Essential Medicine (WHO-EM)

^bRestricted use in USA and Europe

°Phase III clinical trial was halted due to unfavorable renal effects

Phase III clinical trial, include insulin sensitizers, insulin secretagogues, and drugs which increase glucose excretion in the urine (SGLT2 inhibitors), regulate glucose absorption in intestine (amylin analog), and prevent digestion of carbohydrates (Alpha-glucosidase inhibitors) (Table 7.1).

In general, there is a lack of rigorous clinical evidence regarding the skeletal effects of these medications. Fracture is the primary outcome of interest but is a relatively rare outcome, requiring large studies. Randomized controlled trials (RCT) of diabetes medications have not included fracture as a primary endpoint, but increasingly studies are reporting the evidence from fractures identified as adverse events in RCTs. This provides the best evidence available to us regarding the clinical effects of these medications on the skeleton. Large observational studies have also considered the effects of diabetes medications on fracture risk. However, medications may be systematically prescribed to patients with different risk profiles for fracture, making it difficult to distinguish effects of the medications themselves on fracture, a problem known as "allocation bias" or "confounding by indication." Changes in bone mineral density are an important marker for skeletal health, but are not always a consistent predictor of the effects of a medication on fracture risk [21]. In addition, diabetic bone is compromised by other deficits, distinct from BMD, that have not been clearly delineated but may include increased cortical porosity and

greater accumulation of advanced glycation endproducts in bone collagen (see Chap. 9). As research clarifies which secondary markers are important predictors of fracture risk in diabetic patients, it will be essential to clarify how they are affected by diabetic medications.

Biguanides (Metformin)

Metformin is the most commonly used to increase insulin sensitivity in diabetic patients. Biguanides class of drugs decreases hepatic glucose production and increases glucose uptake in muscle. Metformin is considered by the World Health Organization an essential medicine satisfying the criteria of the public health relevance, evidence on efficacy and safety, and comparative cost effectiveness (www. who.int/medicines). Metformin mechanism of insulin sensitization includes activation of hepatic and muscle AMP-activated protein kinase (AMPK), which results in suppression of fatty acid synthesis, stimulation of fatty acid oxidation in liver and increase in muscle glucose uptake [22]. AMPK also decreases expression of sterolregulatory element-binding-protein 1 (SREBP-1), a transcription factor involved in adipocyte differentiation and pathogenesis of insulin resistance, dislipidemia and diabetes. Animal studies indicate that metformin has a positive effect on osteoblast differentiation due to increased activity of osteoblast-specific Runx2 transcription factor via AMPK/USF-1/SHP regulatory cascade [23] and it has a negative effect on osteoclast differentiation and bone loss after ovariectomy by decreasing RANKL and increasing osteoprotegerin levels [24]. Interestingly, in rodent models metformin can prevent the adverse effects of TZDs on bone by either inducing reossification of bone after rosiglitazone treatment or preventing rosiglitazone effects when applied in combination with rosiglitazone [25].

Human studies of the effects of metformin on the skeleton are limited in design and number. The only RCT with a fracture outcome that included metformin is the ADOPT trial, discussed in more detail in the section on "Thiazolidinediones (Rosiglitazone, Pioglitazone)" [26]. Briefly, this trial randomized participants to receive metformin, glyburide (a sulfonylurea), or rosiglitazone; the primary outcome was time to monotherapy failure. Fractures were identified as adverse events. The fracture rates were similar in those randomized to metformin or glyburide. During the first 12 months of ADOPT, changes in the levels of the bone resorption marker CTX were similar in women (difference in 12-months change: +2.0 %) and modestly greater in men (-8.4 %) in those assigned to metformin compared with glyburide [27]. The metformin group had greater decreases in levels of the bone formation marker P1NP (difference in 12-months change: -9.4 % women; -19.5 % men), compared with glyburide.

Several observational studies have reported a lower risk of fracture with metformin use. In a study of the Danish population, metformin use was associated with lower fracture risk, compared with nondiabetic residents [28]. In the Rochester cohort, metformin use was associated with a lower rate of fracture in T2DM patients (adjusted hazard ratio 0.7; 95% CI 0.6–0.96) [29]. However, other studies have found no difference in fracture risk with metformin use [30–35]. Interestingly, results from a study of hip fracture in Scotland suggest that metformin tends to be prescribed to patients with a lower overall risk of fracture while sulfonylureas are prescribed to those at higher risk [33]. This finding suggests that some of the reduction in fracture risk reported with metformin use in observational studies may be due to the underlying prescribing pattern.

Insulin

There are no randomized trials of insulin therapy with fracture or BMD outcomes. Most observational studies have identified increased fracture risk in those using insulin [29–31, 36–38] although others have not found an increased risk [28, 33]. Insulin treatment is also associated with a higher risk of falls [39, 40], and this is likely a contributing factor to the increased fracture risk. Insulin does not appear to have a negative effect on bone; indeed, preclinical studies suggest an anabolic effect. Increased falls and fractures may be the result of more frequent episodes of hypoglycemia and greater frailty due to diabetic complications.

Thiazolidinediones (Rosiglitazone, Pioglitazone)

TZDs increase insulin sensitivity via activation of peroxisome proliferator-activated receptor (PPARy). Two TZDs, rosiglitazone and pioglitazone, have been used clinically since 1999. A number of studies showed superior efficacy of TZDs over other available antidiabetic therapies in the control of diabetic hyperglycemia [41]. However, their prolonged use is associated with several adverse effects. Strong clinical evidence points to the connection between rosiglitazone use and a significant increase in risk of myocardial infarction and death from cardiovascular causes [42]. This association resulted in a recent review of rosiglitazone safety by the FDA and recommendation for its restricted use in the United States. Interestingly, pioglitazone use is associated with a significantly lower risk of death and lower number of myocardial infarction and stroke incidence [43], indicating that cardiovascular effects of TZDs are not a drug class effect, but rather specifically associated with the TZD type. However, increased risk of bladder cancer in long time pioglitazone users resulted in recent restriction of its use by FDA. Both TZDs exhibit drug class properties of fluid retention and weight gain [44]. Although the use of both rosiglitazone and pioglitazone is currently restricted, the new TZDs with better safety profile are in development. Therefore, understanding TZDs mechanism of action on bone is needed in respect to improvement of safety for bone of new line of TZDs.

Although they possess beneficial anti-hyperglycemic profiles, rosiglitazone and pioglitazone use is associated with adverse effects on the skeleton [45, 46]. The crucial clinical evidence of a causal connection between TZD therapy and

increased fracture risk was determined from secondary analyses of results from randomized clinical trials of rosiglitazone and pioglitazone. The first demonstration of increased fracture risk was reported from ADOPT (A Diabetes Outcome Progression Trial), designed to compare time to monotherapy failure of rosiglitazone, metformin and glyburide in recently diagnosed T2DM patients [41]. Because of growing concern, based on rodent models and clinical trials, that TZDs might have a negative effect on bone, the investigators undertook a post hoc analysis of fracture rates in the three groups, using adverse event reports to identify fractures. In 1840 women and 2511 men with a median follow-up of 4.0 years and an average age of 56 (SD 10) years, fracture rates in men did not differ across treatment groups [26, 41]. However, in women, the cumulative incidence of fractures at 5 years was 15.1 % (11.2–19.1) with rosiglitazone, 7.3 % (95 % CI 4.4–10.1) with metformin, and 7.7 % (95 % CI 3.7–11.7) with glyburide, representing hazard ratios of 1.81 (95 % CI 1.17-2.80) and 2.13 (95 % CI 1.30-3.51) for rosiglitazone compared with metformin and glyburide, respectively. Increased fracture rates were seen in the lower and upper limbs. The incidence of hip and clinical vertebral fractures did not differ across treatment assignments, but only four hip and three clinical vertebral fractures were reported in women, as expected in the age range of this trial. Rosiglitazone was associated with higher fracture rates in both pre-and postmenopausal women, suggesting that estrogen status does not modify the effect on bone.

Soon after the ADOPT findings were published, Takeda performed a metaanalysis of pioglitazone trials and reported a similar pattern of increased fracture risk in women, but not men [47]. These observations were subsequently corroborated by other randomized trials. An early meta-analysis of data from ten different randomized controlled trials confirmed that TZD use doubles the risk of fractures exclusively in women [48]. More recently, a meta-analysis of 22 randomized controlled trials, including 896 fracture events, reported increased fracture incidence in women (OR=1.94; 95 % CI 1.60–2.35) but not in men (OR=1.02; 95 % CI 0.83– 1.27) [49]. Effects in women were similar for rosiglitazone (OR=2.10; 95 % CI 1.61–2.51) and pioglitazone (OR=1.73; 95 % CI 1.18–2.55).

Because the TZD trials have included few hip or vertebral fractures, it is necessary to rely on observational studies to assess whether TZD use increases fractures at these particular skeletal sites. A study using registry data in Scotland focused exclusively on hip fracture risk and reported increased risk with greater cumulative TZD use among those with any use (OR per year of exposure 1.18; 95 % CI 1.09–1.28) [33]. Results were similar for pioglitazone and rosiglitazone considered separately. In contrast to reports from randomized trials, increased hip fracture risk was found in men (OR per year of exposure 1.20; 95 % CI 1.03, 1.41) as well as women (OR per year of exposure 1.18; 95 % CI 1.07, 1.29). Those with >4 years of TZD use had OR for hip fracture of 1.94 (95 % CI 1.28, 2.94), compared with those who used a TZD for up to 2 years. Finally, results were similar when evaluated in a subset with adjustment for use of other antidiabetic medications (insulin, metformin, or sulfonylurea). A large observational study using the UK General Practice Research Database also concluded that TZD use increased hip fracture incidence (Rate Ratio 2.09; 95 % CI 1.29–3.40) as well as spine fracture incidence (RR 2.72; 95 % CI

1.29–5.73) [50]. This study reported similar increases in risk of any fracture in men (RR 1.44; 95 % CI 1.18–1.77) and women (RR 1.42; 95 % CI 1.20–1.69).

The principal mechanism underlying increased fracture risk with TZD use appears to be bone loss. A recent meta-analysis of ten randomized clinical trials that assessed change in BMD reported greater bone loss at the lumbar spine, total hip and femoral neck in women randomized to TZD treatment compared with placebo or other antidiabetic medication [49]. Only one trial included men.

Clinical studies of changes in bone turnover markers with TZD treatment have not provided consistent results. In the largest study to date, 12-month changes in serum markers were assessed in 1605 participants in ADOPT [27]. This analysis showed modest but statistically significant increases in levels of resorption marker C-terminal telopeptide (CTX) in women on rosiglitazone therapy compared with glyburide (10.7 % difference, p=0.002) or metformin (7.3 % difference, p=0.029). In men, CTX was elevated in the rosiglitazone group compared with metformin (12.2%, p<0.001) but not compared with glyburide. Both genders had modest reductions in levels of the marker of bone formation P1NP (women -4.4 %, men -14.4 %), but those in the metformin arm experienced greater reductions. For women, changes in P1NP did not differ between the rosiglitazone and glyburide groups while in men losses were greater in the rosiglitazone group. Although rodent models suggest an important role for reduced bone formation as a mechanism of bone loss with TZD treatment, the ADOPT results instead indicate that increases in bone resorption may explain at least in part the increased fracture rate in women on TZD therapy [27].

Smaller trials of rosiglitazone treatment have also reported relative increases in markers of bone resorption compared to placebo [51, 52] and to metformin [53]. However, others have reported a relative reduction in bone formation markers with rosiglitazone treatment, compared with placebo [54] or with diet only treatment [55], and others have reported no difference [56]. For pioglitazone, the largest trial included 156 postmenopausal women with prediabetes and found no differences in bone turn-over markers after 12 months compared with placebo [57]. In contrast, a trial in 71 diabetic men reported relative increases in markers of bone resorption (CTX) and formation (P1NP) in the pioglitazone group compared with metformin [58] while a trial in 86 diabetic men and women found a relative increase in a formation but not a resorption marker with pioglitazone treatment compared with placebo [59].

Taken together, results of available clinical studies indicate the following regarding TZD use (1) women are at increased risk of fractures; however, some studies point to elevated risk in men as well; (2) the increased fracture risk appears to be a class effect of currently available TZDs; (3) bone loss is an underlying mechanism; (4) fracture risk is increased in the extremities and most likely at the hip and spine as well.

Mechanism of TZD-Induced Bone Loss

PPAR γ , an essential regulator of lipid, glucose, and insulin metabolism [10], is a target for TZDs. The PPAR γ protein is expressed in mice and humans in two isoforms, PPAR γ 1 and PPAR γ 2. PPAR γ 1 is expressed in a variety of cell types,

including cells of hematopoietic lineage macrophages and osteoclasts [60], whereas PPAR γ 2 expression is restricted to cells of mesenchymal lineage adipocytes [61]. In bone, PPAR γ 2 plays an important role in regulation of MSC differentiation toward osteoblasts and adipocytes, and the maintenance of bone mass. Activation of the PPAR γ 2 isoform with rosiglitazone converts cells of osteoblast lineage to terminally differentiated adipocytes and irreversibly suppresses both the osteoblast phenotype and osteoblast-specific gene expression. Thus, in MSCs PPAR γ 2 acts as a positive regulator of adipocyte differentiation and a dominant-negative regulator of osteoblast differentiation [11, 62]. In contrast, PPAR γ 1 expressed in HSC promotes osteoclast differentiation and bone resorption [60]. It controls an expression of c-fos protein, an important determinant of osteoclast lineage commitment and development.

An essential role of PPARy in maintenance of bone homeostasis was demonstrated in several animal models of either bone accrual or bone loss depending on the status of PPARy activity [63-68]. In models of bone accrual, a decrease in PPARy activity in either heterozygous PPARy-deficient mice or mice carrying a hypomorphic mutation in the PPARy gene locus led to increased bone mass due to increased quantity of osteoblasts [66, 68]. Interestingly, mice deficient in PPAR γ expression in cells of hematopoietic lineage develop osteopetrosis and are less sensitive to the TZD-induced bone loss than control mice [60]. In contrast, in rodent models of bone loss due to PPARy activation, administration of rosiglitazone resulted in significant decreases in BMD, bone volume, and changes in bone microarchitecture [63, 67, 69]. Observed bone loss was associated with expected changes in the structure and function of bone marrow, which included decreased number of osteoblasts, increased number of adipocytes, and increased support for osteoclastogenesis. The degree of bone loss in response to rosiglitazone correlated with the animal age and the level of PPAR γ expression. In younger animals with less PPAR γ , bone loss was less extensive than in older animals [69]. Moreover, age determined the mechanism by which bone loss occurred. In younger animals it occurred due to decreased bone formation, whereas in older animals due to increased bone resorption [69]. In addition, studies of rosiglitazone effects in estrogen deficient rats showed that bone loss occurred mainly due to increased bone resorption [64]. In conclusion, animal studies suggest that aging and estrogen deficiency confound TZD-induced bone loss and determine its mechanism.

The negative effect of TZDs on osteoblastogenesis includes decreased activity of Runx2, Dlx5, and Osterix, which are osteoblast-specific transcription factors, and decreased activity of osteoblast-specific signaling pathways controlling bone homeostasis, among them Wnt, TGF- β /BMP, and IGF-1 [70, 71]. The effect of TZDs on the expression of genes essential for osteoblast development is strikingly similar to changes observed during aging. Due to the type of bone loss and similarities to aging, some speculate that TZDs may accelerate the aging of bone [69, 72]. The complexity of TZDs effects on bone remodeling resulting from changes on osteoblast and osteoclast differentiation and alterations in bone marrow milieu supporting remodeling are summarized in Fig. 7.3.



Fig. 7.3 PPAR γ activation with TZDs leads to multiple direct and indirect effects in the bone marrow which result in changes in bone cell differentiation, unbalanced bone remodeling and ultimately bone loss. *Anti-OB* PPAR γ activity inhibiting osteoblast differentiation, *pro-AD* PPAR γ activity stimulating adipocyte differentiation, and *pro-OC* PPAR γ activity stimulating osteoclast differentiation

Novel Selective PPARγ Modulators with Beneficial Effect of Insulin Sensitizers and No Effect on Adipocyte Differentiation

The PPAR γ ligand-binding domain contains a large binding pocket capable of encompassing a variety of ligands. This provides a wide array of potential contact points that can result in various PPAR γ post-translational modifications (PTMs), including phosphorylation, acetylation and sumoylation, and differential recruitment of coactivators, which determine specific activities of this nuclear receptor [73]. The molecular studies provide evidence for distinct mechanisms regulating the proadipocytic, antiosteoblastic, and insulin sensitizing activities of PPAR γ and include the levels of Serine 273 and Serine 112 phosphorylation and functional interaction with other proteins such as β -catenin and molecular chaperons FKBP51 and PP5 [74–77].

The concern of TZDs adverse effects has prompted pharmaceutical efforts to develop selective PPAR γ modulators which will retain high potency to treat diabetic disease with minimal adverse effects [78]. The PPAR γ selective activators, with a decreased proadipocytic activity but intact insulin sensitizing activity such as

netoglitazone, INT131, MSDC-0602 and telmisartan do not affect bone mass in mice treated with the therapeutic doses [79–82]. A new class of insulin sensitizers with structural similarities to telmisartan, which block Serine 273 phosphorylation but do not stimulate PPAR γ transcriptional proadipocytic activity, has been recently developed [74, 83], however their safety for bone is not as yet determined.

Sulfonylureas

Sulfonylureas function as insulin secretagogues. This class of drugs activates sulfonylurea receptors on the surface of pancreatic β cells and stimulates exocytosis of insulin from vesicles. In addition, sulfonylureas are associated with greater frequency of hypoglycemia which may increase the risk of falls and fractures [84].

In the ADOPT trial, described earlier, fracture incidence was similar in those randomized to a sulfonylurea (glyburide) versus metformin [41]. The results of observational studies have been inconsistent with reports of increased [34], decreased [28, 31] and no difference [29, 30, 32, 35, 50, 85, 86] in fracture risk among those using a sulfonylurea. As noted in a recent review of current literature [87], although a large number of studies have reported no association with fracture, the majority of these studies were not specifically intended to assess the impact of sulfonylureas in particular on fracture. Allocation bias is an important consideration in observational studies, and results from a study in Scotland suggest that patients using sulfonylureas increase hypoglycemic episodes, an observational study conducted among Kaiser Permanente members did not find an association between sulfonylurea use and incident falls, identified through inpatient and outpatient medical records [88].

Incretin Analogs and DPP4 Protease Inhibitors

This newest class of antidiabetic drugs enhances the mechanism by which enteric hormones stimulate insulin release from β -cells and inhibit glucagon production in the liver [89]. Glucose-dependent insulinotropic peptide (GIP), and glucagon-like peptides (GLP-1 and GLP-2), are released by gut endocrine cells in response to nutrient intake. Bioactivity of incretin hormones is limited by their rapid degradation and inactivation by dipeptidyl peptidase-4 (DPP-4), a serine protease that is present in a soluble form in plasma and is expressed in most tissues [90]. Recently, incretin mimetics (GLP1 receptor agonists) and DPP-4 inhibitors have emerged as a new class of pharmacological agents to enhance incretins action and improve glycemic control in patients with T2DM. Incretin mimetics and DPP-4 inhibitors have a major advantage over other diabetic medications in that glucose control remains stable with little or no rise in HbA1c levels after long periods of use. The side effects common for incretin-based therapies, including incretin receptors

agonists and DPP-4 inhibitors, consist of gastrointestinal, immune system and pancreatic reactions. Since DPP-4 enzyme is known to be involved in the suppression of certain malignancies, particularly in limiting the tissue invasion of tumors, there is a concern that DPP-4 inhibitors may allow some cancers to progress, however clinical data are not as yet available [91–93].

Nutritional hormones are known to be important in bone turnover; as soon as a meal is ingested, bone breakdown is suppressed [94, 95]. Osteoblasts and osteoclasts express receptors for both GIP and GLP incretins. A number of studies indicate that GLP-2 acts mainly as an antiresorptive hormone [96], while GIP can act both as an antiresorptive and anabolic hormone [97, 98]. Mice deficient in GLP-1 receptor develop cortical osteopenia and have more fragile bone as well as increased quantity of osteoclasts and increased bone resorption [99]. GLP-1 receptor signaling may play an essential role in the control of bone resorption indirectly, through a calcitonin-dependent pathway. Calcitonin treatment effectively suppressed bone resorption markers in Glp-1r(-/-) mice, and the GLP-1 receptor agonist exendin-4 increased calcitonin gene expression in the thyroid of wild-type mice [99]. Interestingly, although animal studies showed that DPP-4 inhibitor sitagliptin did not affect bone density, however the absence of DPP-4 in Dpp-4(-/-) mice lead to the greater bone loss after ovariectomy as compared to animals with unaltered DPP-4 expression [100]. In summary, a number of animal studies indicate that incretins have beneficial effects on bone mass and protective effects on bone quality. Therefore, antidiabetic therapies which increase GIP and GLP hormone levels and their bioactivity might exert beneficial effects on human bone.

Since incretin-based therapy is relatively new, the clinical data of its safety for bone is just emerging. The 44-week treatment of T2DM patients with incretin mimetic exenatide did not decrease total body BMD, although it decreased body weight by 6 % [101]. Currently, two meta-analyses of incretin mimetics and fracture outcomes, reported as serious adverse events, have been published. The first metaanalysis included seven trials with placebo or other antidiabetic medications as the comparison group and a total of 19 fractures [102]. There was no difference in fracture incidence between incretin mimetic treatment and the comparator groups (MH OR 0.75; 95 % CI 0.28-2.02), but confidence intervals were wide. A second meta-analysis included 14 trials with 38 fractures and also found no difference in fracture incidence (MH OR 1.05; 95 % CI 0.59-1.87) [103]. However, when examined separately, the investigators found decreased fracture incidence for liraglutide treatment (MH OR 0.38; 95 % CI 0.17-0.87) and increased fracture incidence for exenatide treatment (MH OR 2.09; 95 % CI 1.03-4.21), both compared with placebo or other antidiabetic medications. The reason for this difference is not immediately apparent and may be a chance finding. Liraglutide and exenatide have similar effects on blood glucose and body weight without an increased frequency of hypoglycemia. An important limitation of these trials is their relatively short length for the purposes of assessing fracture risk. Only five of the trials were 52 weeks or longer, and effects on fracture risk that are operating through changes in bone would be expected to develop over periods of a year or more.

A meta-analysis of 28 clinical trials of DPP-4 inhibitors with duration of at least 24 weeks included 63 fractures reported as serious adverse events. Treatment with DPP-4 inhibitors was associated with a reduced risk of fractures (Mantel Haenszel Odds Ratio 0.60; 95 % CI 0.37–0.99) compared to placebo and other treatments [104]. Excluding TZDs or sulfonylurea as comparators yielded similar results (MH OR 0.56; 95 % CI 0.33–0.93). As noted for the trials of incretin mimetics, an important weakness of these results is the short duration of the trials. Only 7 of the 28 trials were 52 weeks or longer. More clinical studies on the effects of incretin mimetics and DPP-4 inhibitors on BMD and fracture risk with stratification according to gender, postmenopausal status, and age are needed.

SGLT2 Inhibitors

In 2013 and 2014 the first of two SGLT2 inhibitors class of diabetes drugs, canagliflozin and dapagliflozin, were approved by FDA for improving glycemic control in T2DM patients in conjunction with diet and exercise. Both drugs are selective and reversible inhibitors of sodium glucose co-transporter 2 (SGLT2), which is responsible for the majority of glucose reabsorption in kidney. Bone safety of dapagliflozin has been evaluated in a randomized trial, adding study drug or placebo to metformin in T2DM patients with inadequate control on metformin. Results have been reported after 50 weeks [105] and 102 weeks [106] of treatment. At 102 weeks, the trial was completed by 140 patients, men and postmenopausal women. The dapagliflozin group lost more weight than the placebo group (difference of -2.42 kg; 95 % CI -3.64, -1.21). In spite of the greater weight loss, no significant differences were identified in changes from baseline in markers of bone formation (P1NP) and bone resorption (CTX). Bone loss was greater at the femoral neck in those treated with dapagliflozin but the difference was not statistically significant (difference -0.94 %; 95 % CI -2.21, 0.35). Differences in BMD changes at the lumbar spine and total hip were smaller and also not statistically significant. There were no significant treatment-by-gender interactions. In a 104-week trial of dapagliflozin to assess efficacy among T2DM patients with moderate renal impairment, 252 patients were randomized to dapagliflozin (5 or 10 mg) or placebo. The treated groups lost weight compared with placebo but glycemic control was not different. More fractures were reported in the treated groups (N=13) than the placebo group (N=0). Dapagliflozin is not recommended for use in patients with moderate renal impairment [107].

There are no available studies of the effects of canagliflozin on bone turnover or BMD. Increased incidence of upper extremity fractures with canagliflozin treatment was reported in the Prescribing Information by the manufacturer [108]. In a metaanalysis of eight clinical trials with longer mean duration of treatment (68 weeks), the incidence rate of fracture was 14.2, 18.7, and 17.6 per 1000 patient-years of exposure to comparator, canagliflozin 100 mg and canagliflozin 300 mg. FDA approval for canagliflozin included a requirement for postmarketing studies to monitor for bone safety [109].
Amylin Analogs

Amylin, also known as Islet Amyloid Polypeptide (IAPP), is a 37-residue peptide hormone produced in pancreatic β -cells. Amylin is co-secreted with insulin and plays a role in glycemic regulation by slowing gastric emptying, promoting satiety and decreasing glucose levels in circulation. Amylin, like insulin, is absent in individuals with T1DM. Amylin belongs to the family of regulatory hormones that are structurally and functionally related to calcitonin, calcitonin gene-related peptide and adrenomedulin, and signals through the calcitonin receptor modified for amylinspecific activity by binding to receptor activity modifying proteins (RAMP) [110]. Cellular studies have shown that amylin may stimulate osteoblast proliferation and may inhibit osteoclast development and activity through increasing cyclic AMP [111]. These activities have been confirmed in several animal studies which showed a positive effect of amylin on trabecular and cortical bone volume [111] and some of them are indicative that the bone response to amylin may differ depending on diabetes status [112].

Human data suggest that there is a functional link between amylin and skeletal health. Amylin levels decrease with aging and correlate inversely with osteoporosis [113]. In addition, reduced amylin levels are associated with low BMD in women with anorexia nervosa and are significant predictors of BMD and of *Z*-scores at the femoral neck and at the total hip in this group of patients [114].

An analog of amylin, pramlintide, had been approved for therapy in 2005 by FDA to treat diabetes. It is used as an adjunctive therapy with insulin in both T1DM and T2DM. Pramlintide allows patients to use less insulin, because it improves hemoglobin 1Ac levels, lowers average blood sugar levels, and substantially reduces blood sugar that occurs in diabetic individuals right after eating.

Although animal studies suggest positive effect of amylin on bone, the clinical studies do not provide supporting results. Pramlintide safety on bone was assessed in a study conducted on patients with T1DM who injected the drug for 12 months. BMD measurements of the lumbar spine by dual-energy X-ray absorptiometry (DXA), and biochemical markers of bone metabolism (serum-calcium, PTH, osteo-calcin, urinary pyridinium cross-links) before and one year after starting pramlintide therapy showed no significant changes. It is concluded that a 1-year pramlintide therapy does not affect bone density or bone metabolism in patients with type 1 diabetes mellitus without osteopenia (based on the markers used) [115].

Alpha-Glucosidase Inhibitors

Alpha-glucosidase inhibitors are saccharides that act as competitive inhibitors of enzymes needed to digest carbohydrates, specifically alpha-glucosidase enzymes in the brush border of the small intestines, which subsequently leads to the reduction of blood sugar levels. Therapy with alpha-glucosidase inhibitors is associated with several side effects which are in close relation to their mechanism of action and include increased levels of carbohydrates in the intestine causing flatulence and diarrhea. There are no available information on bone safety of alpha-glucosidase inhibitors.

Conclusions

In conclusion, the available evidence indicates that antidiabetic therapies may either increase fracture risk (TZDs and insulin), may not affect this risk (sulfonylureas and metformin) or may possibly decrease the risk (DPP-4 inhibitors). From a bone perspective, metformin and sulphonylureas are safer than TZDs; randomised trials have shown that TZDs decrease BMD and increase fracture risk. The mechanism of TZD-induced bone loss includes unbalanced bone remodeling processes resulting from decreased bone formation and increased bone resorption. Animal studies suggest that aging and estrogen deficiency may modify the effects of TZDs on bone and determine the mechanism of bone loss. The emerging potential of incretin-based therapies as sparing or perhaps even beneficial for bones requires systematic clinical assessment in the future.

Acknowledgment This work was supported by grants from NIH/NIA AG028935 and American Diabetes Association's Amaranth Diabetes Fund 1-09-RA-95.

References

- Parfitt AM. Osteonal and hemi-osteonal remodeling: the spatial and temporal framework for signal traffic in adult human bone. J Cell Biochem. 1994;55(3):273–86.
- Teitelbaum SL, Ross FP. Genetic regulation of osteoclast development and function. Nat Rev Genet. 2003;4(8):638–49.
- Jiang Y, Jahagirdar BN, Reinhardt RL, Schwartz RE, Keene CD, Ortiz-Gonzalez XR, et al. Pluripotency of mesenchymal stem cells derived from adult marrow. Nature. 2002;418(6893): 41–9.
- 4. Bianco P, Riminucci M, Gronthos S, Robey PG. Bone marrow stromal stem cells: nature, biology, and potential applications. Stem Cells. 2001;19(3):180–92.
- 5. Aubin JE. Regulation of osteoblast formation and function. Rev Endocr Metab Disord. 2001;2(1):81–94.
- Ducy P, Zhang R, Geoffroy V, Ridall AL, Karsenty G. Osf2/Cbfa1: a transcriptional activator of osteoblast differentiation. Cell. 1997;89(5):747–54.
- Komori T, Yagi H, Nomura S, Yamaguchi A, Sasaki K, Deguchi K, et al. Targeted disruption of Cbfa1 results in a complete lack of bone formation owing to maturational arrest of osteoblasts. Cell. 1997;89(5):755–64.
- Karsenty G. Minireview: transcriptional control of osteoblast differentiation. Endocrinology. 2001;142(7):2731–3.
- Tontonoz P, Hu E, Spiegelman BM. Stimulation of adipogenesis in fibroblasts by PPARgamma 2, a lipid-activated transcription factor [published erratum appears in Cell 1995 Mar 24;80(6):following 957]. Cell. 1994;79(7):1147–56.

- Rosen ED, Spiegelman BM. PPARgamma: a nuclear regulator of metabolism, differentiation, and cell growth. J Biol Chem. 2001;276(41):37731–4.
- Lecka-Czernik B, Gubrij I, Moerman EA, Kajkenova O, Lipschitz DA, Manolagas SC, et al. Inhibition of Osf2/Cbfa1 expression and terminal osteoblast differentiation by PPAR-gamma 2. J Cell Biochem. 1999;74(3):357–71.
- Lian JB, Stein GS, Javed A, van Wijnen AJ, Stein JL, Montecino M, et al. Networks and hubs for the transcriptional control of osteoblastogenesis. Rev Endocr Metab Disord. 2006;7(1–2): 1–16.
- 13. Bonewald LF. The amazing osteocyte. J Bone Miner Res. 2011;26(2):229–38. Pubmed Central PMCID: 3179345.
- Fulzele K, Riddle RC, Digirolamo DJ, Cao X, Wan C, Chen D, et al. Insulin receptor signaling in osteoblasts regulates postnatal bone acquisition and body composition. Cell. 2010;142(2):309–19.
- Ferron M, Wei J, Yoshizawa T, Del Fattore A, Depinho RA, Teti A, et al. Insulin signaling in osteoblasts integrates bone remodeling and energy metabolism. Cell. 2010;142(2):296–308.
- Lee NK, Sowa H, Hinoi E, Ferron M, Ahn JD, Confavreux C, et al. Endocrine regulation of energy metabolism by the skeleton. Cell. 2007;130(3):456–69.
- Clemens TL, Karsenty G. The osteoblast: an insulin target cell controlling glucose homeostasis. J Bone Miner Res. 2011;26(4):677–80.
- Krakauer JC, McKenna MJ, Buderer NF, Rao DS, Whitehouse FW, Parfitt AM. Bone loss and bone turnover in diabetes. Diabetes. 1995;44(7):775–82.
- Reyes-Garcia R, Rozas-Moreno P, Lopez-Gallardo G, Garcia-Martin A, Varsavsky M, Aviles-Perez MD, et al. Serum levels of bone resorption markers are decreased in patients with type 2 diabetes. Acta Diabetol. 2013;50(1):47–52.
- Garcia-Martin A, Rozas-Moreno P, Reyes-Garcia R, Morales-Santana S, Garcia-Fontana B, Garcia-Salcedo JA, et al. Circulating levels of sclerostin are increased in patients with type 2 diabetes mellitus. J Clin Endocrinol Metab. 2012;97(1):234–41.
- 21. Silverman SL, Cummings SR, Watts NB. Recommendations for the clinical evaluation of agents for treatment of osteoporosis: consensus of an expert panel representing the American Society for Bone and Mineral Research (ASBMR), the International Society for Clinical Densitometry (ISCD), and the National Osteoporosis Foundation (NOF). J Bone Miner Res. 2008;23(1):159–65.
- 22. Zhou S, Schuetz JD, Bunting KD, Colapietro AM, Sampath J, Morris JJ, et al. The ABC transporter Bcrp1/ABCG2 is expressed in a wide variety of stem cells and is a molecular determinant of the side-population phenotype. Nat Med. 2001;7(9):1028–34.
- Jang WG, Kim EJ, Bae IH, Lee KN, Kim YD, Kim DK, et al. Metformin induces osteoblast differentiation via orphan nuclear receptor SHP-mediated transactivation of Runx2. Bone. 2011;48(4):885–93.
- Mai QG, Zhang ZM, Xu S, Lu M, Zhou RP, Zhao L, et al. Metformin stimulates osteoprotegerin and reduces RANKL expression in osteoblasts and ovariectomized rats. J Cell Biochem. 2011;112(10):2902–9.
- Sedlinsky C, Molinuevo MS, Cortizo AM, Tolosa MJ, Felice JI, Sbaraglini ML, et al. Metformin prevents anti-osteogenic in vivo and ex vivo effects of rosiglitazone in rats. Eur J Pharmacol. 2011;668(3):477–85.
- Kahn SE, Zinman B, Lachin JM, Haffner SM, Herman WH, Holman RR, et al. Rosiglitazoneassociated fractures in type 2 diabetes: an analysis from A Diabetes Outcome Progression Trial (ADOPT). Diabetes Care. 2008;31(5):845–51.
- Zinman B, Haffner SM, Herman WH, Holman RR, Lachin JM, Kravitz BG, et al. Effect of rosiglitazone, metformin, and glyburide on bone biomarkers in patients with type 2 diabetes. J Clin Endocrinol Metab. 2010;95(1):134–42.
- Vestergaard P, Rejnmark L, Mosekilde L. Relative fracture risk in patients with diabetes mellitus, and the impact of insulin and oral antidiabetic medication on relative fracture risk. Diabetologia. 2005;48(7):1292–9.

- 7 Safety of Antidiabetic Therapies on Bone
 - 29. Melton LJ, Leibson CL, Achenbach SJ, Therneau TM, Khosla S. Fracture risk in type 2 diabetes: update of a population-based study. J Bone Miner Res. 2008;23(8):1334–42.
- 30. Monami M, Cresci B, Colombini A, Pala L, Balzi D, Gori F, et al. Bone fractures and hypoglycemic treatment in type 2 diabetic patients: a case-control study. Diabetes Care. 2008;31(2):199–203. Epub 2007/11/21. Eng.
- 31. Kanazawa I, Yamaguchi T, Yamamoto M, Sugimoto T. Relationship between treatments with insulin and oral hypoglycemic agents versus the presence of vertebral fractures in type 2 diabetes mellitus. J Bone Miner Metab. 2010;28(5):554–60. Epub 2010/02/24. Eng.
- Bilik D, McEwen LN, Brown MB, Pomeroy NE, Kim C, Asao K, et al. Thiazolidinediones and fractures: evidence from translating research into action for diabetes. J Clin Endocrinol Metab. 2010;95(10):4560–5. Epub 2010/07/16. Eng.
- 33. Colhoun HM, Livingstone SJ, Looker HC, Morris AD, Wild SH, Lindsay RS, et al. Hospitalised hip fracture risk with rosiglitazone and pioglitazone use compared with other glucose-lowering drugs. Diabetologia. 2012;55(11):2929–37. Pubmed Central PMCID: 3464390, Epub 2012/09/05. Eng.
- Napoli N, Strotmeyer ES, Ensrud KE, Sellmeyer DE, Bauer DC, Hoffman AR, et al. Fracture risk in diabetic elderly men: the MrOS study. Diabetologia. 2014;57(10):2057–65. Epub 2014/06/09. Eng.
- 35. Tzoulaki I, Molokhia M, Curcin V, Little MP, Millett CJ, Ng A, et al. Risk of cardiovascular disease and all cause mortality among patients with type 2 diabetes prescribed oral antidiabetes drugs: retrospective cohort study using UK general practice research database. BMJ. 2009;339:b4731. Pubmed Central PMCID: 2788912, Epub 2009/12/05. Eng.
- Ivers RQ, Cumming RG, Mitchell P, Peduto AJ. Diabetes and risk of fracture: The Blue Mountains Eye Study. Diabetes Care. 2001;24(7):1198–203.
- Ahmed LA, Joakimsen RM, Berntsen GK, Fonnebo V, Schirmer H. Diabetes mellitus and the risk of non-vertebral fractures: the Tromsø study. Osteoporos Int. 2006;17(4):495–500. Epub 2005/11/12. Eng.
- Schwartz AV, Vittinghoff E, Bauer DC, Hillier TA, Strotmeyer ES, Ensrud KE, et al. Association of BMD and FRAX score with risk of fracture in older adults with type 2 diabetes. JAMA. 2011;305(21):2184–92. Pubmed Central PMCID: NIHMSID, Epub 2011/06/03. Eng.
- 39. Schwartz AV, Hillier TA, Sellmeyer DE, Resnick HE, Gregg E, Ensrud KE, et al. Older women with diabetes have a higher risk of falls: a prospective study. Diabetes Care. 2002;25(10):1749–54.
- Volpato S, Leveille SG, Blaum C, Fried LP, Guralnik JM. Risk factors for falls in older disabled women with diabetes: the women's health and aging study. J Gerontol A Biol Sci Med Sci. 2005;60(12):1539–45.
- Kahn SE, Haffner SM, Heise MA, Herman WH, Holman RR, Jones NP, et al. Glycemic durability of rosiglitazone, metformin, or glyburide monotherapy. N Engl J Med. 2006;355(23):2427–43.
- 42. Nissen SE, Wolski K. Effect of rosiglitazone on the risk of myocardial infarction and death from cardiovascular causes. N Engl J Med. 2007;356(24):2457–71.
- Lincoff AM, Wolski K, Nicholls SJ, Nissen SE. Pioglitazone and risk of cardiovascular events in patients with type 2 diabetes mellitus: a meta-analysis of randomized trials. JAMA. 2007;298(10):1180–8.
- 44. Yki-Jarvinen H. Thiazolidinediones. N Engl J Med. 2004;351(11):1106-18.
- 45. Grey A. Skeletal consequences of thiazolidinedione therapy. Osteoporos Int. 2008;19(2):129–37.
- Lecka-Czernik B. Bone loss in diabetes: use of anti-diabetic thiazolidinediones and secondary osteoporosis. Curr Osteoporos Rep. 2010;8(4):178–84. doi:10.1007/s11914-010-0027-y.
- 47. Takeda. Observation of an increased incidence of fractures in female patients who received long-term treatment with ACTOS[®] (pioglitazone HCl) tablets for type 2 diabetes mellitus (Letter to Health Care Providers). Indianapolis: Takeda Pharmaceuticals North America; 2007.
- Loke YK, Singh S, Furberg CD. Long-term use of thiazolidinediones and fractures in type 2 diabetes: a meta-analysis. CMAJ. 2009;180(1):32–9.

- Zhu ZN, Jiang YF, Ding T. Risk of fracture with thiazolidinediones: an updated meta-analysis of randomized clinical trials. Bone. 2014;68:115–23. Epub 2014/09/01. Eng.
- Douglas IJ, Evans SJ, Pocock S, Smeeth L. The risk of fractures associated with thiazolidinediones: a self-controlled case-series study. PLoS Med. 2009;6(9):e1000154. Pubmed Central PMCID: PMC2741577, Epub 2009/09/30. Eng.
- 51. Gruntmanis U, Fordan S, Ghayee HK, Abdullah SM, See R, Ayers CR, et al. The peroxisome proliferator-activated receptor-gamma agonist rosiglitazone increases bone resorption in women with type 2 diabetes: a randomized, controlled trial. Calcif Tissue Int. 2010;86(5):343–9. Epub 2010/04/01. Eng.
- Harslof T, Wamberg L, Moller L, Stodkilde-Jorgensen H, Ringgaard S, Pedersen SB, et al. Rosiglitazone decreases bone mass and bone marrow fat. J Clin Endocrinol Metab. 2011;96(5):1541–8. Epub 2011/03/04. Eng.
- 53. Bilezikian JP, Josse RG, Eastell R, Lewiecki EM, Miller CG, Wooddell M, et al. Rosiglitazone decreases bone mineral density and increases bone turnover in postmenopausal women with type 2 diabetes mellitus. J Clin Endocrinol Metab. 2013;98(4):1519–28. Epub 2013/03/02. Eng.
- 54. Grey A, Bolland M, Gamble G, Wattie D, Horne A, Davidson J, et al. The peroxisomeproliferator-activated receptor-gamma agonist rosiglitazone decreases bone formation and bone mineral density in healthy postmenopausal women: a randomized, controlled trial. J Clin Endocrinol Metab. 2007;92(4):1305–10.
- 55. Berberoglu Z, Yazici AC, Demirag NG. Effects of rosiglitazone on bone mineral density and remodelling parameters in postmenopausal diabetic women: a 2-year follow-up study. Clin Endocrinol (Oxf). 2010;73(3):305–12. Epub 2010/02/13. Eng.
- 56. Borges JL, Bilezikian JP, Jones-Leone AR, Acusta AP, Ambery PD, Nino AJ, et al. A randomized, parallel group, double-blind, multicentre study comparing the efficacy and safety of Avandamet (rosiglitazone/metformin) and metformin on long-term glycaemic control and bone mineral density after 80 weeks of treatment in drug-naive type 2 diabetes mellitus patients. Diabetes Obes Metab. 2011;13(11):1036–46. Epub 2011/06/21. Eng.
- 57. Bone HG, Lindsay R, McClung MR, Perez AT, Raanan MG, Spanheimer RG. Effects of pioglitazone on bone in postmenopausal women with impaired fasting glucose or impaired glucose tolerance: a randomized, double-blind, placebo-controlled study. J Clin Endocrinol Metab. 2013;98(12):4691–701. Epub 2013/09/24. Eng.
- van Lierop AH, Hamdy NA, van der Meer RW, Jonker JT, Lamb HJ, Rijzewijk LJ, et al. Distinct effects of pioglitazone and metformin on circulating sclerostin and biochemical markers of bone turnover in men with type 2 diabetes mellitus. Eur J Endocrinol. 2012;166(4):711–6. Epub 2012/01/24. Eng.
- 59. Grey A, Bolland M, Fenwick S, Horne A, Gamble G, Drury PL, et al. The skeletal effects of pioglitazone in type 2 diabetes or impaired glucose tolerance: a randomized controlled trial. Eur J Endocrinol. 2014;170(2):257–64. Epub 2013/11/13. Eng.
- Wan Y, Chong LW, Evans RM. PPAR-gamma regulates osteoclastogenesis in mice. Nat Med. 2007;13(12):1496–503.
- Ren D, Collingwood TN, Rebar EJ, Wolffe AP, Camp HS. PPARgamma knockdown by engineered transcription factors: exogenous PPARgamma2 but not PPARgamma1 reactivates adipogenesis. Genes Dev. 2002;16(1):27–32.
- Lecka-Czernik B, Moerman EJ, Grant DF, Lehmann JM, Manolagas SC, Jilka RL. Divergent effects of selective peroxisome proliferator-activated receptor-gamma 2 ligands on adipocyte versus osteoblast differentiation. Endocrinology. 2002;143(6):2376–84.
- 63. Rzonca SO, Suva LJ, Gaddy D, Montague DC, Lecka-Czernik B. Bone is a target for the antidiabetic compound rosiglitazone. Endocrinology. 2004;145(1):401–6.
- 64. Sottile V, Seuwen K, Kneissel M. Enhanced marrow adipogenesis and bone resorption in estrogen-deprived rats treated with the PPARgamma agonist BRL49653 (rosiglitazone). Calcif Tissue Int. 2004;75(4):329–37.
- Soroceanu MA, Miao D, Bai XY, Su H, Goltzman D, Karaplis AC. Rosiglitazone impacts negatively on bone by promoting osteoblast/osteocyte apoptosis. J Endocrinol. 2004;183(1): 203–16.

- 66. Cock TA, Back J, Elefteriou F, Karsenty G, Kastner P, Chan S, et al. Enhanced bone formation in lipodystrophic PPARgamma(hyp/hyp) mice relocates haematopoiesis to the spleen. EMBO Rep. 2004;5(10):1007–12.
- 67. Ali AA, Weinstein RS, Stewart SA, Parfitt AM, Manolagas SC, Jilka RL. Rosiglitazone causes bone loss in mice by suppressing osteoblast differentiation and bone formation. Endocrinology. 2005;146(3):1226–35.
- Akune T, Ohba S, Kamekura S, Yamaguchi M, Chung UI, Kubota N, et al. PPARgamma insufficiency enhances osteogenesis through osteoblast formation from bone marrow progenitors. J Clin Invest. 2004;113(6):846–55.
- Lazarenko OP, Rzonca SO, Hogue WR, Swain FL, Suva LJ, Lecka-Czernik B. Rosiglitazone induces decreases in bone mass and strength that are reminiscent of aged bone. Endocrinology. 2007;148(6):2669–80.
- Shockley KR, Lazarenko OP, Czernik PJ, Rosen CJ, Churchill GA, Lecka-Czernik B. PPARg2 nuclear receptor controls multiple regulatory pathways of osteoblast differentiation from marrow mesenchymal stem cells. J Cell Biochem. 2009;106(2):232–46.
- 71. Lecka-Czernik B, Ackert-Bicknell C, Adamo ML, Marmolejos V, Churchill GA, Shockley KR, et al. Activation of peroxisome proliferator-activated receptor gamma (PPARgamma) by rosiglitazone suppresses components of the insulin-like growth factor regulatory system in vitro and in vivo. Endocrinology. 2007;148(2):903–11.
- Moerman EJ, Teng K, Lipschitz DA, Lecka-Czernik B. Aging activates adipogenic and suppresses osteogenic programs in mesenchymal marrow stroma/stem cells: the role of PPARgamma2 transcription factor and TGF-beta/BMP signaling pathways. Aging Cell. 2004; 3(6):379–89.
- 73. Tontonoz P, Spiegelman BM. Fat and beyond: the diverse biology of PPARgamma. Annu Rev Biochem. 2008;77:289–312.
- 74. Choi JH, Banks AS, Kamenecka TM, Busby SA, Chalmers MJ, Kumar N, et al. Antidiabetic actions of a non-agonist PPARgamma ligand blocking Cdk5-mediated phosphorylation. Nature. 2011;477(7365):477–81.
- 75. Rahman S, Czernik PJ, Lu Y, Lecka-Czernik B. β-catenin directly sequesters adipocytic and insulin sensitizing activities but not osteoblastic activity of PPARgamma2 in marrow mesenchymal stem cells. PLoS One. 2012;7(12):e51746.
- 76. Hinds Jr TD, Stechschulte LA, Cash HA, Whisler D, Banerjee A, Yong W, et al. Protein phosphatase 5 mediates lipid metabolism through reciprocal control of glucocorticoid and PPAR{gamma} receptors. J Biol Chem. 2011;286(10):42911–22.
- 77. Stechschulte LA, Hinds Jr TD, Ghanem SS, Shou W, Najjar SM, Sanchez ER. FKBP51 reciprocally regulates GRalpha and PPARgamma activation via the Akt-p38 pathway. Mol Endocrinol. 2014;28(8):1254–64. Pubmed Central PMCID: 4116593.
- Higgins LS, Depaoli AM. Selective peroxisome proliferator-activated receptor gamma (PPARgamma) modulation as a strategy for safer therapeutic PPARgamma activation. Am J Clin Nutr. 2010;91(1):267S–72.
- 79. Lazarenko OP, Rzonca SO, Suva LJ, Lecka-Czernik B. Netoglitazone is a PPAR-gamma ligand with selective effects on bone and fat. Bone. 2006;38:74–85.
- Lee DH, Huang H, Choi K, Mantzoros C, Kim YB. Selective PPARgamma modulator INT131 normalizes insulin signaling defects and improves bone mass in diet-induced obese mice. Am J Physiol Endocrinol Metab. 2012;302(5):E552–60. Pubmed Central PMCID: 4116349.
- Fukunaga T, Zou W, Rohatgi N, Colca JR, Teitelbaum SL. An insulin-sensitizing thiazolidinedione, which minimally activates PPARγ, does not cause bone loss. J Bone Miner Res. 2015;30(3):481–8.
- Kolli V, Stechschulte LA, Dowling AR, Rahman S, Czernik PJ, Lecka-Czernik B. Partial agonist, telmisartan, maintains PPARgamma serine 112 phosphorylation, and does not affect osteoblast differentiation and bone mass. PLoS One. 2014;9(5):e96323. Pubmed Central PMCID: 4014504.
- Choi JH, Banks AS, Estall JL, Kajimura S, Bostrom P, Laznik D, et al. Anti-diabetic drugs inhibit obesity-linked phosphorylation of PPARgamma by Cdk5. Nature. 2010;466(7305):451–6.

- Johnston SS, Conner C, Aagren M, Ruiz K, Bouchard J. Association between hypoglycaemic events and fall-related fractures in Medicare-covered patients with type 2 diabetes. Diabetes Obes Metab. 2012;14(7):634–43. Epub 2012/02/18. Eng.
- Meier C, Kraenzlin ME, Bodmer M, Jick SS, Jick H, Meier CR. Use of thiazolidinediones and fracture risk. Arch Intern Med. 2008;168(8):820–5.
- Solomon DH, Cadarette SM, Choudhry NK, Canning C, Levin R, Sturmer T. A cohort study of thiazolidinediones and fractures in older adults with diabetes. J Clin Endocrinol Metab. 2009;94(8):2792–8. Pubmed Central PMCID: PMC2730861, Epub 2009/05/28. Eng.
- Lapane KL, Yang S, Brown MJ, Jawahar R, Pagliasotti C, Rajpathak S. Sulfonylureas and risk of falls and fractures: a systematic review. Drugs Aging. 2013;30(7):527–47. Epub 2013/04/24. Eng.
- Huang ES, Karter AJ, Danielson KK, Warton EM, Ahmed AT. The association between the number of prescription medications and incident falls in a multi-ethnic population of adult type-2 diabetes patients: the diabetes and aging study. J Gen Intern Med. 2010;25(2):141–6. Pubmed Central PMCID: 2837501, Epub 2009/12/08. Eng.
- Brucker DJ. The role of gut hormones in glucose homeostasis. J Clin Invest. 2007;117(1): 24–32.
- Deacon CF. Circulation and degradation of GIP and GLP-1. Horm Metab Res. 2004; 36(11–12):761–5.
- Wesley UV, McGroarty M, Homoyouni A. Dipeptidyl peptidase inhibits malignant phenotype of prostate cancer cells by blocking basic fibroblast growth factor signaling pathway. Cancer Res. 2005;65(4):1325–34.
- McGuinness C, Wesley UV. Dipeptidyl peptidase IV (DPPIV), a candidate tumor suppressor gene in melanomas is silenced by promoter methylation. Front Biosci. 2008;13:2435–43.
- Arscott WT, LaBauve AE, May V, Wesley UV. Suppression of neuroblastoma growth by dipeptidyl peptidase IV: relevance of chemokine regulation and caspase activation. Oncogene. 2009;28(4):479–91.
- Clowes JA, Khosla S, Eastell R. Potential role of pancreatic and enteric hormones in regulating bone turnover. J Bone Miner Res. 2005;20(9):1497–506.
- Reid IR, Cornish J, Baldock PA. Nutrition-related peptides and bone homeostasis. J Bone Miner Res. 2006;21(4):495–500.
- 96. Henriksen DB, Alexandersen P, Hartmann B, Adrian CL, Byrjalsen I, Bone HG, et al. Disassociation of bone resorption and formation by GLP-2: a 14-day study in healthy postmenopausal women. Bone. 2007;40(3):723–9.
- Xie D, Zhong Q, Ding KH, Cheng H, Williams S, Correa D, et al. Glucose-dependent insulinotropic peptide-overexpressing transgenic mice have increased bone mass. Bone. 2007; 40(5):1352–60.
- Zhong Q, Itokawa T, Sridhar S, Ding KH, Xie D, Kang B, et al. Effects of glucose-dependent insulinotropic peptide on osteoclast function. Am J Physiol Endocrinol Metab. 2007;292(2): E543–8.
- Yamada C, Yamada Y, Tsukiyama K, Yamada K, Udagawa N, Takahashi N, et al. The murine glucagon-like peptide-1 receptor is essential for control of bone resorption. Endocrinology. 2008;149(2):574–9.
- 100. Kyle KA, Willett TL, Baggio LL, Drucker DJ, Grynpas MD. Differential effects of PPAR-{gamma} activation versus chemical or genetic reduction of DPP-4 activity on bone quality in mice. Endocrinology. 2011;152(2):457–67.
- 101. Bunck MC, Eliasson B, Corner A, Heine RJ, Shaginian RM, Taskinen MR, et al. Exenatide treatment did not affect bone mineral density despite body weight reduction in patients with type 2 diabetes. Diabetes Obes Metab. 2011;13(4):374–7.
- 102. Mabilleau G, Mieczkowska A, Chappard D. Use of glucagon-like peptide-1 receptor agonists and bone fractures: a meta-analysis of randomized clinical trials. J Diabetes. 2014;6(3):260–6. Epub 2013/10/30. Eng.
- 103. Su B, Sheng H, Zhang M, Bu L, Yang P, Li L, et al. Risk of bone fractures associated with glucagon-like peptide-1 receptor agonists' treatment: a meta-analysis of randomized controlled trials. Endocrine. 2015;48(1):107–15. Epub 2014/07/31. Eng.

- 104. Monami M, Dicembrini I, Antenore A, Mannucci E. Dipeptidyl peptidase-4 inhibitors and bone fractures: a meta-analysis of randomized clinical trials. Diabetes Care. 2011;34(11): 2474–6.
- 105. Ljunggren O, Bolinder J, Johansson L, Wilding J, Langkilde AM, Sjostrom CD, et al. Dapagliflozin has no effect on markers of bone formation and resorption or bone mineral density in patients with inadequately controlled type 2 diabetes mellitus on metformin. Diabetes Obes Metab. 2012;14(11):990–9.
- 106. Bolinder J, Ljunggren O, Johansson L, Wilding J, Langkilde AM, Sjostrom CD, et al. Dapagliflozin maintains glycaemic control while reducing weight and body fat mass over 2 years in patients with type 2 diabetes mellitus inadequately controlled on metformin. Diabetes Obes Metab. 2014;16(2):159–69. Epub 2013/08/03. Eng.
- 107. FARXIGA [package insert]. Wilmington: AstraZeneca Pharmaceuticals LP; 2014.
- 108. INVOKANA [package insert]. Titusville: Janssen Pharmaceuticals; 2014.
- 109. Goldenberg MM. Pharmaceutical approval update. P T. 2013;38(8):443–5. Pubmed Central PMCID: 3814443, Epub 2013/11/14. Eng.
- 110. Bronsky J, Prusa R, Nevoral J. The role of amylin and related peptides in osteoporosis. Clin Chim Acta. 2006;373(1–2):9–16.
- 111. Cornish J, Callon KE, Bava U, Kamona SA, Cooper GJ, Reid IR. Effects of calcitonin, amylin, and calcitonin gene-related peptide on osteoclast development. Bone. 2001;29(2):162–8.
- 112. Gutierrez-Rojas I, Lozano D, Nuche-Berenguer B, Moreno P, Acitores A, Ramos-Alvarez I, et al. Amylin exerts osteogenic actions with different efficacy depending on the diabetic status. Mol Cell Endocrinol. 2013;365(2):309–15.
- Bronsky J, Prusa R. Amylin fasting plasma levels are decreased in patients with osteoporosis. Osteoporos Int. 2004;15(3):243–7.
- 114. Wojcik MH, Meenaghan E, Lawson EA, Misra M, Klibanski A, Miller KK. Reduced amylin levels are associated with low bone mineral density in women with anorexia nervosa. Bone. 2010;46(3):796–800. Pubmed Central PMCID: 2824019.
- 115. Borm AK, Klevesath MS, Borcea V, Kasperk C, Seibel MJ, Wahl P, et al. The effect of pramlintide (amylin analogue) treatment on bone metabolism and bone density in patients with type 1 diabetes mellitus. Horm Metab Res. 1999;31(8):472–5.

Chapter 8 Bone Marrow Stem Cells and Bone Turnover in Diabetic Disease

Laura R. McCabe, Sandi Raehtz, Caitlyn VanGelderen, and Naiomy D. Rios-Arce

Introduction

The Centers for Disease Control and Prevention in 2012 indicated that over 29 million Americans have diabetes. Hyperglycemia is a diagnostic indicator of diabetes. Diabetes has two main forms: type 1 (T1D) and type 2 (T2D). In T1D, hyperglycemia occurs as a result of little or no insulin production by pancreatic beta cells, thus glucose is unable to be taken up by cells that have insulin-dependent glucose channels. In T2D, hyperglycemia results from insulin resistance that leads to reduced insulin signaling and glucose uptake in insulin-dependent cells. Longterm diabetes affects multiple organ systems resulting in complications such as neuropathy, nephropathy, and retinopathy as well as less well-known complications such as increased fracture risk. Both T1D and T2D patients are at risk for fractures but the underlying pathophysiology is somewhat different [1-6]. T1D is associated with reduced bone density while T2D is often linked with increased or unaltered bone density. Thus, there are likely separate and overlapping mechanisms. In this chapter we will focus on understanding the effects of T1D and T2D on the bone microenvironment, stem cell maturation and bone remodeling with particular emphasis on osteoblast, osteoclast, and immune cell activity and composition. Understanding the underlying mechanisms of diabetic bone changes is critical for identifying and developing effective therapeutics.

L.R. McCabe, Ph.D. (🖂) • S. Raehtz, M.S. • C. VanGelderen, B.A. • N.D. Rios-Arce, B.S. Departments of Physiology and Radiology, Michigan State University, East Lansing, MI 48824, USA e-mail: mccabel@cns.msu.edu

[©] Springer International Publishing Switzerland 2016 B. Lecka-Czernik, J.L. Fowlkes (eds.), *Diabetic Bone Disease*, DOI 10.1007/978-3-319-16402-1_8

The Bone Marrow Microenvironment

The bone marrow contains many different cell types including stem cells, hematopoetic cells, and immune cells that are in direct contact with the cells that ultimately determine bone density: osteoblasts and osteoclasts. Stem cells are defined as undifferentiated cells capable of self-renewing that have the potential to differentiate into specialized cell types such as red blood cells, bone cells, muscle, adipocytes and immune cells. They are also able of differentiating into cells of non-hematopoietic tissue such as the heart and pancreas [7, 8].

Bone marrow stem cells give rise to both hematopoietic and non-hematopoietic stem cells. Hematopoietic stems cells (HSCs) are important for generating the immune system and renewing blood cells. HSCs can differentiate into lymphoid progenitor cells (i.e., B-lymphocytes, T-lymphocytes, Natural killer (NK) cells) and myeloid progenitor cells (i.e., monocytes, macrophages, Langerhans cells, dendritic cells, megakaryocytes, and granulocytes). HSCs respond to different environmental signals, such as cytokines, to maintain cellular homeostasis and respond to the body's need for certain cell types. Non-hematopoietic mesenchymal stem cells (MSCs) differentiate into osteoblasts, osteocytes, chondrocytes, and adipocytes as well as into myocytes, neurons, and astrocytes in vitro and in vivo. MSCs can also differentiate into endothelial stem cells (ESCs), which are important in blood vessels and lymphatic vessels, and are thought to contribute to the healing of other organs including intestine [9, 10].

The cross talk among cells located within the bone (both non-hematopoietic and hematopoietic) is critical in the regulation of bone remodeling which involves both osteoblasts and osteoclasts. Osteoblasts, derived from MSCs, interact with the hematopoietic precursors through local and systemic factors (i.e., receptor activator of nuclear factor kappa-B ligand (RANKL)) and cells (particularly macrophages and T-cells) to enhance the differentiation and activity of osteoclasts. Osteoclasts, derived from HSCs of the monocyte/macrophage lineage, are induced to differentiate and become active (resorb bone) in the presence of macrophage colony stimulating factor (M-CSF) and RANKL [11] which is further enhanced by the presence of factors such as tumor necrosis factor alpha (TNF- α) [11]. Osteoblasts and osteoclasts work together during bone remodeling to maintain skeletal integrity. This balance is vital to bone health and therefore the two cell types communicate with each other extensively. This cross-talk begins even before the cells are fully mature. Diabetes is one of many diseases that can result in altered bone marrow stem cell maturation leading to changes in bone remodeling.

The Regulation of Osteoblasts and Bone Formation by Diabetes

Regulation of Osteoblast Activity: Normal Physiology

Osteoblast activity (bone formation) is regulated at three main levels: lineage selection, maturation, and apoptosis [12] (Fig. 8.1). Lineage selection, the balance of MSCs maturation into osteoblasts (bone) versus adipocytes (marrow adiposity) [13], is a well-understood mechanism that regulates bone density [14]. MSCs progression to the osteoblast lineage is reduced under conditions of age-related or disease-related osteoporotic conditions as well as with certain pharmacologic therapies such as Rosiglizaone [14-20]. While expression of Runx2 (an osteogenic transcription factor) induces osteogenesis [21], expression and activation of peroxisome proliferator-activated receptor- γ (PPAR- γ , an adipogenic transcription factor) promotes adipogenesis. When PPARy2 is activated (by binding fatty acid ligands) or when PPARy2 is overexpressed, mice display increased marrow adiposity, suppressed osteoblast maturation and bone loss [22-27]. In contrast, PPARy2deficient mice have reduced marrow adiposity and increased bone density [28, 29]. In vitro studies also show that stimulation of MSC towards adipogenesis reduces osteogenesis [30, 31]. Factors that regulate selection of osteogenesis over adipogenesis include transforming growth factor beta (TGF- β), bone morphogenetic proteins (BMPs), cytokines, adipokines, thyroid hormones, metabolic stress, and Wnt signaling [11, 32-41].

Osteoblast activity is also regulated through modifying the rate/extent of cell maturation and death. During early stages of maturation, osteoblasts express collagen I, the major extracellular matrix component in bone [42]. Collagen I provides the foundation for subsequent matrix maturation and mineralization. During late stages of maturation, osteoblasts express osteocalcin, which is carboxylated at three residues to increase affinity to the bone matrix [43]. Not all osteocalcin is carboxylated. The undercarboxylated form can enter into the circulation to stimulate insulin secretion, insulin signaling sensitivity, and glucose homeostasis [43-46]; thus bone may affect and regulate diabetes. BMP and Wnt signaling promote osteoblast differentiation while many pro-inflammatory cytokines such as TNF-α reduce maturation. Once bone mineral is made, osteoblasts become osteocytes and embed within the mineralized bone, become bone lining cells that lie atop the newly formed bone, or undergo apoptosis [13, 21, 47, 48]. Increased osteoblast death is associated with decreased bone density due to the lack of bone forming cells. Metabolic stress (hypoxia, hyperglycemia) and pro-inflammatory cytokines can promote osteoblast apoptosis and bone loss [48-52].



Fig. 8.1 Diabetes significantly affects osteoblast activity at multiple levels. Mesenchymal stem cells can differentiate into pre-osteoblasts or pre-adipocytes as well as other cell types. Runx2 is expressed when mesenchymal stem cells commit to osteogenesis and is required for osteoblast (bone formation cells) lineage selection. During the progression of osteoblast maturation stage, specific genes are expressed including osteocalcin, which is a late stage marker of osteoblast maturation. Following maturation, osteoblasts can undergo apoptosis or become embedded in the bone (osteocytes) during mineralization. Mesenchymal stems cells can also commit to adipogenesis. C/ EBPβ promotes the expression of PPARγ2 which induces expression of other adipogenic genes such as aP2 (a protein that binds fatty acids). Osteoblast activity can be decreased by (1) promoting adipogenesis rather than osteogenesis, (2) inhibiting osteoblast maturation, and (3) increasing osteoblast apoptosis. Factors including TGF-β, Wnts, and BMPs increase osteogenesis, osteoblast maturation, and osteoblast viability. Diabetes can reduce osteoblast activity at all three levels of regulation. Factors such as diabetes-induced inflammation and reduced Wnt signaling can lead to suppressed osteoblast activity through modulation of the three levels of osteoblast regulation. These changes are thought to contribute to the decrease in bone formation seen in diabetes

Effect of T1 Diabetes on Osteoblasts

Decreased osteoblast activity is the key contributor to decreased bone mineral density observed in T1D children and adults [53-58]. Consistent with a decrease in bone mineral density, serum osteocalcin levels are typically lower in diabetic patients compared to nondiabetic subjects [46, 48, 53, 59–73] and are inversely proportional to glycosylated hemoglobin levels, a measure of metabolic control [74, 75]. Levels of serum undercaboxylated osteocalcin, which stimulates insulin secretion [59, 62, 76, 77], are negatively correlated with glycosylated hemoglobin levels in T1D male patients [78]. T1D rodents display bone loss similar to diabetic patients [68, 69, 71-73, 77, 79-82] and are useful models to study mechanisms of T1D bone loss. Specifically, diabetic mice and rats exhibit decreased trabecular bone volume and reduced cortical thickness (in some reports) within 4 weeks of T1D induction. T1D mouse and rat models also display decreased serum osteocalcin levels [64, 71, 73, 77, 80, 83] as well as decreased bone osteocalcin mRNA levels [46, 60, 62, 73, 84]. Histologic analyses indicate that osteoblast surface, mineral apposition and dynamic bone formation rates at trabecular, endosteal, and periosteal sites are reduced in T1D bone [52, 62, 76, 77, 85]. Bone implant integration and fracture healing are also suppressed in diabetic rodent models [72, 85, 86] consistent with an overall reduction in bone formation. Thus, a critical area of investigation is determining at what level (lineage selection, maturation, death) osteoblast activity is modified by T1D.

Lineage Selection

Several lines of evidence support that T1D alters MSC lineage selection. Most notably, mouse models of T1D display an increase in bone marrow adipocyte number and a decrease in the number of surface osteoblasts [14, 16, 18, 19, 45, 60, 87]. In vitro studies further indicate that factors present in T1D patients and mice (i.e., high glucose, pro-inflammatory cytokines) are capable of promoting MSC adipogenesis at the cost of osteogenesis [45, 49, 50, 60, 83, 88–90]. In addition, a recent study in rats found that the osteoprogenitor pool in T1D rat bone marrow was significantly depleted [88], further suggesting decreased lineage selection towards osteogenesis and/or a reduction in overall MSC number. Interestingly, the T1D adiposity is bone marrow specific, as T1D mice have depleted subcutaneous and visceral fat stores and exhibit weight loss [48, 83, 87]. This is consistent with the weight loss seen in T1D patients [87, 91]. Both T1D male and female mice display increased bone marrow adiposity in tibia, femur, and calvaria, and correspondingly display increased levels of adipocyte markers such as PPARy2 and adipocyte protein 2 [aP2, also known as fatty acid binding protein 4 (FABP4)] in bone [45, 62]. T1D-induced marrow adiposity could be either an active or passive shunt of MSCs towards adipogenesis and away from osteogenesis [17-19, 92, 93] (see section on "Potential Mechanisms Contributing to Altered Bone Remodeling").

Several in vivo studies have examined the role of T1D-induced marrow adiposity in mediating bone loss in T1D mouse models. Pharmacologically, an antagonist of PPAR γ (bisphenol-A-diglycidyl ether) was given to T1D mice to inhibit adipogenesis. This approach was successful in preventing T1D-induced marrow adiposity, but T1D bone loss still occurred [23–25, 79]. Genetic manipulations, such as the knockout of C/EBP- β (a transcription factor required early in adipogenesis) have not been effective in preventing T1D bone loss [94, 95]. The lack of an association between bone marrow adiposity and bone loss was also observed in T1D patients [96], though marrow adiposity did correlate with serum lipid levels [96] suggesting a link to hyperlipidemia. Larger clinical studies are needed to determine significant relationships. Consistent with the lack of a link between marrow fat and T1D bone loss, mouse vertebral bone which has reduced density in T1D does not display an increase in marrow adiposity [45]. Taken together, these studies indicate that T1D is associated with changes in MSC lineage selection but T1D bone loss is not the direct result of increased marrow adiposity. Additional pathways regulating osteoblast activity must be involved.

Maturation

Once osteoblast lineage is selected, pre-osteoblasts progressively mature through stages of matrix production and mineralization. Identifying a specific change in maturation is somewhat difficult since altered lineage selection and increased death can also reduce markers of maturation in T1D bone. It is known that bone defect healing is decreased in T1D rats [97]. Similarly, T1D decreases bone formation on titanium and hydroxyapatite implants [80, 98, 99]. Though a reduction in pre-osteoblasts would cause similar outcomes, cell culture studies provide evidence to suggest that T1D reduces osteoblast maturation. Pre-osteoblast cell lines (i.e., MC3T3-E1 cells) cultured under conditions associated with T1D (high glucose or pro-inflammatory cytokines such as TNF α) display reduced levels of maturation markers [100–104]. In many of these conditions there is also evidence of enhanced expression of adipocyte markers (PPAR γ and aP2) [45, 60, 62] suggesting a lineage or functional shift to an adipocyte.

Apoptosis

Increased osteoblast apoptosis can also contribute to reduced T1D osteoblast activity. T1D associated factors, such as reactive oxygen species (ROS), advanced glycation end-products (AGEs) and pro-inflammatory cytokines, are linked to increased osteoblast death [33, 48, 105–107]. In T1D mouse models, the number of osteoblasts stained by terminal deoxynucleotidyl transferase dUTP nick end labeling (TUNEL), a method of detecting cellular apoptosis, was increased twofold compared to levels in control mouse femurs and tibias [52, 108]. The increase in apoptotic osteoblasts was confirmed in spontaneously diabetic *Ins2*^{+/-} mice where the increase was over threefold when compared to control [52]. Similarly, mRNA levels of Bax (a pro-apoptotic protein) were increased in streptozotocin (STZ)-induced diabetic mice and $Ins2^{+/-}$ mice [52]. When osteoblasts (primary cells and cell lines) were cocultured with control versus T1D mouse bone marrow, the T1D marrow caused a twofold increase in osteoblast caspase-3 activity, a marker of cell apoptosis [52]. These findings support a role for the diabetic bone marrow microenvironment in inducing osteoblast death [52].

Effect of T2 Diabetes on Osteoblasts

In clinical studies, T2D patients display an increase, no change or decreased bone density [1–6]. Some studies found T2D to be associated with a decrease in serum osteocalcin levels in men and postmenopausal women [78, 109]. Variations in bone responses are likely a consequence of the complexity of the pathogenesis of T2D. Some T2D patients with above average weight and normal to high bone mineral density (factors typically associated with decreased risk fracture) also have increased risk fracture [110]. T2D can occur in non-obese and obese patients and thus bone changes can be associated with increased load, hyperlipidemia, hyperglycemia, inflammation and/or other complications that can contribute to altered bone density.

Accordingly, there are several T2D mouse models including obese and nonobese. In obese models such as T2D obese Wistar fatty rats, serum osteocalcin and bone osteocalcin levels are decreased as well as bone parameters such as length, strength, and weight [111]. Bone formation and turnover are both decreased in these T2D rodent models [112, 113] In Zucker diabetic fatty (ZDF) rats (harbor a mutation in the leptin receptor, become obese rats and develop T2D at 9 week of age), a 55 % decrease in mineralized matrix formation was shown when bone marrow cells were stimulated by osteogenic media [114]. The same cells showed a 40-80 % reduction in Runx2, osteopontin, BMP-2, and osteocalcin mRNA expression. This data suggests that T2D may decrease lineage selection away from osteogenesis and decrease osteoblast maturation. The ob/ob mouse, a severe obesity model for T2D that harbors a mutation in the leptin gene, displays decreased amounts of trabecular and cortical bone and reduced femoral bone length, but increased trabecular bone volume in the lumbar vertebrae [115-119]. These mice also have decreased bone formation in both long bones and vertebrae [115, 119]. In the db/db T2D model (which harbor a mutation in the leptin receptor and become diabetic at 6-8 weeks of age) P. gingivalis-induced inflammation is exacerbated [112]. Bone formation was impaired and osteocalcin mRNA expression in the calvaria was decreased [112]. Caspase-3 and caspase-8 mRNA levels in the calvaria (apoptotic proteins) were increased over twofold when compared to control [83]. While osteoblast activity is decreased in these models, interpretations are complicated by not being able to distinguish bone response from T2D versus decreased leptin signaling.

Non-obese Goto-Kakizaki mice display reduced cortical bone thickness but this model showed abnormal pancreatic signs in the embryonic stage [33, 111]. Another non-obese T2D model, Torii rats, display suppressed osteoid surface, bone strength, and osteoblast numbers, which are all reversed by treatment with insulin [120]. A spontaneous T2D model (muscle IGF-1-1R-lysine-arginine, MKR) has a mutation in the skeletal muscle IGF-1 that leads to hyperglycemia by 7–8 weeks of age [110, 121–124]. MKR mice have decreased bone volume/total volume at the distal femoral metaphyses, stiffness, and failure load when compared to controls [122].

The Regulation of Osteoclasts and Bone Resorption by Diabetes

Regulation of Osteoclast Activity

Osteoclasts are multinucleated cells responsible for the resorption phase of the bone remodeling process. These cells are derived from HSCs, which reside in the bone marrow. Differentiation of HSCs into active osteoclasts is a multistep process (Fig. 8.2) and is regulated by three main cytokines: macrophage colony stimulating factor (M-CSF), RANKL and osteoprotegerin (OPG). These factors are derived



Fig. 8.2 Model for diabetes induced changes in osteoclast activity. Osteoclastogenesis involves early expression of PU.1 which regulates expression of the receptor for M-CSF. Early osteoclast maturation is induced by M-CSF and RANKL (expressed by osteoblasts). RANKL binds the RANK receptor on osteoclasts to activate NFkB activity, initiating the transcription of genes necessary for bone resorption. Diabetes-associated conditions such as inflammation (TNF α) and hyperlipidemia are known to activate osteoclast activity through increasing osteoclastogenesis and maturation. However, diabetes is typically associated with decreased bone resorption. This likely occurs due to other diabetes-associated factors such as hyperglycemia as well as expression of anti-inflammatory cytokines

from osteoblasts as well as other cells of the immune system including T-cells and B-cells [131]. While RANKL binds to RANK on osteoclasts to activate resorption, OPG (a secreted as well as membrane-bound protein produced by osteoblasts) acts as a decoy for RANKL, preventing it from binding to the transmembrane receptor RANK [137–139]. Both M-CSF and RANKL can increase nuclear factor-kappa (NF-kB) activity to stimulate osteoclast maturation [132–136]. RANKL does this by first inducing TNF receptor-associated factor (TRAF) proteins to start a signaling pathway that ends in the activation of NF-kB. NF-kB is typically bound in the cytoplasm by the inhibitor of kB (IkB). Binding of RANKL induces the phosphorylation of IkB to allow NF-kB to enter the nucleus to stimulate genes required for osteoclast function. Mutations in this signaling pathway lead to defective osteoclast formation and function [140–143]. Similarly, mutations in the gene encoding M-CSF causes osteopetrosis due to the inability of osteoclasts to fully mature.

Other factors involved in osteoclast differentiation include PU.1, which is a transcription factor that is required for the normal formation of osteoclasts. PU.1 regulates the receptor for macrophage colony stimulating actor (M-CSF) and integrin proteins needed for osteoclast maturation and binding of the osteoclast to the bone surface [125–127]. Mice lacking PU.1 die immediately after being born and have symptoms of osteopetrosis, an increase in bone density [128]. PU.1 interacts with other transcription factors (MITF, micro-ophthalmia-associated transcription factor) to regulate genes important for osteoclast function such as tartrate-resistant acid phosphatase (TRAP) and carbonic anhydrase 2 (CA-II) [129, 130].

Effect of T1 Diabetes on Osteoclasts

The majority of rodent studies examining T1D show no change or decreases in markers of osteoclastic resorption [33, 48, 62, 85, 144-147]. However, there is some controversy since a few studies have found increased osteoclast parameters in animals [147]. Clinical studies support no change in T1D osteoclast activity as determined by levels of serum markers of resorption such as deoxypyridinoline and c-terminal telopeptide of type 1 collagen (CTX) [53, 69, 86, 148]. In mice, the typical STZ dosing regimen used to induce T1D does not cause a change in osteoclast parameters (consistent with spontaneous mouse models [60]), however, increasing the dose causes an increase in osteoclast activity possibly due to more rapid onset of T1D and/or inflammation (as seen in the liver with high STZ dose) [67]. As noted previously, T1D increases bone marrow adiposity [60, 62] and thus MSC lineage selection. Not only does this change the overall physical morphology of the bone marrow compartment, but it also reduces the number of MSC maturing to the osteoblasts that are needed for HSC maintenance and maturation to osteoclasts (via M-CSF and RANKL) [149]; this can contribute to reduced osteoclastogenesis in the presence of inflammation. Other contributors to reduced resorption in T1D include hyperglycemia which can directly decrease RANKL-induced osteoclastogenesis [150], a reduced number of osteoblasts which can reduce RANKL levels in bone, and elevated levels of anti-inflammatory cytokines.

Effect of T2 Diabetes on Osteoclasts

T2D effects on patient bone densities are variable with reports indicating lower, the same, or higher bone density than nondiabetic individuals. In T2D animal models, such as the leptin receptor knock out *db/db* mouse, mice display suppression of bone remodeling, with both bone formation and resorption decreased [151]. Yet, osteoclast activity in T2D male patients has been reported to increase [152, 153]. Serum concentrations of TRAP and urine markers for bone resorption, such as CTX, deoxypyridinoline, and N-telopeptide of type I collagen (NTX), are significantly elevated in T2D patients [154–156]. However, T2D patients can also exhibit increased levels of OPG which can reduce the increased resorption caused by T2D [157].

Role of Marrow Immune Cells in the Regulation of Bone Remodeling in Diabetes

Bone Marrow: Bone Remodeling Signaling

Immune cells are capable of regulating osteoblast and osteoclast activity and vice versa. The role of immune cell populations and their effect on bone marrow hematopoiesis has been demonstrated in a variety of pathology conditions associated with bone loss. For example, activated T-cells are involved in regulating bone homeostasis in estrogen deficiency, an effect that was mediated by IFNy [158]. In ovariectomized (ovx) mice, T-cells express high levels of TNF- α (enough to induce osteoclastogenesis) [159]. T-cell deficiency prevents ovx induced bone loss, a benefit lost when mice are given T-cells from WT mice but not T-cells from TNFadeficient mice [160]. Similarly, blockade of TNF- α and IL-1 by Anakinra or Etanercept decreases bone resorption in postmenopausal women [161]. This suggests that an increase in these cytokines can be one of the causes of bone loss. However, TNF- α , IFN γ , and IL-1 are not the only cytokines associated with bone loss. Another cytokine IL-7 promotes osteoclastogenesis by upregulating T- and B-cell-derived RANKL [162]. IL-23 also stimulates the differentiation of human osteoclasts from peripheral blood mononuclear cells (PBMC) in the absence of osteoblasts or exogenous RANKL and in vivo blockade of IL-23 activity prevents inflammation and bone destruction in collagen-induced arthritis in rats [163]. These findings provide a new insight into the interaction between immune system, bone marrow and bone loss.

Another intracellular system that is associated with changes in the bone marrow, immune cells, and bone density is Wnt signaling. Wnt signaling regulates the reciprocal relationship between adipocytes and osteoblast. Part of the mechanism whereby Wnt10b inhibits adipogenesis and stimulates osteoblastogenesis is by suppression of PPAR- γ . In addition, it has been found that expression of Wnt10b in

marrow increases trabecular bone mass and strength [37]. Wnt ligands are also required for hematopoietic progenitor cells (HSPC) expansion and survival. Experiments in ovx mice has shown that T-cell-expressed CD40 ligand (CD40L) enhance T-cell production of Wnt10b which results in activation of stromal cell (SCs) and HSPCs, along with improvement of cytokine production by SCs [164]. In summary, these results recognized the impact of cellular interaction between immune cells, HSPCs and the effect in bone homeostasis.

Effect of T1D on Marrow Cells

T1D is an autoimmune disease characterized by destruction of pancreatic β -cells producing insulin. White blood cells, like T-cells and macrophages, and the interaction and secretion of cytokines, nitric oxide, and free radicals are involved in this autoimmune pathology. Cytokines, such as IFN γ , IL-1 β , and TNF α , can act together to induce beta cell death and can also affect the expression of genes that are protective or harmful for beta cell survival such as signal transducers and activators of transcription 1 (STAT-1) and interferon regulatory factor 1 (IRF-1) [169]. One study examining T-helper cell cytokine profiles indicated higher intracellular TNF α in CD8⁺T cells at the time of T1D diagnosis and higher intracellular TNFa in CD4⁺T lymphocytes in patients at 3 months post-diagnosis, compared with controls [170]. This suggests that there are distinct roles for cytokine release during different stages of T1D. Diabetic mouse models also display elevated TNFa levels in serum and increased TNF α mRNA level/expression in bone [52, 72]. In the STZ-induced diabetic mouse model IL-1 α , IL-6, IFN γ , and TNF α expression are increased in bone with the onset of T1D and during the onset of altered expression of bone phenotype markers [72] (Fig. 8.3). Furthermore, it has been shown that T1D reduces the number of MSCs found at sites of fracture repair and that suppression of TNFa with Pegsunercept treatment restores the level of MSCs to control levels. It was further determined that TNF α is capable of reducing MSC proliferation and increasing MSC apoptosis [171].

Emerging evidence indicates that T1D directly compromises the function of the bone marrow [57]. Increased local cytokine levels in the bone marrow of mice under hyperglycemic conditions can disrupt hematopoiesis by shifting monocytes towards a pro-inflammatory phenotype [167, 172]. This change generated more monocytes and less endothelial progenitor cells (EPCs) in T1D, which contribute to the development of microvascular complications in this condition [166]. This finding was also evident in individuals with non-proliferative diabetic retinopathy [173]. In another experiment using Yorkshire pigs, after STZ-induction of diabetes a significant impairment of bone marrow and circulating EPCs as well as endothelial cells function were observed [174]. Chronic diabetes is also associated with dysfunction in the number and repopulation activity of HSPC, which is likely mediated by alterations in cytokines involved in stem cell renewal and maintenance [165]. Longstanding STZ-induced and spontaneous T1D mice further display impairment



Fig. 8.3 Model for the role of inflammation in diabetes induced bone pathology. Both T1D and T2D conditions can increase bone marrow inflammation. The release of inflammatory cytokines into the bone marrow microenvironment by immune and bone cells can directly impact osteoblast and osteoclast activity which can potentially result in reduced bone turnover (T2D) or bone loss (T1D). TNF-α has extensively been shown to reduce osteoblast viability, lineage selection, and maturation. Similarly, IL-6, IL-8, and IL-1B can negatively affect osteoblast activity. Interestingly, while these cytokines are all capable of promoting osteoclast activity, conditions of diabetes such as hyperglycemia as well as expression of other inhibitory cytokines likely lead to the overall outcome of suppressed osteoclast activity typically seen in diabetes

of colony-forming capability of the bone marrow and an increase in the number of Ly6C^{hi} (pro-inflammatory) monocytes in the peripheral blood [166].

The bone marrow is also the source of cells that contribute to diabetes-associated inflammation. Bone marrow transplantation experiments done in both mice and rats, demonstrate that most of the extrapancreatic proinsulin producing cells (found in liver, adipose tissue, spleen, and thymus) originated from the bone marrow [168]. These cells were also demonstrated to produce the pro-inflammatory cytokine TNF- α which could contribute to damage of the target organs [168]. This suggests that the damage seen in organs such as bone during diabetes may be due to the irregularity in bone marrow stem cells.

Effect of T2D on Marrow Cells

T2D is not an autoimmune disease and is not always associated with bone loss [175]. However, there are changes in hematopoetic profiles. For example, T2D patients have 33 % less circulating progenitor cells (CPCs) and 40 % less circulating EPCs compared with healthy subjects [176]. Peripheral vascular complications of T2D mellitus are associated with an extensively low number of EPCs [176] which could result from alterations in bone marrow function. Altered Wnt signaling is also associated with T2D. A single nucleotide polymorphism locus in the Wnt5b gene conferred susceptibility to T2D by modifying adipocyte function [177]. Wnt signaling is also altered in immune cells. A link between cellular glucose sensing and the Wnt/b-catenin pathway was recently reported in two macrophage cell lines (J774.2 and RAW264.7 cells) [178]. T2D is also accompanied by an increased inflammatory state that can disrupt osteoclast function and survival [67]. However, a small study in T2D patients examining changes in bone turnover markers found that treatment with nonsteroidal anti-inflammatory drugs causes an increase in osteocalcin levels in men (not in women) and increases serum C-terminal telopeptide (s-CTx) [179]. This small study suggests that T2D inflammation decreases bone turnover.

Potential Mechanisms Contributing to Altered Bone Remodeling

Several mechanisms are thought to contribute to diabetes complications and the bone remodeling changes that occur with diabetes. Diabetes is associated with hyperglycemia, hyperlipidemia decreased insulin signaling, decreased IGF-1, oxidative stress, and inflammation; all of which can contribute to suppressed osteoblast activity among other additional regulators including amylin and incretins. We will discuss several of these mechanisms briefly below.

Hyperglycemia

Hyperglycemia has many negative effects on bone and bone formation. High glucose can result in the production of ROS [180, 181], induction of cellular osmotic responses [182, 183], and increased nonenzymatic glycosylation of proteins and DNA [144, 184–188] as well as suppress calciotropic hormones and growth factors [62, 144, 180–185, 187–189], all of which can lead to decreased osteoblast activity. In vitro studies exposing MC3T3-E1 cells (osteoblastic cell line) to high glucose conditions (30 mM) showed a significant decrease in markers of osteoblast maturation, such as alkaline phosphatase and osteocalcin [90, 190]. Even when MC3T3-E1 cells are exposed to hyperglycemic conditions for short time periods, 48 h, osteocalcin mRNA levels are significantly decreased compared to cells cultured under normal (5 mM) glucose levels [90]. MC3T3-E1 cells cultured in high glucose also show decreased mineralization [190, 191]. These in vitro results suggest hyper-glycemia decreases osteoblast maturation.

Hyperglycemia also promotes MSC adipogenesis over osteogenesis [45, 60, 88–90]. Specifically, culturing primary rat osteoblasts under high glucose conditions (25.5 and 35.5 mM) increases levels of adipogenic markers (PPARy and aP2) and decreases levels of osteogenic markers (osteocalcin) [192]. A similar lineage switch was seen in human osteoblastic MG-63 cells cultured under high glucose conditions. The decrease in osteogenesis appears to be dependent on cAMP/protein kinase A (PKA)/extracellular signal-regulated kinase (ERK) signaling [193] and the PI3K/Akt signaling [192]. Hyperglycemia can also reduce osteoblast growth by 40 % [192] and induce osteoblast apoptosis. Both UMR and MC3T3-E1 osteoblasts undergo increased apoptosis when exposed for 24 h to AGE-BSA compared to unmodified BSA [194]. Thus, hyperglycemia can affect osteoblast activity through alterations in MSC lineage selection, osteoblast maturation, and osteoblast death.

Hyperglycemia decreases osteoclastogenesis in vitro. Using RAW264.7 cells and bone marrow macrophages, it was found that high levels of D(+)glucose inhibited osteoclast formation, ROS production (needed for bone resorption), caspase-3 activity (necessary for RANKL-induced differentiation) and migration to bone resorption pits [150]. In addition to this, osteoclasts derived from bone marrow cells extracted from mice which were exposed to high glucose had decreased mRNA expression of RANK and cathepsin K, decreased TRAP activity and decreased resorption activity [195]. Together, these studies indicate that a high glucose environment decreases osteoclast differentiation and activity.

Hyperglycemia also promotes a nonenzymatic reaction between glucose and proteins that results in AGEs. AGEs build up over time in diabetic tissues [196, 197] because they are irreversible modifications [198]. For example, T2D elderly patients display increased urine pentosidine (an AGE) which was associated with increased clinical fracture incidence and increased vertebral fracture prevalence in those with diabetes but not in those without diabetes [199]. Pentosidine, pyrraline and N(carboxymethyl)lysine (CML) are some of the more commonly characterized AGE products whose levels increase in tissues with collagen matrices, particularly pentosidine, under conditions of diabetes [200-202] and aging [199, 203, 204]. AGE modifications can change the structural and functional properties of proteins and correspondingly their levels correlate with diabetes complication severity, including glycosylated hemoglobin levels and bone loss [188, 201, 202, 205-209]. Type 1 collagen, the most abundant constituent of the bone matrix, is a target for AGE [210, 211], which can accumulate and alter the mechanical properties of bone, decreasing its toughness and could therefore contribute to skeletal fragility [212– 214]. AGE increases collagen crosslinking, which can decrease bone strength and ductility, and increase brittleness [209, 214-218]. An increase in AGEs have been linked to a reduction in serum osteocalcin levels and are positively associated with levels of pentosidine which impairs the biomechanical properties of bone [216, 219-221]. AGEs can modify bone cell behavior by interacting with specific cellular

receptors, such as RAGE (receptor for AGEs). When AGEs bind to RAGE on osteoclasts, NF-kB activity and ROS increase [222, 223]. RAGE is also expressed in osteoblasts and AGE treatment of osteoblasts causes decreased osteoblast maturation [184, 222]. In vitro and in vivo studies showed that AGEs increase calvarial periosteal cell apoptosis and induced apoptosis in primary cultures of human or neonatal rat osteoblastic cells. This effect was mediated through RAGE and increased p38 and c-Jun N-terminal kinase (JNK) [224]. The presence of AGE can also induce osteoblast apoptosis through activation of caspase-3 and caspase-8 [105, 108, 197]. Consistent with these findings, deficiency of RAGE results in an increased BMD compared with control mice [225]. Interestingly, in vitro studies using AGE-modified bone slices and osteoclastogenesis assays have shown that resorption was markedly inhibited. This was confirmed by a marked decrease in the release of type I collagen fragments generated by the collagenolytic enzymes secreted by osteoclasts [226]. This decrease in bone resorption is thought to be due not only to the alteration of the structural integrity of the bone matrix by AGEs, but also their effect on osteoclastic differentiation [226]. AGEs also increase inflammation by monocytes and macrophages due to upregulation of TNF α and IL-1 β [227].

Hyperglycemia can also increase oxidative stress. Oxidative stress is triggered through two ways: the mitochondria becoming overloaded with glucose or through AGEs and Polyol signaling [88]. Oxidative DNA damage markers such as 8-hydroxy-deoxyguanosine are increased in hyperglycemic diabetic mice but decreased when the mice are treated with insulin and become euglycemic [120, 228]. This oxidative stress marker is also elevated in T2D rats and could contribute to decreased osteo-blast maturation based on the in vitro findings of oxidative stress negatively affect-ing osteoblast maturation [120, 122, 229, 230]. In addition, hyperglycemia-induced osteoblast apoptosis has been linked with an increase in oxidative stress [51, 231]. ER-stress activates the unfolded protein response mechanism leading to increased C/EBP homologous protein (CHOP) signaling and cellular apoptosis [51]. T1D rats showed an increase in CHOP positive osteoblasts in the femures which also showed decreased bone mineral density [51].

Wnt Signaling

An important regulator of bone formation is Wnt10b [37, 232]. Wnt10b is critical for promoting MSCs toward the osteoblast lineage [232], enhancing osteoblast maturation and suppressing osteoblast apoptosis. Wnt10b activity increases expression of osteogenic transcription factors such as Runx2, Dlx5, and osterix [37]. In addition, targeted overexpression of Wnt10b in mouse bone leads to increased trabecular bone density [234], while Wnt10b knockout mice display significant bone loss [37]. Wnt10b binds its receptors LRP5, LRP6 and frizzled to increase intracellular beta-catenin levels and stimulate gene expression through TCF/LEF transcription factor binding to gene promoter regions. Alterations in the level or in the activity of LRP5 markedly affect bone density. Specifically, activating mutations lead to increased bone density [235] while inactivating mutations or LRP5 deficiency results in osteoporosis [35, 37, 236]. Similarly LRP5 mutations in humans can increase or decrease bone density [237].

Given that diabetes is marked by reduced bone remodeling and reduced osteoblast activity, a role for altered Wnt signaling has been suggested. It was recently found that TNF α transgenic mice display decreased Wnt activity implying a potential interaction between TNF α and the Wnt signaling pathway [238]. As noted previously TNF α is increased with diabetes [227, 228] and its expression is elevated in T1D mouse bone [72]. Correspondingly, Wnt10b expression is decreased in T1D bone (unpublished data, McCabe) suggesting that Wnt10b downregulation could contribute to the T1D bone phenotype [239, 240]. This is supported by decreased β -catenin staining of osteocytes and osteoblasts in diabetic mouse bones [241]. Interestingly, sclerostin, a circulating inhibitor of Wnt signaling made by osteocytes, is demonstrated to be elevated in T2D patient serum [242]. This could function to suppress Wnt10b and overall bone remodeling in T2D. Future studies will further determine the role of Wnts and other regulators of bone remodeling such as BMPs in the pathophysiology of diabetic bone.

Insulin and IGF-1 Signaling

Because this is already discussed in Chap. 1, we will only briefly discuss the role of insulin and insulin growth factor-1 (IGF-1) changes in diabetes that lead to altered osteoblast activity. Osteoblasts express insulin receptors and can readily respond to insulin treatment [33]. IGF-1 binds to insulin growth factor-1 receptors (IGF1R) and can activate osteoblast maturation, proliferation, and increased matrix generation [244]. Bone resorption is also regulated by IGF-1 acting through osteoblasts [244].

A hormone co-secreted with insulin, amylin is also suppressed in TD1 [48, 145]. IGF-1 has been reported to be decreased in T1D patients and animal models [245, 246]. Decreased IGF-1 in serum has been linked to the reduction in bone mineral density, increased adiposity in bone marrow, and decreased osteoblast differentiation. IGF-1 treatment in diabetic rats was able to increase the mineral apposition rate (rate at which new bone is formed) when compared to diabetic controls [65].

To determine if the lack of insulin is responsible for bone loss, insulin receptor knockout mice were analyzed for osteogenic and adipogenic markers. There was no change in osteogenic markers when compared to the control mice but adipogenic markers were decreased [62, 72, 79]. To analyze the effects of insulin-resistance (T2D) on osteoblasts, a mouse strain was generated to reproduce osteoblast insulin-resistance [247]. These mice were then fed a high-fat diet to induce T2D. Circulating osteocalcin levels were decreased and glucose intolerance worsened. The results implicate osteoblast insulin signaling in whole-body glucose homeostasis. Mice that lack insulin receptor signaling in osteoblasts exhibit either no change in osteoclast number [44] or a marked decrease in bone resorption markers [43]. The latter suggests a role for osteoblast insulin signals in enhancing resorption, consistent with studies demonstrating reduced or unaltered resorption in humans and mouse models of T1D. In both studies, OPG was increased and may be involved in suppressing osteoclast activity in T1 diabetics similar to other studies in T1-diabetic mice [67].

Role of Hyperlipidemia

Diabetes is associated with increased serum lipids. Hyperlipidemia has a number of adverse effects, including osteoporosis [250-254]. Hyperlipidemia results in increased amounts of LDL particles which can accumulate in the subendothelial space and are oxidatively modified to increase the generation of lipid oxidation products [255]. These lipid oxidation products can accumulate in bone [256]. Studies examining the effect of hyperlipidemia on bone health revealed increased levels of parathyroid hormone, TNF-a, calcium and phosphorus, and carboxyl-terminal collagen cross-links (a marker of bone resorption) [257]. Diabetic patients exhibit changes in serum lipid profiles along with hyperlipidemia linked with a decrease in metabolic control [258-260]. In female T1D patients, there was a significant correlation between increased serum lipid levels and decreased bone density [96]. Serum lipids can serve as ligands to activate PPARy2 and promote altered MSC lineage selection and thus could play a role in suppressing osteoblast activity. However, in T1D mice, inhibition of PPARy2 activity prevented hyperlipidemia but did not prevent bone loss [79]. This suggests that hyperlipidemia may not play a major role in T1D bone loss but could contribute to the altered lineage selection of MSC.

Studies examining the role of hyperlipidemia during osteoclastogenesis utilized pre-osteoclasts from hyperlipidemic mice (*LDLR*-/- mice) and control mice. After induction with M-CSF and RANKL, there was an increased number of resorption pits indicating an increase in osteoclastic activity [256]. In addition to this, there was also an increase in TRAP activity, indicating increased osteoclast differentiation [256]. Another study using *LDLR*-/- mice showed that serum levels of CTX, a marker for bone resorption was significantly increased in hyperlipidemic mice [257]. Together, these studies indicate a role for hyperlipidemia in osteoclast differentiation and activation. Although hyperlipidemia is associated with diabetes and is linked with osteoclast activity, hyperglycemia, the driving force of diabetes, decreases osteoclast activity and may work to override the effects of hyperlipidemia seen in patients.

Role of Calcitropic Hormones and Leptin

Studies are controversial regarding the effect of T1D on the serum level of vitamin D, which is key for enhancing intestinal calcium absorption and enhancing some stages of osteoblast maturation. While some studies report no changes in vitamin D levels, others report decreased levels in over 75 % of diabetic children and adults in the US northeast [261–263]. However, it is difficult to distinguish T1D associated vitamin D deficiency since more than 50 % of control subjects in winter months experience vitamin D deficiency [261]. In addition to the controversy about vitamin D deficiency, PTH levels in T1D patients are unchanged or decreased and in T2D patients are decreased [71, 72, 77, 264–266].

Another regulator of bone density is leptin [46]. Leptin is produced and secreted by adipocytes and has been shown to regulate lineage selection by stimulating MSC to differentiate into osteoblasts over adipocytes [189, 267, 268]. In T1D patients,

serum leptin levels vary between patients [45, 269] although some studies report a decrease [270, 271]. Chronic treatment of hypoleptinemic T1D mice with leptin did not prevent bone loss but did reduce marrow adiposity [46]. T2D patients are even more complex with various factors affecting leptin levels including insulin secretion and resistance, gender and fat mass [33]. In vitro experiments show that leptin stimulates osteogenesis over adipogenesis [118, 189, 268].

Inflammation

Both obesity (which can result in T2D) and diabetic conditions can increase bone marrow inflammation. Inflammatory cytokines can directly impact osteoblast activity through changes in lineage selection, maturation, and death. T1D patients and mice have increased inflammation [108, 272–274], and elevated expression of inflammatory cytokines have been identified in T1D mouse bone [72]. Similarly, serum proinflammatory cytokines are increased in T2D patients [275–278]. Pro-apoptotic inflammatory cytokines like TNF- α induce osteoblast apoptosis and have been shown to be elevated in both T1D and T2D patients [46, 83, 147]. Elevated TNF- α mRNA levels are increased in T1D mouse whole crushed bone [83, 147], bone marrow [52], and gingival tissue [279]. When TNF- α binding to its receptor is inhibited by Pegsunercept, diabetes-associated osteoblast apoptosis is reduced [49, 280]. TNF- α and other pro-inflammatory cytokines such as IL-6 and IL-1 can also negatively affect osteoblast differentiation/activity and therefore suppress bone formation [281]. For example, in vitro studies show when cultured osteoblasts are treated with inflammatory cytokines (IFN- γ), there is a decrease in alkaline phosphatase activity [282].

Inflammation in the bone marrow favors expansion of osteoclasts and an increase in their activity. This increase in number can not only lead to bone loss, but also lead to impaired HSC maintenance and homeostasis [283]. Osteoclastic bone resorption can be activated by infection and inflammation. Because osteoclasts come from the same lineage as many immune cells, they express many innate immune cell receptors, such as toll-like receptors which can respond to antigen stimulation [284–286]. In addition to antigen stimulation, pro-inflammatory cytokines such as TNFa, IL-1β, and IL-6 can induce osteoclast-mediated resorption [287, 288]. Studies in T1D mice have shown that induction of an immune response by the bacterial antigen LPS induces altered osteoclast differentiation with mature osteoclasts having increased bone resorption activity and increased release of osteoclastogenic factors (cathepsin K and MMP-9) compared to diabetic mice without LPS stimulation [289]. Cytokines secreted from adipose tissue, IL-6, IL-8, and TNF α , can both activate osteoclast bone resorption and decrease bone formation through the suppression of osteoblast differentiation [281]. While inflammatory conditions can directly activate osteoclasts, the effect of inflammation on osteoblasts can indirectly decreases osteoclast maturation and activity due to the extensive cross talk between the two cell types. A study examining the effects of bacterial-induced inflammation in T2D animals (ob/ob) showed that bone loss was not due to increased osteoclastogenesis but was dependent on the decreased levels of bone formation [112].

Summary

Diabetes affects bone remodeling through multiple mechanisms and alterations in multiple cell types. T1D and T2D have similar and different mechanisms at play that result ultimately in increased fracture risk. While we have mentioned several mechanisms here, there is still much to be learned and more contributors to the bone phenotype are likely.

Acknowledgments NIH grants 5R01DK101050-02, 1R01AT007695-01; SR—MSU Genetics Graduate Program; CV—MSU Biochemistry and Molecular Biology Graduate Program; NR—MSU Comparative Medicine and Integrative Biology Graduate Program.

References

- Hanley DA et al. Associations among disease conditions, bone mineral density, and prevalent vertebral deformities in men and women 50 years of age and older: cross-sectional results from the Canadian Multicentre Osteoporosis Study. J Bone Miner Res. 2003;18:784–90. doi:10.1359/jbmr.2003.18.4.784.
- Strotmeyer ES et al. Diabetes is associated independently of body composition with BMD and bone volume in older white and black men and women: the health, aging, and body composition study. J Bone Miner Res. 2004;19:1084–91. doi:10.1359/jbmr.040311.
- 3. Bonds DE et al. Risk of fracture in women with type 2 diabetes: the Women's Health Initiative Observational Study. J Clin Endocrinol Metab. 2006;91:3404–10. doi:10.1210/jc.2006-0614.
- Burghardt AJ et al. High-resolution peripheral quantitative computed tomographic imaging of cortical and trabecular bone microarchitecture in patients with type 2 diabetes mellitus. J Clin Endocrinol Metab. 2010;95:5045–55. doi:10.1210/jc.2010-0226.
- 5. Cummings SR et al. Risk-factors for hip fracture in white women. N Engl J Med. 1995;332:767–73. doi:10.1056/nejm199503233321202.
- 6. Petit MA et al. Bone mass and strength in older men with type 2 diabetes: the Osteoporotic Fractures in Men Study. J Bone Miner Res. 2010;25:285–91. doi:10.1359/jbmr.090725.
- Orlic D et al. Bone marrow stem cells regenerate infarcted myocardium. Pediatr Transplant. 2003;7 Suppl 3:86–8.
- Ianus A, Holz GG, Theise ND, Hussain MA. In vivo derivation of glucose-competent pancreatic endocrine cells from bone marrow without evidence of cell fusion. J Clin Invest. 2003;111:843–50. doi:10.1172/Jci200316502.
- 9. Krause DS et al. Multi-organ, multi-lineage engraftment by a single bone marrow-derived stem cell. Cell. 2001;105:369–77.
- Semont A et al. Mesenchymal stem cells improve small integrity through regulation of endogenous epithelial cell homeostasis. Cell Death Differ. 2010;17:952–61. doi:10.1038/ cdd.2009.187.
- Little RD et al. A mutation in the LDL receptor-related protein 5 gene results in the autosomal dominant high-bone-mass trait. Am J Hum Genet. 2002;70:11–9. doi:10.1086/338450.
- Lian JB, Stein GS, Stein JL, van Wijnen AJ. Osteocalcin gene promoter: unlocking the secrets for regulation of osteoblast growth and differentiation. J Cell Biochem. 1998; 30–31:62–72.
- Cook, David, and Paul Genever. Regulation of mesenchymal stem cell differentiation. Transcriptional and Translational Regulation of Stem Cells. Springer Netherlands, 2013;213–229.

- Nuttall ME, Gimble JM. Controlling the balance between osteoblastogenesis and adipogenesis and the consequent therapeutic implications. Curr Opin Pharmacol. 2004;4:290–4. doi:10.1016/j.coph.2004.03.002.
- Knouff C, Auwerx J. Peroxisome proliferator-activated receptor-gamma calls for activation in moderation: lessons from genetics and pharmacology. Endocr Rev. 2004;25:899–918. doi:10.1210/er.2003-0036.
- Moerman EJ, Teng K, Lipschitz DA, Lecka-Czernik B. Aging activates adipogenic and suppresses osteogenic programs in mesenchymal marrow stroma/stem cells: the role of PPARgamma 2 transcription factor and TGF-beta/BMP signaling pathways. Aging Cell. 2004;3:379–89. doi:10.1111/j.1474-9728.2004.00127.x.
- Jilka RL, Weinstein RS, Takahashi K, Parfitt AM, Manolagas SC. Linkage of decreased bone mass with impaired osteoblastogenesis in a murine model of accelerated senescence. J Clin Invest. 1996;97:1732–40. doi:10.1172/jci118600.
- Kajkenova O et al. Increased adipogenesis and myelopoiesis in the bone marrow of SAMP6, a murine model of defective osteoblastogenesis and low turnover osteopenia. J Bone Miner Res. 1997;12:1772–9. doi:10.1359/jbmr.1997.12.11.1772.
- Verma S, Rajaratnam JH, Denton J, Hoyland JA, Byers RJ. Adipocytic proportion of bone marrow is inversely related to bone formation in osteoporosis. J Clin Pathol. 2002;55:693–8. doi:10.1136/jcp.55.9.693.
- 20. Ferre P. The biology of peroxisome proliferator—activated receptors—relationship with lipid metabolism and insulin sensitivity. Diabetes. 2004;53:S43–50. doi:10.2337/diabetes.53. 2007.S43.
- Komori T. Regulation of osteoblast differentiation by transcription factors. J Cell Biochem. 2006;99:1233–9. doi:10.1002/jcb.20958.
- Lecka-Czernik B et al. Inhibition of Osf2/Cbfa1 expression and terminal osteoblast differentiation by PPAR gamma 2. J Cell Biochem. 1999;74:357–71. doi:10.1002/ (sici)1097-4644(19990901)74:3<357::aid-jcb5>3.0.co;2-7.
- Picard F, Auwerx J. PPAR gamma and glucose homeostasis. Annu Rev Nutr. 2002;22:167– 97. doi:10.1146/annurev.nutr.22.010402.102808.
- Rzonca SO, Suva LJ, Gaddy D, Montague DC, Lecka-Czernik B. Bone is a target for the antidiabetic compound rosiglitazone. Endocrinology. 2004;145:401–6.
- Ali AA et al. Rosiglitazone causes bone loss in mice by suppressing osteoblast differentiation and bone formation. Endocrinology. 2005;146:1226–35.
- Chuang CC, Yang RS, Tsai KS, Ho FM, Liu SH. Hyperglycemia enhances adipogenic induction of lipid accumulation: involvement of extracellular signal-regulated protein kinase 1/2, phosphoinositide 3-kinase/Akt, and peroxisome proliferator-activated receptor gamma signaling. Endocrinology. 2007;148:4267–75. doi:10.1210/en.2007-0179.
- Cho SW et al. Osteoblast-targeted overexpression of PPAR gamma inhibited bone mass gain in male mice and accelerated ovariectomy-induced bone loss in female mice. J Bone Miner Res. 2011;26:1939–52. doi:10.1002/jbmr.366.
- Akune T et al. PPAR gamma insufficiency enhances osteogenesis through osteoblast formation from bone marrow progenitors. J Clin Invest. 2004;113:846–55. doi:10.1172/ jci200419900.
- Cock TA et al. Enhanced bone formation in lipodystrophic PPARgamma(hyp/hyp) mice relocates haematopoiesis to the spleen. EMBO Rep. 2004;5:1007–12. doi:10.1038/sj.embor. 7400254.
- Diascro DD et al. High fatty acid content in rabbit serum is responsible for the differentiation of osteoblasts into adipocyte-like cells. J Bone Miner Res. 1998;13:96–106. doi:10.1359/ jbmr.1998.13.1.96.
- Lecka-Czernik B et al. Divergent effects of selective peroxisome proliferator-activated receptor-gamma 2 ligands on adipocyte versus osteoblast differentiation. Endocrinology. 2002;143:2376–84.
- 32. Napoli N et al. The alliance of mesenchymal stem cells, bone, and diabetes. Int J Endocrinol. 2014;2014:690783. doi:10.1155/2014/690783.

- McCabe LR, Zhang J, Raehtz S. Understanding the skeletal pathology of type 1 and 2 diabetes mellitus. Crit Rev Eukaryot Gene Expr. 2011;21:187–206.
- 34. Thomas T et al. Leptin acts on human marrow stromal cells to enhance differentiation to osteoblasts and to inhibit differentiation to adipocytes. Endocrinology. 1999;140:1630–8.
- Kato M et al. Cbfa1-independent decrease in osteoblast proliferation, osteopenia, and persistent embryonic eye vascularization in mice deficient in Lrp5, a Wnt coreceptor. J Cell Biol. 2002;157:303–14. doi:10.1083/jcb.200201089.
- Cheng SL, Shao JS, Charlton-Kachigian N, Loewy AP, Towler DA. Msx2 promotes osteogenesis and suppresses adipogenic differentiation of multipotent mesenchymal progenitors. J Biol Chem. 2003;278:45969–77. doi:10.1074/jbc.M306972200.
- Bennett CN et al. Regulation of osteoblastogenesis and bone mass by Wnt10b. Proc Natl Acad Sci U S A. 2005;102:3324–9. doi:10.1073/pnas.0408742102.
- Hong JH et al. TAZ, a transcriptional modulator of mesenchymal stem cell differentiation. Science. 2005;309:1074–8.
- 39. Kindblom JM et al. Increased adipogenesis in bone marrow but decreased bone mineral density in mice devoid of thyroid hormone receptors. Bone. 2005;36:607–16.
- Irwin R, LaPres JJ, Kinser S, McCabe LR. Prolyl-hydroxylase inhibition and HIF activation in osteoblasts promotes an adipocytic phenotype. J Cell Biochem. 2007;100:762–72. doi:10.1002/jcb.21083.
- Rawadi G, Vayssière B, Dunn F, Baron R, Roman-Roman S. BMP-2 controls alkaline phosphatase expression and osteoblast mineralization by a Wnt autocrine loop. J Bone Miner Res. 2003;18:1842–53.
- 42. Hauschka P, Mavrakos A, Iafrati M, Doleman S, Klagsbrun M. Growth factors in bone matrix. Isolation of multiple types by affinity chromatography on heparin-Sepharose. J Biol Chem. 1986;261:12665–74.
- 43. Fulzele K et al. Insulin receptor signaling in osteoblasts regulates postnatal bone acquisition and body composition. Cell. 2010;142:309–19. doi:10.1016/j.cell.2010.06.002.
- Lee NK et al. Endocrine regulation of energy metabolism by the skeleton. Cell. 2007;130:456– 69. doi:10.1016/j.cell.2007.05.047.
- Martin LM, McCabe LR. Type I diabetic bone phenotype is location but not gender dependent. Histochem Cell Biol. 2007;128:125–33. doi:10.1007/s00418-007-0308-4.
- Motyl KJ, McCabe LR. Leptin treatment prevents type I diabetic marrow adiposity but not bone loss in mice. J Cell Physiol. 2009;218:376–84. doi:10.1002/jcp.21608.
- Aarden EM, Wassenaar AMM, Alblas MJ, Nijweide PJ. Immunocytochemical demonstration of extracellular matrix proteins in isolated osteocytes. Histochem Cell Biol. 1996;106:495–501. doi:10.1007/s004180050069.
- McCabe LR. Understanding the pathology and mechanisms of type I diabetic bone loss. J Cell Biochem. 2007;102:1343–57. doi:10.1002/jcb.21573.
- 49. Alblowi J et al. High levels of tumor necrosis factor-alpha contribute to accelerated loss of cartilage in diabetic fracture healing. Am J Pathol. 2009;175:1574–85. doi:10.2353/ ajpath.2009.090148.
- 50. Salvi GE et al. Monocytic TNF alpha secretion patterns in IDDM patients with periodontal diseases. J Clin Periodontol. 1997;24:8–16. doi:10.1111/j.1600-051X.1997.tb01178.x.
- Liu W, Zhu X, Wang Q, Wang L. Hyperglycemia induces endoplasmic reticulum stressdependent CHOP expression in osteoblasts. Exp Ther Med. 2013;5:1289–92. doi:10.3892/ etm.2013.978.
- Coe LM, Irwin R, Lippner D, McCabe LR. The bone marrow microenvironment contributes to type I diabetes induced osteoblast death. J Cell Physiol. 2011;226:477–83. doi:10.1002/jcp.22357.
- Kemink SAG, Hermus A, Swinkels L, Lutterman JA, Smals AGH. Osteopenia in insulindependent diabetes mellitus: prevalence and aspects of pathophysiology. J Endocrinol Invest. 2000;23:295–303.
- MunozTorres M, Jodar E, EscobarJimenez F, LopezIbarra PJ, Luna JD. Bone mineral density measured by dual X-ray absorptiometry in Spanish patients with insulin-dependent diabetes mellitus. Calcif Tissue Int. 1996;58:316–9.

- 55. Inzerillo AM, Epstein S. Osteoporosis and diabetes mellitus. Rev Endocr Metab Disord. 2004;5:261–8. doi:10.1023/b:remd.0000032415.83124.20.
- Rix M, Andreassen H, Eskildsen P. Impact of peripheral neuropathy on bone density in patients with type 1 diabetes. Diabetes Care. 1999;22:827–31. doi:10.2337/diacare.22.5.827.
- 57. Forst T et al. Peripheral osteopenia in adult patients with insulin-dependent diabetes-mellitus. Diabet Med. 1995;12:874–9.
- Ivers RQ, Cumming RG, Mitchell P, Peduto AJ. Diabetes and risk of fracture: the Blue Mountains Eye Study. Diabetes Care. 2001;24:1198–203. doi:10.2337/diacare.24.7.1198.
- Bouillon R et al. Influence of age, sex, and insulin on osteoblast function: osteoblast dysfunction in diabetes-mellitus. J Clin Endocrinol Metab. 1995;80:1194–202. doi:10.1210/ jc.80.4.1194.
- Botolin S, McCabe LR. Bone loss and increased bone adiposity in spontaneous and pharmacologically induced diabetic mice. Endocrinology. 2007;148:198–205. doi:10.1210/en.2006-1006.
- Bain S et al. Tetracycline prevents cancellous bone loss and maintains near-normal rates of bone formation in streptozotocin diabetic rats. Bone. 1997;21:147–53. doi:10.1016/ S8756-3282(97)00104-X.
- Botolin S et al. Increased bone adiposity and peroxisomal proliferator-activated receptorgamma 2 expression in type I diabetic mice. Endocrinology. 2005;146:3622–31. doi:10.1210/ en.2004-1677.
- 63. Epstein S et al. Effect of cyclosporine-A on bone-mineral metabolism in experimental diabetes-mellitus in the rat. J Bone Miner Res. 1994;9:557–66.
- 64. Glajchen N et al. Bone-mineral metabolism in experimental diabetes-mellitus: osteocalcin as a measure of bone remodeling. Endocrinology. 1988;123:290–5.
- 65. Verhaeghe J et al. The effects of systemic insulin, insulin-like growth factor-I and growthhormone on bone-growth and turnover in spontaneously diabetic BB rats. J Endocrinol. 1992;134:485–92. doi:10.1677/joe.0.1340485.
- 66. Shires R et al. The effect of streptozotocin-induced chronic diabetes-mellitus on bone and mineral homeostasis in the rat. J Lab Clin Med. 1981;97:231–40.
- Motyl K, McCabe LR. Streptozotocin, type I diabetes severity and bone. Biol Proced Online. 2009;11:296–315. doi:10.1007/s12575-009-9000-5.
- 68. Lumachi F, Camozzi V, Tombolan V, Luisetto G. Bone mineral density, osteocalcin, and bone-specific alkaline phosphatase in patients with insulin-dependent diabetes mellitus. Ann N Y Acad Sci. 2009;1173:E64–7. doi:10.1111/j.1749-6632.2009.04955.x.
- 69. Cakatay U et al. Changes in bone turnover on deoxypyridinoline levels in diabetic patients. Diabetes Res Clin Pract. 1998;40:75–9. doi:10.1016/s0168-8227(98)00025-4.
- Verhaeghe J et al. Effects of long-term diabetes and/or high-dose 17 beta-estradiol on bone formation, bone mineral density, and strength in ovariectomized rats. Bone. 1997;20:421–8. doi:10.1016/s8756-3282(97)00029-x.
- Ishida H et al. Circulating levels and bone contents of bone gamma-carboxyglutamic acidcontaining protein are decreased in streptozocin-induced diabetes: possible marker for diabetic osteopenia. Diabetes. 1988;37:702–6. doi:10.2337/diabetes.37.6.702.
- Motyl KJ et al. Bone inflammation and altered gene expression with type I diabetes early onset. J Cell Physiol. 2009;218:575–83. doi:10.1002/jcp.21626.
- Verhaeghe J et al. Bone and mineral metabolism in BB rats with long-term diabetes: decreased bone turnover and osteoporosis. Diabetes. 1990;39:477–82. doi:10.2337/diabetes.39.4.477.
- Pater A, Sypniewska G, Pilecki O. Biochemical markers of bone cell activity in children with type 1 diabetes mellitus. J Pediatr Endocrinol Metab. 2010;23:81–6.
- Peters AL, Davidson MB, Schriger DL, Hasselblad V. A clinical approach for the diagnosis of diabetes mellitus: an analysis using glycosylated hemoglobin levels. JAMA. 1996;276:1246–52. doi:10.1001/jama.276.15.1246.
- 76. Krakauer JC et al. Bone loss and bone turnover in diabetes. Diabetes. 1995;44:775-82.
- Goodman WG, Hori MT. Diminished bone-formation in experimental diabetes: relationship to osteoid maturation and mineralization. Diabetes. 1984;33:825–31. doi:10.2337/ diabetes.33.9.825.

8 Bone Marrow Stem Cells and Bone Turnover in Diabetic Disease

- 78. Kanazawa I et al. Serum undercarboxylated osteocalcin was inversely associated with plasma glucose level and fat mass in type 2 diabetes mellitus. Osteoporos Int. 2011;22:187–94.
- 79. Botolin S, McCabe LR. Inhibition of PPARgamma prevents type I diabetic bone marrow adiposity but not bone loss. J Cell Physiol. 2006;209:967–76. doi:10.1002/jcp.20804.
- Giglio MJ, Giannunzio G, Olmedo D, Guglielmotti MB. Histomorphometric study of bone healing around laminar implants in experimental diabetes. Implant Dent. 2000;9:143–9. doi:10.1097/00008505-200009020-00006.
- Gooch HL, Hale JE, Fujioka H, Balian G, Hurwitz SR. Alterations of cartilage and collagen expression during fracture healing in experimental diabetes. Connect Tissue Res. 2000;41:81– 91. doi:10.3109/03008200009067660.
- Weiss RE, Reddi AH. Influence of experimental diabetes and insulin on matrix-induced cartilage and bone differentiation. Am J Physiol. 1980;238:E200–7.
- Al-Mashat HA et al. Diabetes enhances mRNA levels of proapoptotic genes and caspase activity, which contribute to impaired healing. Diabetes. 2006;55:487–95. doi:10.2337/ diabetes.55.02.06.db05-1201.
- Herrero S et al. Low bone density with normal bone turnover in ovariectomized and streptozotocin-induced diabetic rats. Calcif Tissue Int. 1998;62:260–5. doi:10.1007/ s002239900427.
- 85. Verhaeghe J et al. Effects of exercise and disuse on bone remodeling, bone mass, and biomechanical competence in spontaneously diabetic female rats. Bone. 2000;27:249–56.
- Bjorgaas M, Haug E, Johnsen HJ. The urinary excretion of deoxypyridinium cross-links is higher in diabetic than in nondiabetic adolescents. Calcif Tissue Int. 1999;65:121–4. doi:10.1007/s002239900668.
- 87. McCabe LR. Switching fat from the periphery to bone marrow: why in type I diabetes? Expert Rev Endocrinol Metab. 2009;4(3):203–7.
- Weinberg E, Maymon T, Moses O, Weinreb M. Streptozotocin-induced diabetes in rats diminishes the size of the osteoprogenitor pool in bone marrow. Diabetes Res Clin Pract. 2014;103:35–41. doi:10.1016/j.diabres.2013.11.015.
- Wright HM et al. A synthetic antagonist for the peroxisome proliferator-activated receptor gamma inhibits adipocyte differentiation. J Biol Chem. 2000;275:1873–7. doi:10.1074/ jbc.275.3.1873.
- Botolin S, McCabe LR. Chronic hyperglycemia modulates osteoblast gene expression through osmotic and non-osmotic pathways. J Cell Biochem. 2006;99:411–24. doi:10.1002/ jcb.20842.
- Lu H, Kraut D, Gerstenfeld LC, Graves DT. Diabetes interferes with the bone formation by affecting the expression of transcription factors that regulate osteoblast differentiation. Endocrinology. 2003;144:346–52.
- Meunier P, Aaron J, Edouard C, Vignon G. Osteoporosis and replacement of cell populations of marrow by adipose tissue: a quantitative study of 84 iliac bone biopsies. Clin Orthop Relat Res. 1971;80:147–54.
- Gimble JM, Zvonic S, Floyd ZE, Kassem M, Nuttall ME. Playing with bone and fat. J Cell Biochem. 2006;98:251–66. doi:10.1002/jcb.20777.
- Iyer VV, Kadakia TB, McCabe LR, Schwartz RC. CCAAT/enhancer-binding protein-beta has a role in osteoblast proliferation and differentiation. Exp Cell Res. 2004;295:128–37. doi:10.1016/j.yexcr.2004.01.004.
- Motyl KJ, Raetz M, Tekalur SA, Schwartz RC, McCabe LR. CCAAT/enhancer binding protein beta-deficiency enhances type 1 diabetic bone phenotype by increasing marrow adiposity and bone resorption. Am J Physiol Regul Integr Comp Physiol. 2011;300:R1250–60. doi:10.1152/ajpregu.00764.2010.
- 96. Slade JM, Coe LM, Meyer RA, McCabe LR. Human bone marrow adiposity is linked with serum lipid levels not T1-diabetes. J Diabetes Complications. 2012;26:1–9.
- Shyng YC, Devlin H, Sloan P. The effect of streptozotocin-induced experimental diabetes mellitus on calvarial defect healing and bone turnover in the rat. Int J Oral Maxillofac Surg. 2001;30:70–4. doi:10.1054/ijom.2000.0004.

- 98. Takeshita F et al. The effects of diabetes on the interface between hydroxyapatite implants and bone in rat tibia. J Periodontol. 1997;68:180–5.
- 99. Siqueira JT, Cavalher-Machado SC, Arana-Chavez VE, Sannomiya P. Bone formation around titanium implants in the rat tibia: role of insulin. Implant Dent. 2003;12:242–51.
- 100. Gilbert L et al. Expression of the osteoblast differentiation factor RUNX2 (Cbfa1/AML3/ Pebp2 alpha A) is inhibited by tumor necrosis factor-alpha. J Biol Chem. 2002;277:2695– 701. doi:10.1074/jbc.M106339200.
- 101. Nanes MS. Tumor necrosis factor-α: molecular and cellular mechanisms in skeletal pathology. Gene. 2003;321:1–15. doi:10.1016/s0378-1119(03)00841-2.
- 102. Romas E, Gillespie MT, Martin TJ. Involvement of receptor activator of NF kappa B ligand and tumor necrosis factor-alpha in bone destruction in rheumatoid arthritis. Bone. 2002;30:340–6. doi:10.1016/s8756-3282(01)00682-2.
- 103. Zhou FH, Foster BK, Zhou XF, Cowin AJ, Xian CJ. TNF-alpha mediates p38 MAP kinase activation and negatively regulates bone formation at the injured growth plate in rats. J Bone Miner Res. 2006;21:1075–88. doi:10.1359/jbmr.060410.
- Christopher MJ, Link DC. Granulocyte colony-stimulating factor induces osteoblast apoptosis and inhibits osteoblast differentiation. J Bone Miner Res. 2008;23:1765–74. doi:10.1359/ jbmr.080612.
- 105. Alikhani M et al. Lipopolysaccharides indirectly stimulate apoptosis and global induction of apoptotic genes in fibroblasts. J Biol Chem. 2003;278:52901–8. doi:10.1074/jbc. M307638200.
- 106. Hock JM et al. Osteoblast apoptosis and bone turnover. J Bone Miner Res. 2001;16:975–84. doi:10.1359/jbmr.2001.16.6.975.
- 107. Manolagas SC. From estrogen-centric to aging and oxidative stress: a revised perspective of the pathogenesis of osteoporosis. Endocr Rev. 2010;31:266–300. doi:10.1210/er.2009-0024.
- Graves DT, Liu R, Oates TW. Diabetes-enhanced inflammation and apoptosis: impact on periodontal pathosis. Periodontol 2000. 2007;45:128–37. doi:10.1111/j.1600-0757.2007.00219.x.
- 109. Kanazawa I et al. Serum osteocalcin level is associated with glucose metabolism and atherosclerosis parameters in type 2 diabetes mellitus. J Clin Endocrinol Metab. 2009;94:45–9.
- Fajardo RJ, Karim L, Calley VI, Bouxsein ML. A review of rodent models of type 2 diabetic skeletal fragility. J Bone Miner Res. 2014;29:1025–40. doi:10.1002/jbmr.2210.
- 111. Ahmad T, Ohlsson C, Saaf M, Ostenson CG, Kreicbergs A. Skeletal changes in type-2 diabetic Goto-Kakizaki rats. J Endocrinol. 2003;178:111–6. doi:10.1677/joe.0.1780111.
- 112. He HB et al. Diabetes causes decreased osteoclastogenesis, reduced bone formation, and enhanced apoptosis of osteoblastic cells in bacteria stimulated bone loss. Endocrinology. 2004;145:447–52. doi:10.1210/en.2003-1239.
- 113. Liu R, Bal HS, Desta T, Behl Y, Graves DT. Tumor necrosis factor-α mediates diabetesenhanced apoptosis of matrix-producing cells and impairs diabetic healing. Am J Pathol. 2006;168:757–64. doi:10.2353/ajpath.2006.050907.
- 114. Hamann C et al. Delayed bone regeneration and low bone mass in a rat model of insulinresistant type 2 diabetes mellitus is due to impaired osteoblast function. Am J Physiol Endocrinol Metab. 2011;301:E1220–8. doi:10.1152/ajpendo.00378.2011.
- 115. Ducy P et al. Leptin inhibits bone formation through a hypothalamic relay: a central control of bone mass. Cell. 2000;100:197–207. doi:10.1016/s0092-8674(00)81558-5.
- 116. Ealey KN, Fonseca D, Archer MC, Ward WE. Bone abnormalities in adolescent leptindeficient mice. Regul Pept. 2006;136:9–13. doi:10.1016/j.regpep.2006.04.013.
- 117. Matsunuma A, Kawane T, Maeda T, Hamada S, Horiuchi N. Leptin corrects increased gene expression of renal 25-hydroxyvitamin D-3-1 alpha-hydroxylase and -24-hydroxylase in leptin-deficient, ob/ob mice. Endocrinology. 2004;145:1367–75. doi:10.1210/en.2003-1010.
- 118. Hamrick MW, Pennington C, Newton D, Xie D, Isales C. Leptin deficiency produces contrasting phenotypes in bones of the limb and spine. Bone. 2004;34:376–83.
- 119. Turner RT et al. Peripheral leptin regulates bone formation. J Bone Miner Res. 2013;28:22–34. doi:10.1002/jbmr.1734.

- 120. Hamada Y et al. Histomorphometric analysis of diabetic osteopenia in streptozotocin-induced diabetic mice: a possible role of oxidative stress. Bone. 2007;40:1408–14. doi:10.1016/j. bone.2006.12.057.
- 121. Fernandez AM et al. Functional inactivation, of the IGF-I and insulin receptors in skeletal muscle causes type 2 diabetes. Genes Dev. 2001;15:1926–34. doi:10.1101/gad.908001.
- 122. Kawashima Y et al. Type 2 diabetic mice demonstrate slender long bones with increased fragility secondary to increased osteoclastogenesis. Bone. 2009;44:648–55. doi:10.1016/j. bone.2008.12.012.
- 123. Li X, Wu X, Camacho R, Schwartz GJ, LeRoith D. Intracerebroventricular leptin infusion improves glucose homeostasis in lean type 2 diabetic MKR mice via hepatic vagal and non-vagal mechanisms. PLoS One. 2011;6:e17058. doi:10.1371/journal.pone.0017058.
- 124. Kim JH et al. Phenotypic characterization of polygenic type 2 diabetes in TALLYHO/JngJ mice. J Endocrinol. 2006;191:437–46. doi:10.1677/joe.1.06647.
- 125. Zhang DE, Hetherington CJ, Chen HM, Tenen DG. The macrophage transcription factor PU.1 directs tissue-specific expression of the macrophage colony-stimulating factor receptor. Mol Cell Biol. 1994;14:373–81.
- Pahl HL et al. The proto-oncogene PU.1 regulates expression of the myeloid-specific CD11b promoter. J Biol Chem. 1993;268:5014–20.
- 127. Del Fattore A, Teti A, Rucci N. Osteoclast receptors and signaling. Arch Biochem Biophys. 2008;473:147–60. doi:10.1016/j.abb.2008.01.011.
- 128. Tondravi MM et al. Osteopetrosis in mice lacking haematopoietic transcription factor PU.1. Nature. 1997;386:81–4. doi:10.1038/386081a0.
- 129. Luchin A et al. The microphthalmia transcription factor regulates expression of the tartrateresistant acid phosphatase gene during terminal differentiation of osteoclasts. J Bone Miner Res. 2000;15:451–60. doi:10.1359/jbmr.2000.15.3.451.
- Luchin A et al. Genetic and physical interactions between Microphthalmia transcription factor and PU.1 are necessary for osteoclast gene expression and differentiation. J Biol Chem. 2001;276:36703–10. doi:10.1074/jbc.M106418200.
- 131. Boyle WJ, Simonet WS, Lacey DL. Osteoclast differentiation and activation. Nature. 2003;423:337–42.
- 132. Yoshida H et al. The murine mutation osteopetrosis is in the coding region of the macrophage colony stimulating factor gene. Nature. 1990;345:442–4. doi:10.1038/345442a0.
- 133. Marks Jr SC, Lane PW. Osteopetrosis, a new recessive skeletal mutation on chromosome 12 of the mouse. J Hered. 1976;67:11–8.
- 134. Wiktor-Jedrzejczak WW, Ahmed A, Szczylik C, Skelly RR. Hematological characterization of congenital osteopetrosis in op/op mouse. Possible mechanism for abnormal macrophage differentiation. J Exp Med. 1982;156:1516–27.
- 135. Arai F et al. Commitment and differentiation of osteoclast precursor cells by the sequential expression of c-Fms and receptor activator of nuclear factor kappaB (RANK) receptors. J Exp Med. 1999;190:1741–54.
- 136. Edwards JR, Mundy GR. Advances in osteoclast biology: old findings and new insights from mouse models. Nat Rev Rheumatol. 2011;7:235–43. doi:10.1038/nrrheum.2011.23.
- 137. Simonet WS et al. Osteoprotegerin: a novel secreted protein involved in the regulation of bone density. Cell. 1997;89:309–19.
- 138. Yasuda H et al. Osteoclast differentiation factor is a ligand for osteoprotegerin/ osteoclastogenesis-inhibitory factor and is identical to TRANCE/RANKL. Proc Natl Acad Sci U S A. 1998;95:3597–602.
- 139. Anderson DM et al. A homologue of the TNF receptor and its ligand enhance T-cell growth and dendritic-cell function. Nature. 1997;390:175–9. doi:10.1038/36593.
- 140. Rothe M, Sarma V, Dixit VM, Goeddel DV. TRAF2-mediated activation of NF-kappa B by TNF receptor 2 and CD40. Science. 1995;269:1424–7.
- 141. Song HY, Regnier CH, Kirschning CJ, Goeddel DV, Rothe M. Tumor necrosis factor (TNF)mediated kinase cascades: bifurcation of nuclear factor-kappaB and c-jun N-terminal kinase

(JNK/SAPK) pathways at TNF receptor-associated factor 2. Proc Natl Acad Sci U S A. 1997;94:9792-6.

- 142. Nakano H et al. TRAF5, an activator of NF-kappaB and putative signal transducer for the lymphotoxin-beta receptor. J Biol Chem. 1996;271:14661–4.
- 143. Ishida T et al. Identification of TRAF6, a novel tumor necrosis factor receptor-associated factor protein that mediates signaling from an amino-terminal domain of the CD40 cytoplasmic region. J Biol Chem. 1996;271:28745–8.
- 144. Locatto ME et al. Growth and development of bone mass in untreated alloxan-diabetic rats: effects of collagen glycosylation and parathyroid activity on bone turnover. Bone Miner. 1993;23:129–44. doi:10.1016/s0169-6009(08)80049-9.
- Horcajada-Molteni MN et al. Amylin and bone metabolism in streptozotocin-induced diabetic rats. J Bone Miner Res. 2001;16:958–65. doi:10.1359/jbmr.2001.16.5.958.
- 146. Kayal RA et al. Diminished bone formation during diabetic fracture healing is related to the premature resorption of cartilage associated with increased osteoclast activity. J Bone Miner Res. 2007;22:560–8. doi:10.1359/jbmr.070115.
- 147. Kaneko H, Sasaki T, Ramamurthy NS, Golub LM. Tetracycline administration normalizes the structure and acid-phosphatase-activity of osteoclasts in streptozotocin-induced diabetic rats. Anat Rec. 1990;227:427–36. doi:10.1002/ar.1092270406.
- 148. Cloos C et al. Urinary glycosylated, free and total pyridinoline and free and total deoxypyridinoline in diabetes mellitus. Clin Endocrinol (Oxf). 1998;48:317–23.
- 149. Keats EC, Dominguez JM, Grant MB, Khan ZA. Switch from canonical to noncanonical Wnt signaling mediates high glucose-induced adipogenesis. Stem Cells. 2014;32:1649–60. doi:10.1002/stem.1659.
- 150. Wittrant Y et al. High d(+)glucose concentration inhibits RANKL-induced osteoclastogenesis. Bone. 2008;42(6):1122–30.
- 151. Liu R et al. Diabetes enhances periodontal bone loss through enhanced resorption and diminished bone formation. J Dent Res. 2006;85:510–4.
- 152. Lacey DL et al. Osteoprotegerin ligand is a cytokine that regulates osteoclast differentiation and activation. Cell. 1998;93:165–76. doi:10.1016/s0092-8674(00)81569-x.
- 153. Bonde M, Fledelius C, Qvist P, Christiansen C. Coated-tube radioimmunoassay for C-telopeptides of type I collagen to assess bone resorption. Clin Chem. 1996;42:1639–44.
- 154. Uebelhart D et al. Effect of menopause and hormone replacement therapy on the urinaryexcretion of pyridinium cross-links. J Clin Endocrinol Metab. 1991;72:367–73.
- 155. Hanson DA et al. A specific immunoassay for monitoring human bone-resorption: quantitation of type-I collagen cross-linked N-telopeptides in urine. J Bone Miner Res. 1992;7:1251–8.
- 156. Suzuki K et al. Osteoclastic function is accelerated in male patients with type 2 diabetes mellitus: the preventive role of osteoclastogenesis inhibitory factor/osteoprotegerin (OCIF/OPG) on the decrease of bone mineral density. Diabetes Res Clin Pract. 2005;68:117–25. doi:10.1016/j.diabres.2004.08.006.
- 157. Tornvig L, Mosekilde L, Justesen J, Falk E, Kassem M. Troglitazone treatment increases bone marrow adipose tissue volume but does not affect trabecular bone volume in mice. Calcif Tissue Int. 2001;69:46–50. doi:10.1007/s002230020018.
- 158. Cenci S et al. Estrogen deficiency induces bone loss by increasing T cell proliferation and lifespan through IFN-gamma-induced class II transactivator. Proc Natl Acad Sci U S A. 2003;100:10405–10. doi:10.1073/pnas.1533207100.
- Cenci S et al. Estrogen deficiency induces bone loss by enhancing T-cell production of TNFalpha. J Clin Invest. 2000;106:1229–37. doi:10.1172/JCI11066.
- 160. Roggia C et al. Up-regulation of TNF-producing T cells in the bone marrow: a key mechanism by which estrogen deficiency induces bone loss in vivo. Proc Natl Acad Sci U S A. 2001;98:13960–5. doi:10.1073/pnas.251534698.
- 161. Charatcharoenwitthaya N, Khosla S, Atkinson EJ, McCready LK, Riggs BL. Effect of blockade of TNF-alpha and interleukin-1 action on bone resorption in early postmenopausal women. J Bone Miner Res. 2007;22:724–9. doi:10.1359/jbmr.070207.

- 162. Weitzmann MN, Cenci S, Rifas L, Brown C, Pacifici R. Interleukin-7 stimulates osteoclast formation by up-regulating the T-cell production of soluble osteoclastogenic cytokines. Blood. 2000;96:1873–8.
- 163. Yago T et al. IL-23 induces human osteoclastogenesis via IL-17 in vitro, and anti-IL-23 antibody attenuates collagen-induced arthritis in rats. Arthritis Res Ther. 2007;9:R96. doi:10.1186/ar2297.
- 164. Li JY et al. Ovariectomy expands murine short-term hemopoietic stem cell function through T cell expressed CD40L and Wnt10B. Blood. 2013;122:2346–57. doi:10.1182/blood-2013-03-487801.
- 165. Orlandi A et al. Long-term diabetes impairs repopulation of hematopoietic progenitor cells and dysregulates the cytokine expression in the bone marrow microenvironment in mice. Basic Res Cardiol. 2010;105:703–12. doi:10.1007/s00395-010-0109-0.
- 166. Hazra S et al. Long-term type 1 diabetes influences haematopoietic stem cells by reducing vascular repair potential and increasing inflammatory monocyte generation in a murine model. Diabetologia. 2013;56:644–53. doi:10.1007/s00125-012-2781-0.
- 167. Fadini GP et al. Monocyte-macrophage polarization balance in pre-diabetic individuals. Acta Diabetol. 2013;50:977–82. doi:10.1007/s00592-013-0517-3.
- 168. Kojima H, Fujimiya M, Terashima T, Kimura H, Chan L. Extrapancreatic proinsulin/insulinexpressing cells in diabetes mellitus: is history repeating itself? Endocr J. 2006;53:715–22.
- Gysemans C, Callewaert H, Overbergh L, Mathieu C. Cytokine signalling in the beta-cell: a dual role for IFNgamma. Biochem Soc Trans. 2008;36:328–33. doi:10.1042/BST0360328.
- Bilgic S et al. Intracytoplasmic cytokine levels and neutrophil functions in early clinical stage of type 1 diabetes. Diabetes Res Clin Pract. 2008;79:31–6. doi:10.1016/j.diabres.2007.06.011.
- 171. Ko KI et al. Diabetes reduces mesenchymal stem cells in fracture healing through a TNFalphamediated mechanism. Diabetologia. 2015;58:633–42. doi:10.1007/s00125-014-3470-y.
- 172. Loomans CJ et al. Differentiation of bone marrow-derived endothelial progenitor cells is shifted into a proinflammatory phenotype by hyperglycemia. Mol Med. 2009;15:152–9. doi:10.2119/molmed.2009.00032.
- 173. Brunner S et al. Correlation of different circulating endothelial progenitor cells to stages of diabetic retinopathy: first in vivo data. Invest Ophthalmol Vis Sci. 2009;50:392–8. doi:10.1167/iovs.08-1748.
- 174. Mohler III ER et al. Diabetes reduces bone marrow and circulating porcine endothelial progenitor cells, an effect ameliorated by atorvastatin and independent of cholesterol. Cytometry A. 2009;75:75–82. doi:10.1002/cyto.a.20691.
- 175. Ma L et al. Association between bone mineral density and type 2 diabetes mellitus: a metaanalysis of observational studies. Eur J Epidemiol. 2012;27:319–32. doi:10.1007/ s10654-012-9674-x.
- 176. Fadini GP et al. Circulating endothelial progenitor cells are reduced in peripheral vascular complications of type 2 diabetes mellitus. J Am Coll Cardiol. 2005;45:1449–57. doi:10.1016/j. jacc.2004.11.067.
- 177. Kanazawa A et al. Association of the gene encoding wingless-type mammary tumor virus integration-site family member 5B (WNT5B) with type 2 diabetes. Am J Hum Genet. 2004;75:832–43. doi:10.1086/425340.
- 178. Anagnostou SH, Shepherd PR. Glucose induces an autocrine activation of the Wnt/betacatenin pathway in macrophage cell lines. Biochem J. 2008;416:211–8. doi:10.1042/ BJ20081426.
- 179. Rubin MR, et al. Effects of the anti-inflammatory drug salsalate on bone turnover in type 2 diabetes mellitus. Endocrine. 2015. doi:10.1007/s12020-015-0535-8.
- 180. Hunt JV, Smith CCT, Wolff SP. Autoxidative glycosylation and possible involvement of peroxides and free-radicals in LDL modification by glucose. Diabetes. 1990;39:1420–4.
- Zayzafoon M, Botolin S, McCabe LR. p38 and activating transcription factor-2 involvement in osteoblast osmotic response to elevated extracellular glucose. J Biol Chem. 2002;277:37212–8. doi:10.1074/jbc.M200129200.

- 182. Zayzafoon M, Stell C, Irwin R, McCabe LR. Extracellular glucose influences osteoblast differentiation and c-jun expression. J Cell Biochem. 2000;79:301–10. doi:10.1002/1097-4644(20001101)79:2<301::aid-jcb130>3.0.co;2-0.
- Brownlee M, Vlassara H, Cerami A. Nonenzymatic glycosylation and the pathogenesis of diabetic complications. Ann Intern Med. 1984;101:527–37.
- 184. Katayama Y, Akatsu T, Yamamoto M, Kugai N, Nagata N. Role of nonenzymatic glycosylation of type I collagen in diabetic osteopenia. J Bone Miner Res. 1996;11:931–7.
- McCarthy AD, Etcheverry SB, Bruzzone L, Cortizo AM. Effects of advanced glycation endproducts on the proliferation and differentiation of osteoblast-like cells. Mol Cell Biochem. 1997;170:43–51. doi:10.1023/a:1006816223292.
- 186. McCarthy AD et al. Non-enzymatic glycosylation of a type I collagen matrix: effects on osteoblastic development and oxidative stress. BMC Cell Biol. 2001;2:16. doi:10.1186/ 1471-2121-2-16.
- McCarthy AD, Etcheverry SB, Cortizo AM. Effect of advanced glycation endproducts on the secretion of insulin-like growth factor-I and its binding proteins: role in osteoblast development. Acta Diabetol. 2001;38:113–22.
- 188. Wautier JL, Guillausseau PJ. Advanced glycation end products, their receptors and diabetic angiopathy. Diabetes Metab. 2001;27:535–42.
- Vasilkova O, Mokhort T, Sharshakova T, Hayashida N, Takamura N. Leptin is an independent determinant of bone mineral density in men with type 2 diabetes mellitus. Acta Diabetol. 2011;48:291–5. doi:10.1007/s00592-011-0266-0.
- 190. Cunha JS, Ferreira VM, Maquigussa E, Naves MA, Boim MA. Effects of high glucose and high insulin concentrations on osteoblast function in vitro. Cell Tissue Res. 2014;358:249–56. doi:10.1007/s00441-014-1913-x.
- 191. Balint E, Szabo P, Marshall CF, Sprague SM. Glucose-induced inhibition of in vitro bone mineralization. Bone. 2001;28:21–8.
- 192. Zhang Y, Yang JH. Activation of the PI3K/Akt pathway by oxidative stress mediates high glucose-induced increase of adipogenic differentiation in primary rat osteoblasts. J Cell Biochem. 2013;114:2595–602. doi:10.1002/jcb.24607.
- 193. Wang W, Zhang X, Zheng J, Yang J. High glucose stimulates adipogenic and inhibits osteogenic differentiation in MG-63 cells through cAMP/protein kinase A/extracellular signal-regulated kinase pathway. Mol Cell Biochem. 2010;338:115–22. doi:10.1007/s11010-009-0344-6.
- 194. Mercer N, Ahmed H, Etcheverry SB, Vasta GR, Cortizo AM. Regulation of advanced glycation end product (AGE) receptors and apoptosis by AGEs in osteoblast-like cells. Mol Cell Biochem. 2007;306:87–94. doi:10.1007/s11010-007-9557-8.
- 195. Xu F et al. Inhibitory effects of high glucose/insulin environment on osteoclast formation and resorption in vitro. J Huazhong Univ Sci Technolog Med Sci. 2013;33:244–9. doi:10.1007/ s11596-013-1105-z.
- 196. Ramasamy R et al. Advanced glycation end products and RAGE: a common thread in aging, diabetes, neurodegeneration, and inflammation. Glycobiology. 2005;15:16R–28. doi:10.1093/ glycob/cwi053.
- 197. Santana RB et al. A role for advanced glycation end products in diminished bone healing in type 1 diabetes. Diabetes. 2003;52:1502–10.
- 198. Dyer DG et al. Accumulation of Maillard reaction-products in skin collagen in diabetes and aging. J Clin Invest. 1993;91:2463–9. doi:10.1172/jci116481.
- 199. Schwartz AV et al. Pentosidine and increased fracture risk in older adults with type 2 diabetes. J Clin Endocrinol Metab. 2009;94:2380–6. doi:10.1210/jc.2008-2498.
- McCance DR et al. Maillard reaction-products and their relation to complications in insulindependent diabetes-mellitus. J Clin Invest. 1993;91:2470–8. doi:10.1172/jci116482.
- Stitt AW. Advanced glycation: an important pathological event in diabetic and age related ocular disease. Br J Ophthalmol. 2001;85:746–53.
- Saito M, Marumo K, Fujii K, Ishioka N. Single-column high-performance liquid chromatographic fluorescence detection of immature, mature, and senescent cross-links of collagen. Anal Biochem. 1997;253:26–32.
- 203. Yamamoto M, Yamaguchi T, Yamauchi M, Yano S, Sugimoto T. Serum pentosidine levels are positively associated with the presence of vertebral fractures in postmenopausal women with type 2 diabetes. J Clin Endocrinol Metab. 2008;93:1013–9. doi:10.1210/jc.2007-1270.
- 204. Sell DR et al. Pentosidine: a molecular marker for the cumulative damage to proteins in diabetes, aging and uremia. Diabetes Metab Rev. 1991;7:239–51.
- 205. Dunn JA, Patrick JS, Thorpe SR, Baynes JW. Oxidation of glycated proteins: age-dependent accumulation of N epsilon-(carboxymethyl)lysine in lens proteins. Biochemistry. 1989;28:9464–8. doi:10.1021/bi00450a033.
- Lyons TJ, Bailie KE, Dyer DG, Dunn JA, Baynes JW. Decrease in skin collagen glycation with improved glycemic control in patients with insulin-dependent diabetes-mellitus. J Clin Invest. 1991;87:1910–5.
- 207. Forbes JM, Cooper ME, Oldfield MD, Thomas MC. Role of advanced glycation end products in diabetic nephropathy. J Am Soc Nephrol. 2003;14:S254–8. doi:10.1097/01.asn. 0000077413.41276.17.
- Lalla E, Lamster IB, Drury S, Fu CF, Schmidt AM. Hyperglycemia, glycoxidation and receptor for advanced glycation endproducts: potential mechanisms underlying diabetic complications, including diabetes-associated periodontitis. Periodontol 2000. 2000;23:50–62. doi:10.1034/j.1600-0757.2000.2230104.x.
- 209. Saito M, Marumo K. Collagen cross-links as a determinant of bone quality: a possible explanation for bone fragility in aging, osteoporosis, and diabetes mellitus. Osteoporos Int. 2010;21:195–214. doi:10.1007/s00198-009-1066-z.
- Sell DR, Monnier VM. Isolation, purification and partial characterization of novel fluorophores from aging human insoluble collagen-rich tissue. Connect Tissue Res. 1989;19:77– 92. doi:10.3109/03008208909016816.
- 211. Hein G et al. Advanced glycation end product modification of bone proteins and bone remodelling: hypothesis and preliminary immunohistochemical findings. Ann Rheum Dis. 2006;65:101–4. doi:10.1136/ard.2004.034348.
- 212. Vashishth D et al. Influence of nonenzymatic glycation on biomechanical properties of cortical bone. Bone. 2001;28:195–201. doi:10.1016/s8756-3282(00)00434-8.
- 213. Wang X, Shen X, Li X, Agrawal CM. Age-related changes in the collagen network and toughness of bone. Bone. 2002;31:1–7. doi:10.1016/s8756-3282(01)00697-4.
- Hernandez CJ et al. Trabecular microfracture and the influence of pyridinium and nonenzymatic glycation-mediated collagen cross-links. Bone. 2005;37:825–32. doi:10.1016/j. bone.2005.07.019.
- 215. Saito M, Fujii K, Mori Y, Marumo K. Role of collagen enzymatic and glycation induced cross-links as a determinant of bone quality in spontaneously diabetic WBN/Kob rats. Osteoporos Int. 2006;17:1514–23. doi:10.1007/s00198-006-0155-5.
- 216. Viguet-Carrin S et al. Contribution of the advanced glycation end product pentosidine and of maturation of type I collagen to compressive biomechanical properties of human lumbar vertebrae. Bone. 2006;39:1073–9. doi:10.1016/j.bone.2006.05.013.
- 217. Saito M, Mori S, Mashiba T, Komatsubara S, Marumo K. Collagen maturity, glycation induced-pentosidine, and mineralization are increased following 3-year treatment with incadronate in dogs. Osteoporos Int. 2008;19:1343–54. doi:10.1007/s00198-008-0585-3.
- 218. Vashishth D. Collagen glycation and its role in fracture properties of bone. J Musculoskelet Neuronal Interact. 2005;5:316.
- 219. Tomasek JJ, Meyers SW, Basinger JB, Green DT, Shew RL. Diabetic and age-related enhancement of collagen-linked fluorescence in cortical bones of rats. Life Sci. 1994; 55:855–61.
- 220. Saito M, Fujii K, Marumo K. Degree of mineralization-related collagen crosslinking in the femoral neck cancellous bone in cases of hip fracture and controls. Calcif Tissue Int. 2006;79:160–8. doi:10.1007/s00223-006-0035-1.
- 221. Silva MJ et al. Type 1 diabetes in young rats leads to progressive trabecular bone loss, cessation of cortical bone growth, and diminished whole bone strength and fatigue life. J Bone Miner Res. 2009;24:1618–27. doi:10.1359/jbmr.090316.

- 222. Kume S et al. Advanced glycation end-products attenuate human mesenchymal stem cells and prevent cognate differentiation into adipose tissue, cartilage, and bone. J Bone Miner Res. 2005;20:1647–58. doi:10.1359/jbmr.050514.
- 223. Miyata T et al. Advanced glycation end products enhance osteoclast-induced bone resorption in cultured mouse unfractionated bone cells and in rats implanted subcutaneously with devitalized bone particles. J Am Soc Nephrol. 1997;8:260–70.
- 224. Alikhani M et al. Advanced glycation end products stimulate osteoblast apoptosis via the MAP kinase and cytosolic apoptotic pathways. Bone. 2007;40:345–53. doi:10.1016/j. bone.2006.09.011.
- 225. Brett J et al. Survey of the distribution of a newly characterized receptor for advanced glycation end-products in tissues. Am J Pathol. 1993;143:1699–712.
- Valcourt U et al. Non-enzymatic glycation of bone collagen modifies osteoclastic activity and differentiation. J Biol Chem. 2007;282:5691–703.
- 227. Chen T-L. The influence of diabetes enhanced inflammation on cell apoptosis and periodontitis. Adv Biosci Biotechnol. 2012;03:712–9. doi:10.4236/abb.2012.326092.
- 228. Fujii H, Hamada Y, Fukagawa M. Bone formation in spontaneously diabetic Torii-newly established model of non-obese type 2 diabetes rats. Bone. 2008;42:372–9. doi:10.1016/j. bone.2007.10.007.
- Bai XC et al. Oxidative stress inhibits osteoblastic differentiation of bone cells by ERK and NF-kappa B. Biochem Biophys Res Commun. 2004;314:197–207. doi:10.1016/j. bbrc.2003.12.073.
- Mody N, Parhami F, Sarafian TA, Demer LL. Oxidative stress modulates osteoblastic differentiation of vascular and bone cells. Free Radic Biol Med. 2001;31:509–19. doi:10.1016/ s0891-5849(01)00610-4.
- 231. van der Kallen CJH, van Greevenbroek MMJ, Stehouwer CDA, Schalkwijk CG. Endoplasmic reticulum stress-induced apoptosis in the development of diabetes: is there a role for adipose tissue and liver? Apoptosis. 2009;14:1424–34. doi:10.1007/s10495-009-0400-4.
- 232. Wend P, Wend K, Krum SA, Miranda-Carboni GA. The role of WNT10B in physiology and disease. Acta Physiol. 2012;204:34–51. doi:10.1111/j.1748-1716.2011.02296.x.
- Bennett CN et al. Regulation of Wnt signaling during adipogenesis. J Biol Chem. 2002;277:30998–1004. doi:10.1074/jbc.M204527200.
- Bennett CN et al. Wnt10b increases postnatal bone formation by enhancing osteoblast differentiation. J Bone Miner Res. 2007;22:1924–32. doi:10.1359/jbmr.070810.
- 235. Babij P et al. High bone mass in mice expressing a mutant LRP5 gene. J Bone Miner Res. 2003;18:960–74.
- Gong YQ et al. LDL receptor-related protein 5 (LRP5) affects bone accrual and eye development. Cell. 2001;107:513–23. doi:10.1016/s0092-8674(01)00571-2.
- 237. Williams BO, Insogna KL. Where Wnts went: the exploding field of Lrp5 and Lrp6 signaling in bone. J Bone Miner Res. 2009;24:171–8. doi:10.1359/jbmr.081235.
- Gilbert LC, Chen H, Lu X, Nanes MS. Chronic low dose tumor necrosis factor-alpha (TNF) suppresses early bone accrual in young mice by inhibiting osteoblasts without affecting osteoclasts. Bone. 2013;56:174–83. doi:10.1016/j.bone.2013.06.002.
- Kayal RA et al. TNF-alpha mediates diabetes-enhanced chondrocyte apoptosis during fracture healing and stimulates chondrocyte apoptosis through FOXO1. J Bone Miner Res. 2010;25:1604–15.
- 240. Liu R, Bal HS, Desta T, Behl Y, Graves DT. Tumor necrosis factor-alpha mediates diabetesenhanced apoptosis of matrix-producing cells and impairs diabetic healing. Am J Pathol. 2006;168:757–64.
- 241. Portal-Nunez S et al. Alterations of the Wnt/beta-catenin pathway and its target genes for the N- and C-terminal domains of parathyroid hormone-related protein in bone from diabetic mice. FEBS Lett. 2010;584:3095–100. doi:10.1016/j.febslet.2010.05.047.
- 242. Gennari L et al. Circulating sclerostin levels and bone turnover in type 1 and type 2 diabetes. J Clin Endocrinol Metab. 2012;97:1737–44. doi:10.1210/jc.2011-2958.

- 243. Longo KA et al. Wnt10b inhibits development of white and brown adipose tissues. J Biol Chem. 2004;279:35503–9.
- 244. Zhang M et al. Osteoblast-specific knockout of the insulin-like growth factor (IGF) receptor gene reveals an essential role of IGF signaling in bone matrix mineralization. J Biol Chem. 2002;277:44005–12. doi:10.1074/jbc.M208265200.
- 245. Jehle PM, Lehle DR, Mohan S, Bohm BO. Serum levels of insulin-like growth factor system components and relationship to bone metabolism in type 1 and type 2 diabetes mellitus patients. J Endocrinol. 1998;159:297–306. doi:10.1677/joe.0.1590297.
- Clark RG. Recombinant human insulin-like growth factor I (IGF-1): risks and benefits of normalizing blood IGF-I concentrations. Horm Res. 2004;62:93–100. doi:10.1159/000080766.
- 247. Wei J et al. Bone-specific insulin resistance disrupts whole-body glucose homeostasis via decreased osteocalcin activation. J Clin Invest. 2014;124:1781–93. doi:10.1172/JCI72323.
- 248. Ogata N et al. Insulin receptor substrate-1 in osteoblast is indispensible for maintaining bone turnover. J Clin Invest. 2000;105:935–43. doi:10.1172/jci9017.
- 249. Shimoaka T et al. Impairment of bone healing by insulin receptor substrate-1 deficiency. J Biol Chem. 2004;279:15314–22. doi:10.1074/jbc.M312525200.
- 250. Wu S, De Luca F. Role of cholesterol in the regulation of growth plate chondrogenesis and longitudinal bone growth. J Biol Chem. 2004;279:4642–7. doi:10.1074/jbc.M305518200.
- 251. Tanko LB, Bagger YZ, Christiansen C. Low bone mineral density in the hip as a marker of advanced atherosclerosis in elderly women. Calcif Tissue Int. 2003;73:15–20.
- 252. Nuzzo V et al. Analysis of skeletal status by quantitative ultrasonometry in a cohort of postmenopausal women with high blood cholesterol without documented osteoporosis. Ultrasound Med Biol. 2009;35:717–22. doi:10.1016/j.ultrasmedbio.2008.11.003.
- 253. Hsu YH et al. Relation of body composition, fat mass, and serum lipids to osteoporotic fractures and bone mineral density in Chinese men and women. Am J Clin Nutr. 2006; 83:146–54.
- 254. Cui LH et al. Association between bone mineral densities and serum lipid profiles of pre- and post-menopausal rural women in South Korea. Osteoporos Int. 2005;16:1975–81. doi:10.1007/s00198-005-1977-2.
- 255. Navab M et al. The Yin and Yang of oxidation in the development of the fatty streak. A review based on the 1994 George Lyman Duff Memorial Lecture. Arterioscler Thromb Vasc Biol. 1996;16:831–42.
- 256. Tintut Y, Morony S, Demer LL. Hyperlipidemia promotes osteoclastic potential of bone marrow cells ex vivo. Arterioscler Thromb Vasc Biol. 2004;24:e6–10. doi:10.1161/01. ATV.0000112023.62695.7f.
- 257. Pirih F et al. Adverse effects of hyperlipidemia on bone regeneration and strength. J Bone Miner Res. 2012;27:309–18.
- 258. Weidman SW, Ragland JB, Fisher JN, Kitabchi AE, Sabesin SM. Effects of insulin on plasma-lipoproteins in diabetic-ketoacidosis: evidence for a change in high-density lipoprotein composition during treatment. J Lipid Res. 1982;23:171–82.
- Winocour PH, Durrington PN, Ishola M, Anderson DC. Lipoprotein abnormalities in insulindependent diabetes-mellitus. Lancet. 1986;1:1176–8.
- 260. Chaturvedi N, Fuller JH, Taskinen MR, Grp EP. Differing associations of lipid and lipoprotein disturbances with the macrovascular and microvascular complications of type 1 diabetes. Diabetes Care. 2001;24:2071–7. doi:10.2337/diacare.24.12.2071.
- Janner M, Ballinari P, Mullis PE, Flueck CE. High prevalence of vitamin D deficiency in children and adolescents with type 1 diabetes. Swiss Med Wkly. 2010;140. doi:10.4414/ smw.2010.13091.
- 262. Baumgarti HJ et al. Changes of vitamin-D3 serum concentrations at the onset of immunemediated type-1 (insulin-dependent) diabetes-mellitus. Diabetes Res. 1991;16:145–8.
- Elder GJ, Mackun K. 25-hydroxyvitamin D deficiency and diabetes predict reduced BMD in patients with chronic kidney disease. J Bone Miner Res. 2006;21:1778–84. doi:10.1359/ jbmr.060803.

- 264. Auwerx J, Dequeker J, Bouillon R, Geusens P, Nijs J. Mineral metabolism and bone mass at peripheral and axial skeleton in diabetes-mellitus. Diabetes. 1988;37:8–12. doi:10.2337/ diabetes.37.1.8.
- 265. Wiske PS, Wentworth SM, Norton JA, Epstein S, Johnston CC. Evaluation of bone mass and growth in young diabetics. Metabolism. 1982;31:848–54. doi:10.1016/0026-0495(82) 90086-5.
- 266. Dobnig H et al. Type 2 diabetes mellitus in nursing home patients: effects on bone turnover, bone mass, and fracture risk. J Clin Endocrinol Metab. 2006;91:3355–63. doi:10.1210/jc.2006-0460.
- Hamrick MW. Leptin, bone mass, and the thrifty phenotype. J Bone Miner Res. 2004;19:1607– 11. doi:10.1359/jbmr.040712.
- Cornish J et al. Leptin directly regulates bone cell function in vitro and reduces bone fragility in vivo. J Endocrinol. 2002;175:405–15. doi:10.1677/joe.0.1750405.
- 269. Luna R et al. High serum leptin levels in children with type 1 diabetes mellitus: contribution of age, BMI, pubertal development and metabolic status. Clin Endocrinol (Oxf). 1999;51:603–10. doi:10.1046/j.1365-2265.1999.00848.x.
- 270. Karagüzel G, Ozdem S, Boz A, Bircan I, Akçurin S. Leptin levels and body composition in children and adolescents with type 1 diabetes. Clin Biochem. 2006;39:788–93. doi:10.1016/j. clinbiochem.2006.02.014.
- 271. Lo HC, Lin SC, Wang YM. The relationship among serum cytokines, chemokine, nitric oxide, and leptin in children with type 1 diabetes mellitus. Clin Biochem. 2004;37:666–72. doi:10.1016/j.clinbiochem.2004.02.002.
- 272. Borst SE. The role of TNF-alpha in insulin resistance. Endocrine. 2004;23:177-82. doi:10.1385/endo:23:2-3:177.
- 273. Nesto RW, Rutter MK. Impact of the atherosclerotic process in patients with diabetes. Acta Diabetol. 2002;39:S22–8.
- 274. Senn JJ, Klover PJ, Nowak IA, Mooney RA. Interleukin-6 induces cellular insulin resistance in hepatocytes. Diabetes. 2002;51:3391–9. doi:10.2337/diabetes.51.12.3391.
- 275. Desfaits AC, Serri O, Renier G. Normalization of plasma lipid peroxides, monocyte adhesion, and tumor necrosis factor-alpha production in NIDDM patients after gliclazide treatment. Diabetes Care. 1998;21:487–93. doi:10.2337/diacare.21.4.487.
- 276. Katsuki A et al. Serum levels of tumor necrosis factor-alpha are increased in obese patients with noninsulin-dependent diabetes mellitus. J Clin Endocrinol Metab. 1998;83:859–62. doi:10.1210/jc.83.3.859.
- 277. Fernandez-Real JM et al. Circulating interleukin 6 levels, blood pressure, and insulin sensitivity in apparently healthy men and women. J Clin Endocrinol Metab. 2001;86:1154–9. doi:10.1210/jc.86.3.1154.
- 278. Mishima Y et al. Relationship between serum tumor necrosis factor-alpha and insulin resistance in obese men with type 2 diabetes mellitus. Diabetes Res Clin Pract. 2001;52:119–23. doi:10.1016/s0168-8227(00)00247-3.
- 279. Lalla E et al. Blockade of RAGE suppresses periodontitis-associated bone loss in diabetic mice. J Clin Invest. 2000;105:1117–24. doi:10.1172/jci8942.
- 280. Salvi GE et al. Inflammatory mediator response as a potential risk marker for periodontal diseases in insulin-dependent diabetes mellitus patients. J Periodontol. 1997;68:127–35.
- 281. Thomas DM, Maher F, Rogers SD, Best JD. Expression and regulation by insulin of glut 3 in UMR 106-01, a clonal rat osteosarcoma cell line. Biochem Biophys Res Commun. 1996;218:789–93. doi:10.1006/bbrc.1996.0140.
- Young N, Mikhalkevich N, Yan Y, Chen D, Zheng WP. Differential regulation of osteoblast activity by Th cell subsets mediated by parathyroid hormone and IFN-gamma. J Immunol. 2005;175:8287–95.
- Halade GV, El Jamali A, Williams PJ, Fajardo RJ, Fernandes G. Obesity-mediated inflammatory microenvironment stimulates osteoclastogenesis and bone loss in mice. Exp Gerontol. 2011;46:43–52. doi:10.1016/j.exger.2010.09.014.

- Jiang J, Zuo J, Hurst IR, Holliday LS. The synergistic effect of peptidoglycan and lipopolysaccharide on osteoclast formation. Oral Surg Oral Med Oral Pathol Oral Radiol Endod. 2003;96:738–43. doi:10.1016/s107921040300502x.
- Dumitrescu AL, Abd El-Aleem S, Morales-Aza B, Donaldson LF. A model of periodontitis in the rat: effect of lipopolysaccharide on bone resorption, osteoclast activity, and local peptidergic innervation. J Clin Periodontol. 2004;31:596–603. doi:10.1111/j.1600-051X.2004.00528.x.
- Matsuo K, Irie N. Osteoclast-osteoblast communication. Arch Biochem Biophys. 2008;473:201–9. doi:10.1016/j.abb.2008.03.027.
- 287. Grey A et al. A role for interleukin-6 in parathyroid hormone-induced bone resorption in vivo. Endocrinology. 1999;140:4683–90. doi:10.1210/endo.140.10.7036.
- 288. O'Gradaigh D, Ireland D, Bord S, Compston JE. Joint erosion in rheumatoid arthritis: interactions between tumour necrosis factor alpha, interleukin 1, and receptor activator of nuclear factor kappaB ligand (RANKL) regulate osteoclasts. Ann Rheum Dis. 2004;63:354–9.
- Catalfamo DL et al. Augmented LPS responsiveness in type 1 diabetes-derived osteoclasts. J Cell Physiol. 2013;228:349–61. doi:10.1002/jcp.24138.

Part II Biomechanics and Bone Quality

Chapter 9 Material Properties of Diabetic Bone

Jeffry S. Nyman and Amy Creecy

Role of Bone Structure in the Diabetes-Related Decrease in Fracture Resistance

As described in a number of review articles [1-3], the hierarchical organization of bone (Fig. 9.1) confers multiple mechanisms of fracture resistance. As such, diabetes-related changes to any level of organization (collagen fibrils at the nanoscale to cortical thickness at the macro-scale) can affect the ability of bone to resist failure [4]. Material properties depend on all aspects of bone organization except the macrostructure. In order to delineate the role of material properties in fracture resistance, bone structure is first defined with respect to diabetes.

Structural properties are derived from the cross-sectional geometry of cortical bone (the analog for trabecular bone is architectural properties) and include the moment of inertia (or second moment of area), endocortical circumference, periosteal circumference, bone cross-sectional area (cortex only), and total cross-sectional area (cortex and marrow space). They are related to one another; but from the perspective of engineering mechanics, moment of inertia and bone cross-sectional area characterize the ability of whole-bone to resist bending/torsion and compression/tension, respectively. Also, an increase in the periosteal circumference—for example, during growth—imparts a greater increase in the principal moment of

Tennessee Valley Healthcare System, Department of Veterans Affairs, Nashville, TN, USA e-mail: jeffry.s.nyman@vanderbilt.edu

A. Creecy, M.S. Department of Biomedical Engineering, Vanderbilt University, Medical Center East, South Tower Suite 4200, Nashville, TN 37232, USA

© Springer International Publishing Switzerland 2016 B. Lecka-Czernik, J.L. Fowlkes (eds.), *Diabetic Bone Disease*, DOI 10.1007/978-3-319-16402-1_9

J.S. Nyman, Ph.D. (🖂)

Department of Orthopaedic Surgery and Rehabilitation, Vanderbilt University Medical Center, South Tower Suite 4200, Nashville, TN 37232, USA



Fig. 9.1 The ability of bone to resist fracture depends on each level of organizational hierarchy from the triple helix of collagen being cross-linked (*left*) and filled with mineral crystals through arrangement of osteons or trabeculae to the shape (*right*)

Fig. 9.2 The cross-sectional area (Ct.Ar) of a femur mid-shaft can be similar between two different rats, as an example, while the moment of inertia (I_{min}) is considerably different. If the material properties are equivalent between the animals, then the strength in compression will be similar but weaker in bending for the diabetic than for the nondiabetic rat



2.70

18.14

Control rat Ct.Ar = 7.39 mm $I_{min} = 8.04 mm$ Diabetic rat Ct.Ar = 7.19 mm I_{min} = 6.58 mm

inertia (I_{min} or I_{max}) than an increase in bone cross-sectional area (Ct.Ar). Given equivalent material properties, normal bone and diabetic bone would break at the same force when loaded in compression if there is no difference in Ct.Ar, but the same diabetic bone would break at a lower force when loaded in bending if the I_{min} is less for the diabetic bone than for the normal bone (Fig. 9.2). That is, two different bones can have the same Ct.Ar but different values of I_{min} or polar moment of inertia $(J = (I_{min} + I_{max})/2)$ in the case of torsion. Also, the moment of inertia varies with the orientation of the bone relative to the bending axis (Fig. 9.3). Therefore, the invariant calculations are often reported in the literature (I_{min} corresponds to bending about the minor axis of the bone cross section or the minimum distance between centroid and the periosteal surface).

From human imaging studies, there is evidence that type 1 diabetes (T1D) affects bone structure in deleterious way [5], especially if glucose is poorly controlled or if T1D starts at a young age [6]. However, since most T1D patients are treated with

Fig. 9.3 As a bone rotates with respect to the axis of bending, the moment of inertia changes. For a full rotation, there is maximum value and minimum value. In bending tests of rodent femurs to determine material properties, the minimum principal moment of inertia corresponds to bending about the medial-lateral axis



insulin, bone structure can normalize with age [7] and structural indices of the hip do not necessarily differ between men $(43.6\pm5.1 \text{ years of age})$ with long-lasting T1D and those without diabetes [8]. For the most part, cortical bone structure of individuals with type 2 diabetes (T2D) does not significantly vary from that of age- and gendermatched nondiabetics as determined by various X-ray computed tomography (CT) scans [9–11], though there are reports of hip strength indices being adversely affected by T2D [12–14]. Additional information on bone structure can be found in Chap. 11.

The consequence of diabetes on size-dependent, structural strength is fairly well established for rodents (see reviews by Nyman [4] and Fajardo et al. [15]), but the mechanism of action is not particularly clear. Whether there is a loss of insulin production or an increase in insulin resistance in rodents, bones become structurally weaker and less rigid [16–19]. Of note though, diabetes develops before skeletal maturation in most rodent models of T2D, which can explain the discrepancy between rodent and human studies of bone structure involving T2D. Differences in bone structure between control mice and mice injected with streptozotocin (STZ), a toxin that decreases insulin production by the pancreas, explain the lower whole bone strength that occurs with the onset of T1D (Fig. 9.4) [19]. The primary cause for the reduction in bone size with diabetes is difficult to delineate in rodents as insulin, leptin, insulin-like growth factor-1 (IGF-1), and glucose levels cannot be independently controlled. Certainly, insulin and IGF-1 signaling are important to bone mass accrual [20, 21]. When the insulin receptor (IR) was conditionally deleted in cells of the osteoblast lineage (Osterix-cre), the femur mid-shafts of the mutant mice were strikingly narrow or slender and thus weaker than those from control littermates [22]. This effect was independent of body weight. Loss of



insulin signaling however is likely not the only cause for deficits in diabetic bone structure. Rats with early onset T2D are hyperinsulinemic before 16 weeks of age, but yet, their bones have a much smaller moment of inertia than do the nondiabetic control rats before and after skeletal maturity [16] suggesting insulin resistance in osteoblasts or the action of other factors contribute to the deficit in cortical bone structure. Also, when human IRs were activated in the pancreas, liver, and brain of mice lacking IR in all cells including osteoblasts, there were no difference in bone properties between genotypes (euglycemic condition) [23]. These mutant mice are known to be hyperinsulinemic [24] and express more IGF-1 receptors in bone (insulin can bind IGF-1 receptors and vice versa) [23]. When a structural deficit in bone does occur with diabetes, it is likely the result of both a loss of anabolic action of insulin/IGF-1 on osteoblasts (especially in early-onset diabetes) and inhibition of bone formation by another mechanism (e.g., oxidative stress due to hyperglycemia).

While poor bone structure (e.g., thinning of cortex) can lower areal bone mineral density (aBMD) as determined by dual-energy X-ray absorptiometry, the increase in fracture risk with T1D is disproportionate to the reduced aBMD [25]. Even though the elevated fracture risk is not as high for individuals with T2D, this increase also occurs independently of aBMD and bone structure [26–28]. Thus, diabetes must cause other changes to bone that lower fracture resistance. What follows is a tutorial on bone mechanics and review of studies reporting whether diabetes affects the material properties of bone.

Engineering Mechanics: Assessing Material Properties of Bone

Engineering materials (e.g., metal alloys, polymers, ceramics) break in many ways, and bone is no different. Intuitively, a bone fracture occurs when a region of the skeleton (e.g., femoral neck or distal radius) experiences a load that exceeds the

maximum force that the bone region can sustain. Perhaps less obvious is that a healthy bone can vield (i.e., experience permanent damage) but not break. As long as that bone does not experience another high load too soon, remodeling removes the microdamage by replacing it with new tissue and returning the bone tissue to its normal material strength. In the case of unhealthy bone that cannot sustain damage after yielding, there is a "brittle" fracture. Through locomotion, bones-especially the femur, tibia, and spine-experience repetitive loads that are well below the yield force. Because there are numerous "flaws" in bone (e.g., lacunae, canaliculi, Haversian canals, Howship's lacuna, and vacancies in the mineral) acting as stress risers, this cyclic loading at a low force causes microdamage formation over time. In the case of trabecular bone, microdamage accumulation from a bout of overloading [29] or fatigue loading [30] lowers the apparent-level strength. For either cortical or trabecular bone, the accumulation of microdamage eventually causes fatigue failure if allowed to proceed unchecked. Clinically, this leads to a stress fracture [31]. Such a bone stress injury has been postulated as a pathological mechanism of stage 0 diabetic Charcot foot [32].

Because materials can fail in a number of ways, such as an overload, insufficient energy dissipation, and fatigue, there are multiple engineering techniques to assess material properties. Comprehensive descriptions of these techniques can be found in several books on mechanical testing of bone or biomaterials [33–35]. Briefly, in all the techniques, force vs. displacement (f vs. d) data is recorded as a bone specimen is loaded to failure (i.e., stretched, compressed, twisted, or flexed). This is not the case for tests characterizing elastic or viscoelastic properties but is necessary to characterize post-yield behavior. After testing the specimen with an instrument that has a load cell (force) and an actuator (displacement), the f vs. d is processed in such way to remove specimen size (or structure) from the mechanical behavior. As a simple example, f vs. d from a monotonic, load-to-failure, tensile test of cortical bone is converted to engineering stress (σ) vs. engineering strain (ε) in which σ is f divided by the cross-sectional area of the specimen and ε is d divided by the original gage length of the extensometer, a device that clips to the specimen and records displacement between two knife edges (Fig. 9.5). The material properties are then determined from the resulting σ vs. ε curve (Fig. 9.5). Similar analysis occurs for bending tests (three point or four point) except that the equations to determine σ vs. ε differ as they are derived from beam theory and use the moment of inertia, instead of cross-sectional area of the specimen [36]. Also, in bending tests, displacement can be recorded from a deflectometer or the instrument's linear variable displacement transducer that is part of the actuator.

With respect to fatigue testing, the primary outcome is fatigue life in which failure can be defined several ways: (1) actual numbers of cycles of loading experienced by the specimen before breaking (N_f), (2) the number of cycles to reach a specified level of creep (permanent deformation or subsidence in the case of compressive testing) or a specified level of modulus loss, and (3) the steady-state creep rate (Fig. 9.6). Unlike most monotonic tests, fatigue tests are run in force control imparting a constant stress-amplitude (σ_a =maximum σ minus the minimum σ during cyclic loading divided by 2). Although difficult in bone studies because the amount of material to generate replicates is limited, characterizing fatigue properties



Fig. 9.5 Using the cross-sectional area of the specimen and gage length of the extensioneter, the resulting force vs. displacement curve (a) from a tensile test is converted to an engineering stress vs. engineering strain curve (b) from which the material properties are determined. Determination of material properties are marked on curve for the bone specimen from a relatively young donor



Fig. 9.6 In a fatigue test, the bone is dynamically loaded (sinusoidal wave profile) below the yield force (**a**). The force vs. displacement data is converted to stress vs. strain data to calculate changes in modulus, as determined by linear elastic beam theory (LEBT), and creep or permanent deformation (**b**). Creep occurs at a constant rate during most of the life of the specimen (**c**). As the number of cycles increases, the modulus decreases during fatigue (**d**)

typically involves loading multiple sets of specimens at different stress amplitudes to generate a σ_a vs. N_f plot or *S*–*N* curve. Alternatively, σ_a can vary across specimens to ensure that the initial maximum strain experienced by each specimen is constant (e.g., 4000 microstrain). Comparing results across fatigue studies is difficult because a number of extrinsic factors affect fatigue life (e.g., frequency of loading, waveform, initial strain, specimen geometry, loading mode, and the ratio of σ_{max} to σ_{min}).

Because fatigue testing does not delineate the ability of bone to resist microcrack initiation from its ability to resist microcrack propagation and accumulation, there is another type of material characterization technique known as fracture toughness testing. Involving a sharp notch that acts as the "worst flaw" in the material, a fracture toughness specimen can be loaded cyclically (fatigue crack growth resistance), monotonically (crack initiation toughness or total nonlinear strain energy release rate), or progressively (R-curve method to measure crack initiation and crack propagation toughness simultaneously).

Fracture toughness tests essentially characterize the ability of a material to resist fracture when a flaw is present. Engineers use this material property to determine what is the tolerable flaw size that a particular structure can have for a given service load without failing. There are acoustic devices to determine the size of flaws or cracks in structures such as airplane wings, but no such method is available for detecting microcracks in bone in vivo. Acoustic techniques such as ultrasound or acoustic emission are however being developed to monitor microdamage progression in in vitro tests [37, 38]. Traditionally, the assessment of microdamage morphology ex vivo involves bulk staining with a chelating agent and fluorescence microscopy [39] and more recently contrast-enhanced micro-CT (μ CT) using an agent (e.g., barium sulfate) with higher X-ray attenuation than bone [40]. Such methods have been used to demonstrate associations between loading mode (compression, tension, torsion) and damage morphology (linear microcracks, diffuse damage) [41, 42]. To the best of our knowledge, differences in microdamage morphology between diabetic and otherwise normal bone has not been investigated.

Combining Material Properties with Bone Structure Using Finite Element Analysis

There is a way to incorporate the contributions of both structure and material to fracture resistance: apply the finite element method to the boundary value problem for a continuum body like bone. Two centuries after the first application of mechanics to bone by Galileo (1638), anatomists and engineers—Ward, Engel, von Meyer, and Culmann—connected the structure and architecture of bone to the stresses acting within bone [43]. While elegant graphical techniques matched bone structure/ architecture to principal stress directions, the basis of Wolff's law (1892), bone geometry is too complex for deriving analytical solutions using the theory of elasticity to determine the displacement response to prescribed boundary conditions. This was overcome in the 1960s for all complex structures when a numerical method



Fig. 9.7 A finite element model can be directly converted from the voxels of a micro-computed tomography scan of a bone. The finite element analysis calculates the strain distribution throughout the bone. In this example, the boundary conditions of the mouse lumbar vertebra simulate compression loading

was developed to solve the equilibrium, strain–displacement, and stress–strain equations comprising the boundary value problem in solid mechanics [44]. Since then, finite element analysis (FEA) has been the cornerstone of engineering design. Beginning in the 1990s, researchers in the field of orthopaedic biomechanics began developing three-dimensional finite element models of bone [45] and bone implants [46]. This became possible with the advent of quantitative CT (QCT) because voxels in the CT scan can be directly converted to elements that discretize the complex bone geometry into a mesh (Fig. 9.7) allowing for the creation of the linear algebra problem that is then solved in an iterative fashion by a finite element solver. Moreover, CT attenuation can be converted to material property definitions when incorporating a hydroxyapatite (HA) phantom into the scan.

With ample evidence that predictions of stiffness and strength by QCT-FEA strongly correlate with experimental measurements as determined by whole bone testing of cadaveric tissue, namely the proximal femur [45, 47, 48], distal radius [49, 50],

and vertebra [51], the methodology has been applied to QCT scans of patients in clinical trials testing the efficacy of drugs to prevent fractures in mainly postmenopausal women (but also men in the MrOS cohort). For example, differences in predicted strength between placebo and therapy have been reported for alendronate (a bisphosphonate) [52, 53], combined teriparatide (recombinant parathyroid hormone 1-34) and alendronate [54], alendronate and then teriparatide [55], teriparatide to treat glucocorticoid-induced osteoporosis [56], odanacatib (cathepsin K inhibitor) [57, 58], denosumab (anti-RANKL antibody) [59], combined odanacatib and teriparatide [60], and combined teriparatide and denosumab [61]. An open question is how well these strength predictions explain the drug-related reduction in fracture incidence after adjusting for aBMD, which of course increases with treatment. QCT-FEAs are capable of fracture discrimination [52, 62-66], though some overlap in predicted strength exists between non-fracture and fracture cases [63, 67]. In a 5-year case-control study (>65 years of age) using baseline QCT scans of the spine and hip to develop the finite element model, the age-adjusted odds ratio for vertebral and hip strength predictions was significant for women and men [68]. That is, women and men with low vertebral strength or low hip strength, as predicted by QCT-FEA, were ~2 or ~4 times more likely, respectively, to suffer a fracture, and these odd ratios were still significant even if adjusted for most aBMD measurements.

To date, there is a paucity of studies investigating whether CT-derived FEA is useful for identifying diabetics who are at risk of a fracture (see Chap. 3). QCT-FEA or even high-resolution (HR)-peripheral CT (pCT)-FEA might not be particularly effective in the case of diabetes, unlike osteoporosis. The power of FEA is its ability to account for the structure in the response of a continuum body to loading or specifically boundary conditions, and as previously mentioned, diabetes does not necessarily affect bone structure. Moreover, the material properties must be defined in the FEA, and they are typically based on assumed empirical relationships between elastic modulus and bone density or strength and bone density acquired from mechanical testing of cadaveric tissue. There are no such relationships derived from diabetic bone. FEA calculates the stress-strain distribution throughout the bone as well as displacement for a given prescribed boundary condition (e.g., side-ways fall in which the trochanter is fixed against a rigid surface and the femoral head experiences a high joint reaction force). It does not actually calculate the strength of the bone. FEA predicted strength requires failure criteria (strain at which elasticity no longer holds), and yield strain or the damage tolerance of diabetic bone relative to normal bone is not yet known.

Age- and Diabetes-Related Changes in Material Properties of Bone

There is little known about the effect of T1D or T2D on the material properties of human bone or bone from nonhuman primates. This is likely to change in the coming years, as there is a growing recognition for the need to understand how diabetes

is increasing fracture risk. The challenge to the assessment of material properties is the procurement of bone samples that can be machined into the proper test geometry. Traditionally, this is done using cadaveric bone, but information about whether the donor had diabetes and for how long is rarely provided by willed body programs and tissue allograft banks. Alternatively, bone biopsies from surgeries or from the iliac crest can be analyzed by other techniques such as μ CT, Raman spectroscopy, Fourier transform infrared spectroscopy, high performance liquid chromatography, and quantitative back-scatter electron microscopy [69] to identify organizational and compositional differences between diabetic and nondiabetic bone. Such information would inform as to whether diabetes affects bone quality. In on-going work, trabecular bone biopsies are being collected from patients undergoing total hip arthroplasty (osteoarthritis, not hip fracture), and those with glycated hemoglobin levels (HbA1c) greater than six tend to have bone with lower apparent strength and higher AGEs [70].

To the best of our knowledge, there are only three studies that reported material properties of human bone from donors with diabetes. In the earliest study (1998), metatarsals were acquired from amputations (30 diabetics and 19 nondiabetics) performed for a variety of reasons (elective, trauma, infection, tumor) [71]. The type of diabetes was not specified nor the duration. An additional set of metatarsals was acquired from 29 cadavers. Tested in the three-point bending, the estimated modulus and strength of the diaphysis (cortical bone) did not differ between the two groups. However, there was an age discrepancy in which the diabetics were 51.3 ± 8 years of age and the normal controls were 72.4 ± 10 years of age, suggesting the effect of diabetes on bone material properties is similar to the effect of aging. In a follow-up study by the same group (2006), beam specimens (nominal dimensions of $30 \times 4 \times 2$ mm) were extracted from the anterior aspect of tibia, which were acquired from amputations (seven diabetics and seven nondiabetics) [72]. Again, there were no differences in the material properties (modulus, strength, and fracture toughness) between the younger diabetic bone and the elderly nondiabetic bone. While an extended period of non-ambulation (unspecified) was part of the exclusion criteria, confounding factors related to the amputation limit the interpretation of these findings. Nonetheless, the idea that diabetes is a form of accelerated aging with respect to material properties of bone is intriguing. In the third study, the fracture toughness of cadaveric bone from one donor with diabetes (duration not specified) was 40 % lower than cadaveric bone from several donors without diabetes [73].

Given the dearth of information about how diabetes affects the material properties of human bone, we turn to the literature reporting how material properties of cadaveric bone change with age. Evidence that the material properties of human cortical bone changed with age began to appear in the literature around the time that the finite element method was being developed [74–76]. However, there were several studies that did not find age-related changes in tensile strength [77, 78] owing to discrepancies in sample size (sometimes less than ten) and storage conditions (frozen vs. fixed). Then, testing 175 tensile specimens from 33 cadaveric femurs acquired from donors spanning 20–89 years of age, Burstein et al. showed that yield and ultimate stress (i.e., material strength) significantly decreased 2.2 % per decade



Fig. 9.8 The toughness of human cortical bone decreases over life-span (adapted from Burstein et al. [79])

and that energy dissipation (i.e., toughness) significantly decreased by 6.8 % per decade (Fig. 9.8) [79].

Age-related decrease in material strength of cortical bone has since been reported by a number of independent groups [80–83], though there is variance in the relationship between strength and age (some elderly donors have strong bones). Material strength can be measured as the stress at yielding when permanent damage begins to form or as the peak stress endured by bone during failure. A primary determinant of this decrease in material strength is an age-related decrease in apparent bone density (wet mass divided by the specimen volume or radiation absorption), but it is not the sole determinant [80] as the rate of decrease in strength is greater than that of density [81]. Of course, apparent bone density is product of intracortical porosity and degree of mineralization (or ash fraction). Both characteristics inversely or directly correlate, respectively, with strength [84, 85]. Therefore, if diabetes causes an increase in porosity or decrease in mineralization, then it would decrease the material strength of bone, thereby increasing fracture risk.

Collagen cross-links via lysyl oxidase are another important determinant of bone strength. When female rats were administered the toxin β -amino-propionitrile (BAPN) to inhibit lysyl oxidase, there was a decrease in material strength of bone that accompanied a decrease in mature enzymatic collagen cross-links (pyridino-lines) without affecting mineral density [86]. Recently, treatment of growing male mice with BAPN resulted in reduced cortical bone strength, reduced fracture toughness, and reduced mature-to-immature cross-link ratio (these properties were positively correlated) without affecting tissue mineral density as determined by μ CT [87]. Diabetes could potentially alter enzymatic collagen cross-linking. The B6 vitamer known as pyridoxamine is a cofactor of lysly oxidase, and diabetes via an oxidative stress pathway could potentially decrease enzymatic collagen cross-links with the progression of T2D in the WBN/Kob rat (material properties of bone were not directly reported) [88]. More information on the contribution of collagen cross-link to bone strength can be found in several review articles [89–91].

With aging, both advanced glycation end products (AGEs) [92, 93] and microdamage [94–98] increase in bone despite continual turnover throughout life. Fatigue loading of cortical bone in vitro to generate microdamage reduces material strength [99, 100]. Conceivably, diabetes alters the matrix of bone in way that favors damage accumulation, especially if remodeling is impaired (turnover marker are generally down in diabetics).

As is the case with material strength, several independent groups have reported that toughness decreases with age [82, 83, 93]. In general, the collagen phase is primarily responsible for bone toughness or the ability of bone to dissipate energy during failure. Much of this energy occurs after yielding such that "brittle" bone has very low post-yield toughness. There are a number of collagen characteristics that partially explain the age-related decrease in human bone toughness: increase in porosity (i.e., loss of collagen) [82], decrease in collagen strength [92], a decrease in isometric shrinkage temperature or the peak rate of contraction of the collagen (i.e., collagen integrity) [101], an increase in secondary osteonal area [96], a decrease in collagen content relative to dry mass of bone [93], an increase in pentosidine [93], and a decrease in bound water [102, 103]. As such, the hyperglycemia that occurs with diabetes can possibly cause bone to become brittle through modifications to the collagen phase. For example, the accumulation of AGEs like pentosidine over the duration of diabetes could decrease the ability of collagen to deform and slide relative to the mineral phase, thereby decreasing toughness (for a review of bone quality in diabetes, see [104]).

The evidence for this possible mechanism of AGE accumulation reducing bone toughness comes from the documented association between pentosidine or AGEs and bone toughness [92] or fracture toughness [105] and in vitro riboslyation experiments [106]. When bovine cortical bone (18–24 months old) was incubated in high concentrations of sugar, namely 0.6 M ribose, for 14 or 38 days to induce nonenzymatic cross-linking, there was a reduction in fracture properties related to post-yield energy dissipation mechanisms, not strength [106, 107]. However, when fetal bovine bone was incubated in 0.2 M ribose for 15 days, there were no differences in material properties, as determined by three-point bending tests, between control and ribosylated bone, despite ribose significantly increasing pentosidine [108]. This suggests that sugar concentration and mineralization status of the tissue influence whether inducing AGEs lowers the material properties of bone. With respect to trabecular bone, which is likely more susceptible to nonenzymatic, glycation-mediated collagen cross-linking than cortical bone [109, 110], in vitro incubation in 0.6 M ribose for 7 days has been reported to lower the difference between strain at maximum stress and strain at yielding (post-yield energy) as determined by compression tests of human trabecular bone [111, 112]. Adding an AGE blocker such as aminoguanidine [108] or *n*-phenacylthiazolium bromide [112] partially prevents ribose-induced AGEs. However, in vivo demonstration that inhibiting diabetes-related increase in AGE prevents a loss of bone toughness has yet to be demonstrated.

As previously mentioned, there are multiple ways to characterize the fracture toughness of human cortical bone. Whether determined as the critical stress state beyond which the crack begins to grow (stress intensity factor K), the nonlinear elastic strain energy dissipated prior to and during fracture (*J*-integral), or the toughness evolution with crack extension (crack resistance curve or *R*-curve), fracture

toughness of human bone decreases with advancing age [73, 83, 92, 113, 114]. Moreover, there are many factors that correlate with fracture toughness of human cortical bone: apparent bone density [115], porosity [96, 116], collagen integrity [101], microdamage [96, 117], bound water [118], and pentosidine [92, 105]. Diabetes could potentially lower fracture toughness and overall fatigue resistance by increasing intracortical porosity and increasing AGEs such that the matrix has less capacity to resist damage.

Differences in Material Properties of Bone Between Nondiabetic and Diabetic Rodents

Different strains of rodents have been used to study the effects of both T1D and T2D on the material properties of the bone, although there are more studies reporting diabetes-related differences in structural properties such as stiffness and peak force [4]. Rodent models of T1D involve impairing pancreatic or β cell function [119]. This can be achieved by injecting a toxin such as alloxan or STZ, infecting transgenic mice with a virus specifically designed to disrupt β cells, and breeding mice that cannot adequately produce insulin (the non-obese diabetic or NOD mouse) [120]. Type 2 diabetic models involve single gene or polygenic mutations [15, 119, 121]. Rodents with single gene mutations become spontaneously diabetic. Examples include mutations in the leptin gene (db/db mouse or ZDF rat), the leptin receptor (ob/ob mouse), and in the IGF-1 receptor gene of muscle cells (MKR mouse). Note that the loss of function in these genes may affect the skeletal phenotype in ways that are independent of diabetic effects. Diabetes in polygenic models of T2D can also spontaneous, but it typically involves diets with moderate to high fat, thereby enuring that frank diabetes occurs. These models vary in the severity of diabetes, and age of diabetic onset often occurs before skeletal maturity. As such, more preclinical and clinical studies are needed to determine which rodents models best represent how diabetes affects bone in humans.

In general, material strength and toughness of bone are lower for rodents with T1D than without diabetes [18, 19, 122–126]. Table 9.1 summarizes some recent studies reporting material properties of bone in primarily STZ-induced models of T1D. Non-fasting glucose levels reproducibly exceed 350 mg/dL in this model with normal levels being less than 100 mg/dL (severity depends on strain of mouse however). There is a lack of information on the material properties of bone in other non-toxin models such as the NOD mouse. Difference in bone toughness between T1D rodents and nondiabetic controls depends on the duration of diabetes [19, 126], which could explain why not all T1D studies report a difference in this material property [18, 123].

As previously mentioned, a recent review by Fajardo et al. [15] comprehensively summarizes the skeletal phenotype of multiple rodent models of T2D. With respect to material properties of bone, material strength and toughness are typically lower, on average, for rodents with diabetes than for the nondiabetic controls [16, 127–129], but the differences are not always statistically significant (Table 9.2).

Strain of rodent SD ^a rat	Age when diabetes started (weeks) 10	Duration of diabetes (weeks) 7	Bone tested in bending Femur Tibia	Material strength (%) NR ^b	Toughness (%) -34 -40	Comments on toughness assessment Work-to- failure in the plastic range (n=12 per group)	Reference [122]
F344 rat	12	12	Ulna	-7.8	-19 (NS)	Work-to- failurec $(n=9-12)$	[18]
			Ulna	-1.6 (NS)	-33		
SD rat	12	12	Femur	NR	+2.6 (NS)	per group)	
			Femur	NR	-26 (NS)		
Wistar rat ^d	10	6	Femur	-5.9 (NS)	-8.9 (NS)	Area under the stress vs. strain curve (<i>n</i> =16 per group)	[126]
		12		-5.9 (NS)	–17.1 (NS)		
Wistar rat	13	8	Femur ^e	-36.7	-58.1	Area under the stress vs. strain curve $(n=7$ per group)	[124]
BALB/c mouse	14	5.7	Femur	-3	NR	Femur cross section assumed to be an ellipsoid (n=5-9 per) group)	[123]
CD-1 mouse	10	5	Femur	-16.8	6 (NS)	Post-yield deflection $(n=7-9 \text{ per group})$	[125]
DBA/2 J mouse	11	10	Femur	-0.2	-11.9 (NS)	Post-yield	[19]
				(NS)		toughness	
		15	-	-9.2	-33.4 (NS)	(n=0-11)	
		18		-10.8	-57.9	per group)	

 Table 9.1
 Summary of the recent literature reporting differences in material properties of bone

 between normal rodents and rodents with type 1 diabetes

Unless otherwise noted, rodents were injected with streptozotocin to induce diabetes. Significant differences indicated in **bold**. Otherwise, the difference (100* (Mean-T1D-Mean-Control)/ Mean-Control) was not statistically significant (NS)

^aSD Sprague–Dawley

^bNR not reported

^cArea under force vs. displacement curve (not independent of structure)

^dInjection of nicotinamide prior to STZ administration

^eBones were tested in tension

^fSpan-adjusted area-under-the-curve after yielding divided by bone cross-sectional area

Strain of rodent	Age when diabetes started (weeks)	Duration of diabetes (weeks)	Bone tested in bending	Material strength (%)	Toughness (%)	Comments on toughness assessment	References
ZDF ^a rat	9–11	2-4	Femur	-10.2 (NS)	-2 (NS)	Post-yield displacement	[16]
		9–11	Femur	-6 (NS)	+6 (NS)	(n=9-10 per)	
		2–4	Tibia	+6.9 (NS)	-31.1 (NS)	group)	
		9–11	Tibia	+8.1 (NS)	-50.6 (NS)		
ZDF rat ^b	9–11	12–14	Femur ^c	-22.8	-15.8 (NS)	Area under the stress vs. strain curve $(n=3-7)$ per group)	[155]
ZDF rat ^d	9–11	22–24	Femur	+7.8 (NS)	-10.1 (NS)	Area under the stress vs. strain curve (<i>n</i> =12 per group)	[17]
CD vs. ZDSD ^e rat ^d	21	10-12	Femur	+1 (NS)	+10.8 (NS)	Area under the stress vs. strain curve $(n = 12-17 \text{ per group})$	[17]
CD vs. ZDSD rat ^d	20	10-12	Femur	-14.9	-50	Area under the stress vs. strain curve $(n=7-11$ per group)	[131]
CD vs. ZDSD rat ^f	20	8-10	Femur	-0.1 (NS)	-6.6 (NS)	Area under the stress vs. strain curve $(n=4-8$ per group)	[129]
C57BL	3	16 ^h	Femur	-26	+8 (NS)	Tested notched	[128]
mouse ^g	15	-		-15	-21	bone to determine fracture toughness (n=14-15 per group)	
ob-ob		10	Femur	NR	-31	Work-to-	[127]
mouse	-	10	-		42.1	nation $(n = 10)$	
db-db		10			-43.1	per group)	
WDN/	12 52	0.0	Eamur	ND	29.4	Work to failure	1001
Kob rat	+3-32	0-9	remur		-30.4	(n=7 per)	[00]
VS.		6-13	-		-42.3	group)	
Wistar		16-25	-		-48.1		
	4	24-33	г	NID	-48	D (11	[120]
IallyHo/ JngJ	4	4	Femur	NK	-59	displacement	[132]
vs. SWR/J		13			-48	(n=0-10 per group)	

 Table 9.2
 Summary of the recent literature reporting differences in material properties of bone between normal rodents and rodents with type 2 diabetes

(continued)

Table 9.2 (continued)

Unless otherwise noted, rodents spontaneously develop diabetes. Significant differences indicated in **bold**. Otherwise, the difference (100* (Mean-T2D-Mean-Control)/Mean-Control) was not statistically significant (NS)

^aZDF Zucker Diabetic fatty

^bAnimals were fed a high fat diet (Purina 5008) starting at 11 weeks of age for the study duration ^cThis was a study investigating the effect of anti-Sclerostin antibody on bone in a diabetic model and included defect in contralateral femur with a slide plate

^dMale rats were fed a high fat diet (Purina 5008) starting at 20 or 21 weeks of age for the entire duration

^eZDSD Zucker Diabetic Sprague Dawley

Female rats were fed a high fat diet (Purina 5SCA) starting at 20 weeks of age for the study duration

^gMale mice were fed a high fat diet (Research Diets) for 16 weeks

^hBlood glucose levels indicated the obese group was likely diabetic

ⁱArea under force vs. displacement curve (not independent of structure)

In the study by Hill-Gallant et al. [129], there was a difference in the estimated compressive strength of cancellous bone (femur metaphysis), but no significant differences in material strength were observed for the femoral mid-shaft when subjected to three-point bending. Using an indentation technique (described in the next section) to assess tissue properties, Hammond et al. [130] reported that the cortical tissue of the tibia mid-shaft from diabetic ZDSD rats had a higher resistance to indentation than the cortical tissue from nondiabetic CD rats, whereas Gallant et al. [131] reported the opposite effect of diabetes on indentation resistance when indenting the femur mid-shaft from similar CD and ZDSD rats. Besides the bone that was tested, the maximum indenting force differed between the studies (5 N vs. 10 N). Also, the rats became diabetic on a high fat diet at approximately 20 weeks of age, but the special diet was only given for 1 week in Hammond et al. study and for entire 12 weeks of diabetes in the Gallant et al. study. In a mouse model of juvenile T2D, Devlin et al. [132] performed a similar indentation test on tibia mid-shafts from 17-week-old, nondiabetic mice (SWR/J strain) and age-matched diabetic mice (TallyHo/JngJ) but using a maximum target force of 2 N (average of three sites). There was lower resistance to indentation with diabetes though method of indentation resistance was not exactly the same as the aforementioned studies. Note that in these indentation studies, the control animals are not littermates to the diabetic animals. Overall, there is a need for more biomechanical studies with sufficient sample size to establish the effect of diabetes on the material- and tissue-level properties of bone. Of the multiple models of T2D, the bone phenotype has only been reported for the Zucker Diabetic Fatty (ZDF), Zucker Diabetic Sprague Dawley (ZDSD), and WBN/Kob rats (Table 9.2).

Assessing Tissue Material Properties with Indentation Techniques

Another way to assess material properties of bone is to use indentation techniques. To avoid confusion, mechanical properties from microindentation or nanoindentation are referred to as intrinsic material properties, as opposed to apparent material properties that were previously described, or simply tissue properties. There are several reviews on the application of microindentation and nanoindentation to bone [133, 134]. Briefly, the techniques involve pressing a very hard tip (e.g., diamond) into a relatively smooth surface of bone tissue and measuring force (P) as a function of penetration depth (h) (Fig. 9.9) such that the contact area of the tip is a known function of h (via calibration using a standard like fused silica). Either the maximum depth that the indenter tip penetrates or the maximum force generated is specified. There are multiple tip geometries in microindentation (Brinell, Kopp, Rockwell, or Vickers) and nanoindentation (Berkovich, Cube corner, Flat punch, Spherical) with Vickers (four-sided pyramid) and Berkovich (three-sided pyramid) being the most commonly used in bone studies. Microindenters measure tissue hardness at a scale of \sim 50 to \sim 200 µm as they are designed to record maximum depth for a given indentation weight (15–300 g). Being more sophisticated with respect to the control of the actuator and isolation of vibration, nanoindenters can provide both elastic and viscoelastic properties at a scale of ~0.1 to ~10 μ m. The elastic properties include tissue hardness and tissue modulus as typically determined by the Oliver-Pharr method [135]. The viscoelastic properties include loss modulus (E'), storage modulus (E"), and loss factor (tan $\delta = E''/E'$) in which the latter characterizes the viscous dampening behavior of a material [136]. Unlike the elastic properties that are derived from the P vs. h curve, the viscoelastic properties are derived from oscillatory loading at a specified frequency (or frequency sweep), displacement amplitude, and depth level in which there is a phase shift (δ) between force oscillation (input)



Fig. 9.9 The tissue hardness and modulus is determined from the peak force and the unloading slope generated during nanoindentation

and the responding displacement oscillation (output) [137]. Note that accurately measuring δ requires proper calibration and proper assumptions about the frame stiffness of the instrument. Other properties such as relaxation time constants and indentation work can be measured using nanoindentation, but hardness, modulus, and loss factor are the most widely reported for bone.

To the best of our knowledge, there are only two studies reporting elastic tissue properties of diabetic bone (none reporting viscoelastic properties). Acquiring femurs from wild-type and leptin-deficient (db/db) mice at 12 weeks of age, Williams et al. [138] cut embedded cross sections of the mid-shaft and polished the surface for nanoindentation. They observed that the dry tissue modulus was 9 % lower for the diabetic db/db mice than for the wild-type mice. In a study by Nyman et al. [19], tibia from normal mice and mice injected with STZ to induce T1D were embedded, sectioned, and polished for nanoindentation. In doing so, they found a 12 % reduction in dry tissue modulus after 18 weeks of diabetes with no significant differences between normal and diabetic bone tissue at 10 or 15 weeks postinjection.

Lower tissue modulus, or lower hardness for that matter, does not mean that the bone is more susceptible to fracture. Nanoindentation properties do not strongly correlate, if at all, with apparent-level modulus of human cortical bone [139] and modulus and strength of mouse bone [140]. Moreover, nanoindentation properties of human bone at the tissue level do not vary with age [139, 141], and for the most part, microindentation properties of bone do not vary with age either, though this has not been extensively analyzed with tissue from donors that span 20-100 years of life [142, 143]. The usefulness of microindentation and nanoindentation is the characterization of tissue at the organizational level of the lamella and then relating the tissue mechanical properties to other tissue properties of composition (degree of bone mineralization [142] or mineral-to-matrix ratio [144]) and organization (collagen orientation [145]). In doing so, there is insight into ultrastructuralmechanical relationships at the length scale in which bone cells interact with the matrix. By performing multiple indents throughout a bone section, an assessment of tissue heterogeneity can also be acquired. In general, a decrease in tissue modulus heterogeneity (i.e., at the nanoscale) causes a decrease in bone toughness [146]. If T2D reduces turnover and increases tissue age in humans, then the tissue modulus and hardness of diabetic bone would increase while the heterogeneity in these properties would decrease, thereby leading to a reduction in fracture resistance.

There is a relatively new microindentation technique called reference point indentation (RPI). Unlike the traditional indentation techniques, RPI does not require any special processing to create a smooth bone surface. Moreover, it can be performed through the periosteum of the tibia mid-shaft of patients in a minimally invasive way (local anesthetic), thereby providing a clinical assessment of indentation properties. Two versions of RPI currently exist: (1) a bench-top instrument (BioDentTM, Active Life Scientific, Inc.) uses an annular reference probe to engage the bone surface so that an inner, stainless steel, test probe with a spherical tip can indent the bone surface over multiple cycles of load–dwell–unload (2 Hz) at a user-specified maximum force (typically 10 N for human bone) [147] and (2) a handheld



Fig. 9.10 In reference point indentation, the depth of the probe tip is measured over multiple cycles of load–dwell–unload (a). Properties related to tissue stiffness, resistance to creep, and energy dissipation can also be measured (b)

device (OsteoProbe[®], Active Life Scientific, Inc.) with a similar spherical, stainless steel tip imparts a one-time impact force of ~ 45 N into the bone [148]. The BioDent records the force vs. displacement curve from which a number of properties can be recorded (Fig. 9.10) including those related to the depth that the indenter tip travels into the bone (e.g., indentation distance increase or IDI and total indentation distance or TID) [149]. In addition, properties can be calculated from each cycle of load-dwell-unload and include the loading stiffness, unloading stiffness, creep during the dwell portion, and the energy dissipated (i.e., area under the force vs. displacement curve). These properties vary over the first 4 cycles of load-dwell-unload and then asymptotically approach a constant value over the remaining 20 cycles [150, 151]. The OsteoProbe on the other hand provides only one measurement, which is the distance that the tip travels into the bone. This indentation distance increase (IDI-Bone) from the impact is then divided into the average distance that the same test probe tip penetrates a plastic standard (PMMA) to give a property called bone material strength index (BMSi=100*IDI-PMMA/IDI-Bone). Because the BioDent indents the bone at a length scale of ~120 μ m (diameter) (depth of \sim 70 µm at 10 N) in dynamic manner while the OsteoProbe indents the bone with impact force at a length scale of 300 μ m (depth of ~150 μ m), the two instruments do not necessarily measure the same tissue properties. That is, their assessment of resistance to indentation may differ.

In the initial experiments using the BioDent, Hansma et al. [147] found that IDI was different between non-irradiated and gamma irradiated bovine cortical bone and between a tibia from a 17-year-old and 79-year-old female donor. These tests involved approximately 100 indents, and IDI was normalized to mean IDI acquired from indenting a PMMA standard block. Subsequent work found that baking bone at 150 °C profoundly increased IDI (lower resistance to indentation), but freeze-thaw cycles had minimal effect on IDI (within biological variance) [152]. At radiation levels used in high-resolution μ CT, there is no effect of irradiation on RPI properties [151]. Also, in a sample of bovine cortical bone, sodium-fluoride

treatment, which affects the collagen-mineral interface, increased normalized IDI and the material strength at the apparent level [153].

Several concepts or "best practices" have emerged with the growing use of the BioDent: (1) acquire the average of multiple indents per bone sample (at least five sites), (2) perform at least ten cycles of loading, (3) indent PMMA standard regularly and possibly normalize data to this reference in order to minimize variance in tip geometry, especially since the probe wears during repeated use, (4) grind irregular surfaces to minimize variance in probe contact, (5) remove data from the average calculation when a pore has been indented (apparent in force vs. displacement curve), (6) apply the same preload according to manufacturer's recommendation, and (7) maintain the same hydration conditions across all bones. Other issues to keep in mind when performing RPI include: differences in properties among groups may depend on selected target force, preconditioning may not be necessary if there is no soft tissue present, and the reference probe may "skip" if not properly anchored into the bone tissue.

In recent studies from independent groups, RPI properties of bone have been found to exhibit anisotropy (dependent on direction of indentation relative to osteonal axis) [149], especially for stiffness (unloading and loading slopes) and average creep indentation distance for human cortical bone [150]. Correlations between IDI (not normalized) and apparent-level materials of bone have also been reported for a target maximum force of 10 N. IDI explained 56 % of the variance in toughness as determined from three-point bending tests of rat femurs acquired from 7 nondiabetics CD male rats (Charles River Labs) and 11 age-matched, diabetic, ZDSD male rats (PreClinOmics, Indianapolis, IN) [131]. Indenting the left femur opposite of the one subjected to whole-bone bending, IDI was an average of five sites on the anterior side. IDI explained 27 % of the variance in the toughness of human cortical bone as determined by three-point bending of machine samples acquired from 34 donors spanning 21-99 years of age [150]. In both studies, IDI was inversely proportional to toughness. IDI and other RPI properties however cannot be interpreted as surrogates of material properties of bone. RPI and bending tests assess bone at different length scales and under different modes of deformation. Thus, factors dictating resistance of bone tissue to indentation may be different than those influencing bone strength, toughness, or fracture toughness. Conceivably, the relationship between IDI and toughness (or strength) could be negative or positive depending on the disease or genetic model. As an example, Carriero et al. [154] found no relationship between normalized IDI and fracture toughness of mouse cortical bone (femur mid-shaft) when bones were collected from several mouse strains including one with osteogenesis imperfecta (OI) and one with defective mineralization (mice lacking a gene critical to phosphate production). In effect, IDI can be high because collagen structure is defective, as in the case of OI relative to normal mice, or high because mineralization was reduced, as in the case of Phospho1-/- relative to Phospho1+/+ mice. Fracture toughness, on the other hand, decreases when collagen integrity is poor (OI) but increases when mineralization decreases (hypomineralization) conferring more ductility to the bone.

Reported differences in RPI properties from clinical studies can be found in Chap. 10. The more recent studies involved the OsteoProbe, and because it is easier to use in a clinical setting than the BiodDent, the expectation is that emerging studies will also use the hand-held device that provides only one measurement. As is this the case with apparent-level material properties, additional preclinical studies are needed to understand what tissue-level properties RPI assess.

Acknowledgments We thank Mathilde Granke and Sasidhar Uppuganti for their efforts in making the figures for this chapter and for reviewing the content in this chapter.

References

- Rho JY, Kuhn-Spearing L, Zioupos P. Mechanical properties and the hierarchical structure of bone. Med Eng Phys. 1998;20(2):92–102.
- 2. Ritchie RO. The conflicts between strength and toughness. Nat Mater. 2011;10(11):817–22.
- Reznikov N, Shahar R, Weiner S. Bone hierarchical structure in three dimensions. Acta Biomater. 2014;10(9):3815–26.
- 4. Nyman JS. Effect of diabetes on the fracture resistance of bone. Clinic Rev Bone Miner Metab. 2012;21.
- Saha MT, Sievänen H, Salo MK, Tulokas S, Saha HH. Bone mass and structure in adolescents with type 1 diabetes compared to healthy peers. Osteoporos Int. 2009;20(8):1401–6.
- Maser RE, Kolm P, Modlesky CM, Beck TJ, Lenhard MJ. Hip strength in adults with type 1 diabetes is associated with age at onset of diabetes. J Clin Densitom. 2012;15(1):78–85.
- Bechtold S, Putzker S, Bonfig W, Fuchs O, Dirlenbach I, Schwarz HP. Bone size normalizes with age in children and adolescents with type 1 diabetes. Diabetes Care. 2007;30(8):2046–50.
- Miazgowski T, Pynka S, Noworyta-Zietara M, Krzyzanowska-Swiniarska B, Pikul R. Bone mineral density and hip structural analysis in type 1 diabetic men. Eur J Endocrinol. 2007;156(1):123–7.
- Melton LJ, Riggs BL, Leibson CL, Achenbach SJ, Camp JJ, Bouxsein ML, et al. A bone structural basis for fracture risk in diabetes. J Clin Endocrinol Metab. 2008;93(12):4804–9.
- Burghardt AJ, Issever AS, Schwartz AV, Davis KA, Masharani U, Majumdar S, et al. Highresolution peripheral quantitative computed tomographic imaging of cortical and trabecular bone microarchitecture in patients with type 2 diabetes mellitus. J Clin Endocrinol Metab. 2010;95(11):5045–55.
- Shu A, Yin MT, Stein E, Cremers S, Dworakowski E, Ives R, et al. Bone structure and turnover in type 2 diabetes mellitus. Osteoporos Int. 2012;23(2):635–41.
- 12. Garg R, Chen Z, Beck T, Cauley JA, Wu G, Nelson D, et al. Hip geometry in diabetic women: implications for fracture risk. Metab Clin Exp. 2012;61(12):1756–62.
- Ishii S, Cauley JA, Crandall CJ, Srikanthan P, Greendale GA, Huang M-H, et al. Diabetes and femoral neck strength: findings from the hip strength across the menopausal transition study. J Clin Endocrinol Metab. 2012;97(1):190–7.
- 14. Hamilton CJ, Jamal SA, Beck TJ, Khaled AS, Adachi JD, Brown JP, et al. Evidence for impaired skeletal load adaptation among Canadian women with type 2 diabetes mellitus: insight into the BMD and bone fragility paradox. Metab Clin Exp. 2013;62(10):1401–5.
- Fajardo RJ, Karim L, Calley VI, Bouxsein ML. A review of rodent models of type 2 diabetic skeletal fragility. J Bone Miner Res. 2014;29(5):1025–40.
- Prisby RD, Swift JM, Bloomfield SA, Hogan HA, Delp MD. Altered bone mass, geometry and mechanical properties during the development and progression of type 2 diabetes in the Zucker diabetic fatty rat. J Endocrinol. 2008;199(3):379–88.

- Reinwald S, Peterson RG, Allen MR, Burr DB. Skeletal changes associated with the onset of type 2 diabetes in the ZDF and ZDSD rodent models. Am J Physiol Endocrinol Metab. 2009;296(4):E765–74.
- Silva MJ, Brodt MD, Lynch MA, McKenzie JA, Tanouye KM, Nyman JS, et al. Type 1 diabetes in young rats leads to progressive trabecular bone loss, cessation of cortical bone growth, and diminished whole bone strength and fatigue life. J Bone Miner Res. 2009; 24(9):1618–27.
- Nyman JS, Even JL, Jo C-H, Herbert EG, Murry MR, Cockrell GE, et al. Increasing duration of type 1 diabetes perturbs the strength-structure relationship and increases brittleness of bone. Bone. 2011;48(4):733–40.
- Rosen CJ, Ackert-Bicknell CL, Adamo ML, Shultz KL, Rubin J, Donahue LR, et al. Congenic mice with low serum IGF-I have increased body fat, reduced bone mineral density, and an altered osteoblast differentiation program. Bone. 2004;35(5):1046–58.
- Fulzele K, Riddle RC, DiGirolamo DJ, Cao X, Wan C, Chen D, et al. Insulin receptor signaling in osteoblasts regulates postnatal bone acquisition and body composition. Cell. 2010; 142(2):309–19.
- Thrailkill K, Bunn RC, Lumpkin C, Wahl E, Cockrell G, Morris L, et al. Loss of insulin receptor in osteoprogenitor cells impairs structural strength of bone. J Diab Res. 2014; 2014:703589.
- Irwin R, Lin HV, Motyl KJ, Mccabe LR. Normal bone density obtained in the absence of insulin receptor expression in bone. Endocrinology. 2006;147(12):5760–7.
- Okamoto H, Obici S, Accili D, Rossetti L. Restoration of liver insulin signaling in Insr knockout mice fails to normalize hepatic insulin action. J Clin Invest. 2005;115(5): 1314–22.
- Strotmeyer ES, Cauley JA. Diabetes mellitus, bone mineral density, and fracture risk. Curr Opin Endocrinol Diabetes Obes. 2007;14(6):429–35.
- Seeley DG, Kelsey J, Jergas M, Nevitt MC. Predictors of ankle and foot fractures in older women. The Study of Osteoporotic Fractures Research Group. J Bone Miner Res. 1996;11(9):1347–55.
- de Liefde II, van der Klift M, de Laet CEDH, van Daele PLA, Hofman A, Pols HAP. Bone mineral density and fracture risk in type-2 diabetes mellitus: the Rotterdam study. Osteoporos Int. 2005;16(12):1713–20.
- Schwartz AV, Vittinghoff E, Bauer DC, Hillier TA, Strotmeyer ES, Ensrud KE, et al. Association of BMD and FRAX score with risk of fracture in older adults with type 2 diabetes. JAMA. 2011;305(21):2184–92.
- Hernandez CJ, Lambers FM, Widjaja J, Chapa C, Rimnac CM. Quantitative relationships between microdamage and cancellous bone strength and stiffness. Bone. 2014;66:205–13.
- Lambers FM, Bouman AR, Rimnac CM, Hernandez CJ. Microdamage caused by fatigue loading in human cancellous bone: relationship to reductions in bone biomechanical performance. PLoS One. 2013;8(12):e83662.
- Burr DB, Forwood MR, Fyhrie DP, Martin RB, Schaffler MB, Turner CH. Bone microdamage and skeletal fragility in osteoporotic and stress fractures. J Bone Miner Res. 1997; 12(1):6–15.
- 32. Chantelau E, Richter A, Schmidt-Grigoriadis P, Scherbaum WA. The diabetic charcot foot: MRI discloses bone stress injury as trigger mechanism of neuroarthropathy. Exp Clin Endocrinol Diabetes. 2006;114(3):118–23.
- 33. Martin RB, Burr DB, Sharkey NA. Skeletal tissue mechanics. New York: Springer; 1998.
- 34. Cowin SC. Bone mechanics handbook. 2nd ed. Boca Raton: CRC Press; 2001.
- Roeder RK. Mechanical characterization of biomaterials. In: Bandhyopadhya A, Bose S, editors. Characterization of biomaterials. Amsterdam: Elsevier Science; 2013.
- 36. Turner CH, Burr DB. Basic biomechanical measurements of bone: a tutorial. Bone. 1993;14(4):595-608.
- Agcaoglu S, Akkus O. Acoustic emission based monitoring of the microdamage evolution during fatigue of human cortical bone. J Biomech Eng. 2013;135(8):81005–8.

- Haupert S, Guérard S, Peyrin F, Mitton D, Laugier P. Non destructive characterization of cortical bone micro-damage by nonlinear resonant ultrasound spectroscopy. PLoS One. 2014;9(1):e83599.
- Lee TC, Mohsin S, Taylor D, Parkesh R, Gunnlaugsson T, O'Brien FJ, et al. Detecting microdamage in bone. J Anat. 2003;203(2):161–72.
- Landrigan MD, Li J, Turnbull TL, Burr DB, Niebur GL, Roeder RK. Contrast-enhanced micro-computed tomography of fatigue microdamage accumulation in human cortical bone. Bone. 2011;48(3):443–50.
- 41. Boyce TM, Fyhrie DP, Glotkowski MC, Radin EL, Schaffler MB. Damage type and strain mode associations in human compact bone bending fatigue. J Orthop Res. 1998;16(3): 322–9.
- 42. Diab T, Vashishth D. Morphology, localization and accumulation of in vivo microdamage in human cortical bone. Bone. 2007;40(3):612–8.
- 43. Koch J. The laws of bone architecture. Am J Anat. 1917;21(2):177-298.
- 44. Williamson CF. A history of the finite element method to the middle 1960's. Ph. D. Thesis, Boston University; 1976.
- Keyak JH, Rossi SA, Jones KA, Skinner HB. Prediction of femoral fracture load using automated finite element modeling. J Biomech. 1998;31(2):125–33.
- Keaveny TM, Bartel DL. Effects of porous coating and collar support on early load transfer for a cementless hip prosthesis. J Biomech. 1993;26(10):1205–16.
- 47. Lengsfeld M, Schmitt J, Alter P, Kaminsky J, Leppek R. Comparison of geometry-based and CT voxel-based finite element modelling and experimental validation. Med Eng Phys. 1998;20(7):515–22.
- Cody DD, Gross GJ, Hou FJ, Spencer HJ, Goldstein SA, Fyhrie DP. Femoral strength is better predicted by finite element models than QCT and DXA. J Biomech. 1999;32(10): 1013–20.
- 49. Pistoia W, van Rietbergen B, Lochmüller E-M, Lill CA, Eckstein F, Rüegsegger P. Estimation of distal radius failure load with micro-finite element analysis models based on threedimensional peripheral quantitative computed tomography images. Bone. 2002;30(6): 842–8.
- 50. Mueller TL, Christen D, Sandercott S, Boyd SK, van Rietbergen B, Eckstein F, et al. Computational finite element bone mechanics accurately predicts mechanical competence in the human radius of an elderly population. Bone. 2011;48(6):1232–8.
- Silva MJ, Keaveny TM, Hayes WC. Computed tomography-based finite element analysis predicts failure loads and fracture patterns for vertebral sections. J Orthop Res. 1998;16(3): 300–8.
- Imai K, Ohnishi I, Matsumoto T, Yamamoto S, Nakamura K. Assessment of vertebral fracture risk and therapeutic effects of alendronate in postmenopausal women using a quantitative computed tomography-based nonlinear finite element method. Osteoporos Int. 2009; 20(5):801–10.
- 53. Burghardt AJ, Kazakia GJ, Sode M, de Papp AE, Link TM, Majumdar S. A longitudinal HR-pQCT study of alendronate treatment in postmenopausal women with low bone density: relations among density, cortical and trabecular microarchitecture, biomechanics, and bone turnover. J Bone Miner Res. 2010;25(12):2558–71.
- 54. Keaveny TM, Hoffmann PF, Singh M, Palermo L, Bilezikian JP, Greenspan SL, et al. Femoral bone strength and its relation to cortical and trabecular changes after treatment with PTH, alendronate, and their combination as assessed by finite element analysis of quantitative CT scans. J Bone Miner Res. 2008;23(12):1974–82.
- 55. Cosman F, Keaveny TM, Kopperdahl D, Wermers RA, Wan X, Krohn KD, et al. Hip and spine strength effects of adding versus switching to teriparatide in postmenopausal women with osteoporosis treated with prior alendronate or raloxifene. J Bone Miner Res. 2013; 28(6):1328–36.
- 56. Farahmand P, Marin F, Hawkins F, Möricke R, Ringe JD, Glüer C-C, et al. Early changes in biochemical markers of bone formation during teriparatide therapy correlate with

improvements in vertebral strength in men with glucocorticoid-induced osteoporosis. Osteoporos Int. 2013;24(12):2971–81.

- 57. Brixen K, Chapurlat R, Cheung AM, Keaveny TM, Fuerst T, Engelke K, et al. Bone density, turnover, and estimated strength in postmenopausal women treated with odanacatib: a randomized trial. J Clin Endocrinol Metab. 2013;98(2):571–80.
- Cheung AM, Majumdar S, Brixen K, Chapurlat R, Fuerst T, Engelke K, et al. Effects of odanacatib on the radius and tibia of postmenopausal women: improvements in bone geometry, microarchitecture, and estimated bone strength. J Bone Miner Res. 2014;29(8): 1786–94.
- 59. Keaveny TM, McClung MR, Genant HK, Zanchetta JR, Kendler D, Brown JP, et al. Femoral and vertebral strength improvements in postmenopausal women with osteoporosis treated with denosumab. J Bone Miner Res. 2014;29(1):158–65.
- 60. Kleerekoper M, Greenspan SL, Lewiecki EM, Miller PD, Kendler DL, Maricic M, et al. Assessing the effects of teriparatide treatment on bone mineral density, bone microarchitecture, and bone strength. J Bone Joint Surg Am. 2014;96(11):e90.
- 61. Tsai JN, Uihlein AV, Burnett-Bowsie SA, Neer RM, Zhu Y, Derrico N, et al. Comparative effects of teriparatide, denosumab, and combination therapy on peripheral compartmental bone density, microarchitecture, and estimated strength: the DATA-HRpQCT study. J Bone Miner Res. 2014;30(1):39–45.
- 62. Melton LJ, Riggs BL, van Lenthe GH, Achenbach SJ, Müller R, Bouxsein ML, et al. Contribution of in vivo structural measurements and load/strength ratios to the determination of forearm fracture risk in postmenopausal women. J Bone Miner Res. 2007;22(9):1442–8.
- 63. Orwoll ES, Marshall LM, Nielson CM, Cummings SR, Lapidus J, Cauley JA, et al. Finite element analysis of the proximal femur and hip fracture risk in older men. J Bone Miner Res. 2009;24(3):475–83.
- 64. Vilayphiou N, Boutroy S, Szulc P, van Rietbergen B, Munoz F, Delmas PD, et al. Finite element analysis performed on radius and tibia HR-pQCT images and fragility fractures at all sites in men. J Bone Miner Res. 2010;26(5):965–73.
- 65. Nishiyama KK, Macdonald HM, Hanley DA, Boyd SK. Women with previous fragility fractures can be classified based on bone microarchitecture and finite element analysis measured with HR-pQCT. Osteoporos Int. 2012;24(5):1733–40.
- 66. Graeff C, Marin F, Petto H, Kayser O, Reisinger A, Peña J, et al. High resolution quantitative computed tomography-based assessment of trabecular microstructure and strength estimates by finite-element analysis of the spine, but not DXA, reflects vertebral fracture status in men with glucocorticoid-induced osteoporosis. Bone. 2013;52(2):568–77.
- 67. Imai K, Ohnishi I, Yamamoto S, Nakamura K. In vivo assessment of lumbar vertebral strength in elderly women using computed tomography-based nonlinear finite element model. Spine. 2008;33(1):27–32.
- Kopperdahl DL, Aspelund T, Hoffmann PF, Sigurdsson S, Siggeirsdottir K, Harris TB, et al. Assessment of incident spine and hip fractures in women and men using finite element analysis of CT scans. J Bone Miner Res. 2014;29(3):570–80.
- 69. Donnelly E. Methods for assessing bone quality: a review. Clin Orthop Relat Res. 2011;469(8):2128–38.
- Warner S, Jo J, Meyers K, DiCarlo E, Lane JM, Donnelly E. The effect of diabetes on bone strength and collagen cross-linking. Annual meeting of the American Society for Bone and Mineral Research, Houston; 2014.
- Fleischli JG, Dpm TJL, Mph LALD, Bs BS, Ms DL, Pe CMAP, et al. The effects of diabetes mellitus on the material properties of human metatarsal bones. J Foot Ankle Surg. 1998;37(3):195–8.
- Fleischli JG, Laughlin TJ, Athanasiou K, Lanctot DR, Lavery L, Wang X, et al. Effect of diabetes mellitus on the material properties of the distal tibia. J Am Podiatr Med Assoc. 2006;96(2):91–5.
- Koester KJ, Barth HD, Ritchie RO. Effect of aging on the transverse toughness of human cortical bone: evaluation by R-curves. J Mech Behav Biomed Mater. 2011;4(7):1504–13.

- Hazama H. Study on the torsional strength of the compact substance of human beings. J Kyoto Pref Med Univ. 1956;60:167–84.
- Melick RA, Miller DR. Variations of tensile strength of human cortical bone with age. Clin Sci. 1966;30(2):243–8.
- Lindahl O, Lindgren AG. Cortical bone in man. II. Variation in tensile strength with age and sex. Acta Orthop Scand. 1967;38(2):141–7.
- 77. Evans FG, Lewbow M. Regional differences in some of the physical properties of the human femur. J Appl Physiol. 1951;3(9):563–72.
- Sedlin ED, Hirsch C. Factors affecting the determination of the physical properties of femoral cortical bone. Acta Orthop Scand. 1966;37(1):29–48.
- Burstein AH, Reilly DT, Martens M. Aging of bone tissue: mechanical properties. J Bone Joint Surg Am. 1976;58(1):82–6.
- Smith CB, Smith DA. Relations between age, mineral density and mechanical properties of human femoral compacta. Acta Orthop Scand. 1976;47(5):496–502.
- Wall JC, Chatterji SK, Jeffery JW. Age-related changes in the density and tensile strength of human femoral cortical bone. Calcif Tissue Int. 1979;27(2):105–8.
- McCalden RW, McGeough JA, Barker MB, Court-Brown CM. Age-related changes in the tensile properties of cortical bone. The relative importance of changes in porosity, mineralization, and microstructure. J Bone Joint Surg Am. 1993;75(8):1193–205.
- 83. Zioupos P, Currey JD. Changes in the stiffness, strength, and toughness of human cortical bone with age. Bone. 1998;22(1):57–66.
- Currey JD. The mechanical consequences of variation in the mineral content of bone. J Biomech. 1969;2(1):1–11.
- McCalden RW, McGeough JA, Court-Brown CM. Age-related changes in the compressive strength of cancellous bone. The relative importance of changes in density and trabecular architecture. J Bone Joint Surg Am. 1997;79(3):421–7.
- Oxlund H, Barckman M, Ortoft G, Andreassen TT. Reduced concentrations of collagen cross-links are associated with reduced strength of bone. Bone. 1995;17(4 Suppl):365S–71.
- McNerny EMB, Gong B, Morris MD, Kohn DH. Bone fracture toughness and strength correlate with collagen cross-link maturity in a dose-controlled lathyrism mouse model. J Bone Miner Res. 2015;30(3):455–64.
- Saito M, Fujii K, Mori Y, Marumo K. Role of collagen enzymatic and glycation induced cross-links as a determinant of bone quality in spontaneously diabetic WBN/Kob rats. Osteoporos Int. 2006;17(10):1514–23.
- Nyman JS, Reyes M, Wang X. Effect of ultrastructural changes on the toughness of bone. Micron. 2005;36(7–8):566–82.
- Viguet-Carrin S, Garnero P, Delmas PD. The role of collagen in bone strength. Osteoporos Int. 2006;17(3):319–36.
- 91. Garnero P. The contribution of collagen crosslinks to bone strength. Bonekey Rep. 2012;1:182.
- 92. Wang X, Shen X, Li X, Agrawal CM. Age-related changes in the collagen network and toughness of bone. Bone. 2002;31(1):1–7.
- Nyman JS, Roy A, Tyler JH, Acuna RL, Gayle HJ, Wang X. Age-related factors affecting the postyield energy dissipation of human cortical bone. J Orthop Res. 2007;25(5):646–55.
- 94. Schaffler MB, Choi K, Milgrom C. Aging and matrix microdamage accumulation in human compact bone. Bone. 1995;17(6):521–5.
- Mori S, Harruff R, Ambrosius W, Burr DB. Trabecular bone volume and microdamage accumulation in the femoral heads of women with and without femoral neck fractures. Bone. 1997;21(6):521–6.
- 96. Zioupos P. Accumulation of in-vivo fatigue microdamage and its relation to biomechanical properties in ageing human cortical bone. J Microsc. 2001;201(2):270–8.
- 97. Norman TL, Little TM, Yeni YN. Age-related changes in porosity and mineralization and in-service damage accumulation. J Biomech. 2008;41(13):2868–73.
- Follet H, Farlay D, Bala Y, Viguet-Carrin S, Gineyts E, Burt-Pichat B, et al. Determinants of microdamage in elderly human vertebral trabecular bone. PLoS One. 2013;8(2):e55232.

- 99. Fletcher L, Codrington J, Parkinson I. Effects of fatigue induced damage on the longitudinal fracture resistance of cortical bone. J Mater Sci Mater Med. 2014;25(7):1661–70.
- Turnbull TL, Baumann AP, Roeder RK. Fatigue microcracks that initiate fracture are located near elevated intracortical porosity but not elevated mineralization. J Biomech. 2014; 47(12):3135–42.
- Zioupos P, Currey JD, Hamer AJ. The role of collagen in the declining mechanical properties of aging human cortical bone. J Biomed Mater Res. 1999;45(2):108–16.
- 102. Nyman JS, Ni Q, Nicolella DP, Wang X. Measurements of mobile and bound water by nuclear magnetic resonance correlate with mechanical properties of bone. Bone. 2008; 42(1):193–9.
- 103. Nyman JS, Gorochow LE, Adam Horch R, Uppuganti S, Zein-Sabatto A, Manhard MK, et al. Partial removal of pore and loosely bound water by low-energy drying decreases cortical bone toughness in young and old donors. J Mech Behav Biomed Mater. 2013;22:136–45.
- 104. Saito M, Marumo K. Bone quality in diabetes. Front Endocrinol (Lausanne). 2013;4:72.
- 105. Zimmermann EA, Schaible E, Bale H, Barth HD, Tang SY, Reichert P, et al. Age-related changes in the plasticity and toughness of human cortical bone at multiple length scales. Proc Natl Acad Sci. 2011;108(35):14416–21.
- Vashishth D, Gibson GJ, Khoury JI, Schaffler MB, Kimura J, Fyhrie DP. Influence of nonenzymatic glycation on biomechanical properties of cortical bone. Bone. 2001;28(2):195–201.
- 107. Willett TL, Sutty S, Gaspar A, Avery N, Grynpas M. In vitro non-enzymatic ribation reduces post-yield strain accommodation in cortical bone. Bone. 2013;52(2):611–22.
- Viguet-Carrin S, Farlay D, Bala Y, Munoz F, Bouxsein ML, Delmas PD. An in vitro model to test the contribution of advanced glycation end products to bone biomechanical properties. Bone. 2008;42(1):139–49.
- 109. Karim L, Tang SY, Sroga GE, Vashishth D. Differences in non-enzymatic glycation and collagen cross-links between human cortical and cancellous bone. Osteoporos Int. 2013; 24(9):2441–7.
- Sroga GE, Siddula A, Vashishth D. Glycation of human cortical and cancellous bone captures differences in the formation of Maillard reaction products between glucose and ribose. PLoS One. 2015;10(2):e0117240.
- 111. Tang SY, Zeenath U, Vashishth D. Effects of non-enzymatic glycation on cancellous bone fragility. Bone. 2007;40(4):1144–51.
- 112. Bradke BS, Vashishth D. N-phenacylthiazolium bromide reduces bone fragility induced by nonenzymatic glycation. PLoS One. 2014;9(7):e103199.
- 113. Yeni YN, Brown CU, Wang Z, Norman TL. The influence of bone morphology on fracture toughness of the human femur and tibia. Bone. 1997;21(5):453–9.
- 114. Nalla RK, Kruzic JJ, Kinney JH, Ritchie RO. Effect of aging on the toughness of human cortical bone: evaluation by R-curves. Bone. 2004;35(6):1240–6.
- 115. Yeni YN, Brown CU, Norman TL. Influence of bone composition and apparent density on fracture toughness of the human femur and tibia. Bone. 1998;22(1):79–84.
- 116. Yeni YN, Norman TL. Calculation of porosity and osteonal cement line effects on the effective fracture toughness of cortical bone in longitudinal crack growth. J Biomed Mater Res. 2000;51(3):504–9.
- 117. Norman TL, Yeni YN, Brown CU, Wang Z. Influence of microdamage on fracture toughness of the human femur and tibia. Bone. 1998;23(3):303–6.
- Granke M, Makowski AJ, Uppuganti S, Does MD, Nyman JS. Identifying novel clinical surrogates to assess human bone fracture toughness. J Bone Miner Res. 2015 doi:10.1002/ jbmr.2452.
- 119. Cefalu WT. Animal models of type 2 diabetes: clinical presentation and pathophysiological relevance to the human condition. ILAR J. 2006;47(3):186–98.
- Van Belle TL, Taylor P, von Herrath MG. Mouse models for type 1 diabetes. Drug Discov Today Dis Models. 2009;6(2):41–5.
- 121. Wang Y-W, Sun G-D, Sun J, Liu S-J, Wang J, Xu X-H, et al. Spontaneous type 2 diabetic rodent models. J Diab Res. 2013;2013:401723.

- 9 Material Properties of Diabetic Bone
- 122. Reddy GK, Stehno-Bittel L, Hamade S, Enwemeka CS. The biomechanical integrity of bone in experimental diabetes. Diabetes Res Clin Pract. 2001;54(1):1–8.
- 123. Motyl KJ, Raetz M, Tekalur SA, Schwartz RC, Mccabe LR. CCAAT/enhancer binding protein β-deficiency enhances type 1 diabetic bone phenotype by increasing marrow adiposity and bone resorption. Am J Physiol Regul Integr Comp Physiol. 2011;300(5):R1250–60.
- 124. Erdal N, Gürgül S, Demirel C, Yildiz A. The effect of insulin therapy on biomechanical deterioration of bone in streptozotocin (STZ)-induced type 1 diabetes mellitus in rats. Diabetes Res Clin Pract. 2012;97(3):461–7.
- 125. Fowlkes JL, Nyman JS, Bunn RC, Jo C, Wahl EC, Liu L, et al. Osteo-promoting effects of insulin-like growth factor I (IGF-I) in a mouse model of type 1 diabetes. Bone. 2013; 57(1):36–40.
- 126. Glorie L, Behets GJ, Baerts L, De Meester I, D'Haese PC, Verhulst A. DPP IV inhibitor treatment attenuates bone loss and improves mechanical bone strength in male diabetic rats. Am J Physiol Endocrinol Metab. 2014;307(5):E447–55.
- 127. Ealey KN, Fonseca D, Archer MC, Ward WE. Bone abnormalities in adolescent leptindeficient mice. Regul Pept. 2006;136(1–3):9–13.
- 128. Ionova-Martin SS, Wade JM, Tang S, Shahnazari M, Ager JW, Lane NE, et al. Changes in cortical bone response to high-fat diet from adolescence to adulthood in mice. Osteoporos Int. 2011;22(8):2283–93.
- 129. Hill Gallant KM, Gallant MA, Brown DM, Sato AY, Williams JN, Burr DB. Raloxifene prevents skeletal fragility in adult female Zucker Diabetic Sprague-Dawley rats. PLoS One. 2014;9(9):e108262.
- 130. Hammond MA, Gallant MA, Burr DB, Wallace JM. Nanoscale changes in collagen are reflected in physical and mechanical properties of bone at the microscale in diabetic rats. Bone. 2014;60:26–32.
- Gallant MA, Brown DM, Organ JM, Allen MR, Burr DB. Reference-point indentation correlates with bone toughness assessed using whole-bone traditional mechanical testing. Bone. 2013;53(1):301–5.
- 132. Devlin MJ, Van Vliet M, Motyl K, Karim L, Brooks DJ, Louis L, et al. Early-onset type 2 diabetes impairs skeletal acquisition in the male TALLYHO/JngJ mouse. Endocrinology. 2014;155(10):3806–16.
- 133. Lewis G, Nyman JS. The use of nanoindentation for characterizing the properties of mineralized hard tissues: state-of-the art review. J Biomed Mater Res B Appl Biomater. 2008; 87(1):286–301.
- 134. Zysset PK. Indentation of bone tissue: a short review. Osteoporos Int. 2009;20(6):1049-55.
- 135. Oliver WC, Pharr GM. An improved technique for determining hardness and elastic modulus using load and displacement sensing indentation experiments. J Mater Res. 1992;7(06): 1564–83.
- 136. Pathak S, Swadener JG, Kalidindi SR, Courtland H-W, Jepsen KJ, Goldman HM. Measuring the dynamic mechanical response of hydrated mouse bone by nanoindentation. J Mech Behav Biomed Mater. 2011;4(1):34–43.
- 137. Oyen ML, Cook RF. A practical guide for analysis of nanoindentation data. J Mech Behav Biomed Mater. 2009;2(4):396–407.
- Williams GA, Callon KE, Watson M, Costa JL, Ding Y, Dickinson M, et al. Skeletal phenotype of the leptin receptor-deficient db/db mouse. J Bone Miner Res. 2011;26(8):1698–709.
- Rho JY, Zioupos P, Currey JD, Pharr GM. Microstructural elasticity and regional heterogeneity in human femoral bone of various ages examined by nano-indentation. J Biomech. 2002;35(2):189–98.
- 140. Silva MJ, Brodt MD, Fan Z, Rho J-Y. Nanoindentation and whole-bone bending estimates of material properties in bones from the senescence accelerated mouse SAMP6. J Biomech. 2004;37(11):1639–46.
- 141. Hoffler CE, Moore KE, Kozloff K, Zysset PK, Goldstein SA. Age, gender, and bone lamellae elastic moduli. J Orthop Res. 2000;18(3):432–7.

- 142. Boivin G, Bala Y, Doublier A, Farlay D, Ste-Marie LG, Meunier PJ, et al. The role of mineralization and organic matrix in the microhardness of bone tissue from controls and osteoporotic patients. Bone. 2008;43(3):532–8.
- 143. Zioupos P, Gresle M, Winwood K. Fatigue strength of human cortical bone: age, physical, and material heterogeneity effects. J Biomed Mater Res A. 2008;86(3):627–36.
- 144. Donnelly E, Boskey AL, Baker SP, van der Meulen MCH. Effects of tissue age on bone tissue material composition and nanomechanical properties in the rat cortex. J Biomed Mater Res A. 2010;92(3):1048–56.
- 145. Donnelly E, Williams RM, Downs SA, Dickinson ME, Baker SP, van der Meulen MCH. Quasistatic and dynamic nanomechanical properties of cancellous bone tissue relate to collagen content and organization. J Mater Res. 2006;21(8):2106–17.
- 146. Tai K, Dao M, Suresh S, Palazoglu A, Ortiz C. Nanoscale heterogeneity promotes energy dissipation in bone. Nat Mater. 2007;6(6):454–62.
- 147. Hansma P, Turner P, Drake B, Yurtsev E, Proctor A, Mathews P, et al. The bone diagnostic instrument II: indentation distance increase. Rev Sci Instrum. 2008;79(6):064303.
- 148. Bridges D, Randall C, Hansma PK. A new device for performing reference point indentation without a reference probe. Rev Sci Instrum. 2012;83(4):044301.
- 149. Rasoulian R, Raeisi Najafi A, Chittenden M, Jasiuk I. Reference point indentation study of age-related changes in porcine femoral cortical bone. J Biomech. 2013;46(10):1689–96.
- 150. Granke M, Coulmier A, Uppuganti S, Gaddy JA, Does MD, Nyman JS. Insights into reference point indentation involving human cortical bone: sensitivity to tissue anisotropy and mechanical behavior. J Mech Behav Biomed Mater. 2014;37(C):174–85.
- 151. Setters A, Jasiuk I. Towards a standardized reference point indentation testing procedure. J Mech Behav Biomed Mater. 2014;34:57–65.
- 152. Kaye B, Randall C, Walsh D, Hansma P. The effects of freezing on the mechanical properties of bone. Open Bone J. 2012;4:14–9.
- 153. Thurner PJ, Erickson B, Turner P, Jungmann R, Lelujian J, Proctor A, et al. The effect of NaF in vitro on the mechanical and material properties of trabecular and cortical bone. Adv Mater. 2009;21(4):451–7.
- 154. Carriero A, Bruse JL, Oldknow KJ, Millán JL, Farquharson C, Shefelbine SJ. Reference point indentation is not indicative of whole mouse bone measures of stress intensity fracture toughness. Bone. 2014;69(C):1–6.
- 155. Hamann C, Rauner M, Höhna Y, Bernhardt R, Mettelsiefen J, Goettsch C, et al. Sclerostin antibody treatment improves bone mass, bone strength, and bone defect regeneration in rats with type 2 diabetes mellitus. J Bone Miner Res. 2013;28(3):627–38.

Chapter 10 Bone Quality in Type 2 Diabetes Mellitus

Joshua N. Farr and Sundeep Khosla

Introduction

Type 2 diabetes mellitus (T2DM) is one of the most common chronic diseases worldwide [1]. As changing lifestyles lead to increased obesity [2], the prevalence of T2DM will continue to grow and the economic public health burden will worsen significantly [1, 3]. Indeed, the direct medical costs of T2DM are estimated at over \$116 billion annually in the USA alone [3]. Globally, 285 million people have T2DM, and this number is predicted to increase to 439 million by 2030 [1].

While there is abundant evidence that patients with T2DM are at significant risk for premature mortality and morbidity from macrovascular disease, retinopathy, nephropathy, and neuropathy [3], emerging evidence suggests that T2DM also has adverse skeletal effects. Indeed, numerous studies have established T2DM as an independent risk factor for fragility fractures at skeletal sites such as the hip, spine, and distal forearm (see Chap. 2) [4–10]. For example, based on existing data, T2DM is associated with a 50–80 % increased extremity fracture risk [4, 8], and a meta-analysis of 12 studies found a relative risk of 1.7 (95 % CI: 1.3–2.2) for hip fracture [10]. These findings are perhaps surprising given that patients with T2DM often have normal or increased dual-energy X-ray absorptiometry (DXA)-derived areal bone mineral density (aBMD) [11], even when normalized for body mass index (BMI) [12]. Nonetheless, data from several large, prospective studies in the USA [13] and Canada [14] have recently demonstrated that T2DM patients have a higher fracture risk (see Chap. 3) for either a given femoral neck aBMD *T*-score and age or a given FRAX probability (defined by the World Health Organization's Fracture

B. Lecka-Czernik, J.L. Fowlkes (eds.), *Diabetic Bone Disease*, DOI 10.1007/978-3-319-16402-1_10

J.N. Farr, Ph.D. • S. Khosla, M.D. (🖂)

Division of Endocrinology, Metabolism, Nutrition and Diabetes, Department of Medicine, Mayo Clinic, 200 First Street SW, Rochester, MN 55905, USA

Robert and Arlene Kogod Center on Aging, Rochester, MN 55905, USA e-mail: farr.joshua@mayo.edu; khosla.sundeep@mayo.edu

[©] Springer International Publishing Switzerland 2016

Risk Algorithm [FRAX[®]] score [15]), suggesting that other factors, independent of aBMD, are likely responsible.

Since the risk for developing T2DM increases with advancing age and therefore frequently coexists with age-related bone loss [16], established risk factors for fragility fractures that occur with normal aging also contribute to fracture risk in patients with T2DM. Nevertheless, there is now considerable evidence that specific fracture risk factors are either exacerbated in patients with T2DM (e.g., obesity, reduced muscle quality, poor balance, and falls [17, 18]) or T2DM specific (e.g., poor glycemic control, T2DM duration, and diabetic medications and complications [4–10]). However, available evidence suggests that these risk factors do not sufficiently explain the increased fracture risk in T2DM patients [19]. Collectively, these findings point to the likelihood that skeletal factors, not captured by DXA, may contribute toward increased fracture risk in T2DM patients, which has led to the hypothesis that an important factor underlying fragility fractures in patients with T2DM is deteriorated "bone quality."

Measuring Aspects of Bone Quality in Patients with T2DM

Historically, progress in understanding how aspects of bone quality might be altered in patients with T2DM has been hampered by methodological barriers. Importantly, however, recent technological advances now allow for the safe quantification of previously inaccessible, but theoretically vital [20], components of bone quality (i.e., structural and material properties that determine skeletal fragility [21]), advances which may enhance our ability to stratify fracture risk in patients with T2DM beyond that provided by DXA.

To test whether bone microarchitecture, an important component of bone quality [20, 21], is altered in patients with T2DM, several studies have used novel imaging techniques (see Chap. 11) such as high-resolution peripheral quantitative computed tomography (HRpQCT), which provides a noninvasive "virtual bone biopsy" of the distal radius and tibia. Because the skeleton is approximately 80 % cortical bone [22], and cortical porosity has emerged as a potentially critical determinant of bone quality and a strong independent predictor of fracture risk [20, 21], there has been considerable interest in assessing cortical bone porosity by HRpQCT in patients with T2DM. In the first such study, Burghardt et al. [23] found that 19 postmenopausal women with T2DM tended to have better trabecular microarchitecture but worse cortical microarchitecture (i.e., higher cortical porosity) at the distal radius, and similar changes at the distal tibia, when compared to 19 nondiabetic postmenopausal control subjects. By contrast, Shu et al. [24] reported no significant differences in trabecular or cortical bone parameters (assessed by HRpQCT) at the radius or tibia in postmenopausal women with T2DM vs. nondiabetic postmenopausal control subjects, although only 14 subjects per group underwent HRpQCT scanning and cortical porosity was not reported. Consistent with the findings of Burghardt and colleagues [23], we found that radial cortical porosity tended to be higher in 30 postmenopausal women with T2DM as compared to 30 age-matched nondiabetic
postmenopausal control subjects after adjustment for BMI, although this difference did not reach statistical significance [25]. Finally, in the only study thus far of postmenopausal women with T2DM and fragility fractures, Patsch et al. [26] found significantly higher cortical porosity at the distal radius in 20 T2DM facture patients as compared to 20 T2DM patients without fracture. Collectively, these relatively small studies in postmenopausal women suggest that cortical bone may be preferentially compromised in patients with T2DM, and that cortical porosity in particular may be of relevance for understanding fracture risk in diabetic patients.

In addition to cortical microarchitecture, the material composition of bone is another important component of bone quality [20, 21]. For many years, however, our understanding of how human bone material properties are affected by aging or disease was limited to cadaveric studies. Despite the inherent limitations of studying cadaveric bone specimens, these early studies were pivotal in identifying deterioration of bone material properties (e.g., increased crack growth and reduced fracture toughness) with advancing age [27, 28], suggesting that worsening bone material properties may contribute to fracture risk. Nonetheless, because of the invasive measures required to assess bone material properties at the time, it was not possible to confirm such findings in human subjects. The recent advent of technology for noninvasively measuring the material properties of bone in vivo in humans, however, has begun to fill this void. To do so, Hansma and colleagues sought to develop an instrument capable of measuring human bone material properties in vivo, a process which required the development, extensive validation, and iterative improvement of a series of instruments [29–33]. The first of these microindentation devices to be utilized in human subjects for the measurement of indices of bone material properties in vivo was the "reference point indentation (RPI) instrument" (Fig. 10.1); other versions were later called the "bone diagnostic instrument" [29–31] and the "tissue diagnostic instrument" [32]. Using this technology, small



Fig. 10.1 Image of the Reference Point Indentation (RPI) Instrument, the first microindentation device to be utilized in human subjects for the measurement of indices of bone material properties in vivo. Reproduced from Hansma et al. (Rev Sci Instrum 2008;79(6):064303) with permission

Fig. 10.2 Image of the OsteoProbe® (Active Life Scientific Inc., Santa Barbara, CA), a handheld microindentation instrument designed for in vivo measurements of the material properties of bone (i.e., bone material strength index [BMSI]) in humans at the midshaft of the non-dominant anterior tibia. Reproduced from Farr et al. (J Bone Miner Res 2014;29:787–95) with permission



studies found that in vivo indices of bone material properties may be worse in hip fracture [34] and atypical femoral fracture [35] patients. While implementation of this technology represented a noteworthy advance, the RPI procedure unfortunately required complete displacement of the periosteum from the bone surface at the site of measurement, a factor which rendered routine clinical assessments of bone material properties using the RPI prototype impractical.

To address these issues, these same investigators developed a new microindentation device that allows for the safe quantification of an index of bone material properties (without the need for a reference probe or displacement of the periosteum) in humans with minimal discomfort. This measure of bone material properties, later coined the "bone material strength index" or BMSI, can now be obtained using the OsteoProbe[®] (Active Life Scientific Inc., Santa Barbara, CA, USA), a small handheld microindentation instrument designed for in vivo bone material property measurements [33, 36]. Figure 10.2 shows the device positioned over the midshaft of the anterior tibia, the optimal testing site (determined by calculating the midpoint from the proximal end of the medial border of the tibial plateau to the distal edge of the medial malleolus). Indeed, this site provides a viable flat surface, away from tendons, ligaments, blood vessels, and nerves, for performing cortical bone material property measurements.

Primary components of the OsteoProbe[®] include an impact mechanism, a displacement transducer, and a sterilized stainless steel disposable probe with a 90° conical tip (375 µm diameter; <10 µm tip sharpness radius). After administration of local anesthesia (1 % lidocaine), the probe is inserted through the skin, soft tissue, and periosteum until resting on the bone surface. While keeping the device perpendicular to the bone surface (within 10°), the measurement is started by slowly depressing the outer housing unit of the instrument, which in turn compresses the internal primary spring until the trigger mechanism initiates an impact. The impact mechanism creates a force (peak of 40 N) to drive the probe a minimal distance into



the bone cortex, while the displacement transducer measures the indentation distance increase (μ m) from impact. Thus, the BMSI is a direct in vivo measure of how well bone resists deformation in response to microindentation.

Accurate and precise in vivo measurements of BMSI are due to four basic operations: (1) pre-load, (2) trigger, (3) impact, and (4) unloading, as described previously [33]. Each operation occurs at different rates as indicated in the time (s; seconds) vs. force (N; newtons) graph shown in Fig. 10.3, which displays the forces imposed on the bone during each measurement cycle. Pre-load (operation 1) occurs over ~1 s when the operator slowly depresses the device's outer housing unit to compress the internal primary spring. This ensures that the test probe is securely anchored to the bone before the primary indentation occurs. Once the pre-load force reaches 10 N, the trigger (operation 2) mechanism is automatically initiated, generating an impact (operation 3) lasting ~0.25 ms (with a peak force of 40 N and a constant impulse rate at 0.01 N s). After impact, unloading (operation 4) results in return of the force to 0 N and completion of the measurement cycle. A displacement transducer measures the indentation distance increase (µm) from impact, which is converted by computer to BMSI, defined as 100 times the ratio of the harmonic mean indentation distance increase from impact from five separate impacts into a polymethyl-methacrylate phantom relative to the indentation distance increase from impact into bone [36]. For each subject, BMSI is calculated as the average of 5-10measurements, generally performed in a circular, rather than linear, fashion at different midshaft tibial sites (separated by >2 mm). It should be noted that the procedure causes minimal discomfort (only during the local anesthesia injection) and no complications have been observed to date. In fact, subjects report feeling only light pressure on the bone, but no pain, during the procedure, which typically takes less than 5 min to perform. Although minimally invasive, it should be noted that this procedure is not intended for patients who have a significant skin disorder, bruising, local edema, or infection, as well as those undergoing treatment for blood clots or severe coagulation defects.

The indentations are very small (on average, <200 μ m), but do create minimal microcracks in the bone surface that are clinically insignificant and only detectable by scanning electron microscopy [33, 36]. Thus, BMSI is a direct measure of fracture resistance because the farther the probe indents the bone (higher the indentation distance from impact), the more easily bone is fractured (lower the BMSI). This technology has now been used in a number of relatively small clinical studies [25, 33, 36–39], and similar microindentation approaches have been utilized in previous studies involving animals [40–42] and humans [34, 35].

There is also considerable interest in how bone material properties relate to other measures of skeletal properties such as whole-bone mechanical properties and bone microarchitecture (see Chap. 9). For example, studies utilizing an animal model of T2DM found strong correlations between impaired bone material properties derived from an in vitro microindentation testing instrument, called the BioDent[®] (Active Life Scientific Inc., Santa Barbara, CA, USA), and reduced bone strength assessed by traditional ex vivo bone mechanical testing techniques (three-point bending and axial compression) at both appendicular and axial skeletal sites [42]. However, given the different loading patterns utilized by the OsteoProbe[®] (single-impact) and BioDent[®] (cyclic loading), these instruments likely provide different measures of bone material properties. Nonetheless, understanding how the outputs of these instruments relate to bone quality and to one another is of great importance toward their application in predicting fracture risk in both preclinical and clinical settings.

Although additional work is needed to answer these questions, some data are starting to emerge. For example, Granke et al. [43] recently tested the extent to which bone parameters derived from the BioDent[®] and OsteoProbe[®] were related to other measures of cortical bone quality in human cadaveric bone specimens. Based on measurements performed by both instruments at the same site, BMSI (derived from the OsteoProbe®) was inversely correlated with local cortical porosity (r=-0.69, P<0.001), whereas none of the BioDent[®] parameters were related to any bone microarchitectural parameters. The authors concluded from these findings that the OsteoProbe® may be more sensitive than the BioDent® for measurement of aspects of cortical bone microarchitecture, in addition to its ability to assess bone tissue material properties [43]. This may reflect the greater force (~40 N) and depth (on average, ~200 µm) of the single-impact loading mechanism used by the OsteoProbe[®] vs. the BioDent[®] (force of ~20 N, depth of ~70 µm). Further, because fractures typically result from a single insult, it is possible that a one-time impact load more accurately reflects how bone responds to trauma as compared to a cyclic loading pattern of modeling. Future studies are needed to test these hypotheses experimentally.

There is also interest in how in vivo measures of bone material properties (e.g., obtained using the OsteoProbe[®]) relate to bone imaging parameters in humans, although this issue has received little attention to date. Interestingly, in T2DM patients [25], BMSI was not significantly (all *P* values >0.05) correlated with any radial or tibial cortical bone parameters (derived from HRpQCT) or any regional aBMD parameters (derived from DXA), suggesting that in patients with T2DM,

bone material properties may be a major predictor of skeletal fragility independent of cortical bone microarchitecture and aBMD. Notwithstanding these preliminary findings, additional studies will be required to test this possibility in larger cohorts of both diabetic and nondiabetic women and men.

Bone Material Properties in Patients with T2DM

Whether patients with T2DM have compromised bone material properties as compared to nondiabetic persons has been a question of long-standing interest [16, 44]. To address this, we recently performed in vivo microindentation testing (using the OsteoProbe[®]) of the tibia to directly measure BMSI in 60 postmenopausal women (age range, 50–80 years): 30 patients diagnosed with T2DM for >10 years and 30 age-matched, nondiabetic women [25]. We also assessed bone microarchitecture of the distal radius and tibia by HRpQCT and regional aBMD by DXA. In addition, we examined the associations of BMSI with circulating glycated hemoglobin A1c levels. Compared to controls, T2DM patients had significantly lower BMSI: unadjusted (-11.7%; P < 0.001; Fig. 10.4a) and following adjustment for BMI (-10.5%;P < 0.001; Fig. 10.4b). These differences remained significant following additional adjustments for potential confounders, including hypertension and diabetic complications (data not shown). Interestingly, the mean glycated hemoglobin level over the previous 10 years was negatively correlated with BMSI (r=-0.41; P=0.026) in patients with T2DM. By contrast, BMSI was not associated with a cross-sectional glycated hemoglobin level in nondiabetic control subjects (r=-0.09, P=0.630). These findings thus represent the first demonstration of compromised bone material properties in patients with T2DM. In addition, our results highlight the potential detrimental effects of prolonged hyperglycemia on bone quality. We infer from these findings that the skeleton warrants recognition as another important target tissue subject to diabetic complications [25].



Fig. 10.4 Unadjusted (**a**) and body mass index (BMI)-adjusted (**b**) comparisons of bone material strength index (BMSI) between patients with type 2 diabetes mellitus (T2DM) and age-matched, nondiabetic controls. Values are shown as mean \pm SE. ****P*<0.001. Adapted and reproduced from Farr et al. (J Bone Miner Res 2014;29:787–95) with permission

A valid concern of the OsteoProbe[®] instrument is safety. Notably, our group [25] and others have extensive experience using the device [25, 33, 36–39] and a closely related prototype [34, 35] in humans. Importantly, no complications have been reported at either the time of measurement or in follow-up. Nevertheless, as noted previously, out of an abundance of caution, patients with a significant skin disorder, bruising, local edema, or infection, as well as those undergoing treatment for blood clots or severe coagulation defects should not undergo this procedure. In addition, although the indentations are very small (on average <200 μ m) and separated by at least 2 mm, it is likely preferable to perform the indentation in a circle rather than linearly down the tibia in order to minimize the extremely improbable, but theoretical [44], risk of stress fracture.

Pathogenesis of Reduced Bone Quality in T2DM

Because patients with T2DM have a higher fracture risk than nondiabetic persons despite having normal or even increased aBMD [4, 13, 14], they likely have impairments in bone quality. Our finding of reduced bone material properties in patients with T2DM [25] is consistent with this hypothesis, and points to important unanswered questions regarding the mechanism(s) of human diabetic bone disease. Studies in T2DM rodent models have identified advanced glycation end products (AGEs) among the primary proximal culprits [45]. Unfortunately, however, there are currently no T2DM animal models that sufficiently recapitulate the skeletal phenotype observed in human patients with adult-onset T2DM [46]. Thus, humans currently serve as the best model system for studying the pathogenesis of diabetic bone disease.

Although additional studies in living human subjects with T2DM are direly needed, as noted previously, most of our knowledge to date of diabetic bone disease in humans has been derived from cadaveric bone specimens. Nevertheless, findings from such studies have been critical in demonstrating that the increased risk of fragility fractures in patients with T2DM may stem from non-enzymatic glycation and collagen cross-links in diabetic bone, partially resulting from the increased production of AGEs. Because AGEs are recognized for their central role in mediating diabetic complications [45], these "bad" intermediate protein products have emerged as potentially viable therapeutic targets for preventing or perhaps reversing diabetic bone disease in humans [47].

AGEs are intermediate protein products that undergo undesirable chemical modifications after excessive glucose exposure [48]. Thus, prolonged high circulating glucose concentrations in T2DM may lead not only to higher glycated hemoglobin levels, but also to the accumulation of AGEs both in the circulation and in bone tissue itself [49]. Increases in AGEs negatively impact type I collagen as well as the generation and survival of osteoblasts in bone [16, 49, 50]. While enzymatic processes normally produce pyridinium crosslinks necessary for the mechanical integrity of collagen, non-enzymatic glycation creates undesired crosslinks that modify the extracellular matrix in bone and decrease bone formation [51–53]. Further, in vitro experiments have demonstrated that AGEs impair bone formation by interfering with osteoblast differentiation, attachment to bone surfaces, function, and survival [54–56]. Thus, the increased production of AGEs in T2DM likely leads to defects in bone formation and ultimately, due to coupling between bone formation and resorption, to low bone turnover.

While many different AGEs have been identified in human tissues [48], the most studied is pentosidine, circulating levels of which have been associated with clinical fractures in T2DM patients [57, 58]. N^{e} -carboxy-methyl-lysine (CML) is another dominant component of total AGEs [59], and its level in circulation has been related to osteoporosis [60] and hip fracture [61]. While commercial immunoassays have been developed to measure pentosidine in human serum, concerns exist regarding assay validity due to numerous factors in blood and urine that interfere with immunoassay standardization [62]. Thus, reverse phase high performance liquid chromatography is currently perhaps the most reliable method for the quantification of pentosidine in the circulation [63, 64].

To date, few studies have measured AGEs in human bone tissue, with most limited to cadaveric specimens obtained from nondiabetic human subjects. Although bone pentosidine is increased in nondiabetic fracture patients relative to non-fracture controls [65, 66] and has been shown to predict vertebral biomechanical properties (independent of BMD) in nondiabetic subjects [67], bone pentosidine levels have not been directly measured in vivo in T2DM patients. Furthermore, while pentosidine has conventionally been used as a surrogate biomarker of total AGE accumulation, it is only a single component of the total AGE content in bone [49]. Indeed, recent evidence suggests that pentosidine may not fully account for the overall influence of glycation on bone tissue [53].

To address this issue, the Vashishth laboratory has developed techniques to directly quantify total AGEs (ng quinine/mg collagen) via a fluorometric assay [68] and pentosidine (mmol/mol collagen) using ultra-high performance liquid chromatography (UPLC) methods [69] in human bone tissue specimens. Using these techniques, they showed that, in nondiabetic human cadaveric bone specimens, total bone AGEs are associated with bone mechanical properties, including yield strain (r=0.54; P<0.01), ultimate strain (r=0.038; P=0.09), and toughness (r=-0.39; P=0.08). In addition, pentosidine is also associated with yield strain (r=0.46; P<0.05) and ultimate strain (r=0.44; P<0.05), but not toughness (r=-0.25; P>0.05) [52]. While these findings provide potential insights into human diabetic bone disease, to our knowledge, no study has examined AGEs in bone tissue specimens obtained from living subjects with T2DM. Such studies may be pivotal toward unveiling the pathogenesis of skeletal fragility in patients with T2DM.

Future Directions

For many years, the lack of appropriate tools for noninvasively assessing critical aspects of bone quality significantly limited our understanding of skeletal deterioration in patients with T2DM. However, recent technological advances in high-resolution image acquisition and analysis, as well as assessment of bone material

properties in vivo offer great potential for improving prediction of fragility fractures in this population. Despite the success and increasingly widespread use of novel tools such as HRpOCT and bone microindentation testing, additional work is needed to understand how the outputs of these instruments improve fracture prediction in patients. Furthermore, although our group previously demonstrated in patients with T2DM that cortical porosity is increased and that BMSI is reduced and correlates inversely with chronic glycemic control as assessed crosssectionally [25], whether cortical porosity and BMSI deteriorate over time to a greater extent in diabetic vs. nondiabetic subjects remains unknown and are important issues that need to be resolved. This will require concurrently evaluating biomechanically relevant components of bone structure and quality in both cross-sectional and longitudinal studies of large numbers of women and men with and without T2DM using identical methodology. In addition, while it is plausible that glycemic control (as assessed by glycated hemoglobin) or AGEs (as measured in the circulation or in bone tissue) are associated with changes in bone material properties over time in patients with T2DM, this remains untested experimentally. Finally, further work is needed to establish the potential causal role of AGEs in mediating diabetic bone disease.

Conclusions and Working Model

In conclusion, considerable evidence indicates that the skeleton needs to be recognized as another important target tissue subject to diabetic complications. Our current understanding of the pathogenesis of skeletal fragility in diabetes suggests a working model (Fig. 10.5), whereby poor glucose control in patients with T2DM leads to increases in AGEs that have negative effects on osteoblasts, which in turn causes a reduction in bone formation. This defect in bone formation subsequently results in low bone turnover in T2DM patients, which prolongs the lifespan of type



Fig. 10.5 Working model of the pathogenesis of fragility fractures in patients with type 2 diabetes mellitus (T2DM). *AGEs* advance glycation endproducts

I collagen in bone, thereby leaving it particularly vulnerable to damage from increased AGEs. Ultimately, this creates a "vicious cycle" that may contribute to reduced bone quality and increased fracture risk in patients with T2DM.

Acknowledgments The authors would like to thank Active Life Scientific for their technical support during the described work, in particular Drs. Paul Hansma and Hal Kopeikin, as well as Mr. Davis Brimer. Active Life Scientific provided the microindentation instrument and probes used for part of this work but had no control over the outcomes or the content of the manuscript. Dr. Khosla is supported by P01 AG004875, R01 AR027065, and UL1 TR000135 from the National Institutes of Health (NIH). Dr. Farr is supported by a career development award from the Robert and Arlene Kogod Center on Aging (Mayo Clinic). The content is solely the responsibility of the authors and does not necessarily represent the official views of the NIH.

Funding: Dr. Khosla is supported by P01 AG004875, R01 AR027065, and UL1 TR000135 from the National Institutes of Health (NIH). Dr. Farr is supported by a career development award from the Robert and Arlene Kogod Center on Aging (Mayo Clinic). The content is solely the responsibility of the authors and does not necessarily represent the official views of the NIH.

Disclosures: The authors have no conflicts of interest with respect to this work.

References

- 1. Shaw JE, Sicree RA, Zimmet PZ. Global estimates of the prevalence of diabetes for 2010 and 2030. Diabetes Res Clin Pract. 2010;87:4–14.
- Flegal KM, Carroll MD, Kit BK, Ogden CL. Prevalence of obesity and trends in the distribution of body mass index among US adults, 1999-2010. JAMA. 2012;307:491–7.
- D'Agostino RB, Nam BH. Evaluation of the performance of survival analysis models: discrimination and calibration measures. In: Balakrishnan N, Rao CR, editors. Handbook of statistics. Amsterdam: Elsevier; 2004. p. 1–25.
- Schwartz AV, Sellmeyer DE, Ensrud KE, Cauley JA, Tabor HK, Schreiner PJ, et al. Older women with diabetes have an increased risk of fracture: a prospective study. J Clin Endocrinol Metab. 2001;86:32–8.
- 5. Ivers RQ, Cumming RG, Mitchell P, Peduto AJ. Diabetes and risk of fracture: the Blue Mountains Eye Study. Diabetes Care. 2001;24:1198–203.
- Strotmeyer ES, Cauley JA, Schwartz AV, Nevitt MC, Resnick HE, Bauer DC, et al. Nontraumatic fracture risk with diabetes mellitus and impaired fasting glucose in older white and black adults: the health, aging, and body composition study. Arch Intern Med. 2005; 165:1612–7.
- De II L, van Der Klift M, De Laet CE, van Daele PL, Hofman A, Pols HA. Bone mineral density and fracture risk in type-2 diabetes mellitus: the Rotterdam Study. Osteoporos Int. 2005;16:1713–20.
- Bonds DE, Larson JC, Schwartz AV, Strotmeyer ES, Robbins J, Rodriguez BL, et al. Risk of fracture in women with type 2 diabetes: the Women's Health Initiative Observational Study. J Clin Endocrinol Metab. 2006;91:3404–10.
- 9. Melton LJ, Leibson CL, Achenbach SJ, Therneau TM, Khosla S. Fracture risk in type 2 diabetes: update of a population-based study. J Bone Miner Res. 2008;23:1334–42.
- 10. Janghorbani M, Van Dam RM, Willett WC, Hu FB. Systematic review of type 1 and type 2 diabetes mellitus and risk of fracture. Am J Epidemiol. 2007;166:495–505.
- 11. Vestergaard P. Discrepancies in bone mineral density and fracture risk in patients with type 1 and type 2 diabetes—a meta-analysis. Osteoporos Int. 2007;18:427–44.

- Strotmeyer ES, Cauley JA, Schwartz AV, Nevitt MC, Resnick HE, Zmuda JM, et al. Diabetes is associated independently of body composition with BMD and bone volume in older white and black men and women: the Health, Aging, and Body Composition Study. J Bone Miner Res. 2004;19:1084–91.
- 13. Schwartz AV, Vittinghoff E, Bauer DC, Hillier TA, Strotmeyer ES, Ensrud KE, et al. Association of BMD and FRAX score with risk of fracture in older adults with type 2 diabetes. JAMA. 2011;305:2184–92.
- 14. Giangregorio LM, Leslie WD, Lix LM, Johansson H, Oden A, McCloskey E, et al. FRAX underestimates fracture risk in patients with diabetes. J Bone Miner Res. 2012;27:301–8.
- Dawson-Hughes B, Tosteson AN, Melton III LJ, Baim S, Favus MJ, Khosla S, et al. Implications of absolute fracture risk assessment for osteoporosis practice guidelines in the USA. Osteoporos Int. 2008;19:449–58.
- Leslie WD, Rubin MR, Schwartz AV, Kanis JA. Type 2 diabetes and bone. J Bone Miner Res. 2012;27:2231–7.
- Schwartz AV, Hillier TA, Sellmeyer DE, Resnick HE, Gregg E, Ensrud KE, et al. Older women with diabetes have a higher risk of falls: a prospective study. Diabetes Care. 2002; 25:1749–54.
- Schwartz AV, Vittinghoff E, Sellmeyer DE, Feingold KR, de Rekeneire N, Strotmeyer ES, et al. Diabetes-related complications, glycemic control, and falls in older adults. Diabetes Care. 2008;31:391–6.
- 19. Vestergaard P, Rejnmark L, Mosekilde L. Diabetes and its complications and their relationship with risk of fractures in type 1 and 2 diabetes. Calcif Tissue Int. 2009;84:45–55.
- 20. Chavassieux P, Seeman E, Delmas PD. Insights into material and structural basis of bone fragility from diseases associated with fractures: how determinants of the biomechanical properties of bone are compromised by disease. Endocr Rev. 2007;28:151–64.
- Seeman E, Delmas PD. Bone quality—the material and structural basis of bone strength and fragility. N Engl J Med. 2006;354:2250–61.
- Bonnick SL. Skeletal anatomy in densitometry. In: Bonnick SL, editor. Bone densitometry in clinical practice. New York: Humana; 1998. p. 35–78.
- Burghardt AJ, Issever AS, Schwartz AV, Davis KA, Masharani U, Majumdar S, et al. Highresolution peripheral quantitative computed tomographic imaging of cortical and trabecular bone microarchitecture in patients with type 2 diabetes mellitus. J Clin Endocrinol Metab. 2010;95:5045–55.
- 24. Shu A, Yin MT, Stein E, Cremers S, Dworakowski E, Ives R, et al. Bone structure and turnover in type 2 diabetes mellitus. Osteoporos Int. 2012;23:635–41.
- Farr JN, Drake MT, Amin S, Melton III LJ, McCready LK, Khosla S. In vivo assessment of bone quality in postmenopausal women with type 2 diabetes. J Bone Miner Res. 2014;29:787–95.
- Patsch JM, Burghardt AJ, Yap SP, Baum T, Schwartz AV, Joseph GB, et al. Increased cortical porosity in type 2 diabetic postmenopausal women with fragility fractures. J Bone Miner Res. 2013;28:313–24.
- Nalla RK, Kruzic JJ, Ritchie RO. On the origin of the toughness of mineralized tissue: microcracking or crack bridging? Bone. 2004;34:790–8.
- Diab T, Condon KW, Burr DB, Vashishth D. Age-related change in the damage morphology of human cortical bone and its role in bone fragility. Bone. 2006;38:427–31.
- Hansma PK, Turner PJ, Fanter GE. Bone diagnostic instrument. Rev Sci Instrum. 2006;77:075105.
- Hansma P, Turner P, Drake B, Yurtsev E, Proctor A, Mathews P, et al. The bone diagnostic instrument II: indentation distance increase. Rev Sci Instrum. 2008;79:064303.
- Randall C, Mathews P, Yurtsev E, Sahar N, Kohn D, Hansma P. The bone diagnostic instrument III: testing mouse femora. Rev Sci Instrum. 2009;80:065108.
- Hansma P, Yu H, Schultz D, Rodriguez A, Yurtsev EA, Orr J, et al. The tissue diagnostic instrument. Rev Sci Instrum. 2009;80:054303.

- Bridges D, Randall C, Hansma PK. A new device for performing reference point indentation without a reference probe. Rev Sci Instrum. 2012;83:044301.
- Diez-Perez A, Guerri R, Nogues X, Caceres E, Pena MJ, Mellibovsky L, et al. Microindentation for in vivo measurement of bone tissue mechanical properties in humans. J Bone Miner Res. 2010;25:1877–85.
- 35. Guerri-Fernandez RC, Nogues X, Quesada Gomex JM, Torres del Pliego E, Puig L, Garcia-Giralt N, et al. Microindentation for in vivo measurement of bone tissue material properties in atypical femoral fracture patients and controls. J Bone Miner Res. 2013;28:162–8.
- Randall C, Bridges D, Guerri R, Nogues X, Puig L, Torres E, et al. Applications of a new handheld reference point indentation instrument measuring bone material strength. J Med Device. 2013;7:410051–6.
- 37. Malgo F, Hamdy NA, Papapoulos S, Appelman-Dijkstra N. Bone material strength as measured by microindentation in vivo is decreased independently of BMD in patients with fractures. ASBMR annual meeting. 2014; Abstract #1064.
- Appelman-Dijkstra N, Malgo F, Papapoulos S, Hamdy NA. Microindentation in vivo captures elements of bone fragility independently of BMD. ASBMR annual meeting. 2014; Abstract #FR0290.
- Sosa DD, Eriksen EF. Patients with stress fractures exhibit impaired bone material properties by microindentation. ASBMR annual meeting. 2014; Abstract #MO0028.
- Rasoulian R, Raeisi Najafi A, Chittenden M, Jasiuk I. Reference point indentation study of age-related changes in porcine femoral cortical bone. J Biomech. 2013;46:1689–96.
- 41. Aref M, Gallant MA, Organ JM, Wallace JM, Newman CL, Burr DB, et al. In vivo reference point indentation reveals positive effects of raloxifene on mechanical properties following 6 months of treatment in skeletally mature beagle dogs. Bone. 2013;56:449–53.
- Gallant MA, Brown DM, Organ JM, Allen MR, Burr DB. Reference-point indentation correlates with bone toughness assessed using whole-bone traditional mechanical testing. Bone. 2013;53:301–5.
- 43. Granke M, Uppuganti S, Manhard MK, Does M, Lee D, Perrien D, et al. Differences in assessment of micro-indentation resistance between BioDent and OsteoProbe. ASBMR annual meeting. 2014; Abstract #SU0023.
- 44. Jepsen KJ, Schlecht SH. Biomechanical mechanisms: resolving the apparent conundrum of why individuals with type II diabetes show increased fracture incidence despite having normal BMD. J Bone Miner Res. 2014;29:784–6.
- 45. Forbes JM, Cooper ME. Mechanisms of diabetic complications. Physiol Rev. 2013;93: 137–88.
- Fajardo RJ, Karim L, Calley VI, Bouxsein ML. A review of rodent models of type 2 diabetic skeletal fragility. J Bone Miner Res. 2014;29:1025–40.
- Goh SY, Cooper ME. Clinical review: the role of advanced glycation end products in progression and complications of diabetes. J Clin Endocrinol Metab. 2008;93:1143–52.
- 48. Ahmed N, Thornalley PJ. Quantitative screening of protein biomarkers of early glycation, advanced glycation, oxidation and nitrosation in cellular and extracellular proteins by tandem mass spectrometry multiple reaction monitoring. Biochem Soc Trans. 2003;31:1417–22.
- Sroga GE, Vashishth D. Effects of bone matrix proteins on fracture and fragility in osteoporosis. Curr Osteoporos Rep. 2012;10:141–50.
- Manolagas SC. From estrogen-centric to aging and oxidative stress: a revised perspective of the pathogenesis of osteoporosis. Endocr Rev. 2010;31:266–300.
- Garnero P, Borel O, Gineyts E, Duboeuf F, Solberg H, Bouxsein ML, et al. Extracellular posttranslational modifications of collagen are major determinants of biomechanical properties of fetal bovine cortical bone. Bone. 2006;38:300–9.
- Karim L, Vashishth D. Heterogeneous glycation of cancellous bone and its association with bone quality and fragility. PLoS One. 2012;7, e35047.
- Karim L, Tang SY, Sroga GE, Vashishth D. Differences in non-enzymatic glycation and collagen cross-links between human cortical and cancellous bone. Osteoporos Int. 2013;24(9): 2441–7.

- McCarthy AD, Uemura T, Etcheverry SB, Cortizo AM. Advanced glycation endproducts interfere with integrin-mediated osteoblastic attachment to a type-I collagen matrix. Int J Biochem Cell Biol. 2004;36:840–8.
- 55. Kume S, Kato S, Yamagishi S, Inagaki Y, Ueda S, Arima N, et al. Advanced glycation endproducts attenuate human mesenchymal stem cells and prevent cognate differentiation into adipose tissue, cartilage, and bone. J Bone Miner Res. 2005;20:1647–58.
- 56. Sanguineti R, Storace D, Monacelli F, Federici A, Odetti P. Pentosidine effects on human osteoblasts in vitro. Ann N Y Acad Sci. 2008;1126:166–72.
- 57. Yamamoto M, Yamaguchi T, Yamauchi M, Yano S, Sugimoto T. Serum pentosidine levels are positively associated with the presence of vertebral fractures in postmenopausal women with type 2 diabetes. J Clin Endocrinol Metab. 2008;93:1013–9.
- Schwartz AV, Garnero P, Hillier TA, Sellmeyer DE, Strotmeyer ES, Feingold KR, et al. Pentosidine and increased fracture risk in older adults with type 2 diabetes. J Clin Endocrinol Metab. 2009;94:2380–6.
- 59. Reddy S, Bichler J, Wells-Knecht KJ, Thorpe SR, Baynes JW. N epsilon-(carboxymethyl) lysine is a dominant advanced glycation end product (AGE) antigen in tissue proteins. Biochemistry. 1995;34:10872–8.
- Hein G, Wiegand R, Lehmann G, Stein G, Franke S. Advanced glycation end-products pentosidine and N epsilon-carboxymethyllysine are elevated in serum of patients with osteoporosis. Rheumatology (Oxford). 2003;42:1242–6.
- Barzilay J, Buzkova P, Zieman SJ, Kizer JR, Djousse L, Ix JH, et al. Circulating levels of carboxy-methyl-lysine (CML) are associated with hip fracture risk: the Cardiovascular Health Study. J Bone Miner Res. 2014;29:1061–6.
- 62. Dorrian CA, Cathcart S, Clausen J, Shapiro D, Dominiczak MH. Factors in human serum interfere with the measurement of advanced glycation endproducts. Cell Mol Biol. 1998;44:1069–79.
- 63. Franke S, Muller A, Sommer M, Busch M, Kientsch-Engel R, Stein G. Serum levels of total homocysteine, homocysteine metabolites and of advanced glycation end-products (AGEs) in patients after renal transplantation. Clin Nephrol. 2003;59:88–97.
- 64. Neumann T, Lodes S, Kastner B, Franke S, Kiehntopf M, Lehmann T, et al. High serum pentosidine but not esRAGE is associated with prevalent fractures in type 1 diabetes independent of bone mineral density and glycaemic control. Osteoporos Int. 2014;25:1527–33.
- Saito M, Fujii K, Mori Y, Marumo K. Role of collagen enzymatic and glycation induced crosslinks as a determinant of bone quality in spontaneously diabetic WBN/Kob rats. Osteoporos Int. 2006;17:1514–23.
- 66. Saito M, Fujii K, Marumo K. Degree of mineralization-related collagen crosslinking in the femoral neck cancellous bone in cases of hip fracture and controls. Calcif Tissue Int. 2006;79:160–8.
- 67. Viguet-Carrin S, Roux JP, Arlot ME, Merabet Z, Leeming DJ, Byrjalsen I, et al. Contribution of the advanced glycation end product pentosidine and of maturation of type I collagen to compressive biomechanical properties of human lumbar vertebrae. Bone. 2006;39:1073–9.
- 68. Tang SY, Zeenath U, Vashishth D. Effects of non-enzymatic glycation on cancellous bone fragility. Bone. 2007;40:1144–51.
- 69. Sroga GE, Vashishth D. UPLC methodology for identification and quantitation of naturally fluorescent crosslinks in proteins: a study of bone collagen. J Chromatogr B Analyt Technol Biomed Life Sci. 2011;879:379–85.

Chapter 11 Imaging of Diabetic Bone Structure

Thomas M. Link and Ursula Heilmeier

Introduction and Background

A number of cohort studies and a comprehensive meta-analysis have found an increased risk of fragility fractures in patients with type 2 diabetes (T2DM) [1–3]. In particular, hip fractures are a cause of substantial morbidity and mortality in older adults and nearly always result in hospitalization. They are fatal in about 20 %, and also produce permanent disability in approximately 50 % [4]. In patients with diabetes, the morbidity and mortality associated with fractures is likely to be even higher related to diabetic complications, frequently found obesity, and possibly also slower fracture healing. Hip fractures are the worst-case scenario as a complication of diabetic bone disease, however, low energy spine, proximal humerus, ankle and feet fractures associated with diabetes are also a cause of substantial morbidity. Schwartz et al. [3] found that insulin-treated diabetics had more than double the risk of foot fractures (multivariate adjusted relative risk (RR) 2.66) compared with non-diabetics and these fractures are a serious and well-recognized complication of diabetes mellitus which may impair the clinical outcome of the patients remarkably.

So far, the most reliable and best validated techniques to assess fracture risk and biomechanical stability of bone are those focusing on *bone mass*, i.e., dual X-ray absorptiometry (DXA) and quantitative computed tomography (QCT). However, for diabetes bone mass, respectively bone mineral density (BMD) measurements have limitations. Previous studies have found average, or increased [2, 5–7] BMD in type 2 diabetes, even after statistically correcting data for effects of larger body size. This paradox of higher BMD but increased fracture risk in type 2 diabetes is not fully explained by more frequent falls or other traditional risk factors for fracture

B. Lecka-Czernik, J.L. Fowlkes (eds.), *Diabetic Bone Disease*, DOI 10.1007/978-3-319-16402-1_11

T.M. Link, M.D., Ph.D. (🖂) • U. Heilmeier, M.D.

Department of Radiology and Biomedical Imaging, University of California, San Francisco, 550, 16th Street, San Francisco, CA 94158, USA e-mail: tmlink@radiology.ucsf.edu

[©] Springer International Publishing Switzerland 2016

in those subjects with diabetes. Type 2 diabetes is associated with a moderate increase in the risk of falling [8], with some evidence that type 2 diabetic women using insulin have two to three times greater risk of falling than those without diabetes. However, in observational studies that have considered reported falls and risk factors for falls and injurious falls, these factors do not account for the association between diabetes and fracture [3]. This suggests that there are additional factors beyond the traditional risk factors of low BMD and more frequent falls that are contributing to fracture risk in older diabetic adults. These include *bone structure and composition* that may be compromised in patients with type 2 diabetes, which may explain the combination of increased fracture risk with higher BMD. The paradox of higher BMD and increased fracture risk in diabetes provides an opportunity to assess bone properties/biomarkers beyond BMD that may contribute to fracture risk. A better understanding of the impact of diabetes on bone strength could provide insights that would guide efforts to prevent fractures in those with diabetes and in older adults generally.

Recently, a number of studies have analyzed cortical bone structure assessed with high-resolution quantitative computed tomography (HR-pQCT) [9, 10], trabecular bone structure and bone marrow composition using magnetic resonance (MR)-based techniques [11–13], quantitative ultrasound (QUS) [14] and trabecular bone score derived from DXA to study diabetic bone disease [15, 16]. This chapter will provide an overview of recent studies characterizing bone structure and composition in an effort to explain increased bone fragility in T2 diabetes.

Imaging Technologies

High-Resolution Quantitative Computed Tomography

HR-pQCT is a new technology to investigate bone architecture analyzing cortical and trabecular bone structure. The dedicated extremity imaging system designed for imaging of trabecular and cortical bone architecture is currently available from a single manufacturer (XtremeCT, Scanco Medical AG, Brüttisellen, Switzerland) and was developed based on experimental MicroCT technology (Scanner and patient set up with a representative distal tibia scan shown in Fig. 11.1). This device has the advantage of substantially higher signal-to-noise ratio (SNR) and spatial resolution compared to Multidetector-CT (MD-CT) and MRI. The nominal isotropic voxel dimension in clinical patients is 82 μ m in the older (generation 1) and 61 μ m in the newer scanners (generation 2) [17]. Furthermore, the effective radiation dose is substantially lower compared to whole body MD-CT, and primarily does not involve critical, radiosensitive organs (effective dose <3 microSv) [18]. The scan time for HR-pQCT is approximately 3 min for each scan of the distal tibia and radius.



Fig. 11.1 HR-pQCT. (a) Demonstrates the correct patient positioning for a lower extremity scan. The patient's lower extremity is immobilized through a carbon fiber cast and is anchored in the scanner to minimize motion. For each scan, the effective patient dose amounts to approximately 3 microSv. Average time per scan is 2.8 min. (b) Displays a representative mid-stack HR-pQCT image of the lower extremity cross section in a healthy patient

Using HR-pQCT, three studies investigated bone architecture in patients with and without diabetes [9, 10, 19]. Shu et al. [19] analyzed a cohort of 14 T2DM postmenopausal women and 14 control postmenopausal women. Standard trabecular and cortical bone parameters were calculated from the HR-pQCT images, which included cortical thickness and density but not cortical porosity. None of the investigated parameters showed significant differences. It should be noted that information on fragility fractures in this cohort was not available.

Burghardt et al. [9] applied HR-pOCT to characterize cortical and trabecular microarchitecture and biomechanics in the peripheral skeleton of female patients with T2DM using a cross-sectional study design. Elderly female patients (age, 62.9 ± 7.7 years) with a history of T2DM (n = 19) and age- and height-matched controls (n=19) were recruited and imaged using HR-pOCT at the distal radius and tibia. The T2DM cohort included patients with and without fragility fractures. Quantitative measures of volumetric (BMD), cross-sectional geometry, trabecular and cortical microarchitecture were calculated. Additionally, compressive mechanical properties were determined by microfinite element analysis. These investigators found that compared to the controls, the T2DM cohort had 10 % higher trabecular volumetric BMD (P < 0.05) adjacent to the cortex and higher trabecular thickness in the tibia (13.8 %; P < 0.05). Cortical porosity differences, however, were consistent with impaired bone strength in T2DM patients and were significant in the radius (>+50 %; P<0.05), whereas pore volume approached significance in the tibia (+118 %; P=0.1). They concluded that, in T2DM patients, the impaired resistance to bending loads may be due to inefficient redistribution of bone mass, characterized by loss of intracortical bone offset by an elevation in trabecular bone density. These results may provide a potential explanation for the inability of standard BMD measures to explain the elevated fracture incidence in patients with T2DM.

In a more recent study comparing T2DM postmenopausal women and agematched control women differences were found in cortical porosity at the distal radius and tibia between T2DM patients with and without fragility fractures [10]. This study provided evidence for the key importance of cortical abnormalities in the increased prevalence of fragility fractures in T2DM patients and the potential role of HR-pQCT as a novel diagnostic imaging test measuring cortical porosity. For this study 80 women (mean age 61.3 ± 5.7 years) were recruited into four groups (n=20per group) with and without diabetes and with and without fragility fractures. Participants underwent DXA and HR-pQCT of the ultradistal and distal radius and tibia. In the HR-pQCT images volumetric bone mineral density, cortical and trabecular structure measures, including cortical porosity, were calculated. Bone strength was estimated using microfinite element analysis (microFEA). At the ultradistal and distal tibia, women with diabetic fractures had substantially greater intracortical pore volume (+52.6 %, p=0.009; +95.4 %, p=0.020), relative porosity (+58.1 %; p=0.005; +87.9 %, p=0.011) and endocortical bone surface (+10.9 %, p=0.031; +11.5 %, 0.019) compared to diabetic women without fracture. At the distal radius T2DM women with fractures had a 4.7-fold greater relative porosity (p < 0.01) than women without fractures. At the ultradistal radius, intracortical pore volume was significantly higher in T2DM women with fractures compared to those



Fig. 11.2 Representative HR-pQCT images of the distal radius in Type 2 diabetic postmenopausal women with and without fragility fracture. Shown are the mid-stack tomograms for a Type 2 diabetic postmenopausal woman without history of fragility fracture (*left*) and a Type 2 diabetic woman with a positive history of fragility fracture (*right*). Note significant differences in cortical porosity between the 2 patients

without fractures (+67.8 %, p=0.018). T2DM women with fractures also displayed larger trabecular heterogeneity (ultradistal radius; +36.8 %, p=0.035), and lower total and cortical BMD (ultradistal tibia: -12.6 %, p=0.031; -6.8 %, p=0.011) than women without fractures. T2DM women with fractures also exhibited significantly higher pore-related deficits in stiffness, failure load and cortical load fraction at the ultradistal and distal tibia, and the distal radius than women with diabetes and no fractures. Figure 11.2 illustrates the findings of increased cortical porosity in a woman with type 2 diabetes and fragility fracture in relation to a diabetic woman without fracture. Interestingly, comparing nondiabetic fracture and control women, only a nonsignificant trend was found with increase in pore volume (+38.9 %, p=0.060) at the ultradistal radius. Overall, similar to the previous study, the results of this study suggested that severe deficits in cortical bone quality may be responsible for fragility fractures in postmenopausal diabetic women.

Magnetic Resonance Based Techniques

A small number of recent studies used magnetic resonance based techniques to investigate bone architecture and bone marrow composition in patients with diabetes [11–13]. Magnetic resonance spectroscopy (MRS) provides information on the biochemical composition of tissues and Li et al. introduced a method to quantify

vertebral bone marrow adiposity with proton MR spectroscopy ((1)H-MRS) at 3 T [20]. In their study, they showed high reproducibility of the technique with an average coefficient of variation of vertebral marrow fat content quantification of 1.7 %. They showed variation of marrow adiposity at different vertebral levels and feasibility for identifying patients with low bone density. In addition to quantifying overall bone marrow fat at 3 T, they were also able to selectively quantify the unsaturation level of the bone marrow fat [21].

Specifically using this technology in patients with T2DM, it was shown that vertebral bone marrow fat content correlated significantly with HbA1c and visceral adipose tissue in T2DM patients [22] and that decreased unsaturated bone marrow lipids were found to be associated with T2DM and fragility fractures [10]. Baum et al. [22] compared vertebral bone marrow fat content quantified with proton MR spectroscopy ((1)H-MRS) with the volume of abdominal adipose tissue, lumbar spine volumetric bone mineral density (vBMD), and blood biomarkers in postmenopausal women with and without T2DM. Thirteen postmenopausal women with T2DM and 13 age- and body mass index-matched healthy controls were included in this study. All subjects underwent (1)H-MRS of L1-L3 to quantify vertebral bone marrow fat content (FC) and unsaturated lipid fraction. OCT was performed to assess vBMD of L1–L3. The volumes of abdominal subcutaneous (SAT), visceral (VAT) and total adipose tissue (TAT) were determined from the OCT images and adjusted for abdominal body volume (SAT(adj)/VAT(adj)/TAT(adj)). Fasting blood tests were also obtained and included plasma glucose and HbA1c. Mean FC showed an inverse correlation with vBMD (r=-0.452; P<0.05) in the whole study population. While mean FC was similar in the diabetic women and healthy controls ($69.3 \pm 7.5 \%$ vs. $67.5 \pm 6.1 \%$; P > 0.05), mean unsaturated lipid fraction was significantly lower in the diabetic group $(6.7 \pm 1.0 \% \text{ vs. } 7.9 \pm 1.6 \%)$; P < 0.05). SAT(adj) and TAT(adj) correlated significantly with mean FC in the whole study population (r=0.538 and r=0.466; P<0.05). In contrast to the control group, significant correlations of mean FC with VAT(adj) and HbA1c were observed in the diabetic group (r=0.642 and r=0.825; P<0.05). This study demonstrated that vertebral bone marrow fat content correlates significantly with SAT(adj), TAT(adj), and lumbar spine vBMD in postmenopausal women with and without T2DM, but with VAT(adj) and HbA1c only in women with T2DM.

Patsch et al. [11] quantified vertebral bone marrow fat content and composition in diabetic and nondiabetic postmenopausal women with fragility fractures and compared these measurements with those of non-fracture controls with and without type 2 diabetes mellitus. Sixty-nine postmenopausal women (mean age 63 ± 5 years) were recruited. Thirty-six patients had spinal and/or peripheral fragility fractures. Seventeen of the fracture patients were diabetic. Thirty-three women were controls without fractures and 16 of these were diabetic. To quantify vertebral bone marrow fat content and composition, patients underwent MR spectroscopy (MRS) of the lumbar spine at 3 T. BMD was determined by DXA of the hip and lumbar spine and QCT of the lumbar spine. To evaluate associations of vertebral marrow fat content and composition with spinal and/or peripheral fragility fractures and diabetes, linear regression models adjusted for age, race, and spine volumetric bone mineral density



Fig. 11.3 *High resolution axial 3 T Magnetic Resonance Image of the distal radius* obtained with a 3D steady state free precession sequence with a spatial resolution of 156 μ m × 208 μ m × 500 μ m. The image shows trabecular bone architecture with bone marrow interlay

(vBMD) by QCT were used. At the lumbar spine, nondiabetic and diabetic fracture patients had lower vBMD than controls and diabetics without fractures (p=0.018; p=0.005). However, areal bone mineral density (aBMD) by DXA did not differ between fracture and non-fracture patients. After adjustment for age, race, and spinal vBMD, the prevalence of fragility fractures was associated with -1.7 % lower unsaturation levels (confidence interval [CI] -2.8 to -0.5 %, p=0.005) and +2.9 % higher saturation levels (CI 0.5–5.3 %, p=0.017). Diabetes was associated with -1.3 % (CI -2.3 to -0.2 %, p=0.018) lower unsaturation and +3.3 % (CI 1.1–5.4 %, p=0.004) higher saturation levels. Diabetics with fractures had the lowest marrow unsaturation and highest saturation levels. In summary, these results demonstrate that altered bone marrow fat composition (lower unsaturation level) is linked with fragility fractures and diabetes. The authors suggested that MRS of spinal bone marrow fat may therefore serve as a novel tool for BMD-independent fracture risk assessment.

In addition to using magnetic resonance based spectroscopy for quantifying tissue composition, magnetic resonance imaging (MRI) can also be used to assess trabecular bone quality. A number of previous studies have shown that high-resolution MRI can directly visualize trabecular bone architecture at the distal radius, tibia, and proximal femur [23–26] (Fig. 11.3). Pritchard et al. were the first to use high-resolution MRI of the distal radius to compare trabecular bone microarchitecture of postmenopausal women with and without type 2 DM [13].

They acquired axial images with a 1 T MRI system and a voxel size of $0.195 \times 0.195 \times 1 \text{ mm}^3$. Image post-processing included geometric, topologic, and stereologic measures including number and size of trabecular bone network holes (marrow spaces), endosteal area, trabecular bone volume fraction, nodal and branch density, and apparent trabecular thickness, separation, and number. They also measured lumbar spine and proximal femur BMD with DXA. They found that T2DM women (n=30, mean \pm SD age 71.0 \pm 4.8 years) had larger holes (+13.3 %; P=0.001) within the trabecular bone network than women without T2DM (n=30, mean \pm SD age 70.7 \pm 4.9 years). Interestingly after adjustment for body mass index, DXA-based lumbar spine BMD did not differ between the diabetes and nondiabetes groups. They concluded that in women with type 2 DM, the average hole size within the trabecular bone network at the distal radius is greater compared to controls and hypothesized that this may explain the elevated fracture risk in women with T2DM.

These investigators also performed follow-up MR studies in a subset of their patients [12]. The aim of this study was to compare 2-year changes in trabecular bone microarchitecture in women with and without T2DM. Using the same technology as in the other study, they analyzed the number and size of trabecular bone holes, bone volume fraction (BV/TV), trabecular thickness (Tb.Th), number (Tb.N) and separation (Tb.Sp), endosteal area, nodal and branch density in 37 women. After adjustment for ethnicity, women with diabetes had a higher percent increase in number of trabecular bone holes compared to controls (10[1]% vs. -7[2]%, p=0.010), however, results were no longer significant after adjustment for multiple comparisons (p=0.090). There were also no differences in the change of the other trabecular bone microarchitecture variables between the two groups. This study provides feasibility data but clearly larger longitudinal studies with longer time intervals are required to study the evolution of T2DM-related bone architecture changes.

In the future, combined MR and HR-pQCT approaches may be used to better study the detailed mechanisms that drive bone changes in T2DM patients, in particular those that drive cortical porosity by fusing high-resolution peripheral quantitative computed tomography images (HR-pQCT) with high-resolution MR images [27]. Recent work showed that it is possible to study bone microarchitectural parameters and characterize intracortical marrow and vascular content, that are otherwise accessible only through destructive procedures [27].

Quantitative Ultrasound

Quantitative ultrasound (QUS) has been shown to measure mechanical properties of bone related to elastic modulus and compressive strength which reflect bone architecture, density, and elasticity [28]. Previous studies have shown that QUS of the calcaneus is effective in differentiating subjects with and without fragility fractures,

that it can be used to monitor treatment in metabolic bone disease and that it provides additional information beyond bone mineral density [29-31]. Based on the results of these previous studies, Yamaguchi et al. [14] investigated the role of calcaneal OUS in assessing fracture risk in T2DM. To test this hypothesis, these investigators measured calcaneal OUS as well as BMD at the lumbar spine, femoral neck, and 1/3 radius in 96 women (mean age 66.6 years old) and 99 men (64.7 years old) with T2DM, and examined their associations with prevalent vertebral fractures. Calcaneal speed of sound measurements were obtained in all patients. In T2DM patients, vertebral fractures were found in 33 women and 45 men. When comparing patients with and without vertebral fractures, there were no significant differences in values of speed of sound or BMD at any site between the groups in either gender. Logistic regression analysis adjusted for age and BMI showed that both OUS and BMD values were not significantly associated with the presence of vertebral fractures in either gender. The authors concluded that these results showed that OUS and BMD are unable to discriminate T2DM patients with and without prevalent vertebral fractures.

Another study, however, suggested that QUS measurements of the calcaneus could be used to differentiate elderly T2DM women with and without low energy fragility fractures at multiple different sites [32]. Patel et al. reported that while the fracture group did not differ significantly from the non-fracture group by age, diabetes-related risk factors or DXA BMD *Z* scores QUS variables were lower in the fracture group (P=0.04). It should be noted, however, that the results were borderline significant. Clearly, larger studies are required in the future to comprehensively assess the role of QUS in assessing fracture risk in patients with T2DM.

Trabecular Bone Score

Trabecular bone score (TBS) is a relatively new texture measure, which is used to analyze DXA images of the lumbar spine [33, 34]. Though DXA of the lumbar spine images provide very limited visualization of bone structure, they are obtained under very standardized conditions, are highly reproducible and potentially provide a texture assessment of combined cortical and trabecular bone [34]. Promising results have been found using TBS in previous studies; in a large cohort of 29,407 women, Hans et al. were able to demonstrate that significantly lower spine TBS and BMD were found in women with major osteoporotic, spine, and hip fractures (all p<0.0001) [35]. Interestingly spine TBS and BMD predicted fractures equally well, and the combination was superior to either measurement alone (p<0.001). The authors concluded that spine TBS predicts osteoporotic fractures and provides information that is independent of spine and hip BMD.

Using this texture measure in patients with T2DM, Leslie et al. [16] found similar results in differentiating women with and without major osteoporotic fractures. These investigators performed a retrospective cohort study using BMD results from a large clinical registry for the province of Manitoba, Canada. They included 29,407 women 50 years old and older with baseline DXA examinations, among whom 2356 had been diagnosed with diabetes. TBS was obtained in each patient and patient records were assessed for incident nontraumatic major osteoporotic fractures. They found that T2DM was associated with higher BMD at all sites but lower lumbar spine TBS in unadjusted and adjusted models (all P < 0.001). Major osteoporotic fractures were identified in 175 women (7.4 %) with and 1493 (5.5 %) without diabetes (P < 0.001). Lumbar spine TBS was a BMD-independent predictor of fracture and predicted fractures in those with diabetes (adjusted hazard ratio 1.27, 95 % CI 1.10–1.46) and without diabetes (hazard ratio 1.31, 95 % CI 1.24–1.38). The investigators concluded that lumbar spine TBS predicts osteoporotic fractures in those with diabetes and captures a larger portion of the diabetes-associated fracture risk than BMD.

In another recently published cross-sectional study, Dhaliwal et al. investigated TBS in a smaller cohort of women aged 30–90 years with T2DM [15]. They found that mean TBS was lower in T2DM (1.228 ± 0.140 vs. 1.298 ± 0.132 , p=0.013), irrespective of age while mean BMD was higher in T2DM (1.150 ± 0.172 vs. 1.051 ± 0.125 , p=0.001). Within the T2DM group, TBS was higher (1.254 ± 0.148) in subjects with good glycemic control (A1c ≤ 7.5 %) compared to those (1.166 ± 0.094 ; p=0.01) with poor glycemic control (A1c > 7.5 %). They concluded that in T2DM, TBS is lower and associated with poor glycemic control. It should be noted, however, that while the results of both of these studies are encouraging TBS is a texture parameter obtained from DXA images, which does not directly measure microarchitecture [33]. It is not entirely clear how TBS works in characterizing fracture risk and the results have to be interpreted cautiously.

Conclusion

Given the paradox of increased BMD and higher fracture risk in T2DM-related bone disease, a number of studies were performed using novel measurements of bone architecture, composition, and quality to better characterize bone abnormalities and fracture risk in T2DM patients. Promising results have been found in particular for HR-pQCT based cortical porosity, which may serve as a potential imaging biomarker for fracture risk in diabetic bone disease. Diabetic fracture and nonfracture groups could also be differentiated with MRS-based quantification of unsaturated lipids and a DXA-based texture measure of the lumbar spine. It should be noted, however, that most of these studies were conducted in relatively small subject numbers and larger scale research studies in this new field are required to investigate new imaging biomarkers that could predict and potentially help to treat the particularly devastating fragility fractures in T2DM patients.

References

- Janghorbani M, Van Dam RM, Willett WC, Hu FB. Systematic review of type 1 and type 2 diabetes mellitus and risk of fracture. Am J Epidemiol. 2007;166(5):495–505. PubMed Epub 2007/06/19. eng.
- Yamamoto M, Yamaguchi T, Yamauchi M, Kaji H, Sugimoto T. Diabetic patients have an increased risk of vertebral fractures independent of BMD or diabetic complications. J Bone Miner Res. 2009;24:702–9. PubMed Epub 2008/12/04.Eng.
- Schwartz AV, Sellmeyer DE, Ensrud KE, Cauley JA, Tabor HK, Schreiner PJ, et al. Older women with diabetes have an increased risk of fracture: a prospective study. J Clin Endocrinol Metab. 2001;86(1):32–8.
- 4. Bone health and osteoporosis: a report of the Surgeon General. Rockville: Office of the Surgeon General; 2004.
- Melton III LJ, Riggs BL, Leibson CL, Achenbach SJ, Camp JJ, Bouxsein ML, et al. A bone structural basis for fracture risk in diabetes. J Clin Endocrinol Metab. 2008;93(12):4804–9. PubMed Epub 2008/09/18. eng.
- Vestergaard P, Rejnmark L, Mosekilde L. Diabetes and its complications and their relationship with risk of fractures in type 1 and 2 diabetes. Calcif Tissue Int. 2009;84(1):45–55. PubMed Epub 2008/12/11. eng.
- 7. Schwartz AV, Sellmeyer DE. Diabetes, fracture, and bone fragility. Curr Osteoporos Rep. 2007;5(3):105–11. PubMed Epub 2007/10/11. eng.
- Schwartz AV, Hillier TA, Sellmeyer DE, Resnick HE, Gregg E, Ensrud KE, et al. Older women with diabetes have a higher risk of falls: a prospective study. Diabetes Care. 2002;25(10): 1749–54.
- Burghardt AJ, Issever AS, Schwartz AV, Davis KA, Masharani U, Majumdar S, et al. Highresolution peripheral quantitative computed tomographic imaging of cortical and trabecular bone microarchitecture in patients with type 2 diabetes mellitus. J Clin Endocrinol Metab. 2010;95(11):5045–55. Pubmed Central PMCID: 2968722. Epub 2010/08/20. eng.
- Patsch JM, Burghardt AJ, Yap SP, Baum T, Schwartz AV, Joseph GB, et al. Increased cortical porosity in type 2 diabetic postmenopausal women with fragility fractures. J Bone Miner Res. 2013;28(2):313–24. Pubmed Central PMCID: 3534818.
- Patsch JM, Li X, Baum T, Yap SP, Karampinos DC, Schwartz AV, et al. Bone marrow fat composition as a novel imaging biomarker in postmenopausal women with prevalent fragility fractures. J Bone Miner Res. 2013;28(8):1721–8. Pubmed Central PMCID: 3720702.
- 12. Pritchard JM, Giangregorio LM, Atkinson SA, Beattie KA, Inglis D, Ioannidis G, et al. Changes in trabecular bone microarchitecture in postmenopausal women with and without type 2 diabetes: a two year longitudinal study. BMC Musculoskelet Disord. 2013;14:114. Pubmed Central PMCID: 3618189.
- 13. Pritchard JM, Giangregorio LM, Atkinson SA, Beattie KA, Inglis D, Ioannidis G, et al. Association of larger holes in the trabecular bone at the distal radius in postmenopausal women with type 2 diabetes mellitus compared to controls. Arthritis Care Res. 2012;64(1):83–91.
- 14. Yamaguchi T, Yamamoto M, Kanazawa I, Yamauchi M, Yano S, Tanaka N, et al. Quantitative ultrasound and vertebral fractures in patients with type 2 diabetes. J Bone Miner Metab. 2011;29(5):626–32.
- 15. Dhaliwal R, Cibula D, Ghosh C, Weinstock RS, Moses AM. Bone quality assessment in type 2 diabetes mellitus. Osteoporos Int. 2014;25(7):1969–73.
- Leslie WD, Aubry-Rozier B, Lamy O, Hans D, Manitoba Bone Density Program. TBS (trabecular bone score) and diabetes-related fracture risk. J Clin Endocrinol Metab. 2013; 98(2):602–9.
- Burghardt AJ, Buie HR, Laib A, Majumdar S, Boyd SK. Reproducibility of direct quantitative measures of cortical bone microarchitecture of the distal radius and tibia by HR-pQCT. Bone. 2010;47(3):519–28. Pubmed Central PMCID: 2926164. Epub 2010/06/22. eng.

- Damilakis J, Adams JE, Guglielmi G, Link TM. Radiation exposure in X-ray-based imaging techniques used in osteoporosis. Eur Radiol. 2010;20(11):2707–14. Pubmed Central PMCID: 2948153.
- Shu A, Yin MT, Stein E, Cremers S, Dworakowski E, Ives R, et al. Bone structure and turnover in type 2 diabetes mellitus. Osteoporos Int. 2012;23(2):635–41. Pubmed Central PMCID: 3690650.
- 20. Li X, Kuo D, Schafer AL, Porzig A, Link TM, Black D, et al. Quantification of vertebral bone marrow fat content using 3 Tesla MR spectroscopy: reproducibility, vertebral variation, and applications in osteoporosis. J Magn Reson Imaging. 2011;33(4):974–9. Pubmed Central PMCID: 3072841. Epub 2011/03/31. eng.
- 21. Li X, Shet K, Rodriguez JP, Pino AM, Kurhanewicz J, Schwartz A, et al. Unsaturation level decreased in bone marrow lipids of postmenopausal women with low bone density using high resolution HRMAS NMR. ASBMR 2012, Minneapolis, MN; 12–15 Oct 2012.
- 22. Baum T, Yap SP, Karampinos DC, Nardo L, Kuo D, Burghardt AJ, et al. Does vertebral bone marrow fat content correlate with abdominal adipose tissue, lumbar spine BMD and blood biomarkers in women with type 2 diabetes mellitus? J Magn Reson Imaging. 2012;35(1): 117–24.
- Wehrli FW, Gomberg BR, Saha PK, Song HK, Hwang SN, Snyder PJ. Digital topological analysis of in vivo magnetic resonance microimages of trabecular bone reveals structural implications of osteoporosis. J Bone Miner Res. 2001;16(8):1520–31.
- Wehrli FW, Leonard MB, Saha PK, Gomberg BR. Quantitative high-resolution magnetic resonance imaging reveals structural implications of renal osteodystrophy on trabecular and cortical bone. J Magn Reson Imaging. 2004;20(1):83–9.
- Krug R, Burghardt AJ, Majumdar S, Link TM. High-resolution imaging techniques for the assessment of osteoporosis. Radiol Clin North Am. 2010;48(3):601–21. Pubmed Central PMCID: 2901255.
- 26. Krug R, Carballido-Gamio J, Banerjee S, Burghardt AJ, Link TM, Majumdar S. In vivo ultrahigh-field magnetic resonance imaging of trabecular bone microarchitecture at 7 T. J Magn Reson Imaging. 2008;27(4):854–9. PubMed Epub 2008/04/03. eng.
- 27. Parrish R, Rivoire J, Han M, Schafer AV, Link TM, Krug R, et al., editors. Multi-modality in vivo imaging identifies marrow and vasculature within pathological cortical porosity. ASBMR, Houston; 2014.
- Guglielmi G, Adams J, Link TM. Quantitative ultrasound in the assessment of skeletal status. Eur Radiol. 2009;19(8):1837–48.
- 29. Gluer CC, Eastell R, Reid DM, Felsenberg D, Roux C, Barkmann R, et al. Association of five quantitative ultrasound devices and bone densitometry with osteoporotic vertebral fractures in a population-based sample: the OPUS Study. J Bone Miner Res. 2004;19(5):782–93.
- Gonnelli S, Cepollaro C, Gennari L, Montagnani A, Caffarelli C, Merlotti D, et al. Quantitative ultrasound and dual-energy X-ray absorptiometry in the prediction of fragility fracture in men. Osteoporos Int. 2005;16(8):963–8.
- Gonnelli S, Cepollaro C, Montagnani A, Martini S, Gennari L, Mangeri M, et al. Heel ultrasonography in monitoring alendronate therapy: a four-year longitudinal study. Osteoporos Int. 2002;13(5):415–21.
- 32. Patel S, Hyer S, Tweed K, Kerry S, Allan K, Rodin A, et al. Risk factors for fractures and falls in older women with type 2 diabetes mellitus. Calcif Tissue Int. 2008;82(2):87–91.
- 33. Bousson V, Bergot C, Sutter B, Levitz P, Cortet B, Scientific Committee of the Groupe de Recherche et d'Information sur les Ostéoporoses. Trabecular bone score (TBS): available knowledge, clinical relevance, and future prospects. Osteoporos Int. 2012;23(5):1489–501.
- 34. Link TM. Screening: Assessing bone structure in the prediction of osteoporotic fractures. Nat Rev Rheumatol. 2012;8(1):6–8.
- Hans D, Goertzen AL, Krieg MA, Leslie WD. Bone microarchitecture assessed by TBS predicts osteoporotic fractures independent of bone density: the Manitoba study. J Bone Miner Res. 2011;26(11):2762–9.

Index

A

Adhesive capsulitis, 82-83 A Diabetes Outcome Progression Trial (ADOPT), 129, 131, 132, 135 Advanced glycosylation end products (AGEs), 10 - 11biomarkers, 118-119 hyperglycemia, 160 material properties, bone, 193, 194 osteoclasts activity, 101-102 Alpha-glucosidase inhibitors, 128, 138-139 Amylin, 128, 138, 159, 162 Ankle fracture, 54 complication rate, 75 incidence, 74 peripheral neuropathy and microvascular complications, 76 plafond fracture, 74 soft tissue injury, 74 stability of, 75 treatment, 75 unstable ankle fracture, 75 Anti-hyperglycemic therapies alpha-glucosidase inhibitors, 138-139 amylin, 138 biguanides (metformin), 129-130 effects, on skeleton, 128 incretin analogs and DPP4 protease inhibitors, 135-137 insulin, 130 randomized controlled trials, 128 SGLT2 inhibitors, 137 sulfonylureas, 135 thiazolidinediones ADOPT, 131

cardiovascular effects, 130 clinical evidence, 130 clinical studies, 132 hip fracture incidence, 131 mechanism of action, 130 PPARγ, 132, 134–135 PPARγ2, 133 rosiglitazone treatment, 132 Apoptosis, 103–104, 152–153 Autoimmune thyroid disease, 12

B

Biguanides (metformin), 129-130 BioDent®, 216 Biomarkers advanced glycation end products (AGEs), 118-119 blood biomarkers, 230 bone remodeling, 115-116 glycemia, 114-115 IGF-1, 117 insulin, 115 PTH, 116-117 sclerostin, 118 sphingosine 1-phosphate (S1P), 118 T2D, in architectural properties, 119 urine and serum pentosidine, 11 Bone marrow bone remodeling signaling, 156-157 hematopoietic stems cells, 148 osteoblasts, 148 osteoclasts, 148 PPARy activation, 134 stem cells, 125, 148

© Springer International Publishing Switzerland 2016 B. Lecka-Czernik, J.L. Fowlkes (eds.), *Diabetic Bone Disease*, DOI 10.1007/978-3-319-16402-1

structure and function, 133 T1D effects, 157-158 T2D effects, 159 vertebral bone marrow fat content and composition, 230 Bone mineral density (BMD), 59-65, 72, 96, 113, 114, 118, 119, 130, 132, 137, 138, 161, 226, 228, 229, 233 BMD Z-score, 26 clinical risk factors, 46 DXA, 54 femoral neck BMD, 49 fracture risk, 5, 27-28 FRAX calculation, 56 screening tools, 46 type 2 diabetes mellitus (T2DM), 26, 30 Bone remodeling, 147-148 biomarkers, 115-116 hyperglycemia (see Hyperglycemia) IL-23, 156 osteoblasts, 126 osteoclasts, 100, 125-126 osteocytes, 126 Wnt ligands, 157 Wnt signaling, 156 Bone structure, 7 diabetic bone structure, imaging of (see Imaging technologies) femur mid-shaft, 184 insulin, 184 insulin receptor (IR), 185 streptozotocin (STZ) model, 186 X-ray computed tomography (CT) scans, 185

С

Carpal tunnel syndrome (CTS), 81-82, 84

D

Diabetes mellitus (DM) orthopaedic condition (*see* Orthopaedic surgery) periodontal disease (*see* Periodontal disease) skeletal fragility, working model, 220–221 T1DM (*see* Type 1 diabetes mellitus (T1DM)) T2DM (*see* Type 2 diabetes mellitus (T2DM)) Dual-energy X-ray absorptiometry, 138, 186, 211 Dupuytren's disease, 71, 80-81

E

Energy metabolism system, 126-127

F

Finite element analysis (FEA) bone geometry, 189 OCT-FEA, 190-191 quantitative CT (OCT), 190 strain distribution, 190 stress-strain distribution, 191 Fracture ankle fracture (see Ankle fracture) hip fracture bone mineral density, 71-72 classification, 71 complications, 73-74 hypoglycemia, 73 incidence, 71 mortality rates, 74 and type 2 diabetes (T2D) age interaction, 28 and aging population, 25 BMD Z-score, 26 body size, 32 bone density, 30-32 case-control studies, 28 diabetes duration, 33 diabetes-related complications, 36-37 falls. 33 fracture incidence, 28-29 glycemic control, 34-36 hypoglycemia, 36 in Netherlands, 27 and prediabetes, 37-38 in Scotland cohort studies, 26 Vestergaard meta-analysis, 26 Fracture risk assessment in diabetes clinical practice guidelines, 60-61 comparative performance, 60 lumbar spine trabecular bone score, 62 - 64MOF, 54 obesity, 54-57 potential risk adjustment, 61-62 predict fracture risk, 58-60 risk factors, 58 screening tools, 58 FRAX

Index

femoral neck BMD, 49 fracture discrimination, 51 high-quality fracture data, 51 input variables, 48 mortality epidemiology, 50 smoking, 49 survival analysis, 49 Garvan Fracture Risk Calculator, 51, 52 OFracture and FRAX, 53 in older women and men, 54 osteoporotic and hip fracture, 52 screenshot for, 53 2012 version, 53 screening tools, 46, 48 Fracture Risk Assessment Tool (FRAX), 47 Frozen shoulder. See Adhesive capsulitis

G

Garvan Fracture Risk Calculator, 47 Gingiva/gingivitis, 99 Glycemia, 114–115

H

Hematopoietic stem cells (HSC), 125 High-resolution peripheral quantitative computed tomography (HRpOCT), 212 Hyperglycemia AGEs, 160 alkaline phosphatase and osteocalcin, 159 calcitropic hormones and leptin, 163-164 hyperlipidemia, 163 inflammation, 164 insulin and IGF-1 signaling, 162 MC3T3-E1 cell culture, 160 MSC adipogenesis, 160 osteogenesis, 160 oxidative stress, 161 RAW264.7 cells and bone marrow macrophages, 160 ROS production, 159 Wnt signaling, 161-162 Hyperlipidemia, 163 Hypoglycemia, 36, 73

I

Imaging technologies HR-pQCT advantages, 226

cortical and trabecular bone structure, 226 cortical porosity, 229 cross-sectional study design, 228 distal radius and tibia, 228 lower extremity scan, 227 microfinite element analysis, 228 scan time, 226 magnetic resonance based techniques, 229-232 quantitative ultrasound, 232-233 trabecular bone score, 233-234 Indentation techniques BioDent records, 201-202 microindentation, 200 nanoindentation properties, 200 Oliver-Pharr method, 199 reference point indentation (RPI), 200, 201 tissue hardness and modulus, 199 Insulin anti-hyperglycemic therapies, 130 biomarkers, 115 bone structure, 184 and IGF-1 signaling, 162 Integrin protein, 155

M

Macrophage colony stimulating factor (M-CSF), 126, 148, 154, 155, 163 Malmo Preventive Project, 37 Material properties, bone advanced glycation end products (AGEs), 193.194 β-amino-propionitrile (BAPN), 193 bone structure (see Bone structure) "brittle" bone, 194 B6 vitamer, 193 cadaveric bone, 192 engineering mechanics "brittle" fracture, 187 engineering stress vs. engineering curve, 188 fatigue life, 187 fatigue testing, 189 flaw size, 189 force vs. displacement data, 188 fracture toughness tests, 189 stress fracture, 187 finite element analysis bone geometry, 189 QCT-FEA, 190-191 quantitative CT (QCT), 190

strain distribution, 190 stress-strain distribution, 191 indentation techniques BioDent records, 201–202 nanoindentation properties, 200 Oliver-Pharr method, 199 reference point indentation (RPI), 200, 201 tissue hardness and modulus, 199 nondiabetic and diabetic rodents, 195–198 sample size and storage conditions, 192 Mesenchymal stem cells (MSCs), 103

0

Obesity, 31, 46, 73, 153, 164 BMI and fracture risk, 55 femoral strength index, 56 fracture discrimination, 57 lean mass and fat mass components, 55 osteoporosis, 54 Oliver-Pharr method, 199 Orthopaedic surgery adhesive capsulitis, 82-83 ankle fracture (see Ankle fracture) CTS, 81-82 Dupuytren's disease, 80-81 hip fracture bone mineral density, 71-72 classification, 71 complications, 73-74 hypoglycemia, 73 incidence, 71 mortality rates, 74 infection, 76-77 plantar ulcerations, 77-78 plantar ulcers, 71 trigger finger, 78-79 Osteoblasts apoptosis, 103–104 bone marrow, 148 bone remodeling, 126 differentiation and function, 102-103 physiology, 149-150 T1 diabetes effects apoptosis, 152-153 lineage selection, 151–152 maturation, 152 T2 diabetes effects, 153-154 Osteoclasts regulation, 154-155 T1 diabetes effects, 155 T2 diabetes effects, 156

Osteoporosis Risk Assessment Instrument (ORAI), 46–48, 58 Osteoporosis Self-Assessment Tool (OST), 47, 48 Osteoprotegerin (OPG), 9, 102, 127, 154–156, 162 Oxidative stress, 96, 161

P

Periodontal disease alveolar bone loss, 100 children and adolescents, 97 diabetes impact, on bone, 96 gingiva/gingivitis, 99 gingival tissues, 97 incidence, 96 metabolic changes, 95-96 osteoblasts apoptosis, 103-104 differentiation and function, 102-103 osteoclasts activity AGEs, 101-102 inflammatory area, 100-101 ROS. 101 pathogenic mechanisms, 98 periodontal ligament/loss of attachment, 100 physiological conditions, 97 Peripheral blood mononuclear cells (PBMC), 156 Plantar ulcerations, 77-78

Q

QFracture, 14, 46, 48, 50, 51 and FRAX, 53 in older women and men, 54 osteoporotic and hip fracture, 52 screenshot for, 53 2012 version, 53 Quantitative ultrasound, 232–233

R

Receptor activator of nuclear factor kappa-B ligand (RANKL), 98, 101, 102, 104, 117, 118, 126, 127, 129, 148, 154–156, 163 Reference point indentation (RPI) instrument, 213 Index

S

Sclerostin, 8, 118, 162 SGLT2 inhibitors, 128, 137 Simple Calculated Risk Estimation Score (SCORE), 46–48 Sphingosine 1-phosphate (S1P), 118 Stem cells bone marrow stem cells, 148 definition, 148 hematopoietic stems cells, 148 Sulfonylureas, 73, 128, 130, 135 Surgical site infections (SSIs), 76

Т

Thiazolidinediones (TZD), 34, 114, 115 ADOPT, 131 cardiovascular effects, 130 clinical evidence, 130 clinical studies, 132 hip fracture incidence, 131 mechanism of action, 130 PPARy, 132, 134–135 PPARy2, 133 rosiglitazone treatment, 132 TNF receptor-associated factor (TRAF), 155 Trabecular bone score (TBS), 62-64, 233-234 Trigger finger, 78–79 Type 1 diabetes mellitus (T1DM) amylin, 138 apoptosis, 152 autoimmune thyroid disease, 12 bone mineral density, 4 celiac autoimmunity/CD, 112 children and adolescents, 7-8 diabetic bone disease prevention, 13 treatment, 12-13 fracture risk BMD decrements, 5 fracture risk studies, 6 long-term glycemic control, 5 skeletal accumulation, 7 gingiva/gingivitis, 99 incidence, 3 inflammation, 164 insulin, 115 marrow cells, 158-159 pathological mechanisms advanced glycosylation end products, 10 - 11bone formation, 8–9 vitamin D deficiency, 9-10

prevalence, 3 PTH responses, 116 sclerostin, 118 Type 2 diabetes mellitus (T2DM) age-related bone loss, 212 architectural properties, 119 BMD, 30 **BMI. 32** BMSI, 216-217 bone material properties, 217-218 extremity fracture risk, 211 and fractures age interaction. 28 and aging population, 25 BMD Z-score, 26 body size, 32 bone density, 30-32 case-control studies, 28 diabetes duration, 33 diabetes-related complications, 36-37 falls, 33 fracture incidence, 28-29 glycemic control, 34-36 hypoglycemia, 36 in Netherlands, 27 and prediabetes, 37-38 in Scotland cohort studies, 26 Vestergaard meta-analysis, 26 health-care databases, 5 hip fracture risk, 26-27 HRpQCT, 212 hyperglycemia, 147 inflammation, role of, 158 insulin, 115 microindentation device, 214 novel imaging techniques, 212 OsteoProbe®, 214 pathogenesis, 218-219 periodontal disease, 101 prevalence, 25 in QFracture, 61 reference point indentation (RPI) instrument, 213 sclerostin, 118 skeletal effects, 211 **TBS**. 64 time vs. force plot, 215

W

Wnt ligands, 157 Wnt signaling, 156